UNITED STATES SECURITIES AND EXCHANGE COMMISSION

Washington, D.C. 20549

FORM 10-K

(Mark One)	
ANNUAL REPORT PURSUANT TO SECTION 13 OR 15(d) OF THE SE	
For the fiscal year ended I	December 31, 2016
OR	
☐ TRANSITION REPORT PURSUANT TO SECTION 13 OR 15(d) OF TH	
For the transition period from	
Commission file nur	nber 1-4448
Baxt	er -
Baxter Interna	tional Inc.
(Exact Name of Registrant as S	pecified in its Charter)
Delaware	36-0781620
(State or Other Jurisdiction of Incorporation or Organization)	(I.R.S. Employer Identification No.)
One Baxter Parkway, Deerfield, Illinois	60015
(Address of Principal Executive Offices)	(Zip Code)
Registrant's telephone number, include	ling area code 224.948.2000
Securities registered pursuant to	Section 12(b) of the Act:
Title of Each Class	Name of Each Exchange on Which Registered
Common stock, \$1.00 par value	New York Stock Exchange Chicago Stock Exchange
Securities registered pursuant to Sec	tion 12(g) of the Act: None
Indicate by check mark if the registrant is a well-known seasoned issuer, as defined in Rule $405\mathrm{e}$	of the Securities Act. Yes ☑ No □
Indicate by check mark if the registrant is not required to file reports pursuant to Section $13\ or\ 1$	$5(d)$ of the Act. Yes \square No \square
Indicate by check mark whether the registrant (1) has filed all reports required to be filed by Sec months (or for such shorter period that the registrant was required to file such reports), and (2) h	
Indicate by check mark whether registrant has submitted electronically and posted on its corpora pursuant to Rule 405 of Regulation S-T during the preceding 12 months (or for such shorter per	
Indicate by check mark if disclosure of delinquent filers pursuant to Item 405 of Regulation S-K knowledge, in definitive proxy or information statements incorporated by reference in Part III o	
Indicate by check mark whether the registrant is a large accelerated filer, an accelerated filer, a neaccelerated filer," "accelerated filer" and "smaller reporting company" in Rule 12b-2 of the Excl	
Large accelerated filer ☑	Accelerated filer
Non-accelerated filer \qed (Do not check if a smaller reporting compa	ny) Smaller reporting company
Indicate by check mark whether the registrant is a shell company (as defined in Rule 12b-2 of the	e Act). Yes □ No ☑
The aggregate market value of the voting common equity held by non-affiliates of the registrant second fiscal quarter), based on the per share closing sale price of \$45.22 on that date and the as and executive officers are affiliates, was approximately \$24 billion. There is no non-voting comregistrant's common stock, \$1.00 par value, outstanding as of January 31, 2017 was 540,082,23	sumption for the purpose of this computation only that all of the registrant's directors non equity held by non-affiliates of the registrant. The number of shares of the 0.
DOCUMENTS INCORPORAT	
Portions of the registrant's definitive 2017 proxy statement for use in connection with its Annual into Part III of this report.	Meeting of Stockholders to be held on May 2, 2017 are incorporated by reference

TABLE OF CONTENTS

		Page Number
Item 1.	<u>Business</u>	1
Item 1A.	Risk Factors	5
Item 1B.	<u>Unresolved Staff Comments</u>	12
Item 2.	<u>Properties</u>	13
Item 3.	<u>Legal Proceedings</u>	14
Item 4.	Mine Safety Disclosures	14
Item 5.	Market for Registrant's Common Equity, Related Stockholder Matters and Issuer Purchases of Equity Securities	16
Item 6.	Selected Financial Data	17
Item 7.	Management's Discussion and Analysis of Financial Condition and Results of Operations	18
Item 7A.	Quantitative and Qualitative Disclosures About Market Risk	39
Item 8.	Financial Statements and Supplementary Data	40
Item 9.	Changes in and Disagreements with Accountants on Accounting and Financial Disclosure	88
Item 9A.	Controls and Procedures	88
Item 9B.	Other Information	88
Item 10.	Directors, Executive Officers and Corporate Governance	89
Item 11.	Executive Compensation	89
Item 12.	Security Ownership of Certain Beneficial Owners and Management and Related Stockholder Matters	89
Item 13.	Certain Relationships and Related Transactions, and Director Independence	90
Item 14.	Principal Accountant Fees and Services	90
Item 15.	Exhibits and Financial Statement Schedules	91

PART I

Item 1. Business.

Company Overview

Baxter International Inc., through its subsidiaries, provides a broad portfolio of essential renal and hospital products, including acute and chronic dialysis; sterile IV solutions; infusion systems and devices; parenteral nutrition therapies; premixed and oncolytic injectables; biosurgery products and anesthetics; drug reconstitution systems; and pharmacy automation, software and services. The company's global footprint and critical nature of its products and services play a key role in expanding access to healthcare in emerging and developed countries. These products are used by hospitals, kidney dialysis centers, nursing homes, rehabilitation centers, doctors' offices and by patients at home under physician supervision. As of December 31, 2016, Baxter manufactured products in over 20 countries and sells them in over 100 countries.

Baxter International Inc. was incorporated under Delaware law in 1931. As used in this report, "Baxter International" means Baxter International Inc. and "Baxter," the "company" or the "Company" means Baxter International and its consolidated subsidiaries (after giving effect to the separation and distribution of Baxalta Incorporated (Baxalta), as further described below), unless the context otherwise requires.

Separation of Baxalta

On July 1, 2015, Baxter completed the distribution of approximately 80.5% of the outstanding common stock of Baxalta to Baxter shareholders (the Distribution). The Distribution was made to Baxter's shareholders of record as of the close of business on June 17, 2015 (the Record Date), who received one share of Baxalta common stock for each Baxter common share held as of the Record Date. As a result of the distribution, Baxalta became an independent public company trading under the symbol "BXLT" on the New York Stock Exchange.

In 2016, Baxter disposed of its remaining 19.5% interest in Baxalta through a series of transactions including debt-for-equity exchanges, an equity-for-equity exchange and a contribution to its U.S. pension plan. As a result of these transactions, the company extinguished approximately \$3.65 billion in company indebtedness, repurchased 11,526,638 Baxter shares and contributed 17,145,570 Baxalta shares to its U.S. pension plan. On June 3, 2016, Baxalta became a wholly-owned subsidiary of Shire plc (Shire).

The local separation of Baxalta's business in certain countries outside the United States did not occur prior to the distribution date due to regulatory requirements, the need to obtain consents from local governmental authorities and other business reasons. Separation of the remaining three countries is expected to occur by 2018.

As a result of the separation, the consolidated statements of income, consolidated balance sheets, consolidated statements of cash flow, and related financial information reflect Baxalta's operations, assets and liabilities, and cash flows as discontinued operations for all periods presented.

Refer to Note 2 in Item 8 of this Annual Report on Form 10-K for additional information regarding the separation of Baxalta.

Business Segments and Products

The company operates in two segments: Hospital Products and Renal.

The Hospital Products business manufactures sterile intravenous (IV) solutions and administration sets, premixed drugs and drug-reconstitution systems, prefilled vials and syringes for injectable drugs, IV nutrition products, parenteral nutrition therapies, infusion pumps, inhalation anesthetics, and biosurgery products. The business also provides products and services related to pharmacy compounding, and drug formulation; sterile IV solutions; infusion systems and devices; parenteral nutrition therapies; premixed and oncolytic injectables; biosurgery products and anesthetics; drug reconstitution systems; and pharmacy automation, software and services.

The Renal business offers a comprehensive portfolio to meet the needs of patients with end-stage renal disease, or irreversible kidney disease and acute kidney injuries, including technologies and therapies for peritoneal dialysis (PD), hemodialysis (HD), continuous renal replacement therapy (CRRT) and additional dialysis services.

For financial information about Baxter's segments and sales franchises, see Note 17 in Item 8 of this Annual Report on Form 10-K.

Sales and Distribution

The company has its own direct sales force and also makes sales to and through independent distributors, drug wholesalers acting as sales agents and specialty pharmacy or other alternate site providers. In the United States, third parties such as Cardinal Health, Inc. warehouse and ship a significant portion of the company's products through their distribution centers. These centers are generally stocked with adequate inventories to facilitate prompt customer service. Sales and distribution methods include frequent contact by sales and customer service representatives, automated communications via various electronic purchasing systems, circulation of catalogs and merchandising bulletins, direct-mail campaigns, trade publication presence and advertising.

International sales are made and products are distributed on a direct basis or through independent distributors or sales agents in more than 100 countries as of December 31, 2016.

International Operations

The majority of the company's revenues are generated outside of the United States and geographic expansion remains a component of the company's strategy. Baxter's international presence includes operations in Europe (including Eastern and Central Europe), the Middle East, Africa, Asia-Pacific, Latin America and Canada. The company is subject to certain risks inherent in conducting business outside the United States. For more information on these risks, see the information under the captions "Risks Related to Baxter's Business—We are subject to risks associated with doing business globally" and "— Changes in foreign currency exchange rates and interest rates could have a material adverse effect on our operating results and liquidity" in Item 1A of this Annual Report on Form 10-K.

For financial information about foreign and domestic operations and geographic information, see Note 17 in Item 8 of this Annual Report on Form 10-K. For more information regarding foreign currency exchange risk, refer to the discussion under the caption entitled "Financial Instrument Market Risk" in Item 7 of this Annual Report on Form 10-K.

Contractual Arrangements

Substantial portions of the company's products are sold through contracts with customers, both within and outside the United States. Some of these contracts have terms of more than one year and place limits on the company's ability to increase prices. In the case of hospitals, governments and other facilities, these contracts may specify minimum quantities of a particular product or categories of products to be purchased by the customer.

In keeping with the increased emphasis on cost-effectiveness in healthcare delivery, many hospitals and other customers of medical products in the United States have joined group purchasing organizations (GPOs), or formed integrated delivery networks (IDNs), to enhance purchasing power. GPOs and IDNs negotiate pricing arrangements with manufacturers and distributors, and the negotiated prices are made available to members. Baxter has purchasing agreements with several of the major GPOs in the United States. GPOs may have agreements with more than one supplier for certain products. Accordingly, in these cases, Baxter faces competition from other suppliers even where a customer is a member of a GPO under contract with Baxter. Purchasing power is similarly consolidated in many other countries. For example, public contracting authorities act as the purchasing entities for the hospitals and other customers of medical products in their region and many hospitals and other customers have joined joint procurement entities and buying consortia. The result is that demand for healthcare products is increasingly concentrated across the company's markets globally.

Raw Materials

Raw materials essential to Baxter's business are purchased from numerous suppliers worldwide in the ordinary course of business. Although most of these materials are generally available, Baxter at times may experience shortages of supply. In an effort to manage risk associated with raw materials supply, Baxter works closely with its suppliers to help ensure availability and continuity of supply while maintaining high quality and reliability. The company also seeks to develop new and alternative sources of supply where beneficial to its overall raw materials procurement strategy.

The company also utilizes long-term supply contracts with some suppliers to help maintain continuity of supply and manage the risk of price increases. Baxter is not always able to recover cost increases for raw materials through customer pricing due to contractual limits and market forces.

In connection with the separation and distribution, Baxter entered into a long-term manufacturing and supply agreement with Baxalta manufactures and supplies Baxter with ARTISS, TISSEEL, FLOSEAL and stand-alone thrombin under the manufacturing and supply agreement, on a cost-plus basis.

Competition and Healthcare Cost Containment

Baxter's Hospital Products and Renal businesses benefit from a number of competitive advantages, including the breadth and depth of their product offerings, as well as strong relationships with customers, including hospitals and clinics, group purchasing organizations, physicians, and patients, many who self-administer the home-based therapies supplied by Baxter. Baxter as a whole benefits from efficiencies and cost advantages resulting from shared manufacturing facilities and the technological advantages of its products.

Although no single company competes with Baxter in all of its businesses, Baxter faces substantial competition in each of its segments from international and domestic healthcare and pharmaceutical companies and providers of all sizes, and these competitors often differ across our businesses. In addition, global and regional competitors continue to expand their manufacturing capacity and sales and marketing channels. Competition is primarily focused on cost-effectiveness, price, service, product performance, and technological innovation. There has been increasing consolidation in the company's customer base and by its competitors, which continues to result in pricing and market pressures.

Global efforts toward healthcare cost containment continue to exert pressure on product pricing. Governments around the world use various mechanisms to control healthcare expenditures, such as price controls, the formation of public contracting authorities, product formularies (lists of recommended or approved products), and competitive tenders which require the submission of a bid to sell products. Sales of Baxter's products are dependent, in part, on the availability of reimbursement by government agencies and healthcare programs, as well as insurance companies and other private payers. In the United States, the federal and many state governments have adopted or proposed initiatives relating to Medicaid and other health programs that may limit reimbursement or increase rebates that Baxter and other providers are required to pay to the state. In addition to government regulation, managed care organizations in the United States, which include medical insurance companies, medical plan administrators, health-maintenance organizations, hospital and physician alliances and pharmacy benefit managers, continue to put pressure on the price and usage of healthcare products. Managed care organizations seek to contain healthcare expenditures, and their purchasing strength has been increasing due to their consolidation into fewer, larger organizations and a growing number of enrolled patients. Baxter faces similar issues outside of the United States. In Europe and Latin America, for example, the government provides healthcare at low cost to patients, and controls its expenditures by purchasing products through public tenders, collective purchasing, regulating prices, setting reference prices in public tenders or limiting reimbursement or patient access to certain products.

Intellectual Property

Patents and other proprietary rights are essential to Baxter's business. Baxter relies on patents, trademarks, copyrights, trade secrets, know-how and confidentiality agreements to develop, maintain and strengthen its competitive position. Baxter owns a number of patents and trademarks throughout the world and has entered into license arrangements relating to various third-party patents and technologies. Products manufactured by Baxter are sold primarily under its own trademarks and trade names. Some products distributed by the company are sold under the company's trade names, while others are sold under trade names owned by its suppliers or partners. Trade secret protection of unpatented confidential and proprietary information is also important to Baxter. The company maintains certain details about its processes, products and technology as trade secrets and generally requires employees, consultants, and business partners to enter into confidentiality agreements. These agreements may be breached and Baxter may not have adequate remedies for any breach. In addition, Baxter's trade secrets may otherwise become known or be independently discovered by competitors. To the extent that Baxter's employees, consultants, and business partners use intellectual property owned by others in their work for the company, disputes may arise as to the rights in related or resulting know-how and inventions.

Baxter's policy is to protect its products and technology through patents and trademarks on a worldwide basis. This protection is sought in a manner that balances the cost of such protection against obtaining the greatest value for the company. Baxter also recognizes the need to promote the enforcement of its patents and trademarks and takes commercially reasonable steps to enforce its patents and trademarks around the world against potential infringers, including judicial or administrative action where appropriate.

Baxter operates in an industry susceptible to significant patent litigation. At any given time, the company is involved as either a plaintiff or defendant in a number of patent infringement and other intellectual property-related actions. Such litigation can result in significant royalty or other payments or result in injunctions that can prevent the sale of products. For more information on patent and other litigation, see Note 16 in Item 8 of this Annual Report on Form 10-K.

Research and Development

Baxter's investment in research and development (R&D), consistent with the company's portfolio optimization and capital allocation strategies, helps fuel its future growth and its ability to remain competitive in each of its business segments. Accordingly, Baxter continues to focus its investment on select R&D programs to enhance future growth through clinical differentiation. Expenditures for Baxter's R&D activities were \$647 million in 2016, \$603 million in 2015 and \$610 million in 2014. These expenditures include costs

associated with R&D activities performed at the company's R&D centers located around the world, which include facilities in Belgium, Sweden, Italy, Germany, China, Japan and the United States, as well as in-licensing, milestone and reimbursement payments made to partners for R&D work performed at non-Baxter locations.

For more information on the company's R&D activities, refer to the discussion under the caption entitled "Strategic Objectives" in Item 7 of this Annual Report on Form 10-K.

Quality Management

Baxter's continued success depends upon the quality of its products. Quality management plays an essential role in determining and meeting customer requirements, preventing defects, facilitating continuous improvement of the company's processes, products and services, and assuring the safety and efficacy of the company's products. Baxter's quality system enables the design, development, manufacturing, packaging, sterilization, handling, distribution and labeling of the company's products to ensure they conform to customer requirements. In order to continually improve the effectiveness and efficiency of the quality system, various measurements, monitoring and analysis methods such as management reviews and internal, external and vendor audits are employed at local and central levels.

Each product that Baxter markets is required to meet specific quality standards, both in packaging and in product integrity and quality. If any of those is determined to be compromised at any time, Baxter endeavors to take corrective and preventive actions designed to ensure compliance with regulatory requirements and to meet customer expectations. For more information on corrective actions taken by Baxter, refer to the discussion under the caption entitled "Certain Regulatory Matters" in Item 7 of this Annual Report on Form 10-K.

Government Regulation

The operations of Baxter and many of the products manufactured or sold by the company are subject to extensive regulation by numerous government agencies, both within and outside the United States. The Food and Drug Administration (FDA) in the United States, the European Medicines Agency (EMA) in Europe, the China Food and Drug Administration (CFDA) in China and other government agencies inside and outside of the United States, administer requirements covering the testing, safety, effectiveness, manufacturing, labeling, promotion and advertising, distribution and post-market surveillance of Baxter's products. The company must obtain specific approval from FDA and non-U.S. regulatory authorities before it can market and sell most of its products in a particular country. Even after the company obtains regulatory approval to market a product, the product and the company's manufacturing processes and quality systems are subject to continued review by FDA and other regulatory authorities globally. State agencies in the United States also regulate the facilities, operations, employees, products and services of the company within their respective states. The company and its facilities are subject to periodic inspections and possible administrative and legal actions by FDA and other regulatory agencies inside and outside the United States. Such actions may include warning letters, product recalls or seizures, monetary sanctions, injunctions to halt manufacture and distribution of products, civil or criminal sanctions, refusal of a government to grant approvals or licenses, restrictions on operations or withdrawal of existing approvals and licenses. As situations require, the company takes steps to ensure safety and efficacy of its products, such as removing products found not to meet applicable requirements from the market and improving the effectiveness of quality systems. For more information on compliance actions taken by the company, refer to the discussion under the caption entitled "Certain Regulatory Matters" in

The company is also subject to various laws inside and outside the United States concerning its relationships with healthcare professionals and government officials, price reporting and regulation, the promotion, sales and marketing of its products and services, the importation and exportation of products, the operation of its facilities and distribution of products. In the United States, the company is subject to the oversight of FDA, Office of the Inspector General within the Department of Health and Human Services (OIG), the Center for Medicare/Medicaid Services (CMS), the Department of Justice (DOJ), Environmental Protection Agency, Department of Defense and Customs and Border Protection in addition to others. The company supplies products and services to healthcare providers that are reimbursed by federally funded programs such as Medicare. As a result, the company's activities are subject to regulation by CMS and enforcement by OIG and DOJ. In each jurisdiction outside the United States, the company's activities are subject to regulation by government agencies including the EMA in Europe, CFDA in China and other agencies in other jurisdictions. Many of the agencies enforcing these laws have increased their enforcement activities with respect to healthcare companies in recent years. These actions appear to be part of a general trend toward increased enforcement activity globally.

Environmental policies of the company require compliance with all applicable environmental regulations and contemplate, among other things, appropriate capital expenditures for environmental protection.

Employees

As of December 31, 2016, Baxter employed approximately 48,000 people.

Available Information

Baxter makes available free of charge on its website at www.baxter.com its Annual Reports on Form 10-K, Quarterly Reports on Form 10-Q, Current Reports on Form 8-K, and amendments to those reports filed or furnished pursuant to Section 13(a) or 15(d) of the Securities Exchange Act of 1934, as amended (Exchange Act), as soon as reasonably practicable after electronically filing or furnishing such material to the Securities and Exchange Commission. In addition, Baxter's Corporate Governance Guidelines, Code of Conduct, and the charters for the committees of Baxter's Board of Directors are available on Baxter's website at www.baxter.com under "About Baxter—About us — Governance." All the foregoing materials will be made available to stockholders in print upon request by writing to: Corporate Secretary, Baxter International Inc., One Baxter Parkway, Deerfield, Illinois 60015. Information contained on Baxter's website shall not be deemed incorporated into, or to be a part of, this Annual Report on Form 10-K.

Item 1A. Risk Factors.

In addition to the other information in this Annual Report on Form 10-K, stockholders or prospective investors should carefully consider the following risk factors. If any of the events described below occurs, our business, financial condition and results of operations and future growth prospects could suffer.

Risks Related to Baxter's Business

We may not achieve our long-term financial improvement goals.

We have begun implementing plans to enhance profitability and returns for our stockholders. These plans include the achievement of certain financial goals (including improved operating margin) in 2017 and beyond. While we are continuing to refine these goals, our plan contemplates significant margin expansion over our long-range plan, which runs through 2020. We have identified certain key strategies to help achieve these targets. These strategies include optimizing our core product portfolio globally, driving operational excellence through the rebasing of our cost structure and various restructuring activities and maximizing the value derived from the allocation of our capital.

As part of these strategies, we continue to evaluate the performance of all of our businesses and may sell or acquire a business or product line or exit a particular market. We are also evaluating our corporate and commercial infrastructure in the interest of streamlining costs while maintaining our commitment to quality and safety. Future divestitures may result in significant write-offs, including those related to goodwill and other intangible assets. Future acquisitions may fail to achieve the desired financial results (including return on investment) and synergies and may not provide the desired market access. The restructuring of our operations may not generate targeted savings or may cause unexpected disruptions to our business. As a result, we may not achieve our targeted financial results, which could have a material adverse effect on our business, financial condition or results of operations.

If we are unable to successfully introduce new products or fail to keep pace with advances in technology, our business, financial condition and results of operations could be adversely affected.

We need to successfully introduce new products to achieve our strategic business objectives. Product development requires substantial investment and there is inherent risk in the research and development process. A successful product development process depends on many factors, including our ability to properly anticipate and satisfy customer needs, adapt to new technologies, obtain regulatory approvals on a timely basis, demonstrate satisfactory clinical results, manufacture products in an economical and timely manner and differentiate our products from those of our competitors. If we cannot successfully introduce new products or adapt to changing technologies, our products may become obsolete and our revenue and profitability could suffer.

Issues with product supply or quality could have an adverse effect upon our business, subject us to regulatory actions and cause a loss of customer confidence in us or our products.

Our success depends upon the availability and quality of our products. The medical products industry is competitive and subject to complex market dynamics and varying demand levels. These levels vary in response to macro-economic conditions, regulatory requirements (including the availability of private or public reimbursement) and seasonality. Additionally the development of new or enhanced products involves a lengthy regulatory process and is capital intensive. As a result, our ability to match our production levels and capacity to market demand is imprecise and may result in a failure to meet market demand or satisfy customer requirements for our products or, alternatively, an oversupply of inventory. Failure to meet market demand may result in customers transitioning to

available competitive products resulting in a loss of market share or customer confidence. In the event of an oversupply, we may be forced to lower our prices or record asset impairment charges or take other action which may adversely affect our business, financial condition and results of operations.

Additionally, quality management plays an essential role in determining and meeting customer requirements, preventing defects, improving the company's products and services and assuring the safety and efficacy of our products. Our future success depends on our ability to maintain and continuously improve our quality management program. While we have a quality system that covers the lifecycle of our products, quality and safety issues may occur with respect to any of our products. A quality or safety issue may result in adverse inspection reports, warning letters, product recalls (either voluntary or required by the FDA or similar governmental authorities in other countries) or seizures, monetary sanctions, injunctions to halt manufacture and distribution of products, civil or criminal sanctions, costly litigation, refusal of a government to grant approvals and licenses, restrictions on operations or withdrawal of existing approvals and licenses. An inability to address a quality or safety issue in an effective and timely manner may also cause negative publicity, a loss of customer confidence in us or our current or future products, which may result in the loss of sales and difficulty in successfully launching new products. Additionally, Baxter has made and continues to make significant investments in assets, including inventory and property, plant and equipment, which relate to potential new products or modifications to existing products. Product quality or safety issues may restrict the company from being able to realize the expected returns from these investments, potentially resulting in asset impairments in the future.

Unaffiliated third party suppliers provide a number of goods and services to our R&D, clinical and manufacturing organizations. Third party suppliers are required to comply with our quality standards. Failure of a third party supplier to provide compliant raw materials or supplies could result in delays, service interruptions or other quality related issues that may negatively impact our business results.

For more information on regulatory matters currently affecting us, refer to the discussion under the caption entitled "Certain Regulatory Matters" in Item 7 of this Annual Report on Form 10-K.

We are subject to a number of existing laws and regulations, non-compliance with which could adversely affect our business, financial condition and results of operations, and we are susceptible to a changing regulatory environment.

As a participant in the healthcare industry, our operations and products, and those of our customers, are regulated by numerous government agencies, both inside and outside the United States. The impact of this on us is direct to the extent we are subject to these laws and regulations, and indirect in that in a number of situations, even though we may not be directly regulated by specific healthcare laws and regulations, our products must be capable of being used by our customers in a manner that complies with those laws and regulations.

The manufacture, distribution, marketing and use of our products are subject to extensive regulation and scrutiny by FDA and other regulatory authorities globally. Any new product must undergo lengthy and rigorous testing and other extensive, costly and time-consuming procedures mandated by FDA and foreign regulatory authorities. Changes to current products may be subject to vigorous review, including additional 510(k) and other regulatory submissions, and approvals are not certain. Our facilities must be approved and licensed prior to production and remain subject to inspection from time to time thereafter. Failure to comply with the requirements of FDA or other regulatory authorities, including a failed inspection or a failure in our adverse event reporting system, could result in adverse inspection reports, warning letters, product recalls or seizures, monetary sanctions, injunctions to halt the manufacture and distribution of products, civil or criminal sanctions, refusal of a government to grant approvals or licenses, restrictions on operations or withdrawal of existing approvals and licenses. Any of these actions could cause a loss of customer confidence in us and our products, which could adversely affect our sales. The requirements of regulatory authorities, including interpretative guidance, are subject to change and compliance with additional or changing requirements or interpretative guidance may subject the company to further review, result in product launch delays or otherwise increase our costs. For information on current regulatory issues affecting us, please refer to the caption entitled "Certain Regulatory Matters" in Item 7 of this Annual Report on Form 10-K. In connection with these issues, there can be no assurance that additional costs or civil and criminal penalties will not be incurred, that additional regulatory actions with respect to the company will not occur, that the company will not face civil claims for damages from purchasers or users, that substantial additional charges or signi

The sales, marketing and pricing of products and relationships that pharmaceutical and medical device companies have with healthcare providers are under increased scrutiny by federal, state and foreign government agencies. Compliance with the Anti-Kickback Statute, False Claims Act, Food, Drug and Cosmetic Act (including as these laws relate to off-label promotion of products) and other healthcare related laws, as well as competition, data and patient privacy and export and import laws, is under increased focus by the agencies charged with overseeing such activities, including FDA, OIG, DOJ and the Federal Trade Commission. The DOJ and the Securities and Exchange Commission have also increased their focus on the enforcement of the U.S. Foreign Corrupt Practices Act

(FCPA), particularly as it relates to the conduct of pharmaceutical and medical product companies. The FCPA and similar anti-bribery laws generally prohibit companies and their employees, contractors or agents from making improper payments to government officials for the purpose of obtaining or retaining business. Healthcare professionals in many countries are employed by the government and consequently may be considered government officials. Foreign governments have also increased their scrutiny of pharmaceutical and medical product companies' sales and marketing activities and relationships with healthcare providers and competitive practices generally. The laws and standards governing the promotion, sale and reimbursement of our products and those governing our relationships with healthcare providers and governments, including the Sunshine Act enacted under the Patient Protection and Affordable Care Act, can be complicated, are subject to frequent change and may be violated unknowingly.

Additionally, the U.S. Department of the Treasury's Office of Foreign Control and the Bureau of Industry and Security at the U.S. Department of Commerce administer laws and regulations that restrict U.S. persons and, in some instances, non-U.S. persons, in conducting activities, transacting business or making investments in certain countries, governments, entities and individuals subject to U.S. economic sanctions. From time to time, certain of our subsidiaries have limited business dealings in countries subject to these sanctions, including Iran, Sudan, Syria, Cuba, and Russia. These dealings represent an insignificant amount of our consolidated revenues and income but expose us to an increased risk of violating applicable sanctions regulations, which are complex and subject to frequent change. Additional restrictions may be enacted, enforced or interpreted in a way that may adversely affect our operations.

We have compliance programs in place, including policies, training and various forms of monitoring, designed to address the risks discussed above. Nonetheless, these programs and policies may not always protect us from conduct by individual employees that violate these laws. Violations or allegations of violations of these laws may result in large civil and criminal penalties, debarment from participating in government programs, diversion of management time, attention and resources and may otherwise have an adverse effect on our business, financial condition and results of operations. For more information related to the company's ongoing government investigations, please refer to Note 16 in Item 8 of this Annual Report on Form 10-K.

The laws and regulations discussed above are broad in scope and subject to evolving interpretations, which could require us to incur substantial cost associated with compliance or to alter one or more of our sales and marketing practices and may subject us to enforcement actions which could adversely affect our business, financial condition and results of operations.

If reimbursement or other payment for our current or future products is reduced or modified in the United States or abroad, including through the implementation of government-sponsored healthcare reform or other similar actions, cost containment measures, or changes to policies with respect to pricing, taxation or rebates, then our business could suffer.

Sales of our products depend, in part, on the extent to which the costs of our products are paid by both public and private payers. These payers include Medicare, Medicaid, and private health care insurers in the United States and foreign governments and third-party payers outside the United States. Public and private payers are increasingly challenging the prices charged for medical products and services. We may continue to experience continued downward pricing pressures from any or all of these payers which could result in an adverse effect on our business, financial condition and operational results.

Global efforts toward healthcare cost containment continue to exert pressure on product pricing. Governments around the world use various mechanisms to control healthcare expenditures such as price controls, the formation of public contracting authorities, product formularies (lists of recommended or approved products), and competitive tenders which require the submission of a bid to sell products. Sales of our products are dependent, in part, on the availability of reimbursement by government agencies and healthcare programs, as well as insurance companies and other private payers. In much of Europe, Latin America, Asia and Australia, for example, the government provides healthcare at low cost to patients, and controls its expenditures by purchasing products through public tenders, collective purchasing, regulating prices, setting reference prices in public tenders or limiting reimbursement or patient access to certain products. Additionally, austerity measures or other reforms by foreign governments may limit, reduce or eliminate payments for our products and adversely affect both pricing flexibility and demand for our products.

For example, in the United States the Patient Protection and Affordable Care Act (PPACA), which was signed into law in March 2010, includes several provisions which impact our businesses in the United States, including increased Medicaid rebates and an expansion of the 340B Drug Pricing Program which provides certain qualified entities, such as hospitals serving disadvantaged populations, with discounts on the purchase of drugs for outpatient use and an excise tax on the sale of certain drugs. The PPACA reduces Medicare and Medicaid payments to hospitals and other providers, which may cause us to experience downward pricing pressure. Members of Congress and the Executive Branch have made statements suggesting plans to seek repeal of all or portions of the PPACA. Because of the continued uncertainty about the implementation of the PPACA, including the potential for legal challenges or repeal of that legislation, we cannot quantify or predict the likely impact of any change in or replacement of the PPACA on our business and the demand for our products.

As a result of these and other measures, including future measures or reforms that cannot be predicted, reimbursement may not be available or sufficient to allow us to sell our products on a competitive basis. Legislation and regulations affecting reimbursement for our products may change at any time and in ways that may be adverse to us. We cannot predict the impact of these pressures and initiatives, or any negative effects of any additional regulations that may affect our business.

There is substantial competition in the product markets in which we operate.

Although no single company competes with us in all of our businesses, we face substantial competition in both of our segments from international and domestic healthcare and pharmaceutical companies and providers of all sizes, and these competitors often differ across our businesses. Competition is primarily focused on cost-effectiveness, price, service, product performance, and technological innovation.

Competition may increase further as additional companies begin to enter our markets or modify their existing products to compete directly with ours. If our competitors respond more quickly to new or emerging technologies and changes in customer requirements or we do not introduce new versions or upgrades to our product portfolio in response to those requirements, our products may be rendered obsolete or non-competitive. If our competitors develop more effective or affordable products, or achieve earlier patent protection or product commercialization than we do, our operations will likely be negatively affected. If we are forced to reduce our prices due to increased competition, our business could become less profitable. The company's sales could be adversely affected if any of its contracts with GPOs, IDNs or other customers are terminated due to increased competition or otherwise.

If our business development activities are unsuccessful, our business could suffer and our financial performance could be adversely affected.

As part of our long-term strategy, we are engaged in business development activities including evaluating acquisitions, joint development opportunities, technology licensing arrangements and other opportunities. These activities may result in substantial investment of the company's resources. Our success developing products or expanding into new markets from such activities will depend on a number of factors, including our ability to find suitable opportunities for acquisition, investment or alliance; whether we are able to complete an acquisition, investment or alliance on terms that are satisfactory to us; the strength of the other company's underlying technology, products and ability to execute its business strategies; any intellectual property and litigation related to these products or technology; and our ability to successfully integrate the acquired company, business, product, technology or research into our existing operations, including the ability to adequately fund acquired in-process research and development projects and to maintain adequate controls over the combined operations. Certain of these activities are subject to antitrust and competition laws, which laws could impact our ability to pursue strategic transactions and could result in mandated divestitures in the context of proposed acquisitions. If we are unsuccessful in our business development activities, we may be unable to meet our financial targets and our financial performance could be adversely affected.

For more information on recent business development activities, see Note 5 in Item 8 of this Annual Report on Form 10-K.

If we are unable to obtain sufficient components or raw materials on a timely basis or if we experience other manufacturing or supply difficulties, our business may be adversely affected.

The manufacture of our products requires the timely delivery of sufficient amounts of quality components and materials. We manufacture our products in approximately 50 manufacturing facilities around the world. We acquire our components and materials from many suppliers in various countries. We work closely with our suppliers to ensure the continuity of supply but we cannot guarantee these efforts will always be successful. Further, while efforts are made to diversify our sources of components and materials, in certain instances we acquire components and materials from a sole supplier. For most of our components and materials for which a sole supplier is used, we believe that alternative sources of supply exist and have made a strategic determination to use a sole supplier. In very limited instances, however, we do rely upon sole supplier relationships for which no alternatives have currently been identified. Although we do carry strategic inventory and maintain insurance to mitigate the potential risk related to any related supply disruption, there can be no assurance that such measures will be effective. Due to the regulatory environment in which we operate, we may be unable to quickly establish additional or replacement sources for some components or materials. A reduction or interruption in supply, and an inability to develop alternative sources for such supply, could adversely affect our ability to manufacture our products in a timely or cost-effective manner, and our ability to make product sales.

Many of our products are difficult to manufacture. This is due to the complex nature of manufacturing pharmaceuticals, including biologics, and devices, as well as the strict regulatory regime governing our manufacturing operations. Variations in the manufacturing process may result in production failures which could lead to launch delays, product shortage, unanticipated costs, lost revenues and damage to our reputation. A failure to identify and address manufacturing problems prior to the release of products to our customers may also result in a quality or safety issue of the type discussed above.

Some of our products are manufactured at a single manufacturing facility or stored at a single storage site. Loss or damage to a manufacturing facility or storage site due to a natural disaster or otherwise could adversely affect our ability to manufacture sufficient quantities of key products or otherwise deliver products to meet customer demand or contractual requirements which may result in a loss of revenue and other adverse business consequences. Because of the time required to approve and license a manufacturing facility a third party manufacturer may not be available on a timely basis to replace production capacity in the event we lose manufacturing capacity or products are otherwise unavailable due to natural disaster, regulatory action or otherwise.

If we are unable to protect our patents or other proprietary rights, or if we infringe the patents or other proprietary rights of others, our competitiveness and business prospects may be materially damaged.

Patent and other proprietary rights are essential to our business. Our success depends to a significant degree on our ability to obtain and enforce patents and licenses to patent rights, both in the United States and in other countries. We cannot guarantee that pending patent applications will result in issued patents, that patents issued or licensed will not be challenged or circumvented by competitors, that our patents will not be found to be invalid or that the intellectual property rights of others will not prevent the company from selling certain products or including key features in the company's products.

The patent position of a healthcare company is often uncertain and involves complex legal and factual questions. Significant litigation concerning patents and products is pervasive in our industry. Patent claims include challenges to the coverage and validity of our patents on products or processes as well as allegations that our products infringe patents held by competitors or other third parties. A loss in any of these types of cases could result in a loss of patent protection or the ability to market products, which could lead to a significant loss of sales, or otherwise materially affect future results of operations. We also rely on trademarks, copyrights, trade secrets and know-how to develop, maintain and strengthen our competitive positions. Third parties may know, discover or independently develop equivalent proprietary information or techniques, or they may gain access to our trade secrets or disclose our trade secrets to the public.

Although our employees, consultants, parties to collaboration agreements and other business partners are generally subject to confidentiality or similar agreements to protect our confidential and proprietary information, these agreements may be breached, and we may not have adequate remedies for any breach. To the extent that our employees, consultants, parties to collaboration agreements and other business partners use intellectual property owned by others in their work for us, disputes may arise as to the rights in related or resulting know-how and inventions.

Furthermore, our intellectual property, other proprietary technology and other sensitive company data is potentially vulnerable to loss, damage or misappropriation from system malfunction, computer viruses, unauthorized access to our data or misappropriation or misuse thereof by those with permitted access and other events. While we have invested to protect our intellectual property and other data, and continue to work diligently in this area, there can be no assurance that our precautionary measures will prevent breakdowns, breaches, cyber incidents or other events. Such events could have a material adverse effect on our reputation, business, financial condition or results of operations.

Misappropriation or other loss of our intellectual property from any of the foregoing would have an adverse effect on our competitive position and may cause us to incur substantial litigation costs.

We are subject to risks associated with doing business globally.

Our operations are subject to risks inherent in conducting business globally and under the laws, regulations and customs of various jurisdictions and geographies. These risks include changes in exchange controls and other governmental actions, loss of business in government and public tenders that are held annually in many cases, increasingly complex labor environments, availability of raw materials, changes in taxation, export control restrictions, changes in or violations of U.S. or local laws, including the FCPA and the United Kingdom Bribery Act, dependence on a few government entities as customers, pricing restrictions, economic and political instability (including instability as it relates to the Euro and currencies in certain emerging market countries), disputes between countries, diminished or insufficient protection of intellectual property, and disruption or destruction of operations in a significant geographic region regardless of cause, including war, terrorism, riot, civil insurrection or social unrest. Failure to comply with, or material changes to, the laws and regulations that affect our global operations could have an adverse effect on our business, financial condition or results of operations.

The 2016 referendum by British voters to exit the European Union (EU) (commonly known as Brexit) has created uncertainties affecting business operations in the EU. The UK government is expected to initiate a process to withdraw from the EU in the coming months. A withdrawal could result in the deterioration of economic conditions, volatility in currency exchange rates (as evidenced by the deterioration in the value of the British pound as compared to the U.S. dollar following the Brexit vote), and increased regulatory complexities. These outcomes could have an adverse effect on our business, financial condition or results of operations.

Changes in foreign currency exchange rates and interest rates could have a material adverse effect on our operating results and liquidity.

We generate the majority of our revenue and profit outside the United States. As a result, our financial results may be adversely affected by fluctuations in foreign currency exchange rates. We cannot predict with any certainty changes in foreign currency exchange rates or our ability to mitigate these risks. We may experience additional volatility as a result of inflationary pressures and other macroeconomic factors in certain emerging market countries. We are also exposed to changes in interest rates, and our ability to access the money markets and capital markets could be impeded if adverse liquidity market conditions occur. A discussion of the financial impact of foreign exchange rate and interest rate fluctuations, and the ways and extent to which we attempt to mitigate such impact is contained under the caption "Financial Instrument Market Risk" in Item 7 of this Annual Report on Form 10-K.

Changes in tax laws or exposure to additional income tax liabilities may have a negative impact on our operating results.

Tax policy reform continues to be a topic of discussion in the United States. Members of the newly installed U.S. Congress, including the Speaker of the House Paul Ryan, have identified comprehensive tax reform as a priority for 2017. A significant change to the tax system in the United States, including changes to the taxation of international income or imported product, could have an adverse effect upon our results of operations. Because we operate in multiple income tax jurisdictions both inside and outside the United States, cross border transactions among our affiliates are a significant part of the manner in which we operate. Although we believe that we transact intercompany business in accordance with arms-length principles, taxing authorities may audit us from time to time, disagree with certain positions we have taken and assess additional taxes. We regularly assess the likely outcomes of these audits in order to determine the appropriateness of our tax provision. However, we may not accurately predict the outcome of these audits, and as a result the actual outcome of these audits may have an adverse impact on our financial results. For more information on ongoing audits, see Note 15 in Item 8 of this Annual Report on Form 10-K.

We are increasingly dependent on information technology systems and subject to privacy and security laws, and our systems and infrastructure face certain risks, including from cyber security breaches and data leakage.

We increasingly rely upon technology systems and infrastructure. Our technology systems are potentially vulnerable to breakdown or other interruption by fire, power loss, system malfunction, unauthorized access and other events. Likewise, data privacy breaches by employees and others with both permitted and unauthorized access to our systems and products may pose a risk that sensitive data (including protected health information (PHI)) may be exposed to unauthorized persons or to the public, or may be permanently lost. The increasing use and evolution of technology, including cloud-based computing, creates additional opportunities for the unintentional dissemination of information, intentional destruction of confidential information stored in our systems, products or in non-encrypted portable media or storage devices. We could also experience a business interruption, information theft of confidential information, or reputational damage from industrial espionage attacks, malware or other cyber incidents, which may compromise our system infrastructure or lead to data leakage, either internally or at our third-party providers or other business partners. As our products continue to evolve, third-parties may attempt to access or obtain proprietary information from our products or systems. Additionally, we must comply with numerous federal and state laws and regulations governing the collection, dissemination, access, use, security and PHI, including The Health Insurance Portability and Accountability Act of 1996 and its implementing privacy and security regulations. While we have invested heavily in the protection of data and information technology and in related training, there can be no assurance that our efforts will prevent significant breakdowns, breaches in our systems or other cyber incidents or ensure compliance with all applicable security and privacy laws, regulations and standards, including with respect to third-party service providers that utilize sensitive personal information, including PHI, on our behalf. Any such breakdown, breach, incident or failure to comply could have a material adverse effect upon our reputation, business, operations or financial condition. In addition, significant implementation issues may arise as we continue to consolidate and outsource certain computer operations and application support activities.

If we fail to attract and retain key employees our business may suffer.

Our ability to compete effectively depends on our ability to attract and retain key employees, including people in senior management, sales, marketing and research positions. Competition for top talent in healthcare can be intense. Our ability to recruit and retain such talent will depend on a number of factors, including hiring practices of our competitors, compensation and benefits, work location, work environment and industry economic conditions. If we cannot effectively recruit and retain qualified employees, our business could suffer.

We are subject to a number of pending lawsuits.

We are a defendant in a number of pending lawsuits. In addition, we may be named as a defendant in future patent, product liability or other lawsuits. These current and future matters may result in a loss of patent protection, reduced revenue, significant liabilities and

diversion of our management's time, attention and resources. Given the uncertain nature of litigation generally, we are not able in all cases to estimate the amount or range of loss that could result from an unfavorable outcome in these current matters. In view of these uncertainties, the outcome of these matters may result in charges in excess of any established reserves, and, to the extent available, liability insurance. We also continue to be self-insured with respect to product liability claims. The absence of third-party insurance coverage for current or future claims increases our potential exposure to unanticipated claims and adverse decisions. Protracted litigation, including any adverse outcomes, may have an adverse impact on the business, operations or financial condition of the company. Even claims without merit could subject us to adverse publicity and require us to incur significant legal fees. See Note 16 in Item 8 of this Annual Report on Form 10-K for more information regarding current lawsuits.

Current or worsening economic conditions may adversely affect our business and financial condition.

The company's ability to generate cash flows from operations could be affected if there is a material decline in the demand for the company's products, in the solvency of its customers or suppliers, or deterioration in the company's key financial ratios or credit ratings. Current or worsening economic conditions may adversely affect the ability of our customers (including governments) to pay for our products and services, and the amount spent on healthcare generally. This could result in a decrease in the demand for our products and services, declining cash flows, longer sales cycles, slower adoption of new technologies and increased price competition. These conditions may also adversely affect certain of our suppliers, which could cause a disruption in our ability to produce our products. We continue to do business with foreign governments in certain countries, including Greece, Spain, Portugal, and Italy, which have experienced deterioration in credit and economic conditions. As of December 31, 2016, the company's net accounts receivable from the public sector in Greece, Spain, Portugal and Italy totaled \$137 million. While global economic conditions have not significantly impacted the company's ability to collect receivables, liquidity issues in certain countries have resulted, and may continue to result, in delays in the collection of receivables and credit losses. These conditions may also impact the stability of the Euro or Yuan. For more information on accounts receivable and credit matters with respect to certain of these countries, refer to the discussion under the caption entitled "Credit Facilities, Access to Capital and Credit Ratings" in Item 7 of this Annual Report on Form 10-K.

We may incur operational difficulties or be exposed to claims and liabilities as a result of the separation and distribution.

On July 1, 2015, we distributed approximately 80.5% of the outstanding shares of Baxalta common stock to Baxter stockholders in connection with the separation of our biopharmaceuticals business. We disposed of our remaining 19.5% stake in Baxalta (Retained Shares) in 2016, in connection with a series of transactions including debt-for-equity exchanges, an equity-for-equity exchange and a contribution to our U.S. pension plan (Retained Shares Transactions). Shire plc (Shire) acquired Baxalta in June 2016, after completion of the last Retained Shares Transaction. In connection with the July 2015 distribution, we entered into a separation and distribution agreement and various other agreements (including a transition services agreement, a tax matters agreement, a long term services agreement, a manufacturing and supply agreement, an employee matters agreement, a trademark license agreement, a Galaxy license agreement, an international commercial operations agreement, a shareholders' and registration rights agreement and certain other commercial agreements) with Baxalta. These agreements govern the separation and distribution and the relationship between the companies going forward, including with respect to potential tax-related losses associated with the separation and distribution and the Retained Shares Transactions. They also provide for the performance of services by each company for the benefit of the other for a period of time (including under the manufacturing and supply agreement pursuant to which Shire now manufactures and sells certain products and materials to us).

The separation and distribution agreement provides for indemnification obligations designed to make Baxalta financially responsible for many liabilities that may exist relating to its business activities, whether incurred prior to or after the distribution, including any pending or future litigation. It is possible that a court would disregard the allocation agreed to between us and Baxalta and require us to assume responsibility for obligations allocated to Baxalta. Third parties could also seek to hold us responsible for any of these liabilities or obligations, and the indemnity rights we have under the separation and distribution agreement may not be sufficient to fully cover all of these liabilities and obligations. Even if we are successful in obtaining indemnification, we may have to bear costs temporarily. In addition, our indemnity obligations to Baxalta may be significant. These risks could negatively affect our business, financial condition or results of operations.

The separation of Baxalta continues to involve a number of risks, including, among other things, the indemnification risks described above and the potential that management's and our employees' attention will be significantly diverted by the provision of transitional services. Certain of the agreements described above provide for the performance of services by each company for the benefit of the other for a period of time. Shire may elect to extend the term for which we provide services to Baxalta under these agreements. If Baxalta is unable to satisfy its obligations under these agreements, including its indemnification obligations, we could incur losses. These arrangements could also lead to disputes over rights to certain shared property and rights and over the allocation of costs and revenues for products and operations. Our inability to effectively manage the separation activities and related events could adversely affect our business, financial condition or results of operations.

There could be significant liability if the separation and distribution or any Retained Shares Transaction is determined to be a taxable transaction. Baxalta has indemnified us for certain potential liabilities that may arise, and such indemnification obligation is guaranteed by Shire, but Baxalta and Shire may be unable to satisfy their indemnification obligations to us in the future.

The separation and distribution and the Retained Shares Transactions (collectively, the Baxter Transactions) qualify for tax-free treatment to Baxter and its stockholders under the Internal Revenue Code of 1986, as amended (the Code). Completion of the separation and distribution was conditioned upon, among other things, the receipt of a private letter ruling from the IRS regarding certain issues relating to the tax-free treatment of the Baxter Transactions. Although the IRS private letter ruling is generally binding on the IRS, the continuing validity of such ruling is subject to the accuracy of factual representations and assumptions made in the ruling. Completion of the distribution was also conditioned upon Baxter's receipt of a tax opinion from KPMG LLP regarding certain aspects of the Baxalta spin-off not covered by the IRS private letter ruling. The opinion was based upon various factual representations and assumptions, as well as certain undertakings made by Baxter and Baxalta. If any of the factual representations or assumptions in the IRS private letter ruling or tax opinion is untrue or incomplete in any material respect, if any undertaking is not complied with, or if the facts upon which the IRS private letter ruling or tax opinion are based are materially different from the actual facts relating to the Baxter Transactions, the opinion or IRS private letter ruling may not be valid. Moreover, opinions of a tax advisor are not binding on the IRS. As a result, the conclusions expressed in the opinion of a tax advisor could be successfully challenged by the IRS.

If the Baxter Transactions are determined to be taxable, Baxter and its stockholders could incur significant tax liabilities. Pursuant to the tax matters agreement, Baxalta agreed to indemnify us for certain tax-related losses incurred if Baxalta's actions cause the separation and distribution and certain related transactions to fail to qualify for tax-free status under the applicable provisions of the Code.

In anticipation of the proposed Baxalta — Shire merger (the Merger), we entered into a letter agreement with Shire and Baxalta (the Letter Agreement). Under the Letter Agreement, Baxalta agreed to indemnify, and Shire agreed to guarantee such indemnity to, Baxter and each of its affiliates and each of their respective officers, directors and employees against certain tax-related losses attributable to or resulting from (in whole or in part) the merger as further described in the Letter Agreement. If the Baxter Transactions are determined to be taxable as a result (in whole or in part) of the merger (for example, if the merger is deemed to be part of a plan (or series of related transactions) that includes the Baxter Transactions), Baxter and its stockholders could incur significant tax liabilities. Although Baxalta and Shire may be required to indemnify Baxter under the tax matters agreement and the Letter Agreement for any such tax liabilities incurred by Baxter, there can be no assurance that the indemnity from Baxalta or the guarantee thereof by Shire will be sufficient to protect us against all or a part of the amount of such liabilities, or that either Baxalta or Shire will be able to fully satisfy their respective obligations.

Even if we ultimately succeed in recovering from Baxalta or Shire any amounts for which we are held liable, we may be temporarily required to bear these costs ourselves, which could negatively affect our business, results of operations and financial condition.

Item 1B. Unresolved Staff Comments.

None.

Item 2. Properties.

The company's corporate offices are owned and located at One Baxter Parkway, Deerfield, Illinois 60015.

Baxter owns or has long-term leases on all of its manufacturing facilities. The company's principal manufacturing facilities by segment are listed below:

Business	Location	Owned/Leased
Hospital Products		
	Shanghai, China	Owned
	Tianjin, China	Owned
	Cartago, Costa Rica	Owned
	Haina, Dominican Republic	Leased
	Halle, Germany	Owned
	Guayama, Puerto Rico	Owned
	Jayuya, Puerto Rico	Leased
	Aibonito, Puerto Rico	Leased
	Sabinanigo, Spain	Owned
	San Vittore, Switzerland	Owned
	Tunis, Tunisia	Owned
	Elstree, United Kingdom	Leased
	Thetford, United Kingdom	Owned
	Hayward, California	Leased
	Irvine, California	Owned
	Englewood, Colorado	Leased
	Round Lake, Illinois	Owned
	Bloomington, Indiana	Owned/Leased(1)
	Cleveland, Mississippi	Leased
	St. Paul, Minnesota	Leased
	Medina, New York	Leased
enal		
	Guangzhou, China	Owned
	Prerov, Czech Republic	Leased
	Meyzieu, France	Owned
	Hechingen, Germany	Leased
	Rostock, Germany	Leased
	Medolla, Italy	Owned
	Sondalo, Italy	Owned
	Miyazaki, Japan	Owned
	Tijuana, Mexico	Owned
	Lund, Sweden	Leased
	Amata, Thailand	Owned
	Liverpool, United Kingdom	Leased Owned
	Opelika, Alabama	Leased
hand (Hamital Duadwata and Danal)	Brooklyn Park, Minnesota	Leased
hared (Hospital Products and Renal)	To an anhhia Avetualia	Owned
	Toongabbie, Australia	
	Lessines, Belgium	Owned
	Sao Paulo, Brazil	Owned
	Alliston, Canada Suzhou, China	Owned Owned
	Cali, Colombia	Owned
	Manesar, India	Owned
	Castlebar, Ireland	Owned
	Grosotta, Italy	Owned
	Marsa, Malta	Owned
	Cuernavaca, Mexico	Owned
	PESA, Mexico	Owned
	Canlubang, Philippines	Leased
	Lublin, Poland	Owned/Leased(1)
	Woodlands, Singapore	Owned/Leased(1) Owned/Leased(2)
	Mountain Home, Arkansas	Owned/Leased(1)
	North Cove, North Carolina	Owned

⁽¹⁾ Includes both owned and leased facilities.

⁽²⁾ Baxter owns the facility located at Woodlands, Singapore and leases the property upon which it rests.

The company also owns or operates shared distribution facilities throughout the world. In the United States and Puerto Rico, there are six shared distribution facilities with the principal facilities located in Memphis, Tennessee; Catano, Puerto Rico; North Cove, North Carolina; and Round Lake, Illinois. Internationally, we have more than 100 shared distribution facilities located in Argentina, Australia, Benelux, Brazil, Brunei, Canada, Chile, China, Colombia, Costa Rica, the Czech Republic, Ecuador, France, Germany, Greece, Guatemala, Hong Kong, India, Ireland, Italy, Japan, Korea, Mexico, New Zealand, Panama, the Philippines, Poland, Portugal, Russia, Singapore, Spain, Sweden, Switzerland, Thailand, Turkey, the United Arab Emirates, the United Kingdom and Venezuela.

The company continually evaluates its plants and production lines and believes that its current facilities plus any planned expansions are generally sufficient to meet its expected needs and expected near-term growth. Expansion projects and facility closings will be undertaken as necessary in response to market needs.

Item 3. Legal Proceedings.

Incorporated by reference to Note 16 in Item 8 of this Annual Report on Form 10-K.

Item 4. Mine Safety Disclosures.

Not Applicable.

Executive Officers of the Registrant

As of February 23, 2017, the following serve as Baxter's executive officers:

José E. Almeida, age 54, is Chairman and Chief Executive Officer, having served in that capacity since January 2016. Between October 2015 and January 2016, Mr. Almeida served as an executive officer of the company. Previously, he served as an operating executive to the Carlye Group L.P. from May 2015 until October 2015. Previously, he served as the Chairman, President and Chief Executive Officer of Covidien plc (Covidien) from March 2012 to January 2015, prior to Medtronic plc's acquisition of Covidien, and President and Chief Executive Officer of Covidien from July 2011 to March 2012. Mr. Almeida served in other executive roles with Covidien (formerly Tyco Healthcare) between April 2004 and June 2011.

Giuseppe Accogli, age 46, is Corporate Vice President and President, Renal. Mr. Accogli joined the company in 2007 as renal business unit director in Italy, and assumed positions of increasing responsibility with the Renal business in Europe, including head of the EMEA region for renal from 2013 to 2015. Prior to joining Baxter, he served as business unit manager and sales and marketing manager for Medtronic, Inc. (Italy) from 2004 to 2007. From 1996 to 2004, he held a series of positions in Europe with Tyco Healthcare – Covidien Ltd., including marketing director and group product director.

Brik V. Eyre, age 53, is Corporate Vice President and President, Hospital Products. Mr. Eyre joined the company in 2008 as general manager for BioPharma Solutions, Baxter's manufacturing and contract services business. He later served as general manager for our U.S. medication delivery business and most recently he was Corporate Vice President and President, Renal. Prior to joining Baxter, he held a variety of senior management positions at Cardinal Health, Inc., including president of Cardinal's PreSource Products and Services business.

Jeanne K. Mason, Ph.D., age 61, is Corporate Vice President, Human Resources. Prior to joining Baxter in May 2006, Dr. Mason was with General Electric from 1988, holding various leadership positions, the most recent of which was with GE Insurance Solutions where she was responsible for global human resource functions.

Scott Pleau, age 51, is Corporate Vice President, Operations. Prior to joining Baxter in June 2016, Mr. Pleau served as vice president of operations for medical devices at Medtronic plc from 2015 to 2016, and at Covidien from 2013 to 2015, prior to Medtronic plc's acquisition of Covidien. From 1995 to 2013, he held several key operations positions at Covidien, including vice president of operations, surgical solutions; vice president of operations, vascular therapies & medical supplies; vice president of engineering; and director of operational quality.

James K. Saccaro, age 44, is Corporate Vice President and Chief Financial Officer and has served in that capacity since June 2015. Mr. Saccaro was Senior Vice President and Chief Financial Officer at Hill-Rom Corporation from December 2013 to July 2014 prior to rejoining Baxter in July 2014 as Special Advisor to the Chief Executive Officer. Prior to that, Mr. Saccaro served as Corporate Vice President and Treasurer of Baxter from 2011 to 2013. He originally joined the company in 2002 as manager of strategy for the company's former BioScience business, and from there moved onto positions of increasing responsibility, including Vice President of

Financial Planning and Vice President of Finance for the company's operations in Europe, Middle East and Africa. Prior to Baxter, he held strategy and business development positions at Clear Channel Communications and the Walt Disney Company.

Marcus Schabacker, M.D., Ph.D., age 53, is Corporate Vice President and Chief Scientific Officer. Dr. Schabacker joined the company in 2011. Prior to his current role, Dr. Schabacker served as Vice President, R&D, Medical Products. Dr. Schabacker held the position of Senior Vice President and Chief Scientific Officer at ConvaTec, Inc. before joining the company. His previous roles include Corporate Vice President R&D at B. Braun Medical and Senior Medical Officer at Mafikeng General Hospital, South Africa.

David P. Scharf, age 49, is Corporate Vice President and General Counsel, having served in this capacity since August 2009. Mr. Scharf joined Baxter in July 2005 and served in advancing leadership roles within the legal department. Prior to joining Baxter, Mr. Scharf was with Guidant Corporation from 2002, in roles of increasing responsibility.

Paul Vibert, age 57, is Corporate Vice President and President, International. Mr. Vibert joined the company in January 2008 as Vice President of Business Development for Asia Pacific. He also served as regional general manager for China and Hong Kong for two years before moving to Ferring Pharmaceuticals, as Senior Vice President, Asia Pacific, from May 2011 to May 2013. He returned to Baxter in May 2013 as President of Western Europe, and assumed his current role in January 2015. Prior to joining Baxter in 2008, Vibert spent 19 years with Abbott Laboratories, where he held various leadership positions.

All executive officers hold office until the next annual election of officers and until their respective successors are elected and qualified.

Item 5. Market for Registrant's Common Equity, Related Stockholder Matters and Issuer Purchases of Equity Securities.

On July 25, 2012, the company announced that its Board of Directors authorized the company to repurchase up to \$2.0 billion of its common stock on the open market or in private transactions. The Board of Directors increased this authority by \$1.5 billion in November 2016. During 2016, the company repurchased approximately 6.3 million shares for \$287 million in cash pursuant to this authority. The remaining authorization under this program totaled approximately \$1.7 billion at December 31, 2016. This program does not have an expiration date.

Additional information required by this item is incorporated by reference to Note 18 in Item 8 of this Annual Report on Form 10-K.

Performance Graph

The following graph compares the change in Baxter's cumulative total shareholder return (including reinvested dividends) on Baxter's common stock with the Standard & Poor's 500 Composite Index and the Standard & Poor's 500 Health Care Index over the past five years. Performance through June 30, 2015 has been adjusted for the Baxalta separation which occurred on July 1, 2015.



Item 6. Selected Financial Data.

See Note 1 of Item 8 for additional details regarding basis of presentation.

as of or for the years ended I	December 31	20162,1	20153,1	20144,1	20135,1	20126,1
Operating Results	Net sales	\$ 10,163	9,968	10,719	9,413	8,626
(in millions)	Income from continuing operations	\$ 4,966	393	457	315	663
	Income (loss) from discontinued operations, net of tax	\$ (1)	575	2,040	1,697	1,663
	Net income	\$ 4,965	968	2,497	2,012	2,326
Balance Sheet	Capital expenditures, continuing operations	\$ 719	911	925	706	622
Information	Total assets	\$ 15,546	20,962	26,138	25,224	20,390
(in millions)	Long-term debt and lease obligations	\$ 2,779	3,922	7,331	8,126	5,580
Common Stock Information	Weighted-average number of common shares outstanding					
	Basic	546	545	542	543	551
	Diluted	551	549	547	549	556
	Income from continuing operations per common share					
	Basic	\$ 9.10	0.72	0.84	0.58	1.20
	Diluted	\$ 9.01	0.72	0.83	0.57	1.19
	Income from discontinued operations per common share					
	Basic	\$ (0.01)	1.06	3.77	3.12	3.02
	Diluted	\$ 0.00	1.04	3.73	3.09	2.99
	Net income per common share					
	Basic	\$ 9.09	1.78	4.61	3.70	4.22
	Diluted	\$ 9.01	1.76	4.56	3.66	4.18
	Cash dividends declared per common share	\$ 0.505	1.270	2.050	1.920	1.570

Refer to the notes to the consolidated financial statements for information regarding other charges and income items.

- Income from continuing operations included charges totaling \$200 million for business optimization, \$111 million related to the Baxalta separation and \$130 million related to Baxter's July 2015 tender offer for certain outstanding indebtedness. Also included were benefits of \$28 million primarily related to adjustments to the COLLEAGUE and SIGMA SPECTRUM infusion pump reserves, \$52 million related to a litigation settlement in which Baxter was the beneficiary and \$20 million relating to the reversal of contingent consideration milestone liabilities.
- Income from continuing operations included charges totaling \$138 million for business optimization, \$68 million for SIGMA Spectrum Infusion Pump product remediation efforts, \$11 million related to the Baxalta separation and \$3 million to account for an additional year of the Branded Prescription Drug Fee in accordance with final regulations issued by the Internal Revenue Service. Also included were benefits of \$1 million related to third-party recoveries and reversals of prior reserves.
- Income from continuing operations included charges totaling \$148 million for business optimization, \$17 million primarily related to remediation efforts associated with modifications to the SIGMA Spectrum Infusion Pump in conjunction with re-filing for 510(k) clearance, \$255 million related to the acquisition and integration of Gambro and losses from the derivative instruments used to hedge the anticipated foreign currency cash outflows and \$25 million related to an upfront payment associated with one of the company's collaboration arrangements. Also included were benefits of \$3 million related to tax and legal reserves associated with VAT matters in Turkey.
- Income from continuing operations included charges totaling \$106 million for business optimization, \$15 million primarily related to business development, and \$170 million primarily related to pension settlement charges and other pension-related items. Also included were benefits of \$23 million primarily related to an adjustment to the COLLEAGUE infusion pump reserve when the company substantially completed its recall activities in the United States and \$91 million for gains related to a decrease in the estimated fair value of acquisition-related contingent payment liabilities.

Income from continuing operations included charges totaling \$409 million for business optimization, \$54 million related to the Baxalta separation, \$149 million of debt extinguishment costs related to the March 2016 debt-for-equity exchange for certain company indebtedness and certain debt redemptions, \$51 million for impairment primarily related to developed technology and \$9 million related to the settlement of an income tax matter in the company's non-wholly owned joint venture in Turkey. Also included were net realized gains of \$4.4 billion related to the Baxalta Retained Shares transactions and a benefit of \$18 million primarily related to adjustments to the COLLEAGUE and SIGMA SPECTRUM infusion pump reserves.

Item 7. Management's Discussion and Analysis of Financial Condition and Results of Operations.

The following commentary should be read in conjunction with the consolidated financial statements and accompanying notes.

EXECUTIVE OVERVIEW

Description of the Company and Business Segments

Baxter International Inc., through its subsidiaries, provides a broad portfolio of essential renal and hospital products, including acute and chronic dialysis; sterile IV solutions; infusion systems and devices; parenteral nutrition therapies; premixed and oncolytic injectables; biosurgery products and anesthetics; drug reconstitution systems; and pharmacy automation, software and services. The company's global footprint and critical nature of its products and services play a key role in expanding access to healthcare in emerging and developed countries. These products are used by hospitals, kidney dialysis centers, nursing homes, rehabilitation centers, doctors' offices and by patients at home under physician supervision.

Separation of Baxalta Incorporated

On July 1, 2015, Baxter completed the distribution of approximately 80.5% of the outstanding common stock of its biopharmaceuticals business, Baxalta Incorporated (Baxalta), to Baxter stockholders (the Distribution). As a result of the separation, the operating results of Baxalta have been reflected as discontinued operations for the years ended December 31, 2016, 2015, and 2014. Refer to Note 2 in Item 8 for additional information regarding the separation of Baxalta. Unless otherwise stated, financial results herein reflect continuing operations.

Segments

Baxter operates under two reportable segments, Hospital Products and Renal. Refer to Note 14 in Item 8 for additional information regarding the company's segments.

The segments and a description of their products and services are as follows:

The Hospital Products business manufactures sterile intravenous (IV) solutions and administration sets, premixed drugs and drug-reconstitution systems, prefilled vials and syringes for injectable drugs, parenteral nutrition therapies, infusion pumps, inhalation anesthetics, and biosurgery products. The business also provides products and services related to pharmacy compounding, and drug formulation; sterile IV solutions; infusion systems and devices; parenteral nutrition therapies; premixed and oncolytic injectables; biosurgery products and anesthetics; drug reconstitution systems; and pharmacy automation, software and services.

The Renal business offers a comprehensive portfolio to meet the needs of patients with end-stage renal disease, or irreversible kidney disease and acute kidney injuries, including technologies and therapies for peritoneal dialysis (PD), hemodialysis (HD), continuous renal replacement therapy (CRRT) and additional dialysis services.

Baxter has approximately 48,000 employees and conducts business in over 100 countries. The company generates approximately 60% of its revenues outside the United States, and maintains approximately 50 manufacturing facilities and over 100 distribution facilities in the United States, Europe, Asia-Pacific, Latin America and Canada.

Financial Results

Baxter's global net sales totaled \$10.2 billion in 2016, an increase of 2% over 2015, including an unfavorable foreign currency impact of two percentage points. International sales totaled \$5.9 billion in 2016, a decrease of 1% compared to 2015, including an unfavorable foreign currency impact of four percentage points. Sales in the United States totaled \$4.3 billion in 2016, an increase of 6% compared to 2015.

Baxter's income from continuing operations for 2016 totaled \$5.0 billion or \$9.01 per diluted share, compared to \$393 million, or \$0.72 per diluted share, in the prior year. Income from continuing operations in 2016 included special items which resulted in a net increase to income from continuing operations of \$3.9 billion, or \$7.05 per diluted share. Income from continuing operations in 2015 included special items which resulted in a net reduction to income from continuing operations of \$362 million, or \$0.66 per diluted share. The company's special items are discussed further in the Results of Operations section below.

Baxter's financial results included R&D expenses totaling \$647 million in 2016, which reflects the company's focus on balancing increased investments to support the company's new product pipeline with efforts to optimize overall R&D spending through continuous evaluation of the portfolio.

The company's financial position remains strong, with operating cash flows from continuing operations totaling \$1.6 billion in 2016. The company has continued to execute on its disciplined capital allocation framework, which is designed to optimize stockholder value creation through reinvestment in the businesses, dividends and targeted share repurchases, as well as acquisitions and other business development initiatives as discussed in the Strategic Objectives section below.

Capital investments totaled \$719 million in 2016 as the company continues to invest across its businesses to support future growth, including additional investments in support of new and existing product capacity expansions. The company's investments in capital expenditures in 2016 were focused on projects that improve production efficiency and enhance manufacturing capabilities to support its strategy of geographic expansion with select investments in growing markets.

The company also continued to return value to its stockholders in the form of dividends. During 2016, the company paid cash dividends to its shareholders totaling \$268 million. Additionally, in 2016 the company repurchased 17.8 million shares through cash repurchases and an equity-for-equity exchange of Retained Shares for outstanding Baxter shares.

Strategic Objectives

Baxter continues to focus on several key objectives to successfully execute its long-term strategy to achieve sustainable growth and deliver enhanced stockholder value. Baxter's diversified and broad portfolio of medical products that treat life-threatening acute or chronic conditions and its global presence are core components of the company's strategy to achieve these objectives. The company is focused on three strategic factors as part of its pursuit of industry leading performance: optimizing its core portfolio globally; operational excellence focused on streamlining the cost structure and enhancing operational efficiency; and following a disciplined and balanced approach to capital allocation.

Optimizing the Core Portfolio Globally

Baxter has categorized its product portfolio into four strategic business groupings. Those groupings include core growth, core return on capital, maintain or manage differently and strategic bets. Within the core growth grouping, Baxter looks to invest for long-term, higher margin growth. Baxter looks to optimize its return on investment and to maintain or enhance its market position with its core return on capital products. Maintain or manage differently products are those for which Baxter looks to sustain or reposition its underlying investment. Finally, the strategic bet grouping includes products for which Baxter is evaluating its market position and investment strategy. These products cover mature and emerging markets. While Baxter has made an initial assignment of each of its product categories to one of the business groupings described above, Baxter continues to evaluate each product category's placement in light of shifting market dynamics and company priorities and may reassign a product category into a different business grouping from time to time.

As part of this portfolio review, Baxter seeks to optimize its position in product areas where the company has a stable, profitable business model, identify and alter investments in products that have reached the end of their life cycles or with respect to which market positions have evolved unfavorably. In the course of doing so, Baxter expects to continue to reallocate capital to more promising opportunities or business groupings, as described above.

As part of this strategy, Baxter is shifting its investments to drive innovation where it has compelling opportunities to serve patients and healthcare professionals while advancing the business and will accelerate the pace in bringing these advances to market. Baxter is in the midst of launching more than 100 products by 2020 in such areas as chronic and acute renal care; smart pump technology; hospital pharmaceuticals and nutritionals; surgical sealants, and more. These comprise a mix of entirely new offerings, marked improvements on existing technologies, and the expansion of current products into new geographies.

Operational Excellence

As part of its pursuit of improved margin performance, Baxter is working to optimize its cost structure, consistent with its emergence as a stand-alone medical products company and as such is critically assessing optimal support levels in light of the company's ongoing portfolio optimization efforts.

The company intends to continue to actively manage its cost structure to help ensure it is committing resources to the highest value uses. Such high value activities include supporting innovation, building out the portfolio, expanding patient access and accelerating growth for the company's stockholders.

Baxter has undertaken a comprehensive review of all aspects of its operations and has already begun to implement changes in line with its business goals.

Maintaining Disciplined and Balanced Capital Allocation

Baxter's capital allocation strategies include the following:

- reinvest in the business, by funding opportunities that are positioned to deliver sustainable growth, support the company's innovation efforts and improve margin performance;
- · return capital to stockholders through stock dividends, to meaningfully increase with earnings growth;
- · targeted share repurchases; and
- identify and pursue accretive M&A opportunities that generate returns above targeted thresholds.

Responsible Corporate Citizen

The company strives for continued growth and profitability, while furthering its focus on acting as a responsible corporate citizen. At Baxter, sustainability means creating lasting social, environmental and economic value by addressing the needs of the company's wide-ranging stakeholder base. Baxter's comprehensive sustainability program is focused on areas where the company is uniquely positioned to make a positive impact. Priorities include providing employees a safe, healthy and inclusive workplace, fostering a culture that drives integrity, strengthening access to healthcare, enhancing math and science education, and driving environmental performance across the product life cycle including development, manufacturing and transport. Baxter and the Baxter International Foundation provide financial support and product donations in support of critical needs, from assisting underserved communities to providing emergency relief for countries experiencing natural disasters.

Throughout 2016 the company continued to implement a range of water conservation strategies and facility-based energy saving initiatives. In the area of product stewardship and life cycle management, Baxter is pursuing efforts such as sustainable design and reduced packaging. Baxter is also responding to the challenges of climate change through innovative greenhouse gas emissions-reduction programs, such as shifting to less carbon-intensive energy sources in manufacturing and transport. Additionally, the company developed new long-term goals to drive continued environmental stewardship while creating healthier, more sustainable communities where Baxter employees work and live.

Risk Factors

The company's ability to sustain long-term growth and successfully execute the strategies discussed above depends in part on the company's ability to manage within an increasingly competitive and regulated environment and to address the other risk factors described in Item 1A of this Annual Report on Form 10-K.

RESULTS OF OPERATIONS

Special Items

The following table provides a summary of the company's special items and the related impact by line item on the company's results of continuing operations for 2016, 2015 and 2014.

years ended December 31 (in millions)		2016		2015		2014
Gross Margin						
Intangible asset amortization expense	\$	(163)	\$	(158)	\$	(168)
Business optimization items ¹		(156)		(38)		11
Intangible asset impairment ²		(51)		_		
Separation-related costs ³		(1)		_		_
Product-related items ⁴		18		28		(64)
Total Special Items	\$	(353)	\$	(168)	\$	(221)
Impact on Gross Margin Ratio		(3.5 pts)		(1.7 pts)		(2.1 pts)
Marketing and Administrative Expenses						
Business optimization items ¹	\$	173	\$	152	\$	115
Separation-related costs ³		53		110		11
Product-related items ⁴		_		_		4
Branded Prescription Drug Fee ⁵						3
Total Special Items	\$	226	\$	262	\$	133
Impact on Marketing and Administrative Expense Ratio		2.3 pts		2.6 pts		1.2 pts
Research and Development Expenses						
Business optimization items ¹	\$	80	\$	13	\$	2
Separation-related costs ³				1		
Total Special Items	\$	80	\$	14	\$	2
Other (Income) Expense, Net						
Business optimization items ¹	\$	_	\$	(3)	\$	25
Net realized gains on Retained Shares transactions ⁶		(4,391)		_		
Loss on debt extinguishment ⁷		149		130		_
Reserve items and adjustments ⁸		_		(52)		1
Business development items ⁹		_		(20)		_
Tax matter ¹⁰		9				
Total Special Items	\$	(4,233)	\$	55	\$	26
Income Tax Expense						
Impact of special items ¹⁰	\$	(314)	\$	(137)	\$	(137)
Total Special Items	\$	(314)	\$	(137)	\$	(137)
Impact on Effective Tax Rate	•	(22.1 pts)	•	(10.4 pts)	•	(12.8 pts)

Intangible asset amortization expense is identified as a special item to facilitate an evaluation of current and past operating performance and is similar to how management internally assesses performance. Additional special items are identified above because they are highly variable, difficult to predict and of a size that may substantially impact the company's reported operations for a period. Management believes that providing the separate impact of the above items on the company's results in accordance with generally accepted accounting principles (GAAP) in the United States may provide a more complete understanding of the company's operations and can facilitate a fuller analysis of the company's results of operations, particularly in evaluating performance from one period to another. This information should be considered in addition to, and not as a substitute for, information prepared in accordance with GAAP.

In 2016, 2015 and 2014, the company's results were impacted by costs associated with the company's execution of certain strategies to optimize its organization and global cost structure on a global basis. These actions included streamlining the company's international operations, rationalizing its manufacturing facilities, reducing its general and administrative infrastructure, re-aligning certain R&D activities and cancelling certain R&D programs. The company recorded net business optimization charges of \$409 million, \$200 million and \$131 million in 2016, 2015 and 2014, respectively. The company's results in 2016 included a net charge of \$285 million related to restructuring activities, \$65 million of costs to implement business optimization programs which primarily included external consulting and project employee costs, \$33 million of accelerated depreciation associated with facilities to be closed, and \$26 million of Gambro integration costs. The \$285 million of

restructuring charges included net \$180 million of employee termination costs, \$54 million of costs related to the discontinuance of the VIVIA home hemodialysis development program, \$47 million of asset impairment charges related to acquired in-process R&D and facility closure costs and \$4 million of other exit costs. The company's results in 2015 included a net charge of \$127 million related to restructuring activities and \$73 million of Gambro integration costs. The \$127 million of net restructuring charges included net \$91 million of employee termination costs, a \$20 million intangible asset impairment and \$16 million of other asset impairments and other exit costs. The company's results in 2014 included \$144 million of Gambro integration costs and a net benefit of \$13 million from adjustments for reserves that are no longer probable of being utilized. Refer to Note 7 in Item 8 for further information regarding these charges and related reserves.

- The company's results in 2016 included a \$51 million asset impairment primarily related to developed technology.
- 3 The company's results in 2016, 2015 and 2014 included costs related to the Baxalta separation of \$54 million, \$111 million and \$11 million, respectively.
- The company's results in 2016 and 2015 included a net benefit of \$18 million and \$28 million, respectively, primarily related to adjustments to the COLLEAGUE and SIGMA SPECTRUM infusion pump reserves. The company's results in 2014 included charges, net of reversals, of \$68 million primarily related to product remediation efforts for the SIGMA SPECTRUM infusion pump. Refer to Note 7 in Item 8 for further information regarding these charges and related reserves.
- The company's results in 2014 included a charge of \$3 million to account for an additional year of the Branded Prescription Drug Fee in accordance with final regulations issued in the third quarter of 2014 by the Internal Revenue Service.
- The company's results in 2016 included net realized gains of \$4.4 billion related to the debt-for-equity exchanges of the company's retained shares in Baxalta for certain indebtedness, the exchange of retained shares in Baxalta for Baxter shares and the contribution of retained shares in Baxalta to Baxter's U.S. pension fund.
- The company's results in 2016 included a net debt extinguishment loss totaling \$149 million related to the March 2016 debt-for-equity exchange for certain company indebtedness and certain debt redemptions. The company's results in 2015 included a loss of \$130 million related to its July 2015 tender offer, for certain of its outstanding indebtedness. Refer to Note 8 in Item 8 for additional information.
- The company's results in 2015 included income of \$52 million related to a litigation settlement in which Baxter was the beneficiary. The company's results in 2014 included income of \$1 million related to third-party recoveries and reversals of prior litigation reserves.
- The company's results in 2015 included a benefit of \$20 million relating to the reversal of contingent consideration milestone liabilities. Refer to Note 5 in Item 8 for further information regarding the company's acquisitions and other arrangements.
- The company's results in 2016 included a net after-tax benefit of \$10 million, related to the settlement of an income tax matter in the company's non-wholly owned joint venture in Turkey. This amount was comprised of \$19 million included in income tax expense offset by \$9 million in non-controlling interest recorded in other income.

Net Sales

						Percent change							
						At actua	al	At consta	nt				
						currency i	ates	currency rates					
years ended December 31 (in millions)	2016		2015		2014	2016	2015	2016	2015				
Renal	\$ 3,855	\$	3,789	\$	4,172	2 %	(9)%	5%	1%				
Hospital Products	6,308		6,179		6,547	2 %	(6)%	4 %	1 %				
Total net sales	\$ 10.163	\$	9,968	\$	10.719	2 %	(7)%	4%	1%				

				Percent change							
				At actua	1	At constant					
				currency r	ates	currency rates					
years ended December 31 (in millions)	2016	2015	2014	2016	2015	2016	2015				
United States	\$ 4,259	\$ 4,001	\$ 3,999	6%	0%	6%	0%				
International	5,904	5,967	6,720	(1)%	(11)%	3 %	2%				
Total net sales	\$ 10,163	\$ 9,968	\$ 10,719	2 %	(7)%	4%	1%				

Net sales for the year ended December 31, 2016 increased 2% at actual currency rates and 4% on a constant currency basis. Net sales for the year ended December 31, 2015 decreased 7% at actual currency rates but increased 1% on a constant currency basis.

Foreign currency unfavorably impacted net sales by two percentage points during 2016 compared to the prior year principally due to the strengthening of the U.S. dollar relative to the British Pound, Mexican Peso, Colombian Peso and the Chinese Yuan, as well as other currencies, partially offset by the weakening of the U.S. dollar relative to the Japanese Yen. Foreign currency unfavorably impacted net sales by eight percentage points during 2015 compared to 2014 principally due to the strengthening of the U.S. Dollar relative to the Euro, Australian Dollar, Colombian Peso, and certain other currencies.

The comparisons presented at constant currency rates reflect comparative local currency sales at the prior year's foreign exchange rates. This measure provides information on the change in net sales assuming that foreign currency exchange rates had not changed between the prior and the current period. The company believes that the non-GAAP measure of change in net sales at constant currency rates, when used in conjunction with the GAAP measure of change in net sales at actual currency rates, may provide a more complete understanding of the company's operations and can facilitate a fuller analysis of the company's results of operations, particularly in evaluating performance from one period to another.

Franchise Net Sales Reporting

The Renal segment includes sales of the company's peritoneal dialysis (PD), hemodialysis (HD) and continuous renal replacement therapies (CRRT) and additional dialysis services.

The Hospital Products segment includes four commercial franchises: Fluid Systems, Integrated Pharmacy Solutions, Surgical Care and Other.

- Fluid Systems includes sales of the company's IV therapies, infusion pumps and administration sets.
- Integrated Pharmacy Solutions includes sales of the company's premixed and oncology drug platforms, nutrition products and pharmacy
 compounding services.
- Surgical Care includes sales of the company's inhaled anesthesia and critical care products as well as biological products and medical devices used in surgical procedures for hemostasis, tissue sealing and adhesion prevention.
- Other includes sales primarily from the company's pharmaceutical partnering business.

The following is a summary of net sales by commercial franchise.

				Percent change							
				At actua		At const					
				currency ra		currency i					
years ended December 31 (in millions)	2016	2015	2014	2016	2015	2016	2015				
Total Renal net sales	\$ 3,855	\$ 3,789	\$ 4,172	2 %	(9)%	5%	1%				
Fluid Systems	\$ 2,300	\$ 2,106	\$ 2,129	9%	(1)%	11%	6%				
Integrated Pharmacy Solutions	2,245	2,297	2,535	(2)%	(9)%	0 %	(2)%				
Surgical Care	1,321	1,323	1,373	0 %	(4)%	1%	3%				
Other	442	453	510	(2)%	(11)%	(2)%	(5)%				
Total Hospital Products net sales	\$ 6,308	\$ 6,179	\$ 6,547	2 %	(6)%	4 %	1%				

Net sales in the Renal segment increased 2% in 2016 from 2015 but decreased 9% in 2015 from 2014. These amounts include an unfavorable foreign currency impact of three percentage points in 2016 and 10 percentage points in 2015. Sales increased 5% on a constant currency basis in 2016, driven by continued global growth of patients, new product launches and improved pricing in the United States in our PD business. PD contributed approximately two percentage points to the growth rate during 2016. In addition, increased sales of the company's CRRT to treat acute kidney injury contributed two percentage points to the growth rate during 2016. Renal net sales are expected to be negatively impacted in 2017 by approximately \$50 million as compared to 2016 due to certain international strategic market exits. Sales increased 1% on a constant currency basis in 2015, driven by continued growth in the number of PD patients globally, which contributed approximately three percentage points, and strong demand in the acute business. These factors were partially offset by lower sales in the chronic in-center HD business, resulting from the decision to forgo certain lower margin sales opportunities, increased austerity measures in Western Europe, and competitive pressures for dialyzers.

Net sales in the Hospital Products segment increased 2% in 2016 and decreased 6% in 2015. Foreign currencies had an unfavorable impact of two percentage points in 2016 and seven percentage points in 2015. Hospital Products net sales are expected to be negatively impacted in 2017 by approximately \$50 million as compared to 2016 due to certain international strategic market exits. Excluding the impact of foreign currency, the principal drivers impacting 2016 net sales growth were the following:

- In the Fluid Systems franchise, sales increased 11% in 2016 on a constant currency basis driven by favorable pricing and volume for IV solutions and increased sales of the SIGMA SPECTRUM pump and the related sets in the United States. Sales increased 6% in 2015 on a constant currency basis driven by increased sales of infusion system products, which contributed approximately four percentage points, including the relaunch of the SIGMA Spectrum infusion pump in the United States, Puerto Rico, and Canada during 2015. Additionally, sales growth in 2015 was impacted by favorable pricing and volume in the United States for the company's IV therapies, which contributed approximately one percentage point.
- In the Integrated Pharmacy Solutions franchise, sales were flat in 2016 on a constant currency basis driven by global demand for the company's nutritional therapies, contributing approximately one percentage point during 2016 and demand for the company's international pharmacy compounding services which contributed approximately one percentage point during 2016. These increases were offset by lower U.S. sales of the company's pharmacy injectable products, as there were government PROTOPAM orders in 2015 that did not reoccur in 2016, contributing approximately one percentage point of decline. In addition, U.S. sales of cyclophosphamide, a generic oncology drug, were approximately \$210 million and \$270 million in 2016 and 2015, respectively, which contributed an approximate three percentage point of decline in 2016. The company expects a significant decline in U.S sales for cyclophosphamide in 2017 due to additional competition in the market. Sales decreased 2% in 2015 on a constant currency basis driven by decreased sales of cyclophosphamide, following a competitor entering the U.S. market in November 2014 which contributed approximately six percentage points. U.S. sales of cyclophosphamide during 2014 were approximately \$450 million. This decline was offset by an increase in revenues from pharmacy compounding services, increased demand for the company's nutritional therapies, and pharmacy injectable products, including approximately \$40 million in sales of PROTOPAM, which contributed two percentage points.
- In the Surgical Care franchise, sales increased 1% in 2016 on a constant currency basis driven by increased demand for international anesthesia products. Sales increased 3% in 2015 on a constant currency basis driven by strong global demand for the company's portfolio of anesthetics products, which contributed three percentage points, offset partially by lower sales of select non-core biosurgery products.
- In the Other franchise, sales decreased 2% in 2016 on a constant currency basis compared to 2015 driven by lower demand for products manufactured by Baxter on behalf of one of its pharmaceutical partners. The company also recognized revenue of \$39 million in 2016 as compared to \$37 million in 2015 related to the company's manufacturing and supply agreement with Baxalta. Sales decreased 5% in 2015 on a constant currency basis compared to 2014 driven by one of the company's pharmaceutical partners electing to self-manufacture products previously contract manufactured by Baxter. This loss of revenue was partially offset by increased sales related to the company's manufacturing and supply agreement with Baxalta.

Gross Margin and Expense Ratios

				Chang	e
years ended December 31 (as a percent of net sales)	2016	2015	2014	2016	2015
Gross margin	40.4%	41.6%	42.7%	(1.2 pts)	(1.1 pts)
Marketing and administrative expenses	27.0 %	31.0%	30.9%	(4.0 pts)	0.1 pts

Gross Margin

The special items previously identified in the above had an unfavorable impact of 3.5, 1.7 and 2.1 percentage points on the gross margin ratio in 2016, 2015 and 2014, respectively. Refer to the Special Items section above for additional detail.

Excluding the impact of the special items, the gross margin ratio increased 0.6 percentage points in 2016. The gross margin ratio was impacted by a positive sales mix, improved pricing in select areas of the portfolio and favorable manufacturing performance, offset by reduced sales of cyclophosphamide in the United States and foreign exchange.

Excluding the impact of the special items, the gross margin ratio in 2015 was unfavorably impacted by decreased sales of cyclophosphamide in the United States, partially offset by an improved product mix in the Renal segment.

Marketing and Administrative Expenses

The special items identified above had an unfavorable impact of 2.3, 2.6 and 1.2 percentage points on the marketing and administrative expenses ratio in 2016, 2015 and 2014, respectively. Refer to the Special Items section above for additional detail.

Excluding the impact of the special items, the marketing and administrative expense ratio decreased 3.7 percentage points in 2016 and was impacted by reduced pension expense, as well as benefits from the company's actions taken to rebase its cost structure and continued focus on expense management, in addition to a reduction to expense under the transition services agreement with Baxalta.

Excluding the impact of the special items, the marketing and administrative expense ratio in 2015 was impacted by the benefits from the company's business optimization actions as the company resets its cost structure, reduced its discretionary spending, and benefited from certain costs charged to Baxalta under the transition services agreement. These benefits were partially offset by increased bad debt expense in emerging markets.

Pension and Other Postemployment Benefit Plan Expense

Expense related to the company's pension and other postemployment benefit plans decreased \$111 million in 2016 primarily due to a change in approach to estimating employer service and interest costs and a \$706 million voluntary, non-cash contribution to the US qualified plan using Retained Shares. Pension and other postemployment benefit plan expense increased \$8 million in 2015 primarily due to a decrease in the discount rate.

Business Optimization Items

Beginning in the second half of 2015, the company has initiated actions to transform the company's cost structure and enhance operational efficiency. These efforts include restructuring the organization, optimizing the manufacturing footprint, R&D operations and supply chain network, employing disciplined cost management, and centralizing and streamlining certain support functions. Through December 31, 2016 the company incurred cumulative pretax costs of \$407 million related to these actions. The costs consisted primarily of employee termination costs, implementation costs, and accelerated depreciation. The company expects to incur additional pretax costs of approximately \$390 million and capital expenditures of \$90 million related to these initiatives by the end of 2018. These costs will primarily include employee termination costs, implementation costs, and accelerated depreciation. The company expects that approximately 10 percent of the charges will be non-cash. These actions in the aggregate are expected to provide future annual pretax savings of approximately \$860 million. The savings from these actions will impact cost of sales, marketing and administrative expenses, and R&D expenses. The company estimates that actions taken through December 31, 2016 have resulted in approximately \$343 million of savings in 2016. Approximately 85 percent of the expected annual pretax savings are expected to be realized by the end of 2018, with the remainder by the end of 2020.

In addition to the programs above, the company recorded additional net business optimization charges of \$125 million in 2016. These charges primarily include employee termination costs, contract termination costs, asset impairments, and Gambro integration costs. Approximately 40% of these costs were non-cash. The company does not anticipate incurring any additional costs related to these programs in the future. The actions in the aggregate are expected to provide future annual pretax savings of approximately \$19 million. The savings from these actions will impact cost of sales, marketing and administrative expenses, and R&D expenses. The company estimates that the actions taken through December 31, 2016, have resulted in approximately \$8 million of savings in the current period. The remaining pretax savings are expected to be realized as the programs are substantially completed by the end of 2017.

Refer to Note 7 in Item 8 for additional information regarding the company's business optimization initiatives.

Research and Development

				_	Percent cha	nge
years ended December 31 (in millions)	2016	2015		2014	2016	2015
Research and development expenses	\$ 647	\$ 603	\$	610	7%	(1)%
as a percent of net sales	6.4 %	6.0%)	5.7%	0.4 pts	0.3 pts

The special items identified above had an unfavorable impact of \$80 million, \$14 million and \$2 million in 2016, 2015 and 2014, respectively.

Excluding the impact of special items, R&D expenses decreased 4% in 2016 primarily due to the optimization of the infrastructure, the exit of certain programs and the impact of foreign currency. R&D expenses in 2015 declined as the company worked to balance increased investments with efforts to optimize its overall R&D expenditures.

Net Interest Expense

Net interest expense was \$66 million, \$126 million and \$145 million in 2016, 2015 and 2014, respectively. The decrease in 2016 was principally driven by lower outstanding debt as a result of the first quarter 2016 debt-for-equity exchanges and reduced coupon rates resulting from the third quarter 2016 debt issuance, partially offset by lower capitalized interest compared to 2015. The decrease in 2015 was principally driven by the debt tender offer completed in July 2015 and the maturity of \$600 million of 4.625% senior unsecured notes in March 2015, partially offset by higher interest on the company's short term revolving credit facility, lower capitalized interest, and lower income from interest rate hedging activities. Refer to Note 3 in Item 8 for a summary of the components of net interest expense for 2016, 2015 and 2014.

Other (Income) Expense, Net

Other (income) expense, net was income of \$4.3 billion in 2016, income of \$105 million in 2015 and expense of \$21 million in 2014. Current year results included net realized gains of \$4.4 billion on the Retained Shares transactions, dividend income of \$16 million from the Retained Shares, and \$28 million of income related to foreign currency fluctuations principally relating to intercompany receivables, payables and monetary assets denominated in a foreign currency. These income items were partially offset by net debt extinguishment losses of \$153 million. The 2015 results were driven primarily by \$52 million of income related to a favorable litigation settlement, \$38 million income from the sale of available-for-sale securities, and \$113 million of income related to foreign currency fluctuations principally relating to intercompany receivables, payables and monetary assets denominated in a foreign currency, partially offset by a \$130 million loss on extinguishment of debt related to the July 2015 debt tender offer.

Segment EBITDA

The company uses income from continuing operations before net interest expense, income tax expense, depreciation and amortization expense (Segment EBITDA), on a segment basis to make resource allocation decisions and assess the ongoing performance of the company's business segments. Refer to Note 17 in Item 8 for additional details regarding the company's segments. The following is a summary of significant factors impacting the segments' financial results.

Renal

Segment EBITDA was \$703 million, \$566 million and \$666 million in 2016, 2015 and 2014, respectively. The increase in 2016 was primarily driven by increased sales and lower marketing and administrative expenses as cost savings were realized from the company's business optimization programs and continued focus on expense management. This was partially offset by unfavorable foreign currency fluctuations, incremental manufacturing and quality costs, and higher allocated R&D costs. EBITDA declined in 2015 due to unfavorable foreign currency fluctuations, the impairment of certain intangible assets and investments in certain quality programs and manufacturing capabilities, partially offset by efficiencies related to the integration of the Gambro business.

Hospital Products

Segment EBITDA was \$2.3 billion, \$2.0 billion and \$2.2 billion in 2016, 2015 and 2014, respectively. The increase in 2016 was driven by increased sales, favorable manufacturing performance, lower allocated R&D costs, and lower marketing and administrative expenses as cost savings were realized from the company's business optimization programs and continued focus on expense management. This growth was partially offset by unfavorable foreign currency fluctuations. EBITDA in 2015 was impacted primarily by unfavorable foreign currency fluctuations and decreased sales of the higher margin cyclophosphamide product. This was offset by a reduction in costs in 2014 related to manufacturing inefficiencies and quality costs.

Corporate and other

Certain income and expense amounts are not allocated to a segment. These amounts are detailed in the table in Note 17 in Item 8 and primarily include net interest expense, foreign exchange fluctuations (principally relating to intercompany receivables, payables and loans denominated in foreign currency), the majority of the foreign currency hedging activities, corporate headquarters costs, international global support costs, stock compensation expense, non-strategic investments and related income and expense, certain employee benefit plan costs as well as certain nonrecurring gains, losses, and other charges (such as business optimization and asset impairments).

Income Taxes

Effective Income Tax Rate

The effective income tax rate for continuing operations was (0.2%) in 2016, 8.2% in 2015 and 6.7% in 2014. The company anticipates that the effective income tax rate from continuing operations, calculated in accordance with GAAP, will be approximately 21.5% in 2017, excluding any impact from tax windfalls or deficiencies attributable to stock compensation exercises as well as additional audit developments or other special items.

The company's effective tax rate differs from the U.S. federal statutory rate each year due to certain operations that are subject to tax incentives, state and local taxes and foreign taxes that are different than the U.S. federal statutory rate. The average foreign effective tax rate on international pre-tax income from continuing operations was 25.2%, 27.5% and 24.1% for the years ended December 31, 2016, 2015 and 2014, respectively. The company's average foreign effective tax rate was lower than the U.S. federal statutory rate as a result of the impact of tax incentives in Puerto Rico, Switzerland and certain other tax jurisdictions outside of the United States, as well as foreign earnings in tax jurisdictions with lower statutory rates than the United States. Adversely impacting the foreign rate were foreign loss-generating operations that did not receive tax benefits due to the losses resulting in, or contributing to, the need for a valuation allowance. In addition, as discussed further below, the company's effective income tax rate can be impacted in each year by discrete factors or events. Refer to Note 15 in Item 8 for further information regarding the company's income taxes.

Factors impacting the company's effective tax rate in 2016 included tax-free net realized gains during the first and second quarter associated with the exchanges of Baxalta Retained Shares for the company's debt and the company's shares as well as tax-free net realized gains associated with the contribution of Baxalta Retained Shares to the company's pension plan. Additionally, the income tax rate for 2016 was favorably impacted by tax benefits from partially settling an IRS (2008-2013) income tax audit, settling a German (2008-2011) income tax audit, resolution of uncertain tax positions related to the company's Turkish joint venture, other transfer pricing matters, and partial settlement of interest expense deductions related to the company's acquisition of Gambro.

Factors adversely impacting the company's effective tax rate in 2015 included charges related to contingent tax matters primarily related to transfer pricing and the separation of Baxalta as well as the need to record valuation allowances for some loss making entities. Partially offsetting the foregoing adverse factors was a benefit from reaching a settlement of a Puerto Rico excise tax matter as well as the U.S. R&D credit resulting from the retroactive reinstatement in December 2015 of the Protecting Americans from Tax Hikes Act of 2015.

Factors impacting the company's effective tax rate in 2014 included the favorable settlement of a portion of the company's contingent tax matter related to operations in Turkey as well as a favorable shift of earnings from high to low tax jurisdictions compared to the prior period. Additionally, the effective tax rate was unfavorably impacted by increases in valuation allowances due to the tax benefit from losses that the company does not believe that it is more likely than not to realize and interest expense related to the company's unrecognized tax benefits.

The company earns a significant amount of its operating income outside the United States, of which a substantial portion is deemed to be indefinitely reinvested in foreign jurisdictions. As a result, most of the company's cash and short-term investments are held by foreign subsidiaries. The company does not intend or foresee a need to repatriate these funds and expects existing domestic cash and short-term investments and cash flows from operations to continue to be sufficient to fund domestic operating activities and cash commitments for investing and financing activities, such as regular quarterly dividends and capital expenditures, for at least the next 12 months and thereafter for the foreseeable future.

If the company should require more capital in the United States than is generated by its domestic operations (e.g., to fund significant discretionary activities such as business acquisitions and share repurchases), the company could elect to repatriate future earnings from foreign jurisdictions or raise capital in the United States through debt or equity issuances. These alternatives could result in higher effective tax rates, increased interest expense or dilution of the company's earnings. The company has borrowed domestically and continues to believe it has the ability to do so at reasonable interest rates.

Income from Continuing Operations and Earnings per Diluted Share

Income from continuing operations was \$5.0 billion in 2016, \$393 million in 2015 and \$457 million in 2014. Income from continuing operations per diluted share was \$9.01 in 2016, \$0.72 in 2015 and \$0.83 in 2014. The significant factors and events causing the net changes from 2015 to 2016 and 2014 to 2015 are discussed above. Additionally, income from continuing operations per diluted share was positively impacted by the repurchase of 17.8 million shares through cash repurchases and an equity-for-equity exchange of Retained Shares for outstanding Baxter shares in 2016, and the repurchase of eight million shares in 2014. Refer to Note 12 in Item 8 for further information regarding the company's stock repurchases.

(Loss) Income from Discontinued Operations

The following table is a summary of the operating results of Baxalta, which have been reflected as discontinued operations for the years ended December 31, 2016, 2015 and 2014.

Years ended December 31 (in millions)	2	016	2015	2014
Net sales	\$	148	\$ 2,895	\$ 6,523
(Loss) income from discontinued operations before income taxes		(10)	752	2,562
Gain on disposal of discontinued operations		19	_	_
Income tax expense		10	177	522
Total (loss) income from discontinued operations	\$	(1)	\$ 575	\$ 2,040

Refer to Note 2 in Item 8 for additional information regarding the separation of Baxalta.

LIQUIDITY AND CAPITAL RESOURCES

The company's cash flows reflect both continuing and discontinued operations.

Cash Flows from Operations — Continuing Operations

Operating cash flows from continuing operations totaled \$1.6 billion in 2016, \$1.3 billion in 2015 and \$1.2 billion in 2014. The cash flows from continuing operations in 2015 were impacted by the receipt of a \$52 million legal settlement as well as \$114 million of payments related to the July 2015 debt tender offer. See below for other factors impacting the increase in cash flows in 2016 and the decrease in cash flows in 2015.

Accounts Receivable

Cash flows relating to accounts receivable increased in 2016 and 2015 as the days sales outstanding decreased in 2016 and increased in 2015. Days sales outstanding were 54.5 days, 56.2 days and 54.2 days for 2016, 2015 and 2014, respectively. Days sales outstanding decreased in 2016 primarily driven by timing of collections in certain international markets. Days sales outstanding increased in 2015 driven by slower collections in the United States.

Inventories

Cash flows relating to inventory improved from an outflow of \$118 million in 2015 to an inflow of \$80 million in 2016, driven by continued working capital improvement initiatives which decreased days inventory on hand by ten days and increased turns significantly as compared to the prior year. The following is a summary of inventories at December 31, 2016 and 2015, as well as inventory turns by segment for 2016, 2015 and 2014. Inventory turns for the year are calculated as the annualized fourth quarter cost of sales divided by the year-end inventory balance.

		Inven	tories	<u> </u>	In		
(in millions, except inventory turn data)	2	2016		2015	2016	2015	2014
Renal	\$	544	\$	605	4.4	3.7	3.6
Hospital Products		885		955	3.9	3.6	3.7
Other		1		44	n/a	n/a	n/a
Total company	\$	1,430	\$	1,604	4.1	3.6	3.7

Other

The changes in accounts payable and accrued liabilities were an outflow of \$197 million in 2016, an inflow of \$236 million in 2015 and a \$37 million outflow in 2014. The change in 2016 was primarily driven by an increase in tax payments primarily due to a tax settlement as well as the timing of payments to suppliers. Refer to Note 15 in Item 8 for additional details regarding the tax settlement. The change in 2015 was primarily driven by the timing of payments to suppliers as well as the timing of tax payments.

Payments related to the execution of the SIGMA SPECTRUM infusion pump recalls as well as the company's business optimization initiatives were \$191 million in 2016, \$112 million in 2015 and \$124 million in 2014. Refer to Note 7 in Item 8 for further information regarding the SIGMA SPECTRUM infusion pump recalls and the business optimization initiatives.

Other balance sheet items had net cash inflows of \$115 million in 2016, and outflows of \$341 million and \$17 million in 2015 and 2014, respectively. In 2016, the company received a U.S. federal income tax refund of \$250 million as a result of carrying back to prior tax years the company's 2015 U.S. tax loss which arose, in significant part, from the funding of the company's defined benefit pension plan with a portion of the Baxalta retained stake. Additionally, cash contributions to the company's pension plans totaled \$66 million, \$157 million and \$74 million in 2016, 2015 and 2014, respectively. The changes during 2015 and 2014 were primarily driven by prepaid expenses and hedging activity.

Cash Flows from Investing Activities — Continuing Operations

Capital Expenditures

Capital expenditures relating to continuing operations totaled \$719 million in 2016, \$911 million in 2015 and \$925 million in 2014. The company's capital expenditures in 2016 and 2015 consisted of targeted investments in projects to support production of PD and IV solutions as well as expansion activities for dialyzers. The decline in capital expenditures over the three years was due to a reduction in spending related to ongoing projects and the completion of certain expansion activities.

Acquisitions and Investments

Net cash outflows related to acquisitions and investments were \$48 million in 2016, \$34 million in 2015 and \$95 million in 2014. The cash outflows in 2016 were driven primarily by the acquisition of the rights to vancomycin from Celerity. The cash outflows in 2015 were driven by the acquisition of the rights to cefazolin injection in GALAXY Container (2g/100mL) from Celerity. The cash outflows in 2014 were driven by the acquisitions of IC Net International Ltd and certain investments.

Refer to Note 5 in Item 8 for further information about the company's significant acquisitions and other arrangements.

Divestitures and Other Investing Activities

Net cash inflows relating to divestitures and other investing activities were \$37 million in 2016, \$84 million in 2015 and \$99 million in 2014. The decrease from 2015 to 2016 was primarily driven by the sale of certain investments and other assets in 2015. Cash inflows in 2014 primarily related to proceeds from the divestiture of Baxter's legacy CRRT business and the sale of certain investments.

Cash Flows from Financing Activities

Debt Issuances, Net of Payments of Obligations

Net cash outflows related to debt and other financing obligations totaled \$56 million in 2016 primarily related to a \$190 million repayment of the company's 0.95% senior unsecured notes that matured in June 2016, a \$130 million repayment of the company's 5.9% senior unsecured notes that matured in September 2016 and the redemption of approximately \$1 billion in aggregate principal amount of senior notes in September 2016, as well as the repayment of other short-term obligations. The company also had \$300 million of net repayments related to its commercial paper program. These cash outflows were partially offset by issuances of debt totaling \$1.6 billion of senior notes in August 2016. Refer to Note 8 in Item 8 for additional details regarding the debt transactions in 2016.

Net cash inflows related to debt and other financing obligations totaled \$2.4 billion in 2015 driven by approximately \$6.9 billion in issuances of debt primarily related to the Baxalta senior notes and borrowings under the company's revolving credit facilities. The company purchased an aggregate of approximately \$2.7 billion in principal amount of its notes in 2015. Additionally, the company repaid \$600 million of 4.625% senior unsecured notes that matured in March 2015 and borrowings under the company's Euro-denominated revolving credit facility. The company issued and redeemed commercial paper throughout the year, and had \$300 million outstanding as of December 31, 2015.

Net cash outflows related to debt and other financing obligations totaled \$113 million in 2014 driven by approximately \$1 billion in repayments, which included \$500 million of floating rate senior unsecured notes that matured in December 2014 as well as \$350 million of 4.0% senior unsecured notes that matured in March 2014. The company issued and redeemed commercial paper throughout the year, and had \$875 million outstanding as of December 31, 2014.

The company's debt instruments discussed above are unsecured and contain certain covenants, including restrictions relating to the company's issuance of secured debt.

Other Financing Activities

In connection with the separation, Baxter transferred \$2.1 billion of cash to Baxalta in 2015.

Cash dividend payments totaled \$0.3 billion in 2016, \$0.9 billion in 2015 and \$1.1 billion in 2014. The decrease in cash dividend payments in 2016 and 2015 was primarily due to the decrease of the quarterly dividend after the separation of Baxalta, from \$0.52 per share for quarterly dividends beginning after May 2014 to \$0.115 per quarterly dividends beginning after July 2015. The Baxter cash dividend was increased to \$0.13 per share for quarterly dividends beginning after May 2016.

Proceeds and realized excess tax benefits from stock issued under employee benefit plans totaled \$325 million, \$200 million and \$369 million in 2016, 2015 and 2014, respectively. Total realized excess tax benefits, which were \$39 million in 2016, \$7 million in 2015 and \$24 million in 2014, are presented in the consolidated statements of cash flows as an inflow in the financing section and an outflow in the operating section.

In 2016, the company executed an equity-for-equity exchange of Retained Shares for 11.5 million outstanding Baxter shares. As authorized by the Board of Directors, the company repurchases its stock depending on the company's cash flows, net debt level and market conditions. In July 2012, the Board of Directors authorized the repurchase of up to \$2.0 billion of the company's common stock. The Board of Directors increased this authority by an additional \$1.5 billion in November 2016. The company paid \$287 million in cash to repurchase approximately 6.3 million shares pursuant to this authority in 2016 and had \$1.7 billion remaining available under this authorization as of December 31, 2016. The company did not repurchase any stock during 2015.

Credit Facilities, Access to Capital and Credit Ratings

Credit Facilities

As of December 31, 2016, the company's U.S. dollar-denominated revolving credit facility and Euro-denominated senior revolving credit facility had a maximum capacity of \$1.5 billion and approximately €200 million, respectively. As of December 31, 2016, the company was in compliance with the financial covenants in these agreements. The non-performance of any financial institution supporting either of the credit facilities would reduce the maximum capacity of these facilities by each institution's respective commitment.

Effective July 1, 2015, the company terminated its \$1.5 billion U.S. dollar-denominated revolving credit facility and €300 million Euro-denominated revolving credit facility, which were set to mature in December 2015, in connection with the separation and distribution. In connection with such terminations, the company entered into credit agreements providing for a senior U.S. dollar-denominated revolving credit facility in an aggregate principal amount of up to \$1.5 billion maturing in 2020, as well as a Euro-denominated senior revolving credit facility in an aggregate principal amount of up to €200 million maturing in 2020. The company may, at its option, seek to increase the aggregate commitment under the new U.S. facility by up to an additional \$750 million. The new facilities enable the company to borrow funds on an unsecured basis at variable interest rates, and contain various covenants, including a maximum net leverage ratio and maximum interest coverage ratio.

Additionally, as of December 31, 2015, the company had a third revolving credit facility, with a maximum capacity of \$1.8 billion, which was scheduled to mature on the earlier of March 28, 2016 and the date on which commitments under the facility have been reduced to zero or terminated in whole pursuant to the terms thereof. On January 27, 2016, Baxter exchanged all 37.6 million shares of Baxalta common stock for the \$1.45 billion aggregate principal amount outstanding under this revolving credit facility. This exchange extinguished all outstanding indebtedness under the facility, at which time the facility was terminated. In connection with the exchange of Baxalta common stock, Baxter recognized \$1.2 billion of realized gains in 2016.

The company also maintains other credit arrangements, as described in Note 8 in Item 8.

Access to Capital

The company intends to fund short-term and long-term obligations as they mature through cash on hand, future cash flows from operations or by issuing additional debt. The company had \$2.8 billion of cash and equivalents as of December 31, 2016, with adequate cash available to meet operating requirements in each jurisdiction in which the company operates. The company invests its excess cash in certificates of deposit and money market funds, and diversifies the concentration of cash among different financial institutions.

The company's ability to generate cash flows from operations, issue debt or enter into other financing arrangements on acceptable terms could be adversely affected if there is a material decline in the demand for the company's products or in the solvency of its

customers or suppliers, deterioration in the company's key financial ratios or credit ratings or other significantly unfavorable changes in conditions. However, the company believes it has sufficient financial flexibility to issue debt, enter into other financing arrangements and attract long-term capital on acceptable terms to support the company's growth objectives.

The company continues to do business with foreign governments in certain countries, including Greece, Spain, Portugal and Italy, which have experienced deterioration in credit and economic conditions. As of December 31, 2016, the company's net accounts receivable from the public sector in Greece, Spain, Portugal and Italy totaled \$137 million.

While these economic conditions have not significantly impacted the company's ability to collect receivables, global economic conditions and liquidity issues in certain countries have resulted, and may continue to result, in delays in the collection of receivables and credit losses.

Credit Ratings

The company's credit ratings at December 31, 2016 were as follows:

	Standard & Poor's	Fitch	Moody's
Ratings			
Senior debt	A-	BBB+	Baa2
Short-term debt	A2	F2	P2
Outlook	Stable	Stable	Stable

Contractual Obligations

As of December 31, 2016, the company had contractual obligations, excluding accounts payable and accrued liabilities, payable or maturing in the following periods.

(in millions)	Total	ss than e year	 ne to e years	hree to re years	ore than we years
Long-term debt and capital lease obligations, including current maturities	\$ 2,794	\$ 3	\$ 6	\$ 700	\$ 2,085
Interest on short- and long-term debt and capital lease obligations 1	1,496	84	169	163	1,080
Operating leases	663	129	193	125	216
Other long-term liabilities ²	443	_	114	43	286
Purchase obligations ³	450	275	153	20	2
Contractual obligations ⁴	\$ 5,846	\$ 491	\$ 635	\$ 1,051	\$ 3,669

- Interest payments on debt and capital lease obligations are calculated for future periods using interest rates in effect at the end of 2016. Projected interest payments include the related effects of interest rate swap agreements. Certain of these projected interest payments may differ in the future based on changes in floating interest rates, foreign currency fluctuations or other factors or events. The projected interest payments only pertain to obligations and agreements outstanding at December 31, 2016. Refer to Note 8 and Note 9 in Item 8 for further discussion regarding the company's debt instruments and related interest rate agreements outstanding at December 31, 2016.
- The primary components of other long-term liabilities in the company's consolidated balance sheet are liabilities relating to pension and other postemployment benefit plans, litigation, and foreign currency hedges. The company projected the timing of the future cash payments based on contractual maturity dates (where applicable) and estimates of the timing of payments (for liabilities with no contractual maturity dates). The actual timing of payments could differ from the estimates.
 - The company contributed \$772 million, \$157 million and \$74 million to its defined benefit pension plans in 2016, 2015 and 2014, respectively. The timing of funding in the future is uncertain and is dependent on future movements in interest rates and investment returns, changes in laws and regulations, and other variables. Therefore, the table above excludes pension plan cash outflows. The pension plan balance included in other long-term liabilities (and excluded from the table above) totaled \$1.5 billion at December 31, 2016.
- Includes the company's significant contractual unconditional purchase obligations. For cancelable agreements, any penalty due upon cancellation is included. These commitments do not exceed the company's projected requirements and are in the normal course of business. Examples include firm commitments for raw material purchases, utility agreements and service contracts.

Excludes contingent liabilities and uncertain tax positions. These amounts have been excluded from the contractual obligations above due to uncertainty regarding the timing and amount of future payments. Refer to Notes 10 and 11 in Item 8 for additional information regarding these commitments.

Off-Balance Sheet Arrangements

Baxter periodically enters into off-balance sheet arrangements. Certain contingencies arise in the normal course of business, and are not recorded in the consolidated balance sheet in accordance with GAAP (such as contingent joint development and commercialization arrangement payments). Also, upon resolution of uncertainties, the company may incur charges in excess of presently established liabilities for certain matters (such as contractual indemnifications). For a discussion of the company's significant off-balance sheet arrangements, refer to Note 10 in Item 8 for information regarding receivable securitizations, Note 11 in Item 8 regarding joint development and commercialization arrangements and indemnifications, and Note 16 in Item 8 regarding legal contingencies.

FINANCIAL INSTRUMENT MARKET RISK

The company operates on a global basis and is exposed to the risk that its earnings, cash flows and equity could be adversely impacted by fluctuations in foreign exchange and interest rates. The company's hedging policy attempts to manage these risks to an acceptable level based on the company's judgment of the appropriate trade-off between risk, opportunity and costs. Refer to Note 9 and Note 10 in Item 8 for further information regarding the company's financial instruments and hedging strategies.

Currency Risk

The company is primarily exposed to foreign exchange risk with respect to revenues generated outside of the United States denominated in the Euro, British Pound, Chinese Yuan, Korean Won, Australian Dollar, Canadian Dollar, Japanese Yen, Colombian Peso, Brazilian Real, Swedish Krona, Mexican Peso, and New Zealand Dollar. The company manages its foreign currency exposures on a consolidated basis, which allows the company to net exposures and take advantage of any natural offsets. In addition, the company uses derivative and nonderivative financial instruments to further reduce the net exposure to foreign exchange. Gains and losses on the hedging instruments offset losses and gains on the hedged transactions and reduce the earnings and stockholders' equity volatility relating to foreign exchange. Financial market and currency volatility may limit the company's ability to cost-effectively hedge these exposures.

The company may use options, forwards and cross-currency swaps to hedge the foreign exchange risk to earnings relating to forecasted transactions denominated in foreign currencies and recognized assets and liabilities. The maximum term over which the company has cash flow hedge contracts in place related to forecasted transactions at December 31, 2016 is 12 months. The company also enters into derivative instruments to hedge certain intercompany and third-party receivables and payables and debt denominated in foreign currencies.

Currency restrictions enacted in Venezuela require Baxter to obtain approval from the Venezuelan government to exchange Venezuelan bolivars for U.S. dollars and require such exchange to be made at the official exchange rate established by the government. In the first quarter of 2016, the Venezuelan government moved from the three-tier exchange rate system to a two-tiered exchange rate system and the official rate for food and medicine imports was adjusted from 6.3 to 10 bolivars per U.S. dollar. This devaluation resulted in a charge of \$9 million during the first quarter of 2016. As of December 31, 2016, the company's exposure to Venezuelan operations was approximately \$12 million.

As part of its risk-management program, the company performs sensitivity analyses to assess potential changes in the fair value of its foreign exchange instruments relating to hypothetical and reasonably possible near-term movements in foreign exchange rates.

A sensitivity analysis of changes in the fair value of foreign exchange option and forward contracts outstanding at December 31, 2016, while not predictive in nature, indicated that if the U.S. Dollar uniformly weakened by 10% against all currencies, on a net-of-tax basis, the net asset balance of \$13 million with respect to those contracts would decrease by \$34 million, resulting in a net liability position. A similar analysis performed with respect to option and forward contracts outstanding at December 31, 2015 indicated that, on a net-of-tax basis, the net asset balance of \$5 million would decrease by \$17 million.

The sensitivity analysis model recalculates the fair value of the foreign exchange option and forward contracts outstanding at December 31, 2016 by replacing the actual exchange rates at December 31, 2016 with exchange rates that are 10% weaker compared to the actual exchange rates for each applicable currency. All other factors are held constant. These sensitivity analyses disregard the possibility that currency exchange rates can move in opposite directions and that gains from one currency may or may not be offset by

losses from another currency. The analyses also disregard the offsetting change in value of the underlying hedged transactions and balances.

Interest Rate and Other Risks

The company is also exposed to the risk that its earnings and cash flows could be adversely impacted by fluctuations in interest rates. The company's policy is to manage interest costs using a mix of fixed- and floating-rate debt that the company believes is appropriate. To manage this mix in a cost-efficient manner, the company periodically enters into interest rate swaps in which the company agrees to exchange, at specified intervals, the difference between fixed and floating interest amounts calculated by reference to an agreed-upon notional amount. The company also periodically uses forward-starting interest rate swaps and treasury rate locks to hedge the risk to earnings associated with fluctuations in interest rates relating to anticipated issuances of term debt.

As part of its risk management program, the company performs sensitivity analyses to assess potential gains and losses in earnings relating to hypothetical movements in interest rates. A 19 basis-point increase in interest rates (approximately 10% of the company's weighted-average interest rate during 2016) affecting the company's financial instruments, including debt obligations and related derivatives, would have an immaterial effect on the company's 2016, 2015 and 2014 earnings and on the fair value of the company's fixed-rate debt as of the end of each fiscal year.

With respect to the company's investments in affiliates, the company believes any reasonably possible near-term losses in earnings, cash flows and fair values would not be material to the company's consolidated financial position.

CHANGES IN ACCOUNTING STANDARDS

Refer to Note 1 in Item 8 for information on changes in accounting standards.

CRITICAL ACCOUNTING POLICIES

The preparation of financial statements in accordance with GAAP requires the company to make estimates and judgments that affect the reported amounts of assets, liabilities, revenues and expenses. A summary of the company's significant accounting policies is included in Note 1 in Item 8. Certain of the company's accounting policies are considered critical because these policies are the most important to the depiction of the company's financial statements and require significant, difficult or complex judgments by the company, often requiring the use of estimates about the effects of matters that are inherently uncertain. Actual results that differ from the company's estimates could have an unfavorable effect on the company's results of operations and financial position. The following is a summary of accounting policies that the company considers critical to the consolidated financial statements.

Revenue Recognition and Related Provisions and Allowances

The company's policy is to recognize revenues from product sales and services when earned. Specifically, revenue is recognized when persuasive evidence of an arrangement exists, delivery has occurred (or services have been rendered), the price is fixed or determinable, and collectability is reasonably assured. For product sales, revenue is not recognized until title and risk of loss have transferred to the customer. The shipping terms for the majority of the company's revenue arrangements are FOB destination. The company sometimes enters into arrangements in which it commits to delivering multiple products or services to its customers. In these cases, total arrangement consideration is allocated to the deliverables based on their relative selling prices. Then the allocated consideration is recognized as revenue in accordance with the principles described above. Selling prices are determined by applying a selling price hierarchy. Selling prices are determined using vendor specific objective evidence (VSOE), if it exists. Otherwise, selling prices are determined using third party evidence (TPE). If neither VSOE nor TPE is available, the company uses its best estimate of selling prices.

Provisions for rebates, chargebacks to wholesalers and distributors, returns, and discounts (collectively, "sales deductions") are provided for at the time the related sales are recorded, and are reflected as a reduction of sales. The sales deductions are based primarily on estimates of the amounts earned or that will be claimed on such sales.

The company periodically and systematically evaluates the collectability of accounts receivable and determines the appropriate reserve for doubtful accounts. In determining the amount of the reserve, the company considers historical credit losses, the past-due status of receivables, payment history and other customer-specific information, and any other relevant factors or considerations.

The company also provides for the estimated costs that may be incurred under its warranty programs when the cost is both probable and reasonably estimable, which is at the time the related revenue is recognized. The cost is determined based on actual company experience for the same or similar products as well as other relevant information. Estimates of future costs under the company's

warranty programs could change based on developments in the future. The company is not able to estimate the probability or amount of any future developments that could impact the reserves, but believes presently established reserves are adequate.

Pension and Other Postemployment Benefit (OPEB) Plans

The company provides pension and other postemployment benefits to certain of its employees. These employee benefit expenses are reported in the same line items in the consolidated income statement as the applicable employee's compensation expense. The valuation of the funded status and net periodic benefit cost for the plans is calculated using actuarial assumptions. These assumptions are reviewed annually, and revised if appropriate. The significant assumptions include the following:

- interest rates used to discount pension and OPEB plan liabilities;
- the long-term rate of return on pension plan assets;
- rates of increases in employee compensation (used in estimating liabilities);
- anticipated future healthcare trend rates (used in estimating the OPEB plan liability); and
- other assumptions involving demographic factors such as retirement, mortality and turnover (used in estimating liabilities).

Selecting assumptions involves an analysis of both short-term and long-term historical trends and known economic and market conditions at the time of the valuation (also called the measurement date). The use of different assumptions would result in different measures of the funded status and net cost. Actual results in the future could differ from expected results. The company is not able to estimate the probability of actual results differing from expected results, but believes its assumptions are appropriate.

The company's key assumptions are listed in Note 13 in Item 8. The most critical assumptions relate to the plans covering U.S. and Puerto Rico employees, because these plans are the most significant to the company's consolidated financial statements.

Discount Rate Assumption

Effective for the December 31, 2016 measurement date, the company utilized discount rates of 4.09% and 3.89% to measure its benefit obligations for the U.S. and Puerto Rico pension plans and OPEB plan, respectively. The company used a broad population of approximately 200 Aa-rated corporate bonds as of December 31, 2016 to determine the discount rate assumption. All bonds were denominated in U.S. Dollars, with a minimum amount outstanding of \$50 million. This population of bonds was narrowed from a broader universe of approximately 700 Moody's Aa rated, non-callable (or callable with make-whole provisions) bonds by eliminating the top 10th percentile and bottom 40th percentile to adjust for any pricing anomalies and to represent the bonds Baxter would most likely select if it were to actually annuitize its pension and OPEB plan liabilities. This portfolio of bonds was used to generate a yield curve and associated spot rate curve to discount the projected benefit payments for the U.S. and Puerto Rico plans. The discount rate is the single level rate that produces the same result as the spot rate curve.

For plans in Canada, Japan, the United Kingdom and the Eurozone, the company uses a method essentially the same as that described for the U.S. and Puerto Rico plans. For the company's other international plans, the discount rate is generally determined by reviewing country- and region-specific government and corporate bond interest rates.

To understand the impact of changes in discount rates on pension and OPEB plan cost, the company performs a sensitivity analysis. Holding all other assumptions constant, for each 50 basis point (i.e., one-half of one percent) increase in the discount rate, global pre-tax pension and OPEB plan cost would decrease by approximately \$36 million, and for each 50 basis point decrease in the discount rate, global pre-tax pension and OPEB plan cost would increase by approximately \$40 million.

Effective January 1, 2016, the company changed its approach used to calculate the service and interest components of net periodic benefit cost. Previously, the company calculated the service and interest components utilizing a single weighted-average discount rate derived from the yield curve used to measure the benefit obligation. The company elected an alternative approach that utilizes a full yield curve approach in the estimation of these components by applying the specific spot rates along the yield curve used in the determination of the benefit obligation to their underlying projected cash flows. The company believes this approach provides a more precise measurement of service and interest costs by improving the correlation between projected benefit cash flows and their corresponding spot rates. The company accounted for this change prospectively as a change in estimate. As a result of this change, the service cost and interest cost for these plans was reduced by \$40 million in 2016 compared to the previous method.

Return on Plan Assets Assumption

In measuring the net periodic cost for 2016, the company used a long-term expected rate of return of 7.00% for the pension plans covering U.S. and Puerto Rico employees. This assumption will decrease to 6.50% in 2017. This assumption is not applicable to the company's OPEB plan because it is not funded.

The company establishes the long-term asset return assumption based on a review of historical compound average asset returns, both company-specific and relating to the broad market (based on the company's asset allocation), as well as an analysis of current market and economic information and future expectations. The current asset return assumption is supported by historical market experience for both the company's actual and targeted asset allocation. In calculating net pension cost, the expected return on assets is applied to a calculated value of plan assets, which recognizes changes in the fair value of plan assets in a systematic manner over five years. The difference between this expected return and the actual return on plan assets is a component of the total net unrecognized gain or loss and is subject to amortization in the future.

To understand the impact of changes in the expected asset return assumption on net cost, the company performs a sensitivity analysis. Holding all other assumptions constant, for each 50 basis point increase (decrease) in the asset return assumption, global pre-tax pension plan cost would decrease (increase) by approximately \$22 million.

Other Assumptions

For the U.S. and Puerto Rico plans, beginning with the December 31, 2014 measurement date, the company used the RP 2014 combined mortality table adjusted to reflect Baxter specific past experience with improvements projected using the generational BB-2D projection scale adjusted to a long term improvement of 0.8% in 2027. For all other pension plans, the company utilized country- and region-specific mortality tables to calculate the plans' benefit obligations. The company periodically analyzes and updates its assumptions concerning demographic factors such as retirement, mortality and turnover, considering historical experience as well as anticipated future trends.

The assumptions relating to employee compensation increases and future healthcare costs are based on historical experience, market trends, and anticipated future company actions. Refer to Note 13 in Item 8 for information regarding the sensitivity of the OPEB plan obligation and the total of the service and interest cost components of OPEB plan cost to potential changes in future healthcare trend rates.

Legal Contingencies

The company is involved in product liability, patent, commercial, regulatory and other legal proceedings that arise in the normal course of business. Refer to Note 16 in Item 8 for further information. The company records a liability when a loss is considered probable and the amount can be reasonably estimated. If the reasonable estimate of a probable loss is a range, and no amount within the range is a better estimate, the minimum amount in the range is accrued. If a loss is not probable or a probable loss cannot be reasonably estimated, no liability is recorded. The company has established reserves for certain of its legal matters. At December 31, 2016, total legal liabilities were \$53 million.

The company's loss estimates are generally developed in consultation with outside counsel and are based on analyses of potential outcomes. With respect to the recording of any insurance recoveries, after completing the assessment and accounting for the company's legal contingencies, the company separately and independently analyzes its insurance coverage and records any insurance recoveries that are probable of occurring at the gross amount that is expected to be collected. In performing the assessment, the company reviews available information, including historical company-specific and market collection experience for similar claims, current facts and circumstances pertaining to the particular insurance claim, the financial viability of the applicable insurance company or companies, and other relevant information.

While the liability of the company in connection with certain claims cannot be estimated and although the resolution in any reporting period of one or more of these matters could have a significant impact on the company's results of operations and cash flows for that period, the outcome of these legal proceedings is not expected to have a material adverse effect on the company's consolidated financial position. While the company believes it has valid defenses in these matters, litigation is inherently uncertain, excessive verdicts do occur, and the company may in the future incur material judgments or enter into material settlements of claims.

Deferred Tax Asset Valuation Allowances and Reserves for Uncertain Tax Positions

The company maintains valuation allowances unless it is more likely than not that all or a portion of the deferred tax asset will be realized. Changes in valuation allowances are included in the company's tax provision in the period of change. In determining whether a valuation allowance is warranted, the company evaluates factors such as prior earnings history, expected future earnings, carryback

and carry forward periods, and tax strategies that could potentially enhance the likelihood of realization of a deferred tax asset. The realizability assessments made at a given balance sheet date are subject to change in the future, particularly if earnings of a subsidiary are significantly higher or lower than expected, or if the company takes operational or tax planning actions that could impact the future taxable earnings of a subsidiary.

In the normal course of business, the company is audited by federal, state and foreign tax authorities, and is periodically challenged regarding the amount of taxes due. These challenges relate to the timing and amount of deductions and the allocation of income among various tax jurisdictions. The company believes its tax positions comply with applicable tax law and the company intends to defend its positions. In evaluating the exposure associated with various tax filing positions, the company records reserves for uncertain tax positions in accordance with GAAP, based on the technical support for the positions, the company's past audit experience with similar situations, and potential interest and penalties related to the matters. The company's results of operations and effective tax rate in a given period could be impacted if, upon final resolution with taxing authorities, the company prevailed in positions for which reserves have been established, or was required to pay amounts in excess of established reserves.

Valuation of Intangible Assets, Including IPR&D

The company acquires intangible assets and records them at fair value. Valuations are generally completed for business acquisitions using a discounted cash flow analysis, incorporating the stage of completion and consideration of market participant assumptions. The most significant estimates and assumptions inherent in a discounted cash flow analysis include the amount and timing of projected future cash flows, the discount rate used to measure the risks inherent in the future cash flows, the assessment of the asset's life cycle, and the competitive and other trends impacting the asset, including consideration of technical, legal, regulatory, economic and other factors. Each of these factors and assumptions can significantly affect the value of the intangible asset.

Acquired in-process R&D (IPR&D) is the value assigned to acquired technology or products under development which have not received regulatory approval and have no alternative future use. Acquired IPR&D included in a business combination is capitalized as an indefinite-lived intangible asset. Development costs incurred after the acquisition are expensed as incurred. Upon receipt of regulatory approval of the related technology or product, the indefinite-lived intangible asset is then accounted for as a finite-lived intangible asset and amortized on a straight-line basis over its estimated useful life. If the R&D project is abandoned, the indefinite-lived asset is charged to expense.

R&D acquired in transactions that are not business combinations is expensed immediately. For such transactions, payments made to third parties on or after regulatory approval are capitalized and amortized over the remaining useful life of the related asset, and are classified as intangible assets.

Due to the inherent uncertainty associated with R&D projects, there is no assurance that actual results will not differ materially from the underlying assumptions used to prepare discounted cash flow analyses, nor that the R&D project will result in a successful commercial product.

Impairment of Assets

Goodwill and other indefinite-lived intangible assets are subject to impairment reviews annually, and whenever indicators of impairment exist. The company assesses goodwill for impairment based on its reporting units, which are the same as its operating segments, Renal and Hospital Products. As of December 31, 2016, the date of the company's annual impairment review, the fair value of the company's reporting units were in excess of their carrying values. The company performs a qualitative assessment of other indefinite-lived intangible assets, including IPR&D, at least annually. If the intangible asset is determined to be more likely than not impaired as a result of the assessment, the company completes a quantitative impairment test. Intangible assets with definite lives and other long-lived assets (such as fixed assets) are reviewed for impairment whenever events or changes in circumstances indicate that the carrying amount of an asset may not be recoverable. Refer to Note 1 in Item 8 for further information. The company's impairment reviews are based on an estimated future cash flow approach that requires significant judgment with respect to future volume, revenue and expense growth rates, changes in working capital use, foreign currency exchange rates, the selection of an appropriate discount rate, asset groupings, and other assumptions and estimates. The estimates and assumptions used are consistent with the company's business plans and when applicable, market participant's views of the company and similar companies. The use of alternative estimates and assumptions could increase or decrease the estimated fair values of the assets, and potentially result in different impacts to the company's results of operations. Actual results may differ from the company's estimates.

Stock-Based Compensation Plans

Stock-based compensation cost is estimated at the grant date based on the fair value of the award, and the cost is recognized as expense ratably over the substantive vesting period. The company's stock compensation costs primarily relate to awards of stock

options, restricted stock units (RSUs), and performance share units (PSUs). The company uses the Black-Scholes model for estimating the fair value of stock options, and significant assumptions include long-term projections regarding stock price volatility, employee exercise, post-vesting termination and prevesting forfeiture behaviors, interest rates and dividend yields. The company's expected volatility assumption is based on a weighted-average of the historical volatility of Baxter's stock and the implied volatility from traded options on Baxter's stock, with historical volatility more heavily weighted. The expected life assumption is primarily based on the vesting terms of the stock option, historical employee exercise patterns and employee post-vesting termination behavior. The risk-free interest rate for the expected life of the option is based on the U.S. Treasury yield curve in effect at the time of grant. The dividend yield reflects historical experience as well as future expectations over the expected life of the option.

The fair value of RSUs is equal to the quoted price of the company's common stock on the date of grant.

PSUs granted in 2016 are based either on adjusted operating margin, return on invested capital (ROIC), or are based upon Baxter stock performance relative to the company's peer group. The vesting condition for such PSUs based on adjusted operating margin or ROIC have annual performance targets set at the beginning of the year for each tranche of the award during the three-year service period. The holder of the adjusted operating margin or ROIC PSUs is entitled to receive a number of shares of common stock equal to a percentage, ranging from 0% to 200%, of the adjusted operating margin or ROIC PSUs granted, depending on the actual results compared to the annual performance targets as such results may be adjusted for individual performance. Compensation cost for the adjusted operating margin or ROIC PSUs is measured based on the fair value of the awards on the date that the specific vesting terms for each tranche of the award are established. The fair value of the awards is determined based on the quoted price of the company's stock on the grant date for each tranche of the award. The compensation cost for adjusted operating margin or ROIC PSUs is adjusted at each reporting date to reflect the estimated probability of achieving the adjusted operating margin or ROIC vesting condition. The probability of achieving the operating margin vesting condition is such that the compensation cost has been adjusted to reflect 200% attainment as of the year ended December 31, 2016. The vesting condition for PSUs based on Baxter stock performance relative to the company's peer group is fair valued using a Monte Carlo model. A Monte Carlo model uses stock price volatility and other variables to estimate the probability of satisfying the market conditions and the resulting fair value of the award. Refer to Note 12 in Item 8 for additional information.

CERTAIN REGULATORY MATTERS

In January 2014, the company received a Warning Letter from FDA primarily directed to quality systems for the company's Round Lake, Illinois, facility, particularly in that facility's capacity as a specification developer for certain of the company's medical devices. This Warning Letter was lifted in February 2017.

The company received a Warning Letter in December 2013 that included observations related to the company's ambulatory infuser business in Irvine, California, which previously had been subject to agency action.

In June 2013, the company received a Warning Letter from FDA regarding operations and processes at its North Cove, North Carolina and Jayuya, Puerto Rico facilities and in November 2015 attended a Regulatory Meeting with FDA concerning the Jayuya facility. The Warning Letter addresses observations related to Current Good Manufacturing Practice (CGMP) violations at the two facilities.

In June 2010, the company received a Warning Letter from FDA in connection with an inspection of its McGaw Park, Illinois facility, which previously supported the Renal franchise. The company's Round Lake facility now provides the related capacity for the Renal franchise. The Warning Letter pertains to the processes by which the company analyzes and addresses product complaints through corrective and preventative action, and reports relevant information to FDA. This Warning Letter was lifted in February 2017.

On October 9, 2014, the company had a Regulatory Meeting with FDA to discuss the Warning Letters described above. At the meeting, the company agreed to work closely with FDA to provide regular updates on its progress to meet all requirements and resolve all matters identified in the Warning Letters described above.

Refer to Item 1A of this Annual Report on Form 10-K for additional discussion of regulatory matters and how they may impact the company.

FORWARD-LOOKING INFORMATION

This annual report includes forward-looking statements. Use of the words "may," "will," "would," "could," "should," "believes," "estimates," "projects," "potential," "expects," "plans," "seeks," "intends," "evaluates," "pursues," "anticipates," "continues," "designs," "impacts," "affects," "forecasts," "target," "outlook," "initiative," "objective," "designed," "priorities," "goal," or the negative of those words or other similar expressions is intended to identify forward-looking statements that represent our current judgment about possible future events. These forward-looking statements may include statements with respect to accounting estimates

and assumptions, litigation-related matters including outcomes, future regulatory filings and the company's R&D pipeline, strategic objectives, credit exposure to foreign governments, potential developments with respect to credit ratings, investment of foreign earnings, estimates of liabilities including those related to uncertain tax positions, contingent payments, future pension plan contributions, costs, discount rates and rates of return, the company's exposure to financial market volatility and foreign currency and interest rate risks, potential tax liability associated with the separation of the company's biopharmaceuticals and medical products businesses (including the 2016 disposition of the company's retained stake in Baxalta), the impact of competition, future sales growth, business development activities, business optimization initiatives, cost saving initiatives, future capital and R&D expenditures, future debt issuances, manufacturing expansion, the sufficiency of the company's facilities and financial flexibility, the adequacy of credit facilities, tax provisions and reserves, the effective tax rate and all other statements that do not relate to historical facts.

These forward-looking statements are based on certain assumptions and analyses made in light of the company's experience and perception of historical trends, current conditions, and expected future developments as well as other factors that the company believes are appropriate in the circumstances. While these statements represent the company's current judgment on what the future may hold, and the company believes these judgments are reasonable, these statements are not guarantees of any events or financial results. Whether actual future results and developments will conform to expectations and predictions is subject to a number of risks and uncertainties, including the following factors, many of which are beyond our control:

- failure to achieve our long-term financial improvement goals;
- demand for and market acceptance risks for and competitive pressures related to new and existing products, and the impact of those products on quality and patient safety concerns;
- product development risks, including satisfactory clinical performance, the ability to manufacture at appropriate scale, and the general unpredictability associated with the product development cycle;
- product quality or patient safety issues, leading to product recalls, withdrawals, launch delays, sanctions, seizures, litigation, or declining sales;
- future actions of FDA, EMA or any other regulatory body or government authority that could delay, limit or suspend product development, manufacturing or sale or result in seizures, recalls, injunctions, monetary sanctions or criminal or civil liabilities;
- failures with respect to the company's compliance programs;
- future actions of third parties, including third-party payers, as healthcare reform and other similar measures are implemented, modified or repealed in the United States and globally;
- the impact of ongoing U.S. healthcare reform and other similar actions undertaken by foreign governments with respect to pricing, reimbursement, taxation and rebate policies;
- additional legislation, regulation and other governmental pressures in the United States or globally, which may affect pricing, reimbursement, taxation and rebate policies of government agencies and private payers or other elements of the company's business;
- the impact of competitive products and pricing, including generic competition, drug reimportation and disruptive technologies;
- global regulatory, trade and tax policies;
- the company's ability to identify business development and growth opportunities and to successfully execute on business development strategies;
- · the company's ability to finance and develop new products or enhancements, on commercially acceptable terms or at all;
- the availability and pricing of acceptable raw materials and component supply;
- inability to create additional production capacity in a timely manner or the occurrence of other manufacturing or supply difficulties;
- the ability to protect or enforce the company's owned or in-licensed patent or other proprietary rights (including trademarks, copyrights, trade secrets and know-how) or patents of third parties preventing or restricting the company's manufacture, sale or use of affected products or technology;
- the impact of any future tax liability with respect to the separation and distribution, including with respect to disposition of the Retained Shares;

- any failure by Baxalta or Shire to satisfy its obligation under the separation agreements, including the tax matters agreement, or the Letter Agreement;
- the impact of global economic conditions on the company and its customers and suppliers, including foreign governments in certain countries in which the company operates;
- fluctuations in foreign exchange and interest rates;
- any changes in law concerning the taxation of income, including income earned outside the United States, which may be a part of comprehensive tax reform;
- actions by tax authorities in connection with ongoing tax audits;
- breaches or failures of the company's information technology systems;
- loss of key employees or inability to identify and recruit new employees;
- the outcome of pending or future litigation;
- the adequacy of the company's cash flows from operations to meet its ongoing cash obligations and fund its investment program; and
- other factors identified elsewhere in this Annual Report on Form 10-K including those factors described in Item 1A and other filings with the Securities and Exchange Commission, all of which are available on the company's website.

Actual results may differ materially from those projected in the forward-looking statements. The company does not undertake to update its forward-looking statements.

Item 7A. Quantitative and Qualitative Disclosures About Market Risk.

Incorporated by reference to the section entitled "Financial Instrument Market Risk" in "Management's Discussion and Analysis of Financial Condition and Results of Operations" in Item 7 of this Annual Report on Form 10-K.

Item 8. Financial Statements and Supplementary Data.

CONSOLIDATED BALANCE SHEETS

as of December 31 (in millions, except share	information)	2016	2015
Current assets	Cash and equivalents	\$ 2,801	\$ 2,213
	Accounts and other current receivables, net	1,691	1,731
	Inventories	1,430	1,604
	Prepaid expenses and other	602	855
	Investment in Baxalta common stock	_	5,148
	Current assets held for disposition	50	245
	Total current assets	6,574	11,796
Property, plant and equipment, net		4,289	4,386
Other assets	Goodwill	2,595	2,687
	Other intangible assets, net	1,111	1,349
	Other	977	744
	Total other assets	4,683	4,780
	Total assets	\$ 15,546	\$ 20,962
Current liabilities	Short-term debt	\$ _	\$ 1,775
	Current maturities of long-term debt and lease obligations	3	810
	Accounts payable and accrued liabilities	2,612	2,666
	Current income taxes payable	126	453
	Current liabilities held for disposition	3	46
	Total current liabilities	2,744	5,750
Long-term debt and lease obligations		2,779	3,922
Other long-term liabilities		1,743	2,425
Equity	Common stock, \$1 par value, authorized 2,000,000,000 shares, issued 683,494,944 shares in 2016 and 2015	683	683
	Common stock in treasury, at cost, 143,890,064 shares in 2016 and 135,839,938 shares in 2015	(7,995)	(7,646)
	Additional contributed capital	5,958	5,902
	Retained earnings	14,200	9,683
	Accumulated other comprehensive (loss) income	(4,556)	224
	Total Baxter shareholders' equity	8,290	8,846
	Noncontrolling interests	(10)	19
	Total equity	8,280	8,865
	Total liabilities and equity	\$ 15,546	\$ 20,962

CONSOLIDATED STATEMENTS OF INCOME

years ended December 31 (in millions, except per share data)		2016		2015	2014
Net sales	\$	10,163	\$	9,968	\$ 10,719
Cost of sales		6,053		5,822	6,138
Gross margin		4,110		4,146	4,581
Marketing and administrative expenses		2,739		3,094	3,315
Research and development expenses		647		603	610
Operating income		724		449	656
Net interest expense		66		126	145
Other (income) expense, net		(4,296)		(105)	21
Income from continuing operations before income taxes		4,954		428	490
Income tax (benefit) expense		(12)		35	33
Income from continuing operations		4,966		393	457
(Loss) income from discontinued operations, net of tax		(1)		575	2,040
Net income	\$	4,965	\$	968	\$ 2,497
Income from continuing operations per common share					
Basic	\$	9.10	\$	0.72	\$ 0.84
Diluted	\$	9.01	\$	0.72	\$ 0.83
(Loss) income from discontinued operations per common share					
Basic	\$	(0.01)	\$	1.06	\$ 3.77
Diluted	\$	0.00	\$	1.04	\$ 3.73
Net income per common share					
Basic	\$	9.09	\$	1.78	\$ 4.61
Diluted	\$	9.01	\$	1.76	\$ 4.56
Weighted-average number of common shares outstanding					
Basic		546		545	542
Diluted	•	551	•	549	547

CONSOLIDATED STATEMENTS OF COMPREHENSIVE INCOME

years ended December 31 (in millions)	2016	2015	2014
Net income	\$ 4,965	\$ 968	\$ 2,497
Other comprehensive (loss) income, net of tax:			
Currency translation adjustments, net of tax benefit of (\$39) in 2016, (\$107) in 2015			
and (\$132) in 2014	(247)	(1,094)	(1,332)
Pension and other employee benefits, net of tax (benefit) expense of (\$36) in 2016, \$104 in 2015 and (\$193) in 2014	(07)	165	(400)
	(97)	103	(400)
Hedging activities, net of tax (benefit) expense of (\$2) in 2016, \$9 in 2015 and \$14 in 2014	(4)	15	24
Available-for-sale securities, net of tax expense (benefit) of zero in 2016, \$6 in 2015			
and (\$2) in 2014	(4,432)	4,438	34
Total other comprehensive income (loss), net of tax	(4,780)	3,524	(1,674)
Comprehensive income	\$ 185	\$ 4,492	\$ 823

CONSOLIDATED STATEMENTS OF CASH FLOWS

years ended December 31 (in millions) (brackets denote cash outflows)	2016	2015	2014
Cash flows from	Net income	\$ 4,965	\$ 968	\$ 2,497
operations	Adjustments to reconcile income from continuing operations to net cash from operating activities:			
	Loss (income) from discontinued operations, net of tax	1	(575)	(2,040)
	Depreciation and amortization	800	759	792
	Deferred income taxes	(302)	(50)	(117)
	Stock compensation	115	126	126
	Realized excess tax benefits from stock issued under employee benefit plans	(39)	(7)	(15)
	Net periodic pension benefit and OPEB costs	116	227	219
	Business optimization items	285	130	(6)
	Net realized gains on Baxalta common stock	(4,387)	_	_
	Infusion pump and other product-related charges	(18)	(28)	93
	Other	264	42	19
	Changes in balance sheet items			
	Accounts and other current receivables, net	15	(4)	(93)
	Inventories	80	(118)	(143)
	Accounts payable and accrued liabilities	(197)	236	(37)
	Business optimization and infusion pump payments	(189)	(112)	(124)
	Other	115	(341)	(17)
	Cash flows from operations – continuing operations	1,624	1,253	1,154
	Cash flows from operations – discontinued operations	30	518	2,061
	Cash flows from operations	1,654	1,771	3,215
Cash flows from	Capital expenditures	(719)	(911)	(925)
investing activities	Acquisitions and investments, net of cash acquired	(48)	(34)	(95)
	Divestitures and other investing activities	37	84	99
	Cash flows from investing activities – continuing operations	(730)	(861)	(921)
	Cash flows from investing activities – discontinued operations	15	(946)	(621)
	Cash flows from investing activities	(715)	(1,807)	(1,542)
Cash flows from	Issuances of debt	1,641	6,868	41
financing activities	Payments of obligations	(1,381)	(3,786)	(1,029)
	Debt extinguishment costs	(16)	(114)	_
	(Decrease) increase in debt with original maturities of three months or less, net	(300)	(575)	875
	Transfer of cash and equivalents to Baxalta	_	(2,122)	_
	Cash dividends on common stock	(268)	(910)	(1,095)
	Proceeds and realized excess tax benefits from stock issued under employee benefit plans	325	200	369
	Purchases of treasury stock	(292)	_	(550)
	Other	(33)	(42)	(13)
	Cash flows from financing activities	(324)	(481)	(1,402)
Effect of foreign excha	ange rate changes on cash and equivalents	(27)	(195)	(79)
Increase (decrease) in		588	(712)	192
Cash and equivalents a	•	2,213	2,925	2,733
Cash and equivalents a		\$ 2,801	\$ 2,213	\$ 2,925
	of non-cash investing and financing activities			
Net proceeds on Retain		\$ 4,387	\$ _	\$ _
	in exchange for Retained Shares	\$ 3,646	\$	\$ _
	ares with Retained Shares	\$ 611	\$ _	\$ _
Other supplemental in				
Interest paid, net of por		\$ 99	\$ 178	\$ 208
Income taxes paid		\$ 500	\$ 466	\$ 726

CONSOLIDATED STATEMENTS OF CHANGES IN EQUITY

		2016	<u> </u>	2015			2014		
as of and for the years ended December 31 (in millions)	Shares	A	Amount	Shares	A	Amount	Shares	A	mount
Common stock									
Balance, beginning and end of year	683	\$	683	683	\$	683	683	\$	683
Common stock in treasury									
Beginning of year	136		(7,646)	141		(7,993)	140		(7,914)
Purchases of common stock	18		(902)	_		_	8		(550)
Stock issued under employee benefit plans and other	(10)		553	(5)		347	(7)		471
End of year	144		(7,995)	136		(7,646)	141		(7,993)
Additional contributed capital									
Beginning of year			5,902			5,853			5,818
Stock issued under employee benefit plans and other			43			49			35
Other			13						
End of year			5,958			5,902			5,853
Retained earnings									
Beginning of year			9,683			13,227			11,852
Net income			4,965			968			2,497
Dividends declared on common stock			(276)			(695)			(1,116)
Stock issued under employee benefit plans			(190)			(90)			(6)
Distribution of Baxalta			18			(3,727)			
End of year			14,200			9,683			13,227
Accumulated other comprehensive income (loss)									
Beginning of year			224			(3,650)			(1,976)
Other comprehensive income (loss)			(4,780)			3,524			(1,674)
Distribution of Baxalta						350			
End of year			(4,556)			224			(3,650)
Total Baxter shareholders' equity		\$	8,290		\$	8,846		\$	8,120
Noncontrolling interests									
Beginning of year		\$	19		\$	36		\$	23
Change in noncontrolling interests			(29)			(17)			13
End of year		\$	(10)		\$	19		\$	36
Total equity		\$	8,280		\$	8,865		\$	8,156

NOTES TO CONSOLIDATED FINANCIAL STATEMENTS

NOTE 1

SUMMARY OF SIGNIFICANT ACCOUNTING POLICIES

Nature of Operations

Baxter International Inc., through its subsidiaries, provides a broad portfolio of essential renal and hospital products, including acute and chronic dialysis; sterile IV solutions; infusion systems and devices; parenteral nutrition therapies; premixed and oncolytic injectables; biosurgery products and anesthetics; drug reconstitution systems; and pharmacy automation, software and services. The company's global footprint and the critical nature of its products and services play a key role in expanding access to healthcare in emerging and developed countries. These products are used by hospitals, kidney dialysis centers, nursing homes, rehabilitation centers, doctors' offices and by patients at home under physician supervision. The company operates in two segments, Renal and Hospital Products, which are described in Note 17.

Use of Estimates

The preparation of the financial statements in conformity with generally accepted accounting principles (GAAP) requires the company to make estimates and assumptions that affect reported amounts and related disclosures. Actual results could differ from those estimates.

Basis of Presentation

The consolidated financial statements include the accounts of Baxter and its majority-owned subsidiaries that Baxter controls, after elimination of intercompany transactions.

On July 1, 2015, Baxter completed the distribution of approximately 80.5% of the outstanding common stock of Baxalta Incorporated (Baxalta), to Baxter shareholders (the Distribution). The Distribution was made to Baxter's shareholders of record as of the close of business on June 17, 2015 (the Record Date), who received one share of Baxalta common stock for each Baxter common share held as of the Record Date. As a result of the Distribution, Baxalta became an independent public company trading under the symbol "BXLT" on the New York Stock Exchange.

In 2016, Baxter disposed of its remaining 19.5% interest in Baxalta through a series of transactions including debt-for-equity exchanges, an equity-for-equity exchange and a contribution to its U.S. pension plan. As a result of these transactions, the company extinguished approximately \$3.65 billion in company indebtedness, repurchased 11,526,638 Baxter shares and contributed 17,145,570 Baxalta shares to its U.S. pension plan. On June 3, 2016, Baxalta became a wholly-owned subsidiary of Shire plc (Shire).

References in this report to Baxalta prior to the Merger closing date refers to Baxalta as a stand-alone public company. References in this report to Baxalta subsequent to the Merger closing date refer to Baxalta as a subsidiary of Shire.

As a result of the separation, the consolidated statements of income, consolidated balance sheets, consolidated statements of cash flow, and related financial information reflect Baxalta's operations, assets and liabilities, and cash flows as discontinued operations for all periods presented. Refer to Note 2 for additional information regarding the separation of Baxalta.

Revenue Recognition

The company recognizes revenues from product sales and services when earned. Specifically, revenue is recognized when persuasive evidence of an arrangement exists, delivery has occurred (or services have been rendered), the price is fixed or determinable, and collectability is reasonably assured. For product sales, revenue is not recognized until title and risk of loss have transferred to the customer. The shipping terms for the majority of the company's revenue arrangements are FOB destination. The recognition of revenue is delayed if there are significant post-delivery obligations, such as training, installation or other services. Provisions for discounts, rebates to customers, chargebacks to wholesalers and returns are provided for at the time the related sales are recorded, and are reflected as a reduction to gross sales to arrive at net sales.

The company sometimes enters into arrangements in which it commits to delivering multiple products or services to its customers. In these cases, total arrangement consideration is allocated to the deliverables based on their relative selling prices. Then the allocated consideration is recognized as revenue in accordance with the principles described above. Selling prices are determined by applying a selling price hierarchy and by using vendor specific objective evidence (VSOE), if it exists. Otherwise, selling prices are determined using third party evidence (TPE). If neither VSOE nor TPE is available, the company uses its best estimate of selling prices.

Accounts Receivable and Allowance for Doubtful Accounts

In the normal course of business, the company provides credit to its customers, performs credit evaluations of these customers and maintains reserves for potential credit losses. In determining the amount of the allowance for doubtful accounts, the company considers, among other items, historical credit losses, the past-due status of receivables, payment histories and other customer-specific information. Receivables are written off when the company determines they are uncollectible. The allowance for doubtful accounts was \$127 million at December 31, 2016 and \$110 million at December 31, 2015.

Product Warranties

The company provides for the estimated costs relating to product warranties at the time the related revenue is recognized. The cost is determined based on actual company experience for the same or similar products, as well as other relevant information. Product warranty liabilities are adjusted based on changes in estimates.

Cash and Equivalents

Cash and equivalents include cash, certificates of deposit and money market funds with an original maturity of three months or less.

Inventories

as of December 31 (in millions)	2016	2015
Raw materials	\$ 319	\$ 374
Work in process	122	142
Finished goods	989	1,088
Inventories	\$ 1,430	\$ 1,604

Inventories are stated at the lower of cost (first-in, first-out method) or market value. Market value for raw materials is based on replacement costs, and market value for work in process and finished goods is based on net realizable value. The company reviews inventories on hand at least quarterly and records provisions for estimated excess, slow-moving and obsolete inventory, as well as inventory with a carrying value in excess of net realizable value.

Property, Plant and Equipment, Net

as of December 31 (in millions)	2016		2015
Land	\$	118	\$ 116
Buildings and leasehold improvements		1,486	1,389
Machinery and equipment		5,551	5,414
Equipment with customers		1,297	1,238
Construction in progress		710	833
Total property, plant and equipment, at cost		9,162	8,990
Accumulated depreciation		(4,873)	(4,604)
Property, plant and equipment (PP&E), net	\$	4,289	\$ 4,386

Depreciation expense is calculated using the straight-line method over the estimated useful lives of the related assets, which range from 20 to 50 years for buildings and improvements and from three to 15 years for machinery and equipment. Leasehold improvements are amortized over the life of the related facility lease (including any renewal periods, if appropriate) or the asset, whichever is shorter. Baxter capitalizes certain computer software and software development costs incurred in connection with developing or obtaining software for internal use as part of machinery and equipment. Capitalized software costs are amortized on a straight-line basis over the estimated useful lives of the software, and are included in depreciation expense. Straight-line and accelerated methods of depreciation are used for income tax purposes. Depreciation expense was \$632 million in 2016, \$597 million in 2015 and \$613 million in 2014. Depreciation expense in 2016 included accelerated depreciation of \$48 million related to business optimization and separation costs.

Acquisitions

Results of operations of acquired companies are included in the company's results of operations as of the respective acquisition dates. The purchase price of each acquisition is allocated to the net assets acquired based on estimates of their fair values at the date of the acquisition. Any purchase price in excess of these net assets is recorded as goodwill. The allocation of purchase price in certain cases may be subject to revision based on the final determination of fair values during the measurement period, which may be up to one year from the acquisition date.

Contingent consideration is recognized at the estimated fair value on the acquisition date. Subsequent changes to the fair value of contingent payments are recognized in earnings as a component of other income (expense), net. Contingent payments related to acquisitions may consist of development, regulatory and commercial milestone payments, in addition to sales-based payments, and are valued using discounted cash flow techniques. The fair value of development, regulatory and commercial milestone payments reflects management's expectations of probability of payment, and increases or decreases as the probability of payment or expectation of timing of payments changes. The fair value of sales-based payments is based upon probability-weighted future revenue estimates and increases or decreases as revenue estimates or expectation of timing of payments changes.

Research and Development

Research and development (R&D) costs, including R&D acquired in transactions that are not business combinations, are expensed as incurred. Pre-regulatory approval contingent milestone obligations to counterparties in collaborative arrangements which include acquired R&D are expensed when the milestone is achieved. Contingent milestone payments made to such counterparties on or after regulatory approval are capitalized and amortized over the remaining useful life of the related product. Amounts capitalized for such payments are included in other intangible assets, net of accumulated amortization.

Acquired in-process R&D (IPR&D) is the value assigned to technology or products under development acquired in a business combination which have not received regulatory approval and have no alternative future use.

Acquired IPR&D is capitalized as an indefinite-lived intangible asset. Development costs incurred after the acquisition are expensed as incurred. Upon receipt of regulatory approval of the related technology or product, the indefinite-lived intangible asset is accounted for as a finite-lived intangible asset and amortized on a straight-line basis over the estimated economic life of the related technology or product, subject to annual impairment reviews as discussed below. If the R&D project is abandoned, the indefinite-lived asset is charged to expense.

Collaborative Arrangements

The company enters into collaborative arrangements in the normal course of business. These collaborative arrangements take a number of forms and structures, and are designed to enhance and expedite long-term sales and profitability growth. These arrangements may provide that Baxter obtain commercialization rights to a product under development, and require Baxter to make upfront payments, contingent milestone payments, profit-sharing, and/or royalty payments. Baxter may be responsible for ongoing costs associated with the arrangements, including R&D cost reimbursements to the counterparty. See above regarding the accounting treatment of upfront and contingent payments. Any royalty and profit-sharing payments during the commercialization phase are expensed as cost of sales when they become due and payable.

Business Optimization Charges

The company records liabilities for costs associated with exit or disposal activities in the period in which the liability is incurred. Employee termination costs are primarily recorded when actions are probable and estimable. Costs for one-time termination benefits in which the employee is required to render service until termination in order to receive the benefits are recognized ratably over the future service period. Refer to the discussion below regarding the accounting for asset impairment charges.

Goodwill, Intangible Assets, and Other Long-Lived Assets

Goodwill is not amortized, but is subject to an impairment review annually and whenever indicators of impairment exist. Goodwill would be impaired if the carrying amount of a reporting unit exceeded the fair value of that reporting unit, calculated as the present value of estimated cash flows discounted using a risk-free market rate adjusted for a market participant's view of similar companies and perceived risks in the cash flows. The implied fair value of goodwill is then determined by subtracting the fair value of all identifiable net assets other than goodwill from the fair value of the reporting unit, with an impairment charge recorded for the excess, if any, of carrying amount of goodwill over the implied fair value.

Indefinite-lived intangible assets, such as IPR&D acquired in business combinations and certain trademarks with indefinite lives, are subject to an impairment review annually and whenever indicators of impairment exist. Indefinite-lived intangible assets are impaired if the carrying amount of the asset exceeded the fair value of the asset.

The company reviews the carrying amounts of long-lived assets, other than goodwill and intangible assets not subject to amortization, for potential impairment when events or changes in circumstances indicate that the carrying amount of an asset may not be recoverable. In evaluating recoverability, the company groups assets and liabilities at the lowest level such that the identifiable cash flows relating to the group are largely independent of the cash flows of other assets and liabilities. The company then compares the carrying amounts of the assets or asset groups with the related estimated undiscounted future cash flows. In the event impairment exists, an impairment charge is recorded as the amount by which the carrying amount of the asset or asset group exceeds the fair value.

Shipping and Handling Costs

Shipping costs, which are costs incurred to physically move product from Baxter's premises to the customer's premises, are classified as marketing and administrative expenses. Handling costs, which are costs incurred to store, move and prepare products for shipment, are classified as cost of sales. Approximately \$311 million in 2016, \$272 million in 2015 and \$319 million in 2014 of shipping costs were classified in marketing and administrative expenses.

Income Taxes

Deferred taxes are recognized for the future tax effects of temporary differences between financial and income tax reporting based on enacted tax laws and rates. The company maintains valuation allowances unless it is more likely than not that the deferred tax asset will be realized. With respect to uncertain tax positions, the company determines whether the position is more likely than not to be sustained upon examination, based on the technical merits of the position. Any tax position that meets the more likely than not recognition threshold is measured and recognized in the consolidated financial statements at the largest amount of benefit that is greater than 50% likely of being realized upon ultimate settlement. The liability relating to uncertain tax positions is classified as current in the consolidated balance sheets to the extent the company anticipates making a payment within one year. Interest and penalties associated with income taxes are classified in the income tax expense line in the consolidated statements of income.

Foreign Currency Translation

Currency translation adjustments (CTA) related to foreign operations are included in other comprehensive income (OCI). For foreign operations in highly inflationary economies, translation gains and losses are included in other expense (income), net, and were not material in 2016, 2015 and 2014.

Derivatives and Hedging Activities

All derivative instruments are recognized as either assets or liabilities at fair value in the consolidated balance sheets and are classified as short-term or long-term based on the scheduled maturity of the instrument. Based upon the exposure being hedged, the company designates its hedging instruments as cash flow or fair value hedges.

For each derivative instrument that is designated and effective as a cash flow hedge, the gain or loss on the derivative is accumulated in accumulated other comprehensive income (AOCI) and then recognized in earnings consistent with the underlying hedged item. Option premiums or net premiums paid are initially recorded as assets and reclassified to OCI over the life of the option, and then recognized in earnings consistent with the underlying hedged item. Cash flow hedges are classified in net sales, cost of sales, and net interest expense, and primarily related to forecasted third-party sales denominated in foreign currencies, forecasted intercompany sales denominated in foreign currencies and anticipated issuances of debt, respectively.

For each derivative instrument that is designated and effective as a fair value hedge, the gain or loss on the derivative is recognized into earnings, offsetting the loss or gain on the underlying hedged item. Fair value hedges are classified in net interest expense, as they hedge the interest rate risk associated with certain of the company's fixed-rate debt.

For derivative instruments that are not designated as hedges, the change in fair value is recorded directly to other expense (income), net.

If it is determined that a derivative or nonderivative hedging instrument is no longer highly effective as a hedge, the company discontinues hedge accounting prospectively. If the company removes the cash flow hedge designation because the hedged forecasted transactions are no longer probable of occurring, any gains or losses are immediately reclassified from AOCI to earnings. Gains or

losses relating to terminations of effective cash flow hedges in which the forecasted transactions are still probable of occurring are deferred and recognized consistent with the income or loss recognition of the underlying hedged items. If the company terminates a fair value hedge, an amount equal to the cumulative fair value adjustment to the hedged items at the date of termination is amortized to earnings over the remaining term of the hedged item.

Derivatives, including those that are not designated as a hedge, are principally classified in the operating section of the consolidated statements of cash flows in the same category as the related consolidated balance sheet account.

Refer to Note 9 for further information regarding the company's derivative and hedging activities.

New Accounting Standards

Recently issued accounting standards not yet adopted

In March 2016, the Financial Accounting Standards Board (FASB) issued Accounting Standards Update (ASU) No. 2016-09, Improvements to Employee Share-Based Payment Accounting, which amends ASC Topic 718, Compensation – Stock Compensation. The updated guidance requires all tax effects related to share-based payments be recorded in income tax expense in the consolidated statement of income. Current guidance requires that the tax effects of deductions in excess of share-based compensation costs (windfall tax benefits) be recorded in additional paid-in capital, and tax deficiencies (shortfalls) be recorded in additional paid-in capital to the extent of previously recognized windfall tax benefits, with the remainder recorded in income tax expense. The new guidance also requires all tax-related cash flows resulting from share-based payments to be reported as operating activities in the consolidated statement of cash flows, rather than the current requirement to present windfall tax benefits as an inflow from financing activities and an outflow from operating activities. The guidance is effective for the company beginning January 1, 2017. The impact of the standard is dependent on the timing and value of award exercises and vesting. The company has evaluated the impact of this standard on its consolidated financial statements for 2016, and determined that net income and operating cash flow for the year would have each increased by approximately \$39 million if the company had adopted the new standard January 1, 2016.

In February 2016, the FASB issued ASU No. 2016-02, Leases (Topic 842). Under the new guidance, lessees are required to recognize a right-of-use asset and a lease liability on the balance sheet for all leases, other than those that meet the definition of a short-term lease. This update will establish a lease asset and lease liability by lessees for those leases classified as operating under current GAAP. Leases will be classified as either operating or finance under the new guidance. Operating leases will result in straight-line expense in the income statement, similar to current operating leases, and finance leases will result in more expense being recognized in the earlier years of the lease term, similar to current capital leases. This ASU is effective for the company beginning January 1, 2019. The company is currently evaluating the impact of this standard on its consolidated financial statements.

In May 2014, the FASB issued ASU No. 2014-09, Revenue from Contracts with Customers (Topic 606), which amends the existing accounting standards for revenue recognition. ASU No. 2014-09 is based on principles that govern the recognition of revenue at an amount an entity expects to be entitled when products are transferred to customers. ASU No. 2014-09 will be effective for the company beginning on January 1, 2018. The standard may be applied retrospectively to each prior period presented or retrospectively with the cumulative effect recognized as of the date of adoption. The company has completed an assessment of the new standard and is in the process of finalizing its detailed implementation plan and evaluating the disclosure requirements under the new standard. Based on the work performed to date, the company does not expect the adoption of the new standard to have a material impact on the consolidated financial statements. The company has not finalized its transition method for adoption.

Recently adopted accounting pronouncements

As of January 1, 2016, the company adopted ASU No. 2015-03, Simplifying the Presentation of Debt Issuance Costs, which amended ASC 835-30, Interest – Imputation of Interest. This guidance requires that debt issuance costs related to a recognized debt liability be presented as a direct deduction from the carrying amount of the related debt liability. As a result of the adoption, the company reclassified debt issuance costs of \$13 million from other assets to long-term debt in the company's consolidated balance sheet as of December 31, 2015. The adoption of this guidance did not impact the company's consolidated statements of income, comprehensive income, changes in equity or cash flows.

As of January 1, 2016, the company adopted ASU No. 2015-05, Intangibles — Goodwill and Other — Internal-Use Software (Subtopic 350-40), Customer's Accounting for Fees Paid in a Cloud Computing Arrangement. This update provides guidance on determining whether a cloud-based computer arrangement includes a software license. If it is determined that the arrangement includes a software license, then the customer is to account for that element in a manner that is consistent with the acquisition of other software licenses. If it is determined that the arrangement does not include a software license, then it is to be accounted for as a service

contract. The company elected to adopt the amendments prospectively to all arrangements entered into or materially modified after the effective date. The adoption of ASU No. 2015-05 did not have a material impact on the company's consolidated financial statements.

As of July 1, 2016, the company adopted ASU No. 2016-15, Statement of Cash Flows (Topic 230). The guidance requires that the cash payments for debt prepayment or debt extinguishment costs be classified as cash outflows for financing activities. As a result of the adoption, in the third quarter of 2016 the company reclassified certain debt repayments and debt extinguishment costs from operating to financing activities which resulted in a decrease in financing cash flows of \$16 million and \$124 million for 2016 and 2015, respectively. The adoption of this guidance did not impact the company's consolidated statements of income, consolidated balance sheets, comprehensive income or changes in equity.

NOTE 2

SEPARATION OF BAXALTA INCORPORATED

After giving effect to the Distribution, the company retained 19.5% of the outstanding common stock, or 131,902,719 shares of Baxalta (Retained Shares). Effective January 27, 2016, Baxter completed a debt-for-equity exchange through the transfer of 37,573,040 Retained Shares in exchange for the extinguishment of the \$1.45 billion aggregate principal amount of indebtedness outstanding under the company's prior U.S. dollar denominated revolving credit facility, which was terminated in connection with the closing of this exchange. On March 16, 2016, the company completed a debt-for-equity exchange, in which Baxter exchanged 63,823,582 Retained Shares for the extinguishment of \$2.2 billion in aggregate principal amount of Baxter indebtedness. On May 6, 2016, the company contributed 17,145,570 Retained Shares to Baxter's U.S. pension fund. On May 26, 2016, the company completed an equity-for-equity exchange by exchanging 13,360,527 Retained Shares for 11,526,638 shares of Baxter. The company held no shares of Baxalta as of December 31, 2016. Refer to Note 10 for additional details regarding these transactions.

The following table is a summary of the operating results of Baxalta, which have been reflected as discontinued operations for the years ended December 31, 2016, 2015 and 2014.

Years ended December 31 (in millions)	2	2016 2015		2015		2014
Major classes of line items constituting income from discontinued operations before income taxes						
Net sales	\$	148	\$	2,895	\$	6,523
Cost of sales		(139)		(1,214)		(2,475)
Marketing and administrative expenses		(20)		(547)		(769)
Research and development expenses		_		(389)		(822)
Other income and expense items that are not major		1		7		105
Total (loss) income from discontinued operations before income taxes		(10)		752		2,562
Gain on disposal of discontinued operations		19		_		_
Income tax expense		10		177		522
Total (loss) income from discontinued operations	\$	(1)	\$	575	\$	2,040

For a portion of Baxalta's operations, the legal transfer of Baxalta's assets and liabilities did not occur with the separation of Baxalta on July 1, 2015 due to the time required to transfer marketing authorizations and other regulatory requirements in certain countries. Under the terms of the International Commercial Operations Agreement (ICOA), Baxalta is subject to the risks and entitled to the benefits generated by these operations and assets until legal transfer; therefore, the net economic benefit and any cash collected by these entities are transferred to Baxalta. Separation of the remaining three countries is expected to occur by 2018.

The assets and liabilities of Baxalta have been classified as held for disposition as of December 31, 2016 and 2015. These amounts consist of the following carrying amounts in each major class.

As of December 31 (in millions)	2016		2015
Carrying amounts of major classes of assets included as part of discontinued operations			
Accounts and other current receivables, net	\$	48	\$ 228
Inventories		_	8
Property, plant, and equipment, net		1	2
Other		1	7
Total assets of the disposal group	\$	50	\$ 245
Carrying amounts of major classes of liabilities included as part of discontinued operations			
Accounts payable and accrued liabilities	\$	3	\$ 46
Total liabilities of the disposal group	\$	3	\$ 46

As of December 31, 2016 and 2015, Baxter recorded a liability of \$46 million and \$190 million, respectively, for its obligation to transfer these net assets to Baxalta. In 2016, the company transferred \$161 million of net assets to Baxalta resulting in a gain of \$19 million, which is recorded within income from discontinued items, net of tax.

Baxter and Baxalta entered into several additional agreements in connection with the July 1, 2015 separation, including a transition services agreement (TSA), separation and distribution agreement, manufacturing and supply agreements (MSA), tax matters agreement, an employee matters agreement, a long-term services agreement, and a shareholder's and registration rights agreement.

Pursuant to the TSA, Baxter and Baxalta and their respective subsidiaries are providing to each other, on an interim, transitional basis, various services. Services being provided by Baxter include, among others, finance, information technology, human resources, quality supply chain and certain other administrative services. The services generally commenced on the Distribution date and are expected to terminate within 24 months (or 36 months in the case of certain information technology services) of the Distribution date. Billings by Baxter under the TSA are recorded as a reduction of the costs to provide the respective service in the applicable expense category, primarily in marketing and administrative expenses, in the consolidated statements of income. In 2016 and 2015, the company recognized approximately \$101 million and \$75 million, respectively, as a reduction to marketing and administrative expenses related to the TSA.

Pursuant to the MSA, Baxalta or Baxter, as the case may be, manufactures, labels, and packages products for the other party. The terms of the agreements range in initial duration from five to 10 years. In 2016 and 2015, Baxter recognized approximately \$39 million and \$37 million, respectively, in sales to Baxalta. In addition, in 2016 and 2015, Baxter recognized approximately \$189 million and \$100 million, respectively, in cost of sales related to purchases from Baxalta pursuant to the MSA. The cash flows associated with these agreements are included in cash flows from operations — continuing operations.

In December 2015, Baxter sold to Baxalta certain assets for approximately \$28 million with no resulting impact to net income.

Cash inflows of \$30 million were reported in cash flows from operations – discontinued operations in 2016. These relate to non-assignable tenders whereby Baxter remains the seller of Baxalta products, transactions related to importation services Baxter provides in certain countries, in addition to trade payables settled by Baxter on Baxalta's behalf after the local separation.

NOTE 3

Prepaid Expenses and Other

SUPPLEMENTAL FINANCIAL INFORMATION

as of December 31 (in millions)	2016	2015
Prepaid value added taxes	\$ 114	\$ 118
Prepaid income taxes	147	302
Other	341	435
Prepaid expenses and other	\$ 602	\$ 855

Other Long-Term Assets

as of December 31 (in millions)	20	016	2015
Deferred income taxes	\$	629	\$ 354
Other long-term receivables		181	176
All other		167	214
Other long-term assets	\$	977	\$ 744

Accounts Payable and Accrued Liabilities

as of December 31 (in millions)	2016	2015
Accounts payable, principally trade	\$ 791	\$ 716
Common stock dividends payable	70	137
Employee compensation and withholdings	542	481
Property, payroll and certain other taxes	143	166
Infusion pump reserves	_	52
Business optimization reserves	153	98
Accrued rebates	206	192
Separation-related reserves	46	190
All other	661	634
Accounts payable and accrued liabilities	\$ 2,612	\$ 2,666

Other Long-Term Liabilities

as of December 31 (in millions)	2016	2015
Pension and other employee benefits	\$ 1,492	\$ 2,041
Deferred tax liabilities	93	195
Litigation reserves	19	24
Business optimization reserves	11	18
Contingent payment liabilities	15	20
All other	113	127
Other long-term liabilities	\$ 1,743	\$ 2,425

Net Interest Expense

years ended December 31 (in millions)	2016	2	015	2014
Interest costs	\$ 107	\$	197 \$	237
Interest costs capitalized	(18)		(51)	(70)
Interest expense	89		146	167
Interest income	(23)		(20)	(22)
Net interest expense	\$ 66	\$	126 \$	145

Other (Income) Expense, net

years ended December 31 (in millions)	2016	2015	2014		
Foreign exchange	\$ (28)	\$ (113)	\$	(8)	
Net loss on debt extinguishment	153	130		_	
Net realized gains on Retained Shares transaction	(4,387)	_		_	
Gain on litigation settlement	_	(52)		_	
Gain on sale of investments and other assets	(3)	(38)		(20)	
All other	(31)	(32)		49	
Other (income) expense, net	\$ (4,296)	\$ (105)	\$	21	

EARNINGS PER SHARE

The numerator for both basic and diluted earnings per share (EPS) is either net income, income from continuing operations, or income from discontinued operations. The denominator for basic EPS is the weighted-average number of common shares outstanding during the period. The dilutive effect of outstanding stock options, restricted stock units (RSUs) and performance share units (PSUs) is reflected in the denominator for diluted EPS using the treasury stock method.

The following table is a reconciliation of basic shares to diluted shares.

years ended December 31 (in millions)	2016	2015	2014
Basic shares	546	545	542
Effect of dilutive securities	5	4	5
Diluted shares	551	549	547

The effect of dilutive securities included unexercised stock options, unvested RSUs and contingently issuable shares related to granted PSUs. All outstanding equity awards were dilutive as of December 31, 2016, while the computation of diluted EPS excluded 18 million and 9 million equity awards in 2015 and 2014, respectively, because their inclusion would have had an anti-dilutive effect on diluted EPS. Refer to Note 12 for additional information regarding items impacting basic shares.

NOTE 5

ACQUISITIONS AND OTHER ARRANGEMENTS

Claris Injectables Limited

In December 2016, Baxter entered into a definitive agreement to acquire Claris Injectables Limited (Claris), a wholly owned subsidiary of Claris Lifesciences Limited, for total consideration of approximately \$625 million. Upon closing, Claris will add capabilities in production of essential generic injectable medicines, such as anesthesia and analgesics, renal, anti-infectives and critical care in a variety of presentations including bags, vials and ampoules. The Boards of Directors of both companies have approved the proposed acquisition, which is expected to close in the second half of 2017. Closing of the deal is subject to satisfaction of regulatory approvals and other closing conditions, including achievement of a specified revenue target.

JW Holdings Corporation

In July 2013, Baxter entered into a collaboration agreement with JW Holdings Corporation (JW Holdings) for parenteral nutritional products containing a novel formulation of omega 3 lipids. Baxter has exclusive rights to co-develop and distribute the products globally, with the exception of Korea. In 2013, Baxter recognized an R&D charge of \$25 million related to an upfront payment. As of December 31, 2016, Baxter had the potential to make future payments of up to \$8 million relating to the achievement of regulatory milestones, in addition to future royalty payments.

Celerity Pharmaceuticals, LLC

In September 2013, Baxter entered into an agreement with Celerity Pharmaceutical, LLC (Celerity), a company of Water Street Healthcare Partners III, LLP, to develop certain acute care generic injectable premix and oncolytic molecules through regulatory approval. Baxter transferred its rights in these molecules to Celerity and Celerity assumed ownership and responsibility for development of the molecules. Baxter is obligated to purchase the individual product rights from Celerity if the products obtain regulatory approval. In 2015, Baxter paid \$14 million to acquire the rights to cefazolin injection in GALAXY Container (2 g/100 mL). In 2016, Baxter paid approximately \$23 million to acquire the rights to vancomycin injection 0.9% Sodium Chloride (Normal Saline) in 500 mg, 750 mg and 1 gram presentations. For both intangible assets, Baxter capitalized the purchase price as an intangible asset and is amortizing the asset over the estimated economic life of 12 years. As of December 31, 2016, Baxter's estimated future payments total up to \$273 million upon Celerity's achievement of specified regulatory approvals.

GOODWILL AND OTHER INTANGIBLE ASSETS, NET

Goodwill

The following table is a summary of the activity in goodwill by segment.

	Hospital						
(in millions)		Renal		Products		Total	
December 31, 2014	\$	445	\$	2,482	\$	2,927	
Additions		_		_		_	
Currency translation and other adjustments		(37)		(203)		(240)	
December 31, 2015	\$	408	\$	2,279	\$	2,687	
Additions		3		_		3	
Currency translation and other adjustments		(14)		(81)		(95)	
December 31, 2016	\$	397	\$	2,198	\$	2,595	

As a result of the separation of Baxalta in July 2015, the goodwill associated with Baxter's former BioScience segment has been eliminated. The remaining goodwill was allocated from the former Medical Products segment to the Renal and Hospital Products segments using the relative fair value approach.

As of December 31, 2016, there were no reductions in goodwill relating to impairment losses.

Other Intangible Assets, Net

The following table is a summary of the company's other intangible assets.

	Devel techno		Otl	ner amortized	Ind	lefinite-lived	
(in millions)	including			angible assets		ingible assets	Total
December 31, 2016							
Gross other intangible assets	\$	1,690	\$	384	\$	57	\$ 2,131
Accumulated amortization		(855)		(165)		_	(1,020)
Other intangible assets, net	\$	835	\$	219	\$	57	\$ 1,111
December 31, 2015							
Gross other intangible assets	\$	1,742	\$	393	\$	86	\$ 2,221
Accumulated amortization		(729)		(143)		_	(872)
Other intangible assets, net	\$	1,013	\$	250	\$	86	\$ 1,349

Intangible asset amortization expense was \$163 million in 2016, \$158 million in 2015 and \$169 million in 2014. The anticipated annual amortization expense for definite-lived intangible assets recorded as of December 31, 2016 is \$138 million in 2017, \$125 million in 2018, \$121 million in 2019, \$118 million in 2020 and \$114 million in 2021.

In 2016, the company recorded an impairment charge of \$27 million related to an indefinite-lived intangible asset (acquired IPR&D) in the company's Renal segment relating to its in-center hemodialysis program. The asset was written down to estimated fair value and recorded in research and development expenses. Additionally, the company recorded an impairment charge of \$51 million, of which \$41 million related to a developed technology asset, relating to the company's Hospital Products segment synthetic bone repair products business which was acquired from ApaTech Limited in 2010. The assets of the business were written down to estimated fair value and recorded in cost of sales.

In 2015, the company recorded impairments of approximately \$10 million related to acquired IPR&D and \$13 million related to developed technology.

INFUSION PUMP AND BUSINESS OPTIMIZATION CHARGES

Infusion Pump Charges

The company had undertaken a field corrective action with respect to the SIGMA Spectrum Infusion Pump, which is predominantly sold in the United States. The United States Food and Drug Administration (FDA) categorized the action as a Class 1 recall during the second quarter of 2014. Remediation primarily included software-related corrections and a replacement pump in a limited number of cases. In 2014, the company recorded a charge of \$93 million related primarily to cash costs associated with remediation efforts and utilized \$4 million in 2014. During 2015, the company refined its expectations relating to the costs associated with the remediation effort and recorded partial reversals of the cash and non-cash reserves totaling \$26 million and \$10 million, respectively. Additionally the company utilized \$13 million of the cash reserves during 2015. In 2016, the company recorded utilization of cash and non-cash reserves of \$22 million and \$3 million, respectively, as well as partial reversals of cash and non-cash reserves of \$11 million and \$1 million, respectively. As of December 31, 2016, the remediation efforts are substantially complete and the remaining costs and reserves are considered immaterial to the company.

From 2005 through 2014, the company recorded total charges and adjustments of \$863 million related to the COLLEAGUE and SYNDEO infusion pumps, including \$700 million of cash costs and \$163 million principally related to asset impairments.

The following table summarizes cash activity and reserve adjustments related to the company's COLLEAGUE and SYNDEO infusion pump reserves through December 31, 2016.

(in millions)	
Charges and adjustments in 2005 through 2014	\$ 700
Utilization in 2005 through 2014	(678)
Reserves at December 31, 2014	22
Reserve adjustments	(7)
Utilization	(8)
Reserves at December 31, 2015	7
Reserve adjustments	(4)
Utilization	(3)
Reserves at December 31, 2016	\$

As of December 31, 2016, the company has completed its field corrective actions related to the COLLEAGUE and SYNDEO infusion pumps.

Business Optimization Charges

Beginning in the second half of 2015, the company has initiated actions to transform the company's cost structure and enhance operational efficiency. These efforts include restructuring the organization, optimizing the manufacturing footprint, R&D operations and supply chain network, employing disciplined cost management, and centralizing and streamlining certain support functions. Through December 31, 2016 the company incurred cumulative pretax costs of \$407 million related to these actions. The costs consisted primarily of employee termination, implementation costs, and accelerated depreciation. The company expects to incur additional pretax costs of approximately \$390 million and capital expenditures of \$90 million through the completion of these initiatives. These costs will primarily include employee termination costs, implementation costs, and accelerated depreciation. Of this amount, the company expects that approximately 10 percent of the charges will be non-cash.

In addition to the programs above, the company recorded additional net business optimization charges of \$125 million in 2016. These charges primarily include employee termination costs, contract termination costs, asset impairments, and Gambro integration costs. Approximately 40% of these costs were non-cash. The company does not anticipate incurring any additional costs related to these programs in the future and expects them to be substantially completed by the end of 2017.

The company recorded the following charges related to business optimization programs in 2016, 2015, and 2014:

years ended December 31 (in millions)	 2016		2015	2014	
Restructuring charges, net	\$ 285	\$	130	\$	(6)
Costs to implement business optimization programs	65		_		_
Gambro integration costs	26		73		144
Accelerated depreciation	33		_		_
Total business optimization charges	\$ 409	\$	203	\$	138

Included in the restructuring charges for 2016 were net employee termination costs of \$180 million which primarily consisted of a global workforce reduction program and \$27 million related to the impairment of acquired IPR&D as described in Note 6. Restructuring charges for 2016 also included \$54 million for costs associated with the discontinuation of the VIVIA home hemodialysis development program. These costs consist of contract termination costs of \$21 million, asset impairments of \$31 million and other exit costs of \$2 million.

Included in the restructuring charges for 2015 were net employee termination costs of \$83 million which primarily related to the global workforce reduction program mentioned above. Additionally, asset impairments of \$13 million and \$29 million were recorded related to a developed technology intangible assets and a manufacturing facility rationalization program, respectively.

Restructuring charges recorded in 2014 primarily related to employee and contract termination costs associated with legacy, non-transformational restructuring initiatives.

The company recorded the following components of restructuring costs in 2016, 2015 and 2014:

	2016							
(in millions)	C	OGS	,	SGA	R	&D	Т	Γotal
Employee termination costs	\$	72	\$	109	\$	13	\$	194
Contract termination costs		9		5		13		27
Asset impairments		38		_		40		78
Reserve adjustments								
Employee termination costs		(1)		(11)		(2)		(14)
Total restructuring charges	\$	118	\$	103	\$	64	\$	285

	2015							
(in millions)	CO	OGS	S	GA	R	&D	7	Total
Employee termination costs	\$	14	\$	86	\$	15	\$	115
Contract termination costs	\$	3	\$	2	\$	_	\$	5
Asset impairment		40		_		2		42
Reserve adjustments								
Employee termination costs		(19)		(10)		(3)		(32)
Total restructuring charges	\$	38	\$	78	\$	14	\$	130

	2014							
(in millions)	COC	iS	S	GA	R	&D	T	otal
Employee termination costs	\$	10	\$	24	\$	_	\$	34
Contract termination costs		2		6		2		10
Asset impairment		2		2		_		4
Reserve adjustments								
Employee termination costs		(19)		(31)		1		(49)
Contract termination costs		(3)		_		_		(3)
Asset impairment		(2)		_		_		(2)
Total restructuring charges	\$	(10)	\$	1	\$	3	\$	(6)

Costs to implement business optimization programs in 2016 were \$65 million. These costs consisted primarily of external consulting and employee salary and related costs. The costs were included within marketing and administrative and R&D expense.

Costs related to the integration of Gambro were included within marketing and administrative expense for all referenced periods.

In 2016, the company recognized accelerated depreciation, primarily associated with facilities to be closed of \$33 million. The costs were recorded in cost of sales.

The following table summarizes cash activity in the reserves related to the company's business optimization initiatives.

(in millions)	
Reserve at December 31, 2013	\$ 244
2014 charges	44
Reserve adjustments	(54)
Utilization in 2014	(88)
CTA	(19)
Reserve at December 31, 2014	127
2015 charges	120
Reserve adjustments	(32)
Utilization in 2015	(89)
CTA	(10)
Reserve at December 31, 2015	116
2016 charges	221
Reserve adjustments	(14)
Utilization in 2016	(164)
CTA	5
Reserve at December 31, 2016	\$ 164

Reserve adjustments primarily relate to employee termination cost reserves established in prior periods.

The company's restructuring reserves of \$164 million as of December 31, 2016 consisted of \$146 million of employee termination costs and the remaining reserves related to contract termination costs. The reserves are expected to be substantially utilized by the end of 2018.

DEBT, CREDIT FACILITIES AND LEASE COMMITMENTS

Debt Outstanding

At December 31, 2016 and 2015, the company had the following debt outstanding:

as of December 31 (in millions)	2016		2015
Line of credit	\$	— \$	1,450
Commercial paper		_	300
Other short-term debt		_	25
Short-term debt	\$	— \$	1,775

	Effective interest		
as of December 31 (in millions)	rate in 20161	20162	20152
5.9% notes due 2016	5.8%	_	301
0.95% notes due 2016	1.2%	_	500
1.85% notes due 2017	2.1%	_	499
5.375% notes due 2018	5.7%	_	501
1.85% notes due 2018	2.1 %	_	747
4.5% notes due 2019	4.5%	_	530
4.25% notes due 2020	4.2%	_	299
Variable-rate loan due 2020	1.0%	294	281
1.7% notes due 2021	1.6%	397	_
2.40% notes due 2022	2.5%	208	211
3.2% notes due 2023	3.1%	_	146
2.6% notes due 2026	2.2%	744	
7.65% debentures due 2027	7.7%	5	5
6.625% debentures due 2028	6.7%	99	100
6.25% notes due 2037	5.8%	265	265
3.65% notes due 2042	3.7%	6	6
4.5% notes due 2043	4.5%	255	255
3.5% notes due 2046	3.0%	439	
Other	_	70	86
Total debt and capital lease obligations		2,782	4,732
Current portion		(3)	(810)
Long-term portion	\$	2,779 \$	3,922

Excludes the effect of any related interest rate swaps.

Significant Debt Issuances

In August 2016, Baxter issued senior notes with a total aggregate principal amount of \$1.6 billion, comprised of \$400 million at a fixed coupon rate of 1.70% due in August 2021, \$750 million at a fixed coupon rate of 2.60% due in August 2026 and \$450 million at a fixed coupon rate of 3.50% due in August 2046.

In June 2015, the company's then wholly-owned subsidiary Baxalta issued senior notes with a total aggregate principal amount of \$5.0 billion. Approximately \$4.0 billion of the related net proceeds were distributed to Baxter in connection with the separation. After the separation, Baxter has no obligations as it relates to the Baxalta senior notes or any other Baxalta indebtedness. Refer to the debt tender offer section below in connection with this debt issuance.

Book values include any discounts, premiums and adjustments related to hedging instruments.

Debt Redemption

In September 2016, Baxter redeemed an aggregate of approximately \$1 billion in principal amount of its 1.850% Senior Notes due 2017, 1.850% Senior Notes due 2018, 5.375% Senior Notes due 2018, 4.500% Senior Notes due 2019, 4.250% Senior Notes due 2020 and 3.200% Senior Notes due 2023. Baxter paid approximately \$1 billion, including accrued and unpaid interest and tender premium, to redeem such notes. As a result of the debt redemptions, the company recognized a loss on extinguishment of debt in the third quarter of 2016 of approximately \$52 million, which is included in other (income) expense, net.

Debt-for-Equity Exchanges

As of December 31, 2015, the company had drawn \$1.45 billion under its \$1.8 billion U.S. dollar-denominated revolving credit facility at a weighted average interest rate of 1.41%. On January 27, 2016, Baxter exchanged Retained Shares for the extinguishment of \$1.45 billion aggregate principal amount outstanding under its \$1.8 billion U.S. dollar-denominated revolving credit facility. This exchange extinguished the indebtedness under the facility, which was terminated in connection with such debt-for-equity exchange. There were no material prepayment penalties or breakage costs associated with the termination of the facility. Baxter recognized a net realized gain of \$1.25 billion related to the Retained Shares exchanged, which is included in other (income) expense, net in 2016.

On March 16, 2016, the company exchanged Retained Shares for the extinguishment of approximately \$2.2 billion in principal amount of its 0.950% Notes due May 2016, 5.900% Notes due August 2016, 1.850% Notes due January 2017, 5.375% Notes due May 2018, 1.850% Notes due June 2018, 4.500% Notes due August 2019 and 4.250% Notes due February 2020 purchased by certain third party purchasers in the previously announced debt tender offers. As a result, the company recognized a net loss on extinguishment of debt totaling \$101 million and a net realized gain of \$2.0 billion on the Retained Shares exchanged, which are included in other (income) expense, net in 2016.

Debt Maturities

In the second quarter of 2016, the company repaid the \$190 million outstanding balance of its 0.95% senior unsecured notes that matured in June 2016. In the third quarter of 2016, the company repaid the \$130 million outstanding balance of its 5.9% senior unsecured notes that matured in September 2016.

Debt Tender Offer

On July 6, 2015 and July 21, 2015 the company purchased an aggregate of approximately \$2.7 billion in principal amount of its 5.900% Notes due September 2016, 6.625% Debentures due February 2028, 6.250% Notes due December 2037, 3.650% Notes due August 2042, 4.500% Notes due June 2043, 3.200% Notes due June 2023, and 2.400% Notes due August 2022 pursuant to a debt tender offer. Baxter paid approximately \$2.9 billion, including accrued and unpaid interest and tender premium, to purchase such notes. As a result of the debt tender offers the company recognized a loss on extinguishment of debt in the third quarter of 2015 of \$130 million, which is included in other (income) expense, net within the Consolidated Statements of Income.

Credit Facilities

Effective July 1, 2015, the company terminated its \$1.5 billion U.S. dollar-denominated revolving credit facility and €300 million Euro denominated revolving credit facility and entered into credit agreements providing for a senior U.S. dollar-denominated revolving credit facility in an aggregate principal amount of up to \$1.5 billion maturing in 2020, as well as a Euro-denominated senior revolving credit facility in an aggregate principal amount of up to €200 million maturing in 2020. As of December 31, 2016 there were zero borrowings outstanding under the company's revolving credit facilities. The facilities enable the company to borrow funds on an unsecured basis at variable interest rates, and contain various covenants, including a maximum net leverage ratio and maximum interest coverage ratio.

The company also maintains other credit arrangements, which totaled \$271 million at December 31, 2016 and \$307 million at December 31, 2015. There were no borrowings outstanding under these arrangements at December 31, 2016 and \$25 million of borrowings outstanding at December 31, 2015.

At December 31, 2016, the company was in compliance with the financial covenants in these agreements. The non-performance of any financial institution supporting any of the credit facilities would reduce the maximum capacity of these facilities by each institution's respective commitment.

Commercial Paper

During 2016, the company issued and redeemed commercial paper, and no commercial paper was outstanding at December 31, 2016. There was a balance of \$300 million outstanding at December 31, 2015 with a weighted-average interest rate of 0.6%.

Leases

The company leases certain facilities and equipment under capital and operating leases expiring at various dates. The leases generally provide for the company to pay taxes, maintenance, insurance and certain other operating costs of the leased property. Most of the operating leases contain renewal options. For the years ending December 31, 2016, 2015, and 2014 operating lease rent expense was \$174 million, \$184 million and \$203 million, respectively.

Future Minimum Lease Payments and Debt Maturities

			Debt maturities
	(Operating	and capital
as of and for the years ended December 31 (in millions)		leases	leases
2017	\$	129	\$ 3
2018		108	3
2019		85	3
2020		67	297
2021		58	403
Thereafter		216	2,085
Total obligations and commitments		663	2,794
Discounts, premiums, and adjustments relating to hedging instruments		_	(12)
Total debt and lease obligations	\$	663	\$ 2,782

NOTE 9

DERIVATIVE INSTRUMENTS AND HEDGING ACTIVITY

Foreign Currency and Interest Rate Risk Management

The company operates on a global basis and is exposed to the risk that its earnings, cash flows and equity could be adversely impacted by fluctuations in foreign exchange and interest rates. The company's hedging policy attempts to manage these risks to an acceptable level based on the company's judgment of the appropriate trade-off between risk, opportunity and costs.

The company is primarily exposed to foreign exchange risk with respect to recognized assets and liabilities, forecasted transactions and net assets denominated in the Euro, British Pound, Chinese Yuan, Korean Won, Australian Dollar, Canadian Dollar, Japanese Yen, Colombian Peso, Brazilian Real, Swedish Krona, Mexican Peso and New Zealand Dollar. The company manages its foreign currency exposures on a consolidated basis, which allows the company to net exposures and take advantage of any natural offsets. In addition, the company uses derivative and nonderivative instruments to further reduce the net exposure to foreign exchange. Gains and losses on the hedging instruments offset losses and gains on the hedged transactions and reduce the earnings and equity volatility resulting from foreign exchange. Financial market and currency volatility may limit the company's ability to cost-effectively hedge these exposures.

The company is also exposed to the risk that its earnings and cash flows could be adversely impacted by fluctuations in interest rates. The company's policy is to manage interest costs using a mix of fixed- and floating-rate debt that the company believes is appropriate. To manage this mix in a cost-efficient manner, the company periodically enters into interest rate swaps in which the company agrees to exchange, at specified intervals, the difference between fixed and floating interest amounts calculated by reference to an agreed-upon notional amount.

The company does not hold any instruments for trading purposes and none of the company's outstanding derivative instruments contain credit-risk-related contingent features.

Cash Flow Hedges

The company may use options, including collars and purchased options, forwards and cross-currency swaps to hedge the foreign exchange risk to earnings relating to forecasted transactions and recognized assets and liabilities. The company periodically uses

forward-starting interest rate swaps and treasury rate locks to hedge the risk to earnings associated with movements in interest rates relating to anticipated issuances of debt. Certain other firm commitments and forecasted transactions are also periodically hedged. Cash flow hedges primarily related to forecasted intercompany sales denominated in foreign currencies, and anticipated issuances of debt.

The notional amounts of foreign exchange contracts were \$561 million and \$378 million as of December 31, 2016 and 2015, respectively. The company did not have any interest rate contracts designated as cash flow hedges outstanding at December 31, 2016 and 2015. The maximum term over which the company has cash flow hedge contracts in place related to forecasted transactions at December 31, 2016 is 12 months.

Fair Value Hedges

The company uses interest rate swaps to convert a portion of its fixed-rate debt into variable-rate debt. These instruments hedge the company's earnings from changes in the fair value of debt due to fluctuations in the designated benchmark interest rate.

The total notional amount of interest rate contracts designated as fair value hedges was \$200 million and \$1.3 billion as of December 31, 2016 and 2015, respectively.

Dedesignations

If it is determined that a derivative or nonderivative hedging instrument is no longer highly effective as a hedge, the company discontinues hedge accounting prospectively. If the company removes the cash flow hedge designation because the hedged forecasted transactions are no longer probable of occurring, any gains or losses are immediately reclassified from AOCI to earnings. Gains or losses relating to terminations of effective cash flow hedges in which the forecasted transactions are still probable of occurring are deferred and recognized consistent with the loss or income recognition of the underlying hedged items.

There were no hedge dedesignations in 2016, 2015 or 2014 resulting from changes in the company's assessment of the probability that the hedged forecasted transactions would occur.

If the company terminates a fair value hedge, an amount equal to the cumulative fair value adjustment to the hedged items at the date of termination is amortized to earnings over the remaining term of the hedged item. In 2016, the company terminated a total notional value of \$765 million of interest rate contracts in connection with the March 2016 debt tender offers, resulting in a \$34 million reduction to the debt extinguishment loss. In 2015, the company terminated \$1.65 billion of interest rate contracts in connection with the July debt tender offers, which resulted in a \$33 million reduction to the debt extinguishment loss. There were no fair value hedges terminated during 2014.

<u>Undesignated Derivative Instruments</u>

The company uses forward contracts to hedge earnings from the effects of foreign exchange relating to certain of the company's intercompany and third-party receivables and payables denominated in a foreign currency. These derivative instruments are generally not formally designated as hedges and the terms of these instruments generally do not exceed one month.

The total notional amount of undesignated derivative instruments was \$822 million as of December 31, 2016 and \$580 million as of December 31, 2015.

Gains and Losses on Derivative Instruments

The following table summarizes the gains and losses on the company's derivative instruments for the years ended December 31, 2016, 2015, and 2014.

	Gain (loss) recognized in OCI		Location of gain (loss) in		ed from	1							
(in millions)	2	016	2	015	2014 income		income statement	2016		2015		20	014
Cash flow hedges													
Interest rate contracts	\$	_	\$		\$		Other (income) expense, net	\$	9	\$	_	\$	(1)
Foreign exchange contracts		_		(1)		1	Net sales		_		_		1
Foreign exchange contracts		1		4		51	Cost of sales		(3)		47		13
Total	\$	1	\$	3	\$	52		\$	6	\$	47	\$	13

	Gain (loss) recognized in income								
(in millions)		2016			2015		2014		
Fair value hedges									
Interest rate contracts	Net interest expense \$		9	\$	(43)	\$		68	
Undesignated derivative instruments									
Foreign exchange contracts	Other (income) expense, net \$		4	\$	(13)	\$		49	

For the company's fair value hedges, equal and offsetting losses of \$9 million, gains of \$43 million and losses of \$68 million were recognized in net interest expense in 2016, 2015 and 2014, respectively, as adjustments to the underlying hedged items, fixed-rate debt. Ineffectiveness related to the company's cash flow and fair value hedges for the year ended December 31, 2016 was not material.

The following table summarizes net-of-tax activity in AOCI, a component of shareholders' equity, related to the company's cash flow hedges.

as of and for the years ended December 31 (in millions)	20	16	20	015	20	014
Accumulated other comprehensive income (loss) balance at beginning of year	\$	7	\$	34	\$	10
Gain in fair value of derivatives during the year		1		4		32
Amount reclassified to earnings during the year		(5)		(31)		(8)
Accumulated other comprehensive income balance at end of year	\$	3	\$	7	\$	34

As of December 31, 2016, \$3 million of deferred, net after-tax gains on derivative instruments included in AOCI are expected to be recognized in earnings during the next 12 months, coinciding with when the hedged items are expected to impact earnings.

Fair Values of Derivative Instruments

The following table summarizes the classification and fair value amounts of derivative instruments reported in the consolidated balance sheet as of December 31, 2016.

	Derivatives in asset posit	Derivatives in liability po	atives in liability positions			
(in millions)	Balance sheet location	Fair value		Balance sheet location	Fair	value
Derivative instruments designated as hedges						
Interest rate contracts	Other long-term assets	\$	7	Other long-term liabilities	\$	_
Foreign exchange contracts	Prepaid expenses and other		22	Accounts payable and accrued liabilities		1
Total derivative instruments designated as hedges		\$	29		\$	1
Undesignated derivative instruments						
	Prepaid expenses			Accounts payable		
Foreign exchange contracts	and other	\$	1	and accrued liabilities	\$	2
Total derivative instruments		\$	30		\$	3

The following table summarizes the classification and fair value amounts of derivative instruments reported in the consolidated balance sheet as of December 31, 2015.

	Derivatives in asset posit	ions		Derivatives in liability po	sitions	
(in millions)	Balance sheet location	Fair	r value	Balance sheet location	Fair	value
Derivative instruments designated as hedges						
Interest rate contracts	Other long-term assets	\$	46	Other long-term liabilities	\$	_
Foreign exchange contracts	Prepaid expenses and other		9	Accounts payable and accrued liabilities		1
Total derivative instruments designated as hedges		\$	55		\$	1
Undesignated derivative instruments						
	Prepaid expenses			Accounts payable		
Foreign exchange contracts	and other	\$	1	and accrued liabilities	\$	1
Total derivative instruments		\$	56		\$	2

While the company's derivatives are all subject to master netting arrangements, the company presents its assets and liabilities related to derivative instruments on a gross basis within the consolidated balance sheets. Additionally, the company is not required to post collateral for any of its outstanding derivatives. The following table provides information on the company's derivative positions as if they were presented on a net basis, allowing for the right of offset by counterparty.

	 December	r 31,	2016	December 3	31, 2015		
(in millions)	Asset		Liability	Asset	I	Liability	
Gross amounts recognized in the consolidated balance sheet	\$ 30	\$	3	\$ 56	\$		2
Gross amount subject to offset in master netting arrangements not offset in the							
consolidated balance sheet	(3)		(3)	(2)		((2)
Total	\$ 27	\$	_	\$ 54	\$	_	_

NOTE 10

FINANCIAL INSTRUMENTS AND RELATED FAIR VALUE MEASUREMENTS

Receivable Securitizations

For trade receivables originated in Japan, the company has entered into agreements with financial institutions in which the entire interest in and ownership of the receivable is sold. The company continues to service the receivables in its Japanese securitization arrangement. Servicing assets or liabilities are not recognized because the company receives adequate compensation to service the sold receivables. The Japanese securitization arrangement includes limited recourse provisions, which are not material.

The following is a summary of the activity relating to the securitization arrangement.

as of and for the years ended December 31 (in millions)	2016	2015	2014
Sold receivables at beginning of year	\$ 81 \$	104	\$ 114
Proceeds from sales of receivables	348	361	464
Cash collections (remitted to the owners of the receivables)	(367)	(384)	(459)
Effect of currency exchange rate changes	6	_	(15)
Sold receivables at end of year	\$ 68 \$	81	\$ 104

The net losses relating to the sales of receivables were immaterial for each year.

Concentrations of Credit Risk

The company invests excess cash in certificates of deposit or money market funds and diversifies the concentration of cash among different financial institutions. With respect to financial instruments, where appropriate, the company has diversified its selection of counterparties, and has arranged collateralization and master-netting agreements to minimize the risk of loss.

The company continues to do business with foreign governments in certain countries, including Greece, Spain, Portugal and Italy, which have experienced deterioration in credit and economic conditions. As of December 31, 2016 and 2015, the company's net accounts receivable from the public sector in Greece, Spain, Portugal and Italy totaled \$137 million and \$211 million, respectively.

Global economic conditions and liquidity issues in certain countries have resulted, and may continue to result, in delays in the collection of receivables and credit losses. Global economic conditions, governmental actions and customer-specific factors may require the company to re-evaluate the collectability of its receivables and the company could potentially incur additional credit losses. These conditions may also impact the stability of the Euro.

Fair Value Measurements

The fair value hierarchy under the accounting standard for fair value measurements consists of the following three levels:

- Level 1 Quoted prices in active markets that the company has the ability to access for identical assets or liabilities;
- Level 2 Quoted prices for similar instruments in active markets, quoted prices for identical or similar instruments in markets that are not active, and model-based valuations in which all significant inputs are observable in the market; and
- Level 3 Valuations using significant inputs that are unobservable in the market and include the use of judgment by the company's
 management about the assumptions market participants would use in pricing the asset or liability.

The following table summarizes the bases used to measure financial assets and liabilities that are carried at fair value on a recurring basis in the consolidated balance sheets.

			Basis of fair value measurement							
(in millions)	Decen	ce as of other 31,		Quoted prices in active markets for dentical assets (Level 1)		Significant other observable inputs (Level 2)		Significant unobservable inputs (Level 3)		
Assets										
Foreign currency hedges	\$	23	\$	_	\$	23	\$	_		
Interest rate hedges		7		_		7		_		
Available-for-sale securities		9		9		_		_		
Total assets	\$	39	\$	9	\$	30	\$	_		
Liabilities										
Foreign currency hedges	\$	3	\$	_	\$	3	\$	_		
Contingent payments related to acquisitions		19		_		_		19		
Total liabilities	\$	22	\$	_	\$	3	\$	19		

			Basis of fair value measurement							
(in millions)	Dec	ance as of ember 31, 2015	Quoted prices in active markets for identical assets (Level 1)	o	ignificant other bservable inputs (Level 2)	1	Significant unobservable inputs (Level 3)			
Assets										
Foreign currency hedges	\$	10	\$ _	\$	10	\$				
Interest rate hedges		46	_		46		_			
Available-for-sale securities		5,162	14		5,148					
Total assets	\$	5,218	\$ 14	\$	5,204	\$	_			
Liabilities										
Foreign currency hedges	\$	2	\$ _	\$	2	\$	_			
Contingent payments related to acquisitions		20	_		_		20			
Total liabilities	\$	22	\$	\$	2	\$	20			

As of December 31, 2016, cash and equivalents of \$2.8 billion included money market funds of approximately \$1.4 billion, which would be considered Level 2 in the fair value hierarchy.

For assets that are measured using quoted prices in active markets, the fair value is the published market price per unit multiplied by the number of units held, without consideration of transaction costs. The investment in the Retained Shares of \$5.1 billion as of December 31, 2015 was categorized as a Level 2 security as these securities were not registered as of that date. The value of this investment is based on Baxalta's common stock price as of December 31, 2015, which represents an identical equity instrument registered under the Securities Act of 1933, as amended. The company disposed of the remainder of its Retained Shares in Baxalta in the first half of 2016, as described below. The majority of the derivatives entered into by the company are valued using internal valuation techniques as no quoted market prices exist for such instruments. The principal techniques used to value these instruments are discounted cash flow and Black-Scholes models. The key inputs are considered observable and vary depending on the type of derivative, and include contractual terms, interest rate yield curves, foreign exchange rates and volatility.

Contingent payments related to acquisitions consist of commercial milestone payments and sales-based payments, and are valued using discounted cash flow techniques. The fair value of commercial milestone payments reflects management's expectations of probability of payment, and increases as the probability of payment increases or expectation of timing of payments is accelerated. The fair value of sales-based payments is based upon probability-weighted future revenue estimates, and increases as revenue estimates increase, probability weighting of higher revenue scenarios increase or expectation of timing of payment is accelerated.

The following table is a reconciliation of the fair value measurements that use significant unobservable inputs (Level 3), which consist of contingent payments related to acquisitions.

(in millions)	Contingent payments
Fair value as of December 31, 2014 \$	45
Additions	_
Payments	(3)
Net gains recognized in earnings	(22)
Fair value as of December 31, 2015	20
Additions	_
Payments	(1)
Net gains recognized in earnings	<u> </u>
Fair value as of December 31, 2016	19

The following table provides information relating to the company's investments in available-for-sale equity securities.

(in millions)	Amoi	tized cost	Unr	realized gains	Un	realized losses	Fair va	alue
December 31, 2016	\$	13	\$	_	\$	4	\$	9
December 31, 2015	\$	732	\$	4,430	\$	_	\$	5,162

Book Values and Fair Values of Financial Instruments

In addition to the financial instruments that the company is required to recognize at fair value in the consolidated balance sheets, the company has certain financial instruments that are recognized at historical cost or some basis other than fair value. For these financial instruments, the following table provides the values recognized in the consolidated balance sheets and the approximate fair values.

	Book values				Approximate fair values			
as of December 31 (in millions)		2016		2015		2016		2015
Assets								
Investments	\$	31	\$	21	\$	31	\$	21
Liabilities								
Short-term debt		_		1,775		_		1,775
Current maturities of long-term debt and lease obligations		3		810		3		818
Long-term debt and lease obligations		2,779		3,922		2,756		4,077

The following table summarizes the bases used to measure the approximate fair value of the financial instruments as of December 31, 2016 and 2015.

			Basis of fair value measurement					
(in millions)	Dece	nce as of mber 31, 2016		Quoted prices in active markets for dentical assets (Level 1)		Significant other observable inputs (Level 2)		Significant unobservable inputs (Level 3)
Assets								
Investments	\$	31	\$	_	\$	_	\$	31
Total assets	\$	31	\$	_	\$	_	\$	31
Liabilities								
Short-term debt	\$	_	\$	_	\$	_	\$	_
Current maturities of long-term debt and lease obligations		3		_		3		_
Long-term debt and lease obligations		2,756		_		2,756		_
Total liabilities	\$	2,759	\$	_	\$	2,759	\$	

			Basis of fair value measurement					
(in millions)	Dece	nce as of ember 31, 2015		Quoted prices in active markets for identical assets (Level 1)		Significant other observable inputs (Level 2)		Significant unobservable inputs (Level 3)
Assets								
Investments	\$	21	\$	_	\$	2	\$	19
Total assets	\$	21	\$		\$	2	\$	19
Liabilities								
Short-term debt	\$	1,775	\$	_	\$	1,775	\$	_
Current maturities of long-term debt and lease obligations		818		_		818		_
Long-term debt and lease obligations		4,077		_		4,077		_
Total liabilities	\$	6,670	\$	_	\$	6,670	\$	

Investments in 2016 and 2015 include certain cost method investments and held-to-maturity debt securities.

The fair value of held-to-maturity debt securities is calculated using a discounted cash flow model that incorporates observable inputs, including interest rate yields, which represents a Level 2 basis of fair value measurement. In determining the fair value of cost method investments, the company takes into consideration recent transactions, as well as the financial information of the investee, which represents a Level 3 basis of fair value measurement.

The estimated fair values of current and long-term debt were computed by multiplying price by the notional amount of the respective debt instrument. Price is calculated using the stated terms of the respective debt instrument and yield curves commensurate with the company's credit risk. The carrying values of the other financial instruments approximate their fair values due to the short-term maturities of most of these assets and liabilities.

In 2016, the company recorded net \$4.4 billion of realized gains within other expense (income), net related to exchanges of available-for-sale equity securities, which represented gains from the Retained Shares transactions. On May 6, 2016, Baxter made a voluntary non-cash contribution of 17,145,570 Retained Shares to the company's U.S. pension fund. The company recorded \$611 million of realized gains within other (income) expense, net related to the contribution of Retained Shares. On May 26, 2016, Baxter completed an exchange of 13,360,527 Retained Shares for 11,526,638 outstanding shares of Baxter common stock. The company recorded \$537 million of realized gains within other (income) expense, net related to the exchange of the Retained Shares. The company held no shares of Baxalta as of December 31, 2016. Refer to the debt-for-equity exchange section in Note 8 for discussion related to the first quarter 2016 Retained Shares transactions. In 2015, the company recorded income of \$38 million, in other expense (income), net related to equity method investments, which primarily represented gains from the sale of certain investments as well as distributions from funds that sold portfolio companies.

COMMITMENTS AND CONTINGENCIES

Collaborative and Other Arrangements

Refer to Note 5 for information regarding the company's unfunded contingent payments associated with collaborative and other arrangements.

Indemnifications

During the normal course of business, Baxter makes indemnities, commitments and guarantees pursuant to which the company may be required to make payments related to specific transactions. Indemnifications include: (i) intellectual property indemnities to customers in connection with the use, sales or license of products and services; (ii) indemnities to customers in connection with losses incurred while performing services on their premises; (iii) indemnities to vendors and service providers pertaining to claims based on negligence or willful misconduct; (iv) indemnities involving the representations and warranties in certain contracts; (v) contractual indemnities related to the separation and distribution as set forth in certain of the agreements entered into in connection with such transactions (including the separation and distribution agreement and the tax matters agreement); and (vi) contractual indemnities for its directors and certain of its executive officers for services provided to or at the request of Baxter. In addition, under Baxter's Amended and Restated Certificate of Incorporation, and consistent with Delaware General Corporation Law, the company has agreed to indemnify its directors and officers for certain losses and expenses upon the occurrence of certain prescribed events. The majority of these indemnities, commitments and guarantees do not provide for any limitation on the maximum potential for future payments that the company could be obligated to make. To help address some of these risks, the company maintains various insurance coverages. Based on historical experience and evaluation of the agreements, the company does not believe that any significant payments related to its indemnities will occur, and therefore the company has not recorded any associated liabilities.

Legal Contingencies

Refer to Note 16 for a discussion of the company's legal contingencies.

NOTE 12

SHAREHOLDERS' EQUITY

Stock-Based Compensation

The company's stock-based compensation generally includes stock options, restricted stock units (RSUs), performance share units (PSUs) and purchases under the company's employee stock purchase plan. Shares issued relating to the company's stock-based plans are generally issued out of treasury stock.

Approved in 2015, the Baxter International Inc. 2015 Incentive Plan provided for 35 million additional shares of common stock available for issuance with respect to awards for participants. As of December 31, 2016, approximately 46 million authorized shares are available for future awards under the company's stock-based compensation plans.

Stock Compensation Expense

Stock compensation expense was \$115 million, \$126 million and \$126 million in 2016, 2015 and 2014, respectively. The related tax benefit recognized was \$34 million in 2016, \$38 million in 2015 and \$41 million in 2014.

Stock compensation expense is recorded at the corporate level and is not allocated to the segments. Approximately 70% of stock compensation expense is classified in marketing and administrative expenses, with the remainder classified in cost of sales and R&D expenses. Costs capitalized in the consolidated balance sheets at December 31, 2016 and 2015 were not material.

Stock compensation expense is based on awards expected to vest, and therefore has been reduced by estimated forfeitures.

Stock Options

Stock options are granted to employees and non-employee directors with exercise prices equal to 100% of the market value on the date of grant. Stock options granted to employees generally vest in one-third increments over a three-year period. Stock options granted to non-employee directors generally cliff-vest one year from the grant date. Stock options typically have a contractual term of

10 years. The grant-date fair value, adjusted for estimated forfeitures, is recognized as expense on a straight-line basis over the substantive vesting period.

The fair value of stock options is determined using the Black-Scholes model. The weighted-average assumptions used in estimating the fair value of stock options granted during each year, along with the weighted-average grant-date fair values, were as follows:

years ended December 31	2016		2015	2014
Expected volatility		20%	20%	24%
Expected life (in years)		5.5	5.5	5.5
Risk-free interest rate		1.4%	1.7%	1.7%
Dividend yield		1.2%	2.9%	2.8%
Fair value per stock option	\$	7 \$	9 \$	12

The following table summarizes stock option activity for the year ended December 31, 2016 and the outstanding stock options as of December 31, 2016.

Ontions		average exercise	Weighted- average remaining contractual term		Aggregate intrinsic value
35,799	\$	34.16	())		, 33 9 9
6,939	\$	39.70			
(7,705)	\$	31.50			
(1,831)	\$	38.14			
(126)	\$	32.87			
33,076	\$	35.73	6.2	\$	284,821
32,502	\$	35.67	6.2	\$	281,976
18,707	\$	33.71	4.7	\$	198,873
	6,939 (7,705) (1,831) (126) 33,076 32,502	Options 35,799 \$ 6,939 \$ (7,705) \$ (1,831) \$ (126) \$ 33,076 \$ 32,502 \$	Options exercise price 35,799 \$ 34.16 6,939 \$ 39.70 (7,705) \$ 31.50 (1,831) \$ 38.14 (126) \$ 32.87 33,076 \$ 35.73 32,502 \$ 35.67	Options Weighted-average exercise price Weighted-contractual term (in years) 35,799 \$ 34.16 6,939 \$ 39.70 (7,705) \$ 31.50 (1,831) \$ 38.14 (126) \$ 32.87 33,076 \$ 35.73 6.2 32,502 \$ 35.67 6.2	Weighted-average exercise price (in years) 35,799

The aggregate intrinsic value in the table above represents the difference between the exercise price and the company's closing stock price on the last trading day of the year. The total intrinsic value of options exercised in 2016, 2015 and 2014 were \$162 million, \$43 million and \$114 million, respectively.

As of December 31, 2016, \$45 million of unrecognized compensation cost related to stock options is expected to be recognized as expense over a weighted-average period of approximately 1.3 years.

RSUs

RSUs are granted to employees and non-employee directors. RSUs granted to employees generally vest in one-third increments over a three-year period. RSUs granted to non-employee directors generally cliff-vest one year from the grant date. The grant-date fair value, adjusted for estimated forfeitures, is recognized as expense on a straight-line basis over the substantive vesting period. The fair value of RSUs is determined based on the number of shares granted and the close price of the company's common stock on the date of grant.

The following table summarizes nonvested RSU activity for the year ended December 31, 2016.

		Weighted-
		average
		grant-date
(share units in thousands)	Share units	fair value
Nonvested RSUs as of January 1, 2016	3,078	\$ 37.62
Granted	1,229	\$ 40.32
Vested	(1,271)	\$ 23.14
Forfeited	(338)	\$ 33.20
Nonvested RSUs as of December 31, 2016	2,698	\$ 32.90

As of December 31, 2016 \$49 million of unrecognized compensation cost related to RSUs is expected to be recognized as expense over a weighted-average period of approximately 1.4 years. The weighted-average grant-date fair value of RSUs in 2016, 2015 and 2014 was \$40.32, \$66.65 and \$71.22, respectively. The fair value of RSUs vested in 2016, 2015 and 2014 was \$50 million, \$73 million and \$62 million, respectively.

PSUs

The company's annual equity awards stock compensation program for senior management includes the issuance of PSUs based on adjusted operating margin, ROIC, as well as market conditions. The vesting condition for adjusted operating margin or ROIC PSUs is set at the beginning of the year for each tranche of the award during the three-year service period. Compensation cost for the adjusted operating margin or ROIC PSUs is measured based on the fair value of the awards on the date that the specific vesting terms for each tranche of the award are established. The fair value of the awards is determined based on the quoted price of the company's stock on the grant date for each tranche of the award. The compensation cost for adjusted operating margin or ROIC PSUs is adjusted at each reporting date to reflect the estimated probability of achieving the vesting condition.

The fair value for PSUs based on market conditions is determined using a Monte Carlo model. The assumptions used in estimating the fair value of these PSUs granted during the period, along with the grant-date fair values, were as follows:

years ended December 31	2016	2015	2014
Baxter volatility	20%	19%	20%
Peer group volatility	17%-51%	16%-38%	13%-58%
Correlation of returns	0.22-0.73	0.24-0.55	0.23-0.66
Risk-free interest rate	1.0%	1.1%	0.7%
Fair value per PSU	\$ 51	\$ 46	\$ 57

Unrecognized compensation cost related to all unvested PSUs of \$9 million at December 31, 2016 is expected to be recognized as expense over a weighted-average period of 1.9 years.

The following table summarizes nonvested PSU activity for the year ended December 31, 2016.

		Weighted- average grant-date
(share units in thousands)	Share units	fair value
Nonvested PSUs as of January 1, 2016	300	\$ 36.11
Granted	282	\$ 45.83
Vested	(260)	\$ 34.42
Forfeited	(44)	\$ 42.27
Nonvested PSUs as of December 31, 2016	278	\$ 46.82

Realized Excess Income Tax Benefits and the Impact on the Statements of Cash Flows

Realized excess tax benefits associated with stock compensation are presented in the consolidated statements of cash flows as an outflow within the operating section and an inflow within the financing section. Realized excess tax benefits from stock-based compensation related to continuing operations were \$39 million, \$7 million and \$15 million in 2016, 2015 and 2014, respectively.

Employee Stock Purchase Plan

Nearly all employees are eligible to participate in the company's employee stock purchase plan. The employee purchase price is 85% of the closing market price on the purchase date.

The Baxter International Inc. Employee Stock Purchase Plan provides for 10 million shares of common stock available for issuance to eligible participants, of which approximately five million shares were available for future purchases as of December 31, 2016.

During 2016, 2015, and 2014, the company issued approximately 1.0 million, 1.1 million and 0.8 million shares, respectively, under the employee stock purchase plan. The number of shares under subscription at December 31, 2016 totaled approximately 1.0 million.

Cash Dividends

Total cash dividends declared per common share for 2016, 2015, and 2014 were \$0.51, \$1.27 and \$2.05, respectively.

A quarterly dividend of \$0.115 per share (\$0.46 on an annualized basis) was declared in February 2016 and was paid in April 2016. Quarterly dividends of \$0.13 per share (\$0.52 on an annualized basis) were declared in May and August of 2016 and were paid in June and October of 2016, respectively. Baxter's board of directors declared a quarterly dividend of \$0.13 per share in November of 2016, which was paid in January of 2017.

Stock Repurchase Programs

As authorized by the board of directors, the company repurchases its stock depending on the company's cash flows, net debt level and market conditions. The company repurchased 6.3 million shares for \$287 million in cash in 2016 and 8 million shares for \$600 million in cash in 2014. The company did not repurchase shares in 2015. In July 2012, the board of directors authorized the repurchase of up to \$2 billion of the company's common stock. The board of directors increased this authority by an additional \$1.5 billion in November 2016. \$1.7 billion remained available as of December 31, 2016.

NOTE 13

RETIREMENT AND OTHER BENEFIT PROGRAMS

The company sponsors a number of qualified and nonqualified pension plans for eligible employees. The company also sponsors certain unfunded contributory healthcare and life insurance benefits for substantially all domestic retired employees. Newly hired employees in the United States and Puerto Rico are not eligible to participate in the pension plans but receive a higher level of company contributions in the defined contribution plans.

In the second quarter of 2016, the company made a \$706 million voluntary, non-cash contribution to the qualified U.S. pension plan using Retained Shares. Refer to Note 2 for additional information regarding Retained Share transactions.

In the second quarter of 2015, in connection with the transfer of liabilities and assets from a combined Baxter pension or Other Postemployment Benefits (OPEB) plan to a newly created Baxalta pension or OPEB plan, the company remeasured pension and OPEB liabilities and assets for several of its plans. The remeasurement resulted in a reduction to pension and OPEB obligations of \$220 million, with an offset to AOCI.

In July 2014, a change was made to postemployment medical benefits for retirees who are age 65 or older to provide eligible retirees and their dependents a subsidy to be utilized on a medical insurance exchange. This change was accounted for as a significant plan amendment. Accordingly, the postemployment benefit obligation was remeasured using a discount rate of 4.30% as of July 31, 2014. Prior to the effect of the separation, the plan amendment resulted in a reduction to the postemployment benefit obligation of \$124 million, which was partially offset by a \$44 million actuarial loss for the change in discount rate. The corresponding \$80 million recognized in AOCI will be amortized as a reduction to net periodic benefit cost over approximately 11 years.

Reconciliation of Pension and OPEB Plan Obligations, Assets and Funded Status

The benefit plan information in the table below pertains to all of the company's pension and OPEB plans, both in the United States and in other countries.

	Pension benefits			OPEB				
as of and for the years ended December 31 (in millions)	'-	2016		2015		2016		2015
Benefit obligations								
Beginning of period	\$	5,423	\$	6,331	\$	266	\$	551
Service cost		93		128		2		4
Interest cost		183		211		8		14
Participant contributions		5		7		_		_
Actuarial (gain)/loss		298		(105)		10		(261)
Benefit payments		(234)		(214)		(20)		(21)
Settlements		(6)		(3)		_		
Plan amendments		_		(2)		(23)		_
Separation of Baxalta		_		(821)		_		(21)
Foreign exchange and other		(45)		(109)				
End of period		5,717		5,423		243		266
Fair value of plan assets								
Beginning of period		3,698		4,197		_		
Actual return on plan assets		309		(41)		_		_
Employer contributions		752		157		20		21
Participant contributions		5		7		_		_
Benefit payments		(234)		(214)		(20)		(21)
Settlements		(6)		(3)		_		_
Separation of Baxalta		_		(347)		_		
Foreign exchange and other		(23)		(58)				_
End of period		4,501		3,698		_		
Funded status at December 31	\$	(1,216)	\$	(1,725)	\$	(243)	\$	(266)
Amounts recognized in the consolidated balance sheets								
Noncurrent asset	\$	42	\$	47	\$	_	\$	_
Current liability		(23)		(22)		(19)		(19)
Noncurrent liability		(1,235)		(1,750)		(224)		(247)
Net liability recognized at December 31	\$	(1,216)	\$	(1,725)	\$	(243)	\$	(266)

Accumulated Benefit Obligation Information

The pension obligation information in the table above represents the projected benefit obligation (PBO). The PBO incorporates assumptions relating to future compensation levels. The accumulated benefit obligation (ABO) is the same as the PBO except that it includes no assumptions relating to future compensation levels. The ABO for all of the company's pension plans was \$5.4 billion and \$5.1 billion at the 2016 and 2015 measurement dates, respectively.

The information in the funded status table above represents the totals for all of the company's pension plans. The following table is information relating to the individual plans in the funded status table above that have an ABO in excess of plan assets.

as of December 31 (in millions)	20	16	2015
ABO	\$	5,153 \$	4,855
Fair value of plan assets		4.190	3,374

The following table is information relating to the individual plans in the funded status table above that have a PBO in excess of plan assets (many of which also have an ABO in excess of assets, and are therefore also included in the table directly above).

as of December 31 (in millions)	2016	2015			
PBO	\$ 5,523	\$	5,244		
Fair value of plan assets	4,265		3,472		

Expected Net Pension and OPEB Plan Payments for the Next 10 Years

(in millions)	Pension benefits	OPEB
2017	\$ 237	\$ 19
2018	246	20
2019	261	19
2020	270	18
2021	282	18
2022 through 2026	1,566	78
Total expected net benefit payments for next 10 years	\$ 2,862	\$ 172

The expected net benefit payments above reflect the company's share of the total net benefits expected to be paid from the plans' assets (for funded plans) or from the company's assets (for unfunded plans). The federal subsidies relating to the Medicare Prescription Drug, Improvement and Modernization Act are not expected to be significant.

Amounts Recognized in AOCI

The pension and OPEB plans' gains or losses, prior service costs or credits, and transition assets or obligations not yet recognized in net periodic benefit cost are recognized on a net-of-tax basis in AOCI and will be amortized from AOCI to net periodic benefit cost in the future. The company utilizes the average future working lifetime as the amortization period for prior service.

The following table is a summary of the pre-tax losses included in AOCI at December 31, 2016 and December 31, 2015.

(in millions)	Pension benefits			OPEB		
Actuarial loss (gain)	\$	1,885	\$	(89)		
Prior service credit and transition obligation		(5)		(103)		
Total pre-tax loss recognized in AOCI at December 31, 2016	\$	1,880	\$	(192)		
Actuarial loss (gain)	\$	1,762	\$	(107)		
Prior service credit and transition obligation		(5)		(94)		
Total pre-tax loss recognized in AOCI at December 31, 2015	\$	1,757	\$	(201)		

Refer to Note 14 for the net-of-tax balances included in AOCI as of each of the year-end dates. The following table is a summary of the net-of-tax amounts recorded in OCI relating to pension and OPEB plans.

years ended December 31 (in millions)	2016	2015	2014
Gain (loss) arising during the year, net of tax expense (benefit) of (\$72) in 2016, \$44 in			
2015 and (\$240) in 2014	\$ (191) \$	45	\$ (494)
Distribution to Baxalta, net of tax expense of \$73	_	198	_
Amortization of loss to earnings, net of tax expense of \$36 in 2016, \$61 in 2015 and \$47			
in 2014	94	120	94
Pension and other employee benefits (loss) gain	\$ (97) \$	363	\$ (400)

In 2016 and 2015, OCI activity for pension and OPEB plans was primarily related to actuarial gains and losses. In 2014, OCI activity for pension and OPEB plans was related to actuarial losses as well as the OPEB plan amendment referenced above.

Amounts Expected to be Amortized from AOCI to Net Periodic Benefit Cost in 2017

With respect to the AOCI balance at December 31, 2016, the following table is a summary of the pre-tax amounts expected to be amortized to net periodic benefit cost in 2017.

(in millions)	Pensio	on benefits	OPEB
Actuarial loss/(gain)	\$	162 \$	(11)
Prior service credit and transition obligation		_	(15)
Total pre-tax amount expected to be amortized from AOCI to net pension and OPEB cost in 2017	\$	162 \$	(26)

Net Periodic Benefit Cost – Continuing Operations

years ended December 31 (in millions)	2	016	2015	2014
Pension benefits				
Service cost	\$	93	\$ 128	\$ 130
Interest cost		183	211	242
Expected return on plan assets		(298)	(270)	(269)
Amortization of net losses and other deferred amounts		149	192	144
Settlement losses		2	2	1
Net pension costs related to discontinued operations		_	(43)	(55)
Net periodic pension benefit cost	\$	129	\$ 220	\$ 193
OPEB				
Service cost	\$	2	\$ 4	\$ 5
Interest cost		8	14	25
Amortization of net loss and prior service credit		(19)	(11)	(3)
Curtailment		(4)	_	_
Net OPEB costs related to discontinued operations		_	_	(1)
Net periodic OPEB cost	\$	(13)	\$ 7	\$ 26

Weighted-Average Assumptions Used in Determining Benefit Obligations at the Measurement Date

	Pension ben	efits	OPEB		
	2016	2015	2016	2015	
Discount rate					
U.S. and Puerto Rico plans	4.09%	4.36%	3.89 %	4.12%	
International plans	2.05%	2.60%	n/a	n/a	
Rate of compensation increase					
U.S. and Puerto Rico plans	3.75%	3.75%	n/a	n/a	
International plans	3.08%	3.37%	n/a	n/a	
Annual rate of increase in the per-capita cost	n/a	n/a	6.25%	6.50%	
Rate decreased to	n/a	n/a	5.00%	5.00%	
by the year ended	n/a	n/a	2022	2022	

The assumptions above, which were used in calculating the December 31, 2016 measurement date benefit obligations, will be used in the calculation of net periodic benefit cost in 2017.

Effective January 1, 2016, the company changed its approach used to calculate the service and interest components of net periodic benefit cost. Previously, the company calculated the service and interest components utilizing a single weighted-average discount rate derived from the yield curve used to measure the benefit obligation. The company has elected an alternative approach that utilizes a full yield curve approach in the estimation of these components by applying the specific spot rates along the yield curve used in the determination of the benefit obligation to their underlying projected cash flows. The company believes this approach provides a more precise measurement of service and interest costs by improving the correlation between projected benefit cash flows and their corresponding spot rates. The company accounted for this change prospectively as a change in estimate.

Weighted-Average Assumptions Used in Determining Net Periodic Benefit Cost

	Pe	nsion benefits			OPEB	
	2016	2015	2014	2016	2015	2014
Discount rate						
U.S. and Puerto Rico plans	4.36%	4.00%	4.85%	4.12%	3.95%	4.90%
International plans	2.60%	2.26%	3.41%	n/a	n/a	n/a
Expected return on plan assets						
U.S. and Puerto Rico plans	7.00%	7.25%	7.50%	n/a	n/a	n/a
International plans	6.07%	6.20%	6.02%	n/a	n/a	n/a
Rate of compensation increase						
U.S. and Puerto Rico plans	3.75%	3.76%	3.80%	n/a	n/a	n/a
International plans	3.37%	3.33%	3.29%	n/a	n/a	n/a
Annual rate of increase in the per-capita cost	n/a	n/a	n/a	6.50%	6.00%	6.25%
Rate decreased to	n/a	n/a	n/a	5.00%	5.00%	5.00%
by the year ended	n/a	n/a	n/a	2022	2019	2019

The 2015 actuarial gain for the OPEB plan was primarily related to adjustments to the assumptions for retirees who are age 65 and older and receive a subsidy to be utilized on a medical insurance exchange.

The company establishes the expected return on plan assets assumption primarily based on a review of historical compound average asset returns, both company-specific and relating to the broad market (based on the company's asset allocation), as well as an analysis of current market and economic information and future expectations. The company plans to use a 6.50% assumption for its U.S. and Puerto Rico plans for 2017.

Effect of a One-Percent Change in Assumed Healthcare Cost Trend Rate on the OPEB Plan

	One percent increase				One perc		
years ended December 31 (in millions)		2016	2015		2016	2015	
Effect on total of service and interest cost components of OPEB cost	\$	— \$	1	1 \$	_	\$	(1)
Effect on OPEB obligation	\$	— \$	1	l \$	_	\$	(4)

Pension Plan Assets

An investment committee of members of senior management is responsible for supervising, monitoring and evaluating the invested assets of the company's funded pension plans. The investment committee, which meets at least quarterly, abides by documented policies and procedures relating to investment goals, targeted asset allocations, risk management practices, allowable and prohibited investment holdings, diversification, use of derivatives, the relationship between plan assets and benefit obligations, and other relevant factors and considerations.

The investment committee's policies and procedures include the following:

- Ability to pay all benefits when due;
- Targeted long-term performance expectations relative to applicable market indices, such as Russell, MSCI EAFE, and other indices;
- Targeted asset allocation percentage ranges (summarized below), and periodic reviews of these allocations;
- Diversification of assets among third-party investment managers, and by geography, industry, stage of business cycle and other measures;

- Specified investment holding and transaction prohibitions (for example, private placements or other restricted securities, securities that are not traded in a sufficiently active market, short sales, certain derivatives, commodities and margin transactions);
- Specified portfolio percentage limits on holdings in a single corporate or other entity (generally 5% at time of purchase, except for holdings in U.S. government or agency securities);
- Specified average credit quality for the fixed-income securities portfolio (at least A- by Standard & Poor's or A3 by Moody's);
- Specified portfolio percentage limits on foreign holdings; and
- Periodic monitoring of investment manager performance and adherence to the investment committee's policies.

Plan assets are invested using a total return investment approach whereby a mix of equity securities, debt securities and other investments are used to preserve asset values, diversify risk and exceed the planned benchmark investment return. Investment strategies and asset allocations are based on consideration of plan liabilities, the plans' funded status and other factors, such as the plans' demographics and liability durations. Investment performance is reviewed by the investment committee on a quarterly basis and asset allocations are reviewed at least annually.

Plan assets are managed in a balanced portfolio comprised of two major components: return-seeking investments and liability hedging investments. The target allocations for plan assets are 53% in return-seeking investments and 46% in liability hedging investments and other holdings. The documented policy includes an allocation range based on each individual investment type within the major components that allows for a variance from the target allocations of approximately two to five percentage points depending on the investment type. Return-seeking investments primarily include common stock of U.S. and international companies, common/collective trust funds, mutual funds, and partnership investments. Liability hedging investments and other holdings primarily include cash, money market funds with an original maturity of three months or less, U.S. and foreign government and governmental agency issues, corporate bonds, municipal securities, hedge funds, derivative contracts and asset-backed securities.

While the investment committee provides oversight over plan assets for U.S. and international plans, the summary above is specific to the plans in the United States. The plan assets for international plans are managed and allocated by the entities in each country, with input and oversight provided by the investment committee. The plan assets for the U.S. and international plans are included in the table below.

The following tables summarize the bases used to measure the pension plan assets and liabilities that are carried at fair value on a recurring basis.

		Basis of fair value measurement							
(in millions)	 ance at er 31, 2016	Quoted prices in active markets for identical assets (Level 1)	Significant other observable inputs (Level 2)	Significant unobservable inputs (Level 3)	Measured at NAV				
Assets									
Fixed income securities									
Cash and cash equivalents	\$ 443	\$ 16	\$ 427	\$ —	\$ —				
U.S. government and government agency issues	457	_	457	_					
Corporate bonds	850	13	837	_	_				
Equity securities									
Common stock:									
Large cap	545	545	_	_	_				
Mid cap	371	371	_	_	_				
Small cap	 94	94	_	_	<u> </u>				
Total common stock	1,010	1,010	_	_	_				
Mutual funds	 336	118	218	_					
Common/collective trust funds	900	_	143	6	751				
Partnership investments	388	_	_	_	388				
Other holdings	117	10	97	10	_				
Collateral held on loaned securities	126	_	126	_	_				
Liabilities									
Collateral to be paid on loaned securities	(126)	(37)	(89)	_	<u> </u>				
Fair value of pension plan assets	\$ 4,501	\$ 1,130	\$ 2,216	\$ 16	\$ 1,139				

		Basis of fair value measurement								
(in millions)	 ance at er 31, 2015	in active markets for c identical assets		Significant other observable inputs (Level 2)		Significant unobservable inputs (Level 3)			sured at	
Assets										
Fixed income securities										
Cash and cash equivalents	\$ 166	\$	32	\$	134	\$	_	\$	_	
U.S. government and government agency issues	347		_		347		_		_	
Corporate bonds	632		_		632		—		_	
Equity securities										
Common stock:										
Large cap	880		880		_		—		_	
Mid cap	359		359		_		—		_	
Small cap	111		111		_		_		_	
Total common stock	 1,350		1,350				_			
Mutual funds	 328		112		216					
Common/collective trust funds	565		_		131		6		428	
Partnership investments	209		_		_		_		209	
Other holdings	101		4		95		2		_	
Collateral held on loaned securities	208		_		208				_	
Liabilities										
Collateral to be paid on loaned securities	(208)		(60)		(148)		_			
Fair value of pension plan assets	\$ 3,698	\$	1,438	\$	1,615	\$	8	\$	637	

The following table is a reconciliation of changes in fair value measurements that used significant unobservable inputs (Level 3).

a		Common/collective		Other
(in millions)	Total	trust funds		holdings
Balance at December 31, 2014	\$ 8	\$	6	\$ 2
Actual return on plan assets still held at year end			_	_
Actual return on plan assets sold during the year	_		—	_
Purchases, sales and settlements	_		_	
Balance at December 31, 2015	8		6	2
Actual return on plan assets still held at year end	_		_	_
Actual return on plan assets sold during the year	_		_	_
Purchases, sales and settlements	8		_	8
Balance at December 31, 2016	\$ 16	\$	6	\$ 10

The assets and liabilities of the company's pension plans are valued using the following valuation methods:

Investment category	Valuation methodology
Cash and cash equivalents	These largely consist of a short-term investment fund, U.S. Dollars and foreign currency. The fair value of the short-term investment fund is based on the net asset value
U.S. government and government agency issues	Values are based on reputable pricing vendors, who typically use pricing matrices or models that use observable inputs
Corporate bonds	Values are based on reputable pricing vendors, who typically use pricing matrices or models that use observable inputs
Common stock	Values are based on the closing prices on the valuation date in an active market on national and international stock exchanges
Mutual funds	Values are based on the net asset value of the units held in the respective fund which are obtained from national and international exchanges or based on the net asset value of the underlying assets of the fund provided by the fund manager
Common/collective trust funds	Values are based on the net asset value of the units held at year end
Partnership investments	Values are based on the estimated fair value of the participation by the company in the investment as determined by the general partner or investment manager of the respective partnership
Other holdings	The value of these assets vary by investment type, but primarily are determined by reputable pricing vendors, who use pricing matrices or models that use observable inputs
Collateral held on loaned securities	Values are based on the net asset value per unit of the fund in which the collateral is invested
Collateral to be paid on loaned securities	Values are based on the fair value of the underlying securities loaned on the valuation date

Expected Pension and OPEB Plan Funding

The company's funding policy for its pension plans is to contribute amounts sufficient to meet legal funding requirements, plus any additional amounts that the company may determine to be appropriate considering the funded status of the plans, tax deductibility, the cash flows generated by the company, and other factors. Volatility in the global financial markets could have an unfavorable impact on future funding requirements. The company has no obligation to fund its principal plans in the United States in 2017. The company continually reassesses the amount and timing of any discretionary contributions. The company expects to make a contribution of at least \$20 million to its Puerto Rico pension plan and at least a \$40 million contribution to its foreign pension plans in 2017. The company expects to have net cash outflows relating to its OPEB plan of approximately \$19 million in 2017.

The following table details the funded status percentage of the company's pension plans as of December 31, 2016, including certain plans that are unfunded in accordance with the guidelines of the company's funding policy outlined above.

	United States and Puerto Rico				International					
	Qualified		Nonqualified		Funded		Unfunded			
as of December 31, 2016 (in millions)	plans			plan	n plan		lans plans		Total	
Fair value of plan assets	\$	3,828		n/a	\$	673		n/a	\$	4,501
PBO		4,296	\$	214		843	\$	363		5,716
Funded status percentage		89%		n/a		80%		n/a		79%

U.S. Defined Contribution Plan

Most U.S. employees are eligible to participate in a qualified defined contribution plan. Expense recognized by the company was \$50 million in 2016, \$46 million in 2015 and \$54 million in 2014.

NOTE 14

ACCUMULATED OTHER COMPREHENSIVE INCOME

Comprehensive income includes all changes in shareholders' equity that do not arise from transactions with stockholders, and consists of net income, CTA, pension and other employee benefits, unrealized gains and losses on cash flow hedges and unrealized gains and losses on available-for-sale equity securities. The following table is a net-of-tax summary of the changes in AOCI by component for the years ended December 31, 2016 and 2015.

	Pension and						Available-			
		other employee				Hedging for-sale				
(in millions)		CTA		benefits		activities securities		ecurities	Total	
Gains (losses)										
Balance as of December 31, 2015	\$	(3,191)	\$	(1,064)	\$	7	\$	4,472	\$	224
Other comprehensive income before reclassifications		(247)		(191)		1		104		(333)
Amounts reclassified from AOCI		_		94		(5)		(4,536)		(4,447)
Net other comprehensive (loss) income		(247)		(97)		(4)		(4,432)		(4,780)
Balance as of December 31, 2016	\$	(3,438)	\$	(1,161)	\$	3	\$	40	\$	(4,556)

	Pension and other employee				Available- Hedging for-sale				
(in millions)	CTA		benefits		activities		securities		Total
Gains (losses)									
Balance as of December 31, 2014	\$	(2,323)	\$	(1,427)	\$	34	\$	66	\$ (3,650)
Other comprehensive income before reclassifications		(1,094)		45		46		4,461	3,458
Amounts reclassified from AOCI		_		120		(31)		(23)	66
Net other comprehensive (loss) income		(1,094)		165		15		4,438	3,524
Distribution to Baxalta		226		198		(42)		(32)	350
Balance as of December 31, 2015	\$	(3,191)	\$	(1,064)	\$	7	\$	4,472	\$ 224

The following table is a summary of the amounts reclassified from AOCI to net income during the years ended December 31, 2016 and 2015.

	AOCI	(a)		
20	16	2	2015	Location of impact in income statement
\$	(130)	\$	(181)	
	(130)		(181)	Total before tax
	36		61	Tax benefit
\$	(94)	\$	(120)	Net of tax
\$	9	\$	_	Other (income) expense, net
	(3)		47	Cost of sales

Total before tax

Total before tax Tax benefit

Total net of tax

Other (income) expense, net

Tax expense

Net of tax

Net of tax

47

(16)

31

38

38

(15)

23

(66)

Amounts reclassified from

6

4,536

4,536

4,536

4,447

\$

\$

\$

\$

\$

\$

(a)	Amounts in parentheses indicate reductions to net income.

Amortization of pension and other employee benefits items

Actuarial losses and other(b)

Gains (losses) on hedging activities Interest rate contracts

Foreign exchange contracts

Total reclassification for the period

Gain on available-for-sale equity securities

Available-for-sale securities

(in millions)

Refer to Note 9 for additional information regarding hedging activity and Note 13 for additional information regarding the amortization of pension and other employee benefits items.

⁽b) These AOCI components are included in the computation of net periodic benefit cost disclosed in Note 13.

NOTE 15

INCOME TAXES

Income from Continuing Operations Before Income Tax Expense by Category

years ended December 31 (in millions)	2	016	2015	2014
United States	\$	3,906	\$ (738)	\$ (803)
International		1,048	1,166	1,293
Income from continuing operations before income taxes	\$	4,954	\$ 428	\$ 490

Income Tax Expense Related to Continuing Operations

years ended December 31 (in millions)	2016	2015	2014
Current			
United States			
Federal	\$ 10 \$	(251)	\$ (305)
State and local	(3)	(6)	(44)
International	282	345	398
Current income tax expense	289	88	49
Deferred			
United States			
Federal	(286)	(9)	63
State and local	3	(20)	7
International	(18)	(24)	(86)
Deferred income tax expense	(301)	(53)	(16)
Income tax (benefit) expense	\$ (12) \$	35	\$ 33

Deferred Tax Assets and Liabilities

as of December 31 (in millions)	2	2016		2015
Deferred tax assets				
Accrued expenses	\$	377	\$	389
Retirement benefits		411		352
Tax credits and net operating losses		747		547
Valuation allowances		(150)		(135)
Total deferred tax assets		1,385		1,153
Deferred tax liabilities				
Subsidiaries' unremitted earnings		145		147
Asset basis differences		704		847
Total deferred tax liabilities		849	•	994
Net deferred tax asset	\$	536	\$	159

At December 31, 2016, the company had U.S. operating loss carryforwards totaling \$887 million and tax credit carryforwards totaling \$332 million. The U.S. operating loss carryforwards expire between 2018 and 2036 and the tax credits expire between 2017 and 2034. At December 31, 2016, the company had foreign operating loss carryforwards totaling \$1.4 billion and foreign tax credit carryforwards totaling \$48 million. Of these foreign amounts, \$38 million expires in 2017, \$18 million expires in 2018, \$20 million expires in 2019, \$41 million expires in 2020, \$28 million expires in 2021, \$293 million expires after 2021 and \$1 billion has no expiration date. Realization of these operating loss and tax credit carryforwards depends on generating sufficient taxable income in future periods. A valuation allowance of \$150 million and \$135 million was recorded at December 31, 2016 and 2015, respectively, to reduce the deferred tax assets associated with net operating loss and tax credit carryforwards, because the company does not believe it is more likely than not that these assets will be fully realized prior to expiration. The company will continue to evaluate the need for additional valuation allowances and, as circumstances change, the valuation allowance may change.

Income Tax Expense Related to Continuing Operations Reconciliation

years ended December 31 (in millions)	2016	2015		2014
Income tax expense at U.S. statutory rate	\$ 1,734	\$ 150	\$	174
Retained shares tax free exchange gains	(1,587)	_		_
Tax incentives	(126)	(133)		(105)
State and local taxes	1	(13)		(24)
Foreign tax expense (benefit)	5	11		(16)
Valuation allowances	3	5		7
Contingent tax matters	(48)	9		(18)
Branded Prescription Drug Fee	1	1		4
Deferred tax charge on intangible intra-group transfers	13	14		14
R&D tax credit	(2)	(4)		(1)
Puerto Rico excise tax credit	(5)	(9)		
Other factors	(1)	4		(2)
Income tax (benefit) expense	\$ (12)	\$ 35	\$	33

The company recognized deferred US income tax expense of \$35 million during 2016 relating to 2016 earnings outside the United States that are not deemed indefinitely reinvested. The company continues to evaluate whether to indefinitely reinvest earnings in certain foreign jurisdictions as it continues to analyze the company's global financial structure. Currently, management intends to continue to reinvest past earnings in several jurisdictions outside of the United States indefinitely, and therefore has not recognized U.S. income tax expense on these earnings. U.S. federal and state income taxes, net of applicable credits, on these foreign unremitted earnings from continuing operations of \$9.3 billion as of December 31, 2016 would be approximately \$2.6 billion. As of December 31, 2015 the foreign unremitted earnings from continuing operations and U.S. federal and state income tax amounts were \$8.5 billion and \$2.4 billion, respectively.

Effective Income Tax Rate — Continuing Operations

The effective income tax rate for continuing operations was \$(0.2%) in 2016, 8.2% in 2015 and 6.7% in 2014. As detailed in the income tax expense reconciliation table above, the company's effective tax rate differs from the U.S. federal statutory rate each year due to certain operations that are subject to tax incentives, state and local taxes, and foreign taxes that are different than the U.S. federal statutory rate. In addition, the effective tax rate can be impacted each period by discrete factors and events.

Factors impacting the company's effective tax rate in 2016 included tax-free net realized gains during the first and second quarter associated with the exchanges of Baxalta Retained Shares for the company's debt and the company's shares as well as tax-free net realized gains associated with the contribution of Baxalta Retained Shares to the company's pension plan. Additionally, the income tax rate for 2016 was favorably impacted by tax benefits from partially settling an IRS (2008-2013) income tax audit, settling a German (2008-2011) income tax audit, resolution of uncertain tax positions related to the company's former Turkish JV and other miscellaneous TP matters including partial settlement of interest expense deductions related to the company's acquisition of Gambro.

Factors adversely impacting the company's effective tax rate in 2015 included charges related to contingent tax matters primarily related to transfer pricing and separation of Baxalta as well as the need to record valuation allowances for some loss making entities. Partially offsetting the foregoing adverse factors was a benefit from reaching a settlement of the Puerto Rico excise tax matter as well as the retroactive reinstatement in December 2015 of the US R&D credit resulting from the Protecting Americans from Tax Hikes Act of 2015.

Factors impacting the company's effective tax rate in 2014 included the favorable settlement of a portion of the company's contingent tax matter related to operations in Turkey as well as a favorable shift of earnings from high to low tax jurisdictions compared to the prior period. Additionally, the effective tax rate was unfavorably impacted by increases in valuation allowances in respect of the tax benefit from losses that the company does not believe that it is more likely than not to realize and interest expense related to the company's unrecognized tax benefits.

Unrecognized Tax Benefits

The company classifies interest and penalties associated with income taxes in the income tax expense line in the consolidated statements of income. Net interest and penalties recorded during 2016, 2015 and 2014 were \$6 million, \$3 million and \$12 million, respectively. The liability recorded at December 31, 2016 and 2015 related to interest and penalties was \$11 million and \$56 million, respectively. The decrease in the liability for interest and penalties was due, in part, to settlement of audits in the United States,

Germany and Italy, as described in the Examination of Tax Returns section below. The total amount of unrecognized tax benefits that, if recognized, would impact the effective tax rate is approximately \$82 million.

The following table is a reconciliation of the company's unrecognized tax benefits, including those related to discontinued operations for the years ended December 31, 2016, 2015 and 2014.

as of and for the years ended (in millions)	2	016	2015	2014
Balance at beginning of the year	\$	191	\$ 206	\$ 287
Increase associated with tax positions taken during the current year		7	24	41
Decrease associated with tax positions taken during a prior year		(31)	(26)	(27)
Settlements		(75)	(3)	(82)
Decrease associated with lapses in statutes of limitations		(10)	(10)	(13)
Balance at end of the year	\$	82	\$ 191	\$ 206

Of the gross unrecognized tax benefits, \$74 million and \$209 million were recognized as liabilities in the consolidated balance sheets as of December 31, 2016 and 2015, respectively. Baxter has recorded net indemnification receivables from Baxalta in the amount of \$28 million and \$93 million as of December 31, 2016 and 2015, respectively, related to the unrecognized tax benefits for which Baxter is the primary obligor but economically relate to Baxalta operations. Additionally, in the table above amounts related to 2015 included as a decrease a gross liability transferred to Baxalta in the amount of \$10 million for which Baxalta is the primary obligor.

None of the positions included in the liability for uncertain tax positions related to tax positions for which the ultimate deductibility is highly certain but for which there is uncertainty about the timing of such deductibility.

Tax Incentives

The company has received tax incentives in Puerto Rico, Switzerland, Dominican Republic, Costa Rica and certain other taxing jurisdictions outside the United States. The financial impact of the reductions as compared to the statutory tax rates is indicated in the income tax expense reconciliation table above. The tax reductions as compared to the local statutory rate favorably impacted earnings from continuing operations per diluted share by \$0.23 in 2016, \$0.24 in 2015 and \$0.20 in 2014. The Puerto Rico grant provides that the company's manufacturing operations are and will be partially exempt from local taxes until the year 2018.

Examinations of Tax Returns

During 2016, Baxter paid approximately \$303 million to partially settle a U.S. federal income tax audit for the period 2008-2013. Additionally, the company settled a German income tax audit for the period 2008-2011 and settled an Italian audit for the period 2010-2012. As a result of these settlements, the company reduced its gross unrecognized tax benefits by \$75 million. Pursuant to the tax matters agreement with Baxalta, Baxalta paid the company approximately \$37 million related to its tax indemnity obligations in respect of its portion of the settled gross unrecognized tax benefits. See Note 2 for additional details regarding the separation of Baxalta.

As of December 31, 2016, Baxter had ongoing audits in the United States, Austria, Sweden and other jurisdictions. Baxter expects to reduce the amount of its liability for uncertain tax positions within the next 12 months by \$10 million due principally to the resolution of transfer pricing disputes in several jurisdictions. While the final outcome of these matters is inherently uncertain, the company believes it has made adequate tax provisions for all years subject to examination.

NOTE 16

LEGAL PROCEEDINGS

Baxter is involved in product liability, patent, commercial, and other legal matters that arise in the normal course of the company's business. The company records a liability when a loss is considered probable and the amount can be reasonably estimated. If the reasonable estimate of a probable loss is a range, and no amount within the range is a better estimate, the minimum amount in the range is accrued. If a loss is not probable or a probable loss cannot be reasonably estimated, no liability is recorded. As of December 31, 2016 and 2015, the company's total recorded reserves with respect to legal matters were \$53 million and \$29 million, respectively, and the total related receivables were \$10 million and \$5 million, respectively.

Baxter has established reserves for certain of the matters discussed below. The company is not able to estimate the amount or range of any loss for certain contingencies for which there is no reserve or additional loss for matters already reserved. While the liability of the company in connection with the claims cannot be estimated and the resolution thereof in any reporting period could have a significant

impact on the company's results of operations and cash flows for that period, the outcome of these legal proceedings is not expected to have a material adverse effect on the company's consolidated financial position. While the company believes that it has valid defenses in these matters, litigation is inherently uncertain, excessive verdicts do occur, and the company may incur material judgments or enter into material settlements of claims.

In addition to the matters described below, the company remains subject to the risk of future administrative and legal actions. With respect to governmental and regulatory matters, these actions may lead to product recalls, injunctions, and other restrictions on the company's operations and monetary sanctions, including significant civil or criminal penalties. With respect to intellectual property, the company may be exposed to significant litigation concerning the scope of the company's and others' rights. Such litigation could result in a loss of patent protection or the ability to market products, which could lead to a significant loss of sales, or otherwise materially affect future results of operations.

General litigation

On July 31, 2015, Davita Healthcare Partners, Inc. filed suit against Baxter Healthcare Corporation in the District Court of the State of Colorado regarding an ongoing commercial dispute relating to the provision of peritoneal dialysis products. A bench trial concluded in third quarter 2016 and the parties are awaiting the court's decision.

In November 2016, a purported antitrust class action complaint seeking monetary and injunctive relief was filed in the United States District Court for the Northern District of Illinois. The complaint alleges a conspiracy among manufacturers of IV solutions to restrict output and affect pricing in connection with a shortage of such solutions. Similar parallel actions subsequently were filed. In January 2017, a single consolidated complaint covering these matters was filed in the Northern District of Illinois. The New York Attorney General has requested that Baxter provide information regarding business practices in the IV saline industry. The company is cooperating with the New York Attorney General.

Other

In the fourth quarter of 2012, the company received two investigative demands from the United States Attorney for the Western District of North Carolina for information regarding its quality and manufacturing practices and procedures at its North Cove facility. In January 2017, the parties resolved this matter by entering into a deferred prosecution agreement and a civil settlement whereby the company agreed to pay approximately \$18 million and implement certain enhanced compliance measures.

In December 2016, the company received a civil investigative demand from the Commercial Litigation Branch of the United States Department of Justice primarily relating to contingent discount arrangements for, and other promotion of, the company's TISSEEL and ARTISS products. The company is cooperating in this matter.

NOTE 17

SEGMENT INFORMATION

Baxter's two segments are strategic businesses that are managed separately as each business develops, manufactures and markets distinct products and services. The segments and a description of their products and services are as follows:

The **Renal** business provides products and services to treat end-stage renal disease, or irreversible kidney failure, along with other renal therapies. The Renal business offers a comprehensive portfolio to meet the needs of patients across the treatment continuum, including technologies and therapies for peritoneal dialysis (PD), hemodialysis (HD), continuous renal replacement therapy (CRRT) and additional dialysis services.

The **Hospital Products** business manufactures intravenous (IV) solutions and administration sets, premixed drugs and drug-reconstitution systems, oncology injectable drugs, IV nutrition products, infusion pumps, inhalation anesthetics and biosurgery products. The business also provides products and services related to pharmacy compounding, drug formulation and packaging technologies.

The company uses income from continuing operations before net interest expense, income tax expense, depreciation and amortization expense (Segment EBITDA), on a segment basis to make resource allocation decisions and assess the ongoing performance of the company's business segments. Intersegment sales are eliminated in consolidation.

Certain items are maintained at Corporate and are not allocated to a segment. They primarily include most of the company's debt and cash and equivalents and related net interest expense, foreign exchange fluctuations (principally relating to intercompany receivables, payables and loans denominated in a foreign currency) and the majority of the foreign currency hedging activities, corporate headquarters costs, stock compensation expense, nonstrategic investments and related income and expense, certain employee benefit plan costs as well as certain nonrecurring gains, losses, and other charges (such as business optimization, integration and separation-related costs, and asset impairments). Financial information for the company's segments is as follows:

for the years ended December 31 (in millions)	2016	2015			2014
Net sales					
Renal	\$ 3,855	\$	3,789	\$	4,172
Hospital Products	6,308		6,179		6,547
Total net sales	\$ 10,163	\$	9,968	\$	10,719
EBITDA					
Renal	\$ 703	\$	566	\$	666
Hospital Products	2,273		1,998		2,237
Total segment EBITDA	\$ 2,976	\$	2,564	\$	2,903
(in millions)	2016		2015		2014
Total assets					
Renal	\$ 4,315	\$	4,609	\$	4,928
Hospital Products	6,407		6,632		6,915
Total segment assets	\$ 10,722	\$	11,241	\$	11,843

The following table is a reconciliation of segment EBITDA to income from continuing operations before income taxes per the consolidated statements of income.

for the years ended December 31 (in millions)	2016		2015	2014
Total segment EBITDA	\$	2,976	\$ 2,564	\$ 2,903
Reconciling items				
Depreciation and amortization		(800)	(759)	(792)
Stock compensation		(115)	(126)	(126)
Net interest expense		(66)	(126)	(145)
Restructuring charges, net		(285)	(130)	6
Certain foreign exchange fluctuations and hedging activities		34	197	37
Net realized gains on Retained Shares transactions		4,387	_	_
Net loss on debt extinguishment		(153)	(130)	_
Other Corporate items		(1,024)	(1,062)	(1,393)
Income from continuing operations before income taxes	\$	4,954	\$ 428	\$ 490

The following table is a reconciliation of segment assets to consolidated total assets per the consolidated balance sheets.

as of December 31 (in millions)	2016	2015	2014
Total segment assets	\$ 10,722	\$ 11,241	\$ 11,843
Cash and equivalents	2,801	2,213	2,925
Deferred income taxes	629	354	531
PP&E, net	805	932	1,039
Assets held for disposition	50	245	9,363
Other Corporate assets	539	5,977	437
Consolidated total assets	\$ 15,546	\$ 20,962	\$ 26,138

Geographic Information

Net sales are based on product shipment destination and assets are based on physical location.

years ended December 31 (in millions)	2016		2015		2014
Net sales					
United States	\$ 4,259	\$	4,001	\$	3,999
Europe	2,697		2,774		3,257
Asia-Pacific	2,029		1,972		2,079
Latin America and Canada	1,178		1,221		1,384
Consolidated net sales	\$ 10,163	\$	9,968	\$	10,719

as of December 31 (in millions)	2016		2015		2014
PP&E, net					
United States	\$ 1,751	\$	1,746	\$	1,625
Europe	1,166		1,298		1,466
Asia-Pacific	752		757		753
Latin America and Canada	620		585		590
Consolidated PP&E, net	\$ 4,289	\$	4,386	\$	4,434

Net Sales by Franchise

The following table represents net sales by commercial franchise.

years ended December 31		2016		2016		2015	2014
Total Renal ¹	\$	3,855	\$	3,789	\$ 4,172		
Fluid Systems ²		2,300		2,106	2,129		
Integrated Pharmacy Solutions ³		2,245		2,297	2,535		
Surgical Care ⁴		1,321		1,323	1,373		
Other ⁵		442		453	510		
Total Hospital Products	\$	6,308	\$	6,179	\$ 6,547		

- 1 The Renal segment is presented as a separate commercial franchise and includes sales of the company's PD, HD and CRRT.
- 2 Principally includes IV therapies, infusion pumps, and administration sets.
- Includes sales of the company's premixed and oncology drug platforms, nutrition products and pharmacy compounding services.
- 4 Includes sales of the company's inhaled anesthesia products as well as biological products and medical devices used in surgical procedures for hemostasis, tissue sealing and adhesion prevention.
- 5 Principally includes sales from the company's pharmaceutical partnering business.

NOTE 18
QUARTERLY FINANCIAL RESULTS AND MARKET FOR THE COMPANY'S STOCK (UNAUDITED)

years ended December 31 (in millions, except per share data)	Firs	t quarter	Sec	ond quarter	Th	ird quarter	quarter Fourth quarter		Fourth quarter		Full year
2016											
Net sales	\$	2,375	\$	2,585	\$	2,558	\$	2,645	\$ 10,163		
Gross margin ¹		965		972		1,071		1,102	4,110		
Income from continuing operations ¹		3,387		1,212		127		240	4,966		
Income from continuing operations per common share ¹											
Basic		6.17		2.21		0.23		0.44	9.10		
Diluted		6.13		2.19		0.23		0.44	9.01		
(Loss) income from discontinued operations, net of tax		(7)		_		3		3	(1)		
(Loss) income from discontinued operations per common share											
Basic		(0.01)		0.00		0.01		0.01	(0.01)		
Diluted		(0.01)		0.00		0.01		0.00	0.00		
Net income!		3,380		1,212		130		243	4,965		
Net income per common share ¹											
Basic		6.16		2.21		0.24		0.45	9.09		
Diluted		6.12		2.19		0.24		0.44	9.01		
Cash dividends declared per common share		0.115		0.13		0.13		0.13	0.505		
Market price per common share											
High		41.28		46.39		49.03		49.16	49.16		
Low		34.76		41.31		45.09		43.63	34.76		
<u>2015</u>											
Net sales	\$	2,403	\$	2,475	\$	2,487	\$	2,603	\$ 9,968		
Gross margin ²		1,019		1,021		1,034		1,072	4,146		
Income from continuing operations ²		134		74		2		183	393		
Income from continuing operations per common share ²											
Basic		0.25		0.14		0.00		0.33	0.72		
Diluted		0.24		0.13		0.00		0.33	0.72		
Income from discontinued operations, net of tax		296		258		(1)		22	575		
Income from discontinued operations per common share											
Basic		0.54		0.47		0.00		0.04	1.06		
Diluted		0.54		0.47		0.00		0.04	1.04		
Net income ²		430		332		1		205	968		
Net income per common share ²											
Basic		0.79		0.61		0.00		0.37	1.78		
Diluted		0.78		0.60		0.00		0.37	1.76		
Cash dividends declared per common share		0.52		0.52		0.115		0.115	1.27		
Market price per common share											
High ³		38.97		39.05		43.44		38.79	43.44		
Low ³		35.84		34.59		33.25		32.18	32.18		

- The first quarter of 2016 included benefits of \$3.1 billion related to business optimization, separation-related costs, Retained Stake transactions, a loss on debt extinguishment, and product-related items. The second quarter of 2016 included benefits of \$1.0 billion related to business optimization, separation-related costs, Retained Stake transactions, and asset impairment. The third quarter of 2016 included charges of \$155 million related to business optimization, separation-related costs, a loss on debt extinguishment, and a tax matter. The fourth quarter of 2016 included charges of \$47 million related to business optimization, separation-related costs, and reserve items and adjustments.
- The first quarter of 2015 included charges of \$29 million related to business optimization, Gambro integration costs, and separation-related costs. The second quarter of 2015 included benefits of \$5 million related to business optimization, Gambro integration costs, separation-related costs, and tax and legal reserves. The third quarter of 2015 included charges of \$191 million related to business optimization, Gambro integration costs, separation-related costs, a loss on debt extinguishment, and product-related items. The fourth quarter of 2015 included \$17 million related to business optimization, Gambro integration costs, product-related items, separation-related costs, and reserve items and adjustments.
- 3 All stock prices for periods preceding the July 1, 2015 separation of Baxalta are adjusted to reflect the high or low adjusted closing price for the period.

Report of Independent Registered Public Accounting Firm

To the Board of Directors and Shareholders of Baxter International Inc.:

In our opinion, the consolidated financial statements listed in the index appearing under Item 15(1) present fairly, in all material respects, the financial position of Baxter International Inc. and its subsidiaries at December 31, 2016 and 2015, and the results of their operations and their cash flows for each of the three years in the period ended December 31, 2016 in conformity with accounting principles generally accepted in the United States of America. In addition, in our opinion, the financial statement schedule listed in the index appearing under Item 15(2) of this Form 10-K presents fairly, in all material respects, the information set forth therein when read in conjunction with the related consolidated financial statements. Also in our opinion, the Company maintained, in all material respects, effective internal control over financial reporting as of December 31, 2016, based on criteria established in Internal Control — Integrated Framework (2013) issued by the Committee of Sponsoring Organizations of the Treadway Commission (COSO). The Company's management is responsible for these financial statements and financial statement schedule, for maintaining effective internal control over financial reporting and for its assessment of the effectiveness of internal control over financial reporting, included in Management's Report on Internal Control over Financial Reporting included in Item 9A. Our responsibility is to express opinions on these financial statements and on the Company's internal control over financial reporting based on our integrated audits. We conducted our audits in accordance with the standards of the Public Company Accounting Oversight Board (United States). Those standards require that we plan and perform the audits to obtain reasonable assurance about whether the financial statements are free of material misstatement and whether effective internal control over financial reporting was maintained in all material respects. Our audits of the financial statements included examining, on a test basis, evidence supporting the amounts and disclosures in the financial statements, assessing the accounting principles used and significant estimates made by management, and evaluating the overall financial statement presentation. Our audit of internal control over financial reporting included obtaining an understanding of internal control over financial reporting, assessing the risk that a material weakness exists, and testing and evaluating the design and operating effectiveness of internal control based on the assessed risk. Our audits also included performing such other procedures as we considered necessary in the circumstances. We believe that our audits provide a reasonable basis for our opinions.

A company's internal control over financial reporting is a process designed to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles. A company's internal control over financial reporting includes those policies and procedures that (i) pertain to the maintenance of records that, in reasonable detail, accurately and fairly reflect the transactions and dispositions of the assets of the company; (ii) provide reasonable assurance that transactions are recorded as necessary to permit preparation of financial statements in accordance with generally accepted accounting principles, and that receipts and expenditures of the company are being made only in accordance with authorizations of management and directors of the company; and (iii) provide reasonable assurance regarding prevention or timely detection of unauthorized acquisition, use, or disposition of the company's assets that could have a material effect on the financial statements.

Because of its inherent limitations, internal control over financial reporting may not prevent or detect misstatements. Also, projections of any evaluation of effectiveness to future periods are subject to the risk that controls may become inadequate because of changes in conditions, or that the degree of compliance with the policies or procedures may deteriorate.

/s/ PricewaterhouseCoopers LLP Chicago, Illinois February 23, 2017

Item 9. Changes in and Disagreements with Accountants on Accounting and Financial Disclosure.

None.

Item 9A. Controls and Procedures.

Evaluation of Disclosure Controls and Procedures

Baxter carried out an evaluation, under the supervision and with the participation of its Disclosure Committee and management, including the Chief Executive Officer and Chief Financial Officer, of the effectiveness of Baxter's disclosure controls and procedures (as defined in Rules 13a-15(e) and 15d-15(e) under the Exchange Act) as of December 31, 2016. Baxter's disclosure controls and procedures are designed to ensure that information required to be disclosed by Baxter in the reports it files or submits under the Exchange Act is recorded, processed, summarized and reported on a timely basis and that such information is communicated to management, including the Chief Executive Officer, Chief Financial Officer and its Board of Directors, to allow timely decisions regarding required disclosure.

Based on that evaluation the Chief Executive Officer and Chief Financial Officer concluded that the company's disclosure controls and procedures were effective as of December 31, 2016.

Management's Assessment of Internal Control Over Financial Reporting

Management is responsible for establishing and maintaining adequate internal control over financial reporting, as defined in Rules 13a-15(f) and 15d-15(f) under the Securities Exchange Act of 1934, as amended. The company's internal control over financial reporting is a process designed under the supervision of the principal executive and financial officers, and effected by the board of directors, management and other personnel, to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with accounting principles generally accepted in the United States of America.

Management performed an assessment of the effectiveness of the company's internal control over financial reporting as of December 31, 2016. In making this assessment, management used the framework in *Internal Control-Integrated Framework (2013)* issued by the Committee of Sponsoring Organizations of the Treadway Commission. Based on that assessment under the framework in *Internal Control-Integrated Framework (2013)*, management concluded that the company's internal control over financial reporting was effective as of December 31, 2016.

The effectiveness of the company's internal control over financial reporting as of December 31, 2016 has been audited by PricewaterhouseCoopers LLP, an independent registered public accounting firm, as stated in their report which appears herein.

Changes in Internal Control over Financial Reporting

There have been no changes in Baxter's internal control over financial reporting (as such term is defined in Rules 13a-15(f) and 15d-15(f) under the Exchange Act) during the quarter ended December 31, 2016 that have materially affected, or are reasonably likely to materially affect, Baxter's internal control over financial reporting.

Item 9B. Other Information.

None.

PART III

Item 10. Directors, Executive Officers and Corporate Governance.

Refer to information under the captions entitled "Corporate Governance at Baxter International Inc. — Proposal 1 — Election of Directors," — Directors Continuing in Office," "— Board of Directors — Nomination of Directors," "— Committees of the Board — Audit Committee," "— Board Responsibilities — Code of Conduct," and "Ownership of Our Stock — Section 16(a) Beneficial Ownership Reporting Compliance" in Baxter's definitive proxy statement to be filed with the Securities and Exchange Commission and delivered to stockholders in connection with the Annual Meeting of Stockholders to be held on May 2, 2017 (the Proxy Statement), all of which information is incorporated herein by reference. Also refer to information regarding executive officers of Baxter under the caption entitled "Executive Officers of the Registrant" in Part I of this Annual Report on Form 10-K.

Item 11. Executive Compensation.

Refer to information under the captions entitled "Executive Compensation," and "Corporate Governance at Baxter International—Director Compensation" in the Proxy Statement, all of which information is incorporated herein by reference.

Item 12. Security Ownership of Certain Beneficial Owners and Management and Related Stockholder Matters.

The following table provides information relating to shares of common stock that may be issued under Baxter's existing equity compensation plans as of December 31, 2016.

					Number of Shares	
					Remaining	
	Number of Shares				Available for	
	to be Issued upon				Future Issuance	
	Exercise of		Weighted-Average		Under Equity	
	Outstanding		Exercise Price of		Compensation Plans	
	Options,		Outstanding		(Excluding	
	Warrants and		Options, Warrants		Shares Reflected in	
Plan Category	Rights(a)		and Rights(b)		Column(a))(c)	
Equity Compensation Plans Approved by Shareholders	35,252,613	(1)	\$ 35.95	(2)	45,829,969	(3)
Equity Compensation Plans Not Approved by Shareholders	1,140,770	(4)	\$ 30.06		_	
Total	36,393,383	(5)	\$ 35.73	(2)	45,829,969	

- (1) Excludes purchase rights under the Employee Stock Purchase Plan. Under the Employee Stock Purchase Plan, eligible employees may purchase shares of common stock through payroll deductions of up to 15 percent of base pay at a purchase price equal to 85 percent of the closing market price on the purchase date (as defined by the Employee Stock Purchase Plan). A participating employee may not purchase more than \$25,000 in fair market value of common stock under the Employee Stock Purchase Plan in any calendar year and may withdraw from the Employee Stock Purchase Plan at any time
- (2) Restricted stock units and performance share units are excluded when determining the weighted-average exercise price of outstanding options.
- (3) Includes (i) 5,032,670 shares of common stock available for purchase under the Employee Stock Purchase Plan; (ii) 119,171 shares of common stock available under the 2007 Incentive Plan; (iii) 7,228,968 shares of common stock available under the 2011 Incentive Plan; and (iv) 33,449,160 shares of common stock available under the 2015 Incentive Plan.
- (4) Includes shares of common stock issuable upon exercise of options granted under the 2001 Incentive Compensation Program. These shares were made available pursuant to an amendment thereto not approved by shareholders. These additional shares were approved by the company's board of directors, not the company's shareholders, although the company shareholders have approved the 2001 Incentive Compensation Program.
- (5) Includes outstanding awards of 33,076,401 stock options, which have a weighted-average exercise price of \$35.73 and a weighted-average remaining term of 6.2 years, 2,697,906 shares of common stock issuable upon vesting of restricted stock units, and 277,743 shares of common stock reserved for issuance in connection with performance share unit grants.

Refer to information under the captions entitled "Ownership of Our Stock — Security Ownership by Directors and Executive Officers" and "— Security Ownership by Certain Beneficial Owners" in the Proxy Statement, all of which information is incorporated herein by reference.

Item 13. Certain Relationships and Related Transactions, and Director Independence.

Refer to the information under the first paragraph of the caption entitled "Corporate Governance—at Baxter International Inc.—Board of Directors" and the captions entitled "Corporate Governance at Baxter International Inc.—Directors—Director Independence" and "Corporate Governance at Baxter International Inc.—Other Corporate Governance Information—Certain Relationships and Related Person Transactions" in the Proxy Statement, all of which information is incorporated herein by reference.

Item 14. Principal Accountant Fees and Services.

Refer to the information under the caption entitled "Audit Matters — Audit and Non-Audit Fees" and "—Pre-Approval of Audit and Permissible Non-Audit Fees" in the Proxy Statement, all of which information is incorporated herein by reference.

PART IV

Item 15. Exhibits and Financial Statement Schedules.

The following documents are filed as a part of this report:

		Number
(1)	Financial Statements:	
	Consolidated Balance Sheets	40
	Consolidated Statements of Income	41
	Consolidated Statements of Comprehensive Income	42
	Consolidated Statements of Cash Flows	43
	Consolidated Statements of Changes in Equity	44
	Notes to Consolidated Financial Statements	45
	Report of Independent Registered Public Accounting Firm	87
(2)	Schedules required by Article 12 of Regulation S-X:	
	Schedule II — Valuation and Qualifying Accounts	96
	All other schedules have been omitted because they are not applicable or not required.	
(3)	Exhibits required by Item 601 of Regulation S-K are listed in the Exhibit Index, which is incorporated herein by reference. Exhibits in the Exhibit Index marked with a "C" in the left margin constitute management contracts or compensatory plans or arrangements contemplated by Item 15(b) of Form 10-K.	

SIGNATURES

Pursuant to the requirements of Section 13 or 15(d) of the Securities Exchange Act of 1934, the registrant has duly caused this report to be signed on its behalf by the undersigned, thereunto duly authorized.

BAXTER INTERNATIONAL INC.

By: /s/ José E. Almeida José E. Almeida

Chairman and Chief Executive Officer

DATE: February 23, 2017

Pursuant to the requirements of the Securities Exchange Act of 1934, this report has been signed below by the following persons on behalf of the registrant and in the capacities indicated on February 23, 2017.

Signature	Title	
/s/ José E. Almeida José E. Almeida.	Chairman and Chief Executive Officer (principal executive officer)	
/s/ James K. Saccaro James K. Saccaro	Corporate Vice President and Chief Financial Officer (principal financial officer)	
/s/ Caroline D. Karp Caroline D. Karp	Corporate Vice President and Controller (principal accounting officer)	
/s/ Thomas F. Chen Thomas F. Chen	Director	
/s/ John D. Forsyth John D. Forsyth	Director	
/s/ James R. Gavin III, M.D., Ph.D. James R. Gavin III, M.D., Ph.D.	Director	
/s/ Peter S. Hellman Peter S. Hellman	Director	
/s/ Munib Islam Munib Islam	Director	
/s/ Michael F. Mahoney Michael F. Mahoney	Director	
Stephen N. Oesterle, M.D.	Director	
/s/ Carole J. Shapazian Carole J. Shapazian	Director	
/s/ Thomas T. Stallkamp Thomas T. Stallkamp	Director	
/s/ K.J. Storm K.J. Storm	Director	
/s/ Albert P. L. Stroucken Albert P. L. Stroucken	Director	
	92	

EXHIBIT INDEX

Number and Description of Exhibit

- 2.1 Separation and Distribution Agreement (incorporated by reference to Exhibit 2.1 to the Company's Current Report on Form 8-K, filed on July 7, 2015).
- 3.1 Amended and Restated Certificate of Incorporation (incorporated by reference to Exhibit 3.1 to the Company's Current Report on Form 8-K, filed on May 10, 2013).
- 3.2 Certificate of Amendment to the Amended and Restated Certificate of Incorporation dated May 3, 2016 (incorporated by reference to Exhibit 3.1 to the Company's Current Report on Form 8-K, filed on May 4, 2016).
- 3.3 Bylaws, as amended and restated on December 18, 2015 (incorporated by reference to Exhibit 3.1 to the Company's Current Report on Form 8-K, filed on December 18, 2015).
- 4.1 Form of Common Stock Certificate of the Company (incorporated by reference to Exhibit(a) to the Company's Registration Statement on Form S-16 (Registration No. 02-65269), filed on August 17, 1979).
- 4.2 Indenture, dated August 8, 2006, between the Company and J.P. Morgan Trust Company, National Association, as Trustee (incorporated by reference to Exhibit 4.1 to the Company's Current Report on Form 8-K, filed on August 9, 2006).
- 4.3 First Supplemental Indenture, dated August 8, 2006, between the Company and J.P. Morgan Trust Company, National Association, as Trustee (including form of 5.90% Senior Note due 2016) (incorporated by reference to Exhibit 4.2 to the Company's Current Report on Form 8-K, filed on August 9, 2006).
- 4.4 Second Supplemental Indenture, dated December 7, 2007, between the Company and The Bank of New York Trust Company, N.A. (as successor in interest to J.P. Morgan Trust Company, National Association), as Trustee (including form of 6.250% Senior Note due 2037) (incorporated by reference to Exhibit 4.1 to the Company's Current Report on Form 8-K, filed on December 7, 2007).
- 4.5 Eighth Supplemental Indenture, dated August 13, 2012, between the Company and The Bank of New York Mellon Trust Company, N.A. (as successor in interest to J.P. Morgan Trust Company, National Association), as Trustee (including forms of 2.400% Senior Notes due 2022 and 3.650% Senior Notes due 2042) (incorporated by reference to Exhibit 4.1 to the Company's Current Report on Form 8-K, filed on August 13, 2012).
- 4.6 Ninth Supplemental Indenture, dated June 11, 2013, between the Company and The Bank of New York Mellon Trust Company, N.A. (as successor in interest to J.P. Morgan Trust Company, National Association), as Trustee (including forms of 0.950% Senior Notes due 2016 and 4.500% Senior Notes due 2043) (incorporated by reference to Exhibit 4.1 to the Company's Current Report on Form 8-K, filed on June 11, 2013).
- 4.7 Tenth Supplemental Indenture, dated August 13, 2016, between the Company and The Bank of New York Mellon Trust Company, N.A., as Trustee (including forms of 1.700% Senior Notes due 2021, 2.600% Senior Notes due 2026 and 3.500% Senior Notes due 2046) (incorporated by reference to Exhibit 4.2 to the Company's Current Report on Form 8-K, filed on August 15, 2016).
- 10.1 Five-Year Credit Agreement, dated as of July 1, 2015, among Baxter International Inc. as Borrower, JPMorgan Chase Bank, National Association, as Administrative Agent and certain other financial institutions named therein (incorporated by reference to Exhibit 10.4 to the Company's Current Report on Form 8-K, filed on July 7, 2015).
- Amendment No. 1 to the Five-Year Credit Agreement, dated as of October 26, 2015, among Baxter International Inc. as Borrower, JPMorgan Chase Bank, National Association, as Administrative Agent and certain other financial institutions named therein (incorporated by reference to Exhibit 10.1 to the Company's Current Report on Form 8-K, filed on October 27, 2015).
- 10.3 Credit Agreement, dated as of July 1, 2015, among Baxter Healthcare SA and Baxter World Trade SPRL, as Borrowers, J.P. Morgan Europe Limited, as Administrative Agent and certain other financial institutions named therein (incorporated by reference to Exhibit 10.5 to the Company's Current Report on Form 8-K, filed on July 7, 2015).
- Amendment No. 1 to the Credit Agreement, dated as of October 26, 2015, among Baxter Healthcare SA and Baxter World Trade SPRL, as Borrowers, J.P. Morgan Europe Limited, as Administrative Agent and certain other financial institutions named therein (incorporated by reference to Exhibit 10.2 to the Company's Current Report on Form 8-K, filed on October 27, 2015),

Number and Description of Exhibit

10.5	Employee Matters Agreement, dated as of June 30, 2015, by and between Baxter International Inc. and Baxalta Incorporated (incorporated by reference to Exhibit 10.1 to the Company's Current Report on Form 8-K, filed on July 7, 2015).
10.6	Tax Matters Agreement, dated as of June 30, 2015, by and between Baxter International Inc. and Baxalta Incorporated (incorporated by reference to Exhibit 10.2 to the Company's Current Report on Form 8-K, filed on July 7, 2015).
10.7	Shareholder's and Registration Rights Agreement, dated as of June 30, 2015, by and between Baxter International Inc. and Baxalta Incorporated (incorporated by reference to Exhibit 10.3 to the Company's Current Report on Form 8-K, filed on July 7, 2015).
10.8	Support Agreement, dated as of September 29, 2015, by and among Baxter International Inc., Third Point LLC, Third Point Partners L.P., Third Point Partners Qualified L.P., Third Point Offshore Master Fund L.P., Third Point Ultra Master Fund L.P., Third Point Reinsurance Co. Ltd., Third Point Advisors LLC, Third Point Advisors II LLC, Daniel S. Loeb and Munib Islam (incorporated by reference to Exhibit 10.1 to the Company's Current Report on Form 8-K, filed on September 30, 2015).
C 10.9	Form of Indemnification Agreement entered into with directors and officers (incorporated by reference to Exhibit 19.4 to the Company's Quarterly Report on Form 10-Q, filed on November 14, 1986).
C 10.10	Baxter International Inc. 2007 Incentive Plan (incorporated by reference to Appendix A to the Company's Definitive Proxy Statement on Schedule 14A, filed on March 20, 2007).
C 10.11	Baxter International Inc. Equity Plan for the 2007 Incentive Plan (incorporated by reference to Exhibit 10.1 to the Company's Current Report on Form 8-K, filed on March 16, 2007).
C 10.12	Baxter International Inc. 2011 Incentive Plan (incorporated by reference to Appendix B to the Company's Definitive Proxy Statement on Schedule 14A, filed on March 18, 2011).
C 10.13	Baxter International Inc. Equity Plan for the 2011 Incentive Plan (incorporated by reference to Exhibit 10.1 to the Company's Quarterly Report on Form 10-Q, filed on May 3, 2011).
C 10.14	Baxter International Inc. 2015 Incentive Plan (incorporated by reference to Appendix A to the Company's Definitive Proxy Statement on Schedule 14A, filed on March 25, 2015).
C 10.15	Baxter International Inc. Equity Plan for the 2015 Incentive Plan (incorporated by reference to Exhibit 10.6 to the Company's Current Report on Form 8-K, filed on July 7, 2015).
C 10.16	Baxter International Inc. Equity Plan for José E. Almeida under the 2015 Incentive Plan (incorporated by reference to Exhibit 10.2 to the Company's Current Report on Form 8-K, filed on October 29, 2015).
C 10.17	Baxter International Inc. Directors' Deferred Compensation Plan (amended and restated effective January 1, 2009) and Amendment No. 1 thereto effective January 1, 2012 (incorporated by reference to Exhibit 10.9 to the Company's Annual Report on Form 10-K filed on February 23, 2012).
C 10.18	Offer Letter between Baxter International Inc. and José E. Almeida, dated as of October 28, 2015 (incorporated by reference to Exhibit 10.1 to the Company's Current Report on Form 8-K, filed on October 29, 2015).
C 10.19	Form of Severance Agreement entered into with executive officers (incorporated by reference to Exhibit 10.11 to the Company's Annual Report on Form 10-K, filed on February 21, 2014).
C 10.20*	Baxter International Inc. and Subsidiaries Pension Plan (amended and restated effective January 1, 2016).
C 10.21*	First Amendment to Baxter International Inc. and Subsidiaries Pension Plan (dated as of December 21, 2016).
C 10.22*	Baxter International Inc. and Subsidiaries Supplemental Pension Plan (amended and restated effective January 1, 2015).
C 10.23*	Baxter International Inc. and Subsidiaries Deferred Compensation Plan (amended and restated effective January 1, 2015).

Number and Description of Exhibit

C 10.24*	First Amendment to the Baxter International Inc. and Subsidiaries Deferred Compensation Plan (effective May 11, 2016).				
C 10.25*	Second Amendment to the Baxter International Inc. and Subsidiaries Deferred Compensation Plan (effective December 21, 2016).				
	Baxter International Inc. Employee Stock Purchase Plan (as amended and restated effective July 1, 2011) (incorporated by reference to Appendix A to the Company's Definitive Proxy Statement on Schedule 14A, filed on March 18, 2011).				
C 10.26	Appendix A to the Company's Definitive Proxy Statement on Schedule 14A, filed on March 18, 2011).				
C 10.27*	First Amendment to Baxter International Inc. Employee Stock Purchase Plan (dated as of July 15, 2016).				
C 10.28*	Baxter International Inc. Non-Employee Director Compensation Plan (as amended and restated effective January 1, 2017).				
10.29	Letter Agreement, dated as of January 11, 2016, by and among Baxter International Inc., Baxalta Incorporated and Shire plc. (Incorporated by reference to Exhibit 10.1 to the Company's Current Report on Form 8-K, filed on January 11, 2016).				
12*	Computation of Ratio of Earnings to Fixed Charges.				
21*	Subsidiaries of Baxter International Inc.				
23*	Consent of PricewaterhouseCoopers LLP.				
31.1*	Certification of Chief Executive Officer pursuant to Rules 13a-14(a) and 15d-14(a) of the Securities Exchange Act of 1934, as amended.				
31.2*	Certification of Chief Financial Officer pursuant to Rules 13a-14(a) and 15d-14(a) and 15d-14(a) of the Securities Exchange Act of 1934, as amended.				
32.1*	Certification of Chief Executive Officer pursuant to 18 U.S.C. Section 1350, as adopted pursuant to Section 906 of the Sarbanes-Oxley Act of 2002.				
32.2*	Certification of Chief Financial Officer pursuant to 18 U.S.C. Section 1350, as adopted pursuant to Section 906 of the Sarbanes-Oxley Act of 2002.				
101.INS*	XBRL Instance Document				
101.SCH*	XBRL Taxonomy Extension Schema Document				
101.CAL*	XBRL Taxonomy Extension Calculation Linkbase Document				
101.LAB*	XBRL Taxonomy Extension Label Linkbase Document				
101.PRE*	XBRL Taxonomy Extension Presentation Linkbase Document				
101.DEF*	XBRL Taxonomy Extension Definition Linkbase Document				

^{*} Filed herewith.

C Management contract or compensatory plan or arrangement.

SCHEDULE II

	Additions							
Valuation and Qualifying Accounts (in millions)	Balance at beginning of period		Charged to costs and expenses	costs and to other		ı	Balance at end of period	
Year ended December 31, 2016:								
Allowance for doubtful accounts	\$	110	16	11	(10)	\$	127	
Deferred tax asset valuation allowance	\$	135	16	3	(4)	\$	150	
Year ended December 31, 2015:								
Allowance for doubtful accounts	\$	119	30	(10)	(29)	\$	110	
Deferred tax asset valuation allowance	\$	129	30	(16)	(8)	\$	135	
Year ended December 31, 2014:								
Allowance for doubtful accounts	\$	148	1	(15)	(15)	\$	119	
Deferred tax asset valuation allowance	\$	128	10	(6)	(3)	\$	129	

⁽¹⁾ (2) Valuation accounts of acquired or divested companies and foreign currency translation adjustments. Amounts include adjustments related to the divestiture of the BioSciences business.

Reserves are deducted from assets to which they apply.

BAXTER INTERNATIONAL INC. AND SUBSIDIARIES PENSION PLAN

(Amended and Restated Effective as of January 1, 2016)

ARTICLE I. INTRODUCTION2

- 1.1. The Plan2
- 1.2. Plan Objectives2
- 1.3. Benefits of Participants Terminating Employment prior to Effective Date2
- 1.4. Supplements2
- 1.5. Baxalta Incorporated Spin-Off2

ARTICLE II. DEFINITIONS4

- 2.1. "Accrued Benefit'4
- 2.2. "Actuarial Equivalent"6
- 2.3. "Administrative Committee"8
- 2.4. "American"8
- 2.5. "Average Monthly Compensation"8
- 2.6. "Beneficiary"11
- 2.7. "Benefit Reduction Factors"11
- 2.8. "Board of Directors"12
- 2.9. "Code"12
- 2.10. "Company"12
- 2.11. "Compensation"12
- 2.12. "Computation Period"16
- 2.13. "Death Benefit"16
- 2.14. "Disability"16
- 2.15. "Early Retirement Date"17
- 2.16. "Effective Date"17
- 2.17. "Eligible Employee"17
- 2.18. "Employee"18
- 2.19. "Employer"18
- 2.20. "Employment Date"18
- 2.21. "Entry Date"18
- 2.22. "ERISA"19
- 2.23. "Excluded Division"19
- 2.24. "Highly Compensated Employee"19
- 2.25. "Hour of Service"19
- 2.26. "Investment Committee"21
- 2.27. "Investment Manager"21
- 2.28. "Joint and 50 Percent Survivor Annuity"21
- 2.29. "Lump Sum"21
- 2.30. "Non-forfeitable"22
- 2.31. "Normal Retirement Date"22
- 2.32. "One-Year Break in Service"22
- 2.33. "Participant"22
- 2.34. "Participating Employer"; "Non-Participating Employer"22
- 2.35. "Participating Employer Contributions"22
- 2.36. "Payment Date"22
- 2.37. "Pension Benefit"22
- 2.38. "Plan"23

- i -

	TABLE	OF CONTENTS PAGE				
	2.39.	"Plan Year"23				
	2.40.	"Points"23				
	2.41.	"Primary Social Security Benefit"23				
	2.42.	"Projected Benefit Service"25				
	2.43.	"Single Life Annuity"25				
	2.44.	"Spouse"25				
	2.45.	"Termination of Employment"25				
	2.46.	"Trust"26				
	2.47.	"Trust Agreement"26				
	2.48.	"Trust Fund"26				
	2.49.	"Trustee"26				
	2.50.	"Year of Service"26				
ARTICL	E III. PAR	TICIPATION30				
·····	3.1.	Becoming a Participant30				
	3.2.	Ceasing to Be a Participant31				
	3.3.	Reemployment31				
	3.4.	Change of Job Status32				
	3.5.	Transfers32				
	3.6.	International Employees32				
	3.7.	Reemployment of Veterans32				
ADTICI		GIBILITY FOR AND AMOUNT OF PENSION BENEFITS33				
AKTICL	4.1.	Normal Retirement33				
	4.1.	Late Retirement33				
	4.2.	Late Retirements				
	4.3.	Early Retirement33				
	4.4.	Disability Retirement34				
	4.5.	Deferred Vested Benefit35				
ARTICLE V. SPECIAL VESTING PROVISIONS FOR DIVESTED EMPLOYEES37						
	5.1.	Accelerated Vesting37				
	5.2.	Re-employment37				
ARTICL		NEFITS AFTER DEATH38				
	6.1.	Death after Payment Date38				
	6.2.	Death before Payment Date38				
	6.3.	Death Benefits Payable to Spouses38				
	6.4.	Survivor Benefits Payable to Non-Spouse Beneficiaries39				
	6.5.	Death Benefits for Disability Retirees40				
	6.6.	Pre-Retirement Death Benefits Payable in a Lump Sum40				
	6.7.	Designation of Beneficiary40				
	6.8.	Incapacitated Participants or Beneficiaries41				
	6.9.	Death During Military Service41				
	- ii -					

ARTICLE VII. FORM AND PAYMENT OF PENSION BENEFITS AND DEATH BENEFITS42

- 7.1. Normal Form of Payment42
- 7.2. Optional Forms of Payment42
- 7.3. Lump Sum Cash-Out43
- 7.4. Rules as to Election and Discontinuance of Optional Forms of Payment44
- 7.5. Special Payment Limitations47
- 7.6. Effect of Prior Lump Sums on Pension Benefits48
- 7.7. Effect of Participant Resuming Employment after Benefits Commence 48
- 7.8. Special One Time Election for Immediate Payment of Certain Deferred Vested Benefits 50

ARTICLE VIII. PLAN COMMITTEES53

- 8.1. Membership of Administrative and Investment Committees53
- 8.2. Administrative Committee Powers and Duties53
- 8.3. Investment Committee Powers and Duties 55
- 8.4. Conflicts of Interest56
- 8.5. Compensation; Reimbursement56
- 8.6. Standard of Care 56
- 8.7. Action by Committees 57
- 8.8. Resignation or Removal of Committee Member 57
- 8.9. Uniform Application of Rules by Administrative Committee 57
- 8.10. Claims Procedure 57
- 8.11. Correction of Errors 58

ARTICLE IX. TRUST, THE TRUSTEE AND PLAN FINANCING60

- 9.1. Trust Agreement60
- 9.2. Selection of Trustee60
- 9.3. Trustee's Duties60
- 9.4. Trust Income60
- 9.5. Expenses 60
- 9.6. Trust Entity60
- 9.7. Funding Policy60
- 9.8. Participating Employer Contributions60
- 9.9. Forfeitures61
- 9.10. Exclusive Benefit of Participants61
- 9.11. Benefits Payable Only from Trust Fund62

ARTICLE X. ADOPTION AND WITHDRAWAL FROM PLAN63

- 10.1. Procedure for Adoption63
- 10.2. Procedure for Withdrawal63

ARTICLE XI. AMENDMENT AND TERMINATION64

- 11.1. Amendments64
- 11.2. Termination66
- 11.3. Disposition of Fund on Termination66
- 11.4. Disposition Medium67

- iii -

ARTICLE XII. SPECIAL TOP-HEAVY RULES68

- 12.1. Application68
- 12.2. Special Terms68
- 12.3. Vested Percentage71
- 12.4. Minimum Benefit72
- 12.5. Maximum Benefit Accrual73
- 12.6. Termination of Top-Heavy Status73

ARTICLE XIII. MISCELLANEOUS PROVISIONS74

- 13.1. Company Merger74
- 13.2. Plan Merger74
- 13.3. Nonalienation of Benefits74
- 13.4. Qualified Domestic Relations Orders74
- 13.5. No Employment Guarantee 75
- 13.6. Termination of Employment75
- 13.7. Limitation on Vesting75
- 13.8. No Duplication of Benefits75
- 13.9. Source of Benefits75
- 13.10. Reduction for Overpayment75
- 13.11. Limitations on Pension Benefits Payable to Highly Compensated Participants75
- 13.12. Maximum Pensions76
- 13.13. Funding-Based Limitations on Benefits76
- 13.14. Indemnity 79
- 13.15. Gender and Number80
- 13.16. Severability80
- 13.17. Headings80
- 13.18. Uniform and Nondiscriminatory Treatment80
- 13.19. Applicable Law80
- 13.20. Action by the Participating Employer80
- 13.21. Participant Litigation 80

SUPPLEMENT A Early Payment Factors % of Age 65 Accrued Benefit Payable for Commencement from Ages 44-65XX*

SUPPLEMENT B Special Retirement ProgramXX*

SUPPLEMENT C Excluded DivisionsXX*

SUPPLEMENT D Provisions Applicable to Participants Transferred to AllegianceXX*

SUPPLEMENT E Entitlement Codes for Non-Resident Aliens with U.S.-Source Income Excluded from Participation in the PlanXX*

SUPPLEMENT F Regression FactorsXX*

SUPPLEMENT G Provisions Applicable to Employees of Sybron CorporationXX*

SUPPLEMENT H Average Monthly Compensation for Participants Incurring Termination of Employment Prior to January 1, 1998XX*

SUPPLEMENT I Pre-retirement Death Benefits for Terminations Prior to January 1, 1990XX*

SUPPLEMENT J Transfer of Assets from BOC Group Cash Balance Retirement PlanXX*

SUPPLEMENT K Transfer of Assets from the Clintec Pension PlanXX*

SUPPLEMENT L Merger of the Immuno-U.S., Inc. Defined Benefit PlanXX*

SUPPLEMENT M Special One Time Election for Immediate Payment of Certain Deferred Vested BenefitsXX*

*Supplement has been redacted

BAXTER INTERNATIONAL INC. AND SUBSIDIARIES PENSION PLAN

(Amended and Restated Effective January 1, 2016)

ARTICLE I. Introduction

- 1.1. The Plan. Effective September 1, 1960, American Hospital Supply Corporation established the American Hospital Supply Corporation Retirement Plan, which was amended and restated as of January 1, 1985. American Hospital Supply Corporation merged with Baxter International Inc. (then known as "Baxter Travenol Laboratories, Inc."), and, effective as of January 1, 1986, the plan name was changed to Baxter Travenol Laboratories, Inc. and Subsidiaries Pension Plan. The Plan was amended to provide for participation by Eligible Employees of Baxter International Inc. and its participating units. The Plan has been further amended from time to time including an amendment, effective June 9, 1988, which changed the name of the Plan to Baxter International Inc. and Subsidiaries Pension Plan (the "Plan"). The Plan was amended and restated effective January 1, 1992, again amended and restated, on September 30, 1996, January 1, 2007, January 1, 2010, January 1, 2015 and most recently restated effective January 1, 2016. This document incorporates all previous amendments adopted since the most recent restatement, and constitutes an amendment, restatement and continuation of the Plan as in effect immediately prior to the Effective Date, subject to any transition dates and other effective dates set forth in this document.
- **1.2. Plan Objectives**. The Plan is a defined benefit pension plan maintained by Baxter International Inc. to assist in providing Participants with retirement benefits.
- 1.3. Benefits of Participants Terminating Employment prior to Effective Date. The benefits payable with respect to Participants whose benefits were in pay status prior to the Effective Date and who do not again become Eligible Employees will in all respects continue to be governed by the terms and conditions of the Plan as in effect at the time such benefits commenced. Except to the extent expressly provided to the contrary in an applicable Supplement or elsewhere in the Plan, the Accrued Benefits, eligibility, vesting and other rights of other Participants incurring a Termination of Employment prior to the Effective Date will continue to be governed by the terms and conditions of the Plan as in effect at the time of such Termination of Employment.
- **1.4. Supplements.** Supplements to the Plan may be adopted, attached to and incorporated in the Plan at any time. The provisions of any such Supplements will have the same effect that such provisions would have if they were included within the basic text of the Plan. Supplements will specify the persons affected and will supersede the other provisions of the Plan to the extent necessary to eliminate inconsistencies between the Plan provisions and the provisions of such Supplements.
- **1.5. Baxalta Incorporated Spin-Off.** In connection with the Company's spin-off of its biopharmaceutical business, the Company underwent an internal reorganization and

incorporated Baxalta Incorporated ("Baxalta") as a subsidiary of the Company, and Baxalta was spun-off on July 1, 2015 (the "Spin-Off Date") pursuant to a Separation and Distribution Agreement (the "Agreement"), and thereupon became an independent, publicly traded corporation which owns and operates the biopharmaceutical business previously owned and operated by the Company.

As described in the Agreement and the Employee Matters Agreement between the Company and Baxalta, Baxalta established, effective as of May 1, 2015, the Baxalta Incorporated Pension Plan (the "Baxalta Plan"), a qualified defined benefit plan that generally mirrors the Plan, to provide for the benefits of the following Employees who were Participants immediately prior to the date such Employee transfers to employment with Baxalta from the Company:

- (a) An Employee (including an Employee who is on an approved leave of absence from the Company) who transfers to employment with Baxalta from the Company on or before the Spin-Off Date; or
- (b) An Employee who transfers to employment with Baxalta from the Company in accordance with the Transition Services Agreement.

As described in the Agreement and the Employee Matters Agreement, the assets and liabilities in the Plan allocable to an Employee who transferred to employment with Baxalta as described in this Section (a "Baxalta Participant"), were transferred to the Baxalta Plan as of May 1, 2015, in accordance with Code Section 414(l) (and the regulations promulgated thereunder) and ERISA Section 4044, as applicable. As a result of such, a Baxalta Participant shall cease to have any benefit under the Plan following his transfer of employment to Baxalta, and shall instead participate under the Baxalta Plan; provided, however, that the benefit of any Baxalta Participant who terminated employment between May 1, 2015, and the Spin-Off Date were transferred from the Baxalta Plan back to the Plan. For the avoidance of doubt, an Employee who transfers to employment with Baxalta as described in this Section and is subsequently rehired by the Company is not eligible to participate in the Plan.

ARTICLE II. Definitions

The following terms, whenever used in the capitalized form, will have the meanings set forth below unless the context clearly indicates otherwise:

- **2.1.** "Accrued Benefit" means, subject to (c) below and the limitations of Sections 13.11 and 13.12, a monthly amount payable to or for the benefit of a Participant commencing on the Payment Date for his Normal Retirement Pension Benefit under Section 4.1 (or if later, the Payment Date for his Late Retirement Pension Benefit described in Section 4.2), equal to the greater of (a) or (b):
 - (a) General Formula. A Single Life Annuity in an amount equal to (i) minus (ii) multiplied by (iii):
 - (i) an amount equal to one and three-quarters percent (1-3/4 percent) of the Participant's Average Monthly Compensation multiplied by his Projected Benefit Service; *minus*
 - (ii) an amount equal to one and three-quarters percent (1-3/4 percent) of the Participant's Primary Social Security Benefit multiplied by his Projected Benefit Service. In no event will the amount calculated under this paragraph (ii) exceed 60 percent of the Participant's Primary Social Security Benefit; *multiplied by*
 - (iii) an amount equal to a fraction, the numerator of which is the Participant's actual Years of Service and the denominator of which is the Participant's Projected Benefit Service.

For Participants who terminate employment on or after December 31, 1997 with a Non-forfeitable Accrued Benefit, the benefit provided under this subsection will be increased as necessary so that the monthly payment is not less than \$100.

- (b) Alternate Formulae. The Accrued Benefit of a Participant will be the greater of the Accrued Benefit described in (a) above or, if applicable, the greatest Accrued Benefit payable as a Single Life Annuity for such Participant which is determined as follows:
 - (i) American Merger Benefit. Subject to the conditions described in the Plan as in effect on November 25, 1985 (the "Merger Date"), each Participant who was a Participant on the Merger Date is entitled to a fully vested benefit (referred to as the "American Merger Benefit") which is payable in lieu of any other benefits under the Plan if such American Merger Benefit is larger than the benefits otherwise payable to or for the benefit of such Participant hereunder. Such American Merger Benefit will be determined under the terms of the Plan as in effect on the Merger Date

and will be an amount determined as of the Merger Date equal to (A) minus (B) multiplied by (C):

- (A) an amount equal to four and one-quarter percent (41/4 percent) of the Participant's Average Monthly Compensation as of the Merger Date (under the terms of the Plan as then in effect) multiplied by his Projected Benefit Service as of the Merger Date; *minus*
- (B) an amount equal to one and three-quarters percent (1¾ percent) of the Participant's Primary Social Security Benefit as of the Merger Date (under the terms of the Plan as then in effect) multiplied by his Projected Benefit Service as of such date (under the terms of the Plan as then in effect). In no event will the amount calculated under this subparagraph (B) exceed 60 percent of the Participant's Primary Social Security Benefit; *multiplied by*
- (C) an amount equal to a fraction, the numerator of which is the Participant's actual Years of Service as of the Merger Date (under the terms of the Plan as then in effect) and the denominator of which is the Participant's Projected Benefit Service determined as of the Merger Date (under the terms of the Plan as then in effect).
- (ii) Alternative Benefit under Applicable Supplement. The Accrued Benefit of certain groups of Participants which are frozen or otherwise limited or modified may be preserved as a minimum benefit in accordance with an applicable Supplement.
- (iii) **No Diminishment of Benefit Amount after Early Retirement.** In no event will the Accrued Benefit of a Participant eligible for a normal retirement benefit be less than the largest early retirement benefit the Participant could have received if he had retired on his Early Retirement Date.
- (iv) **Top-Heavy Minimum Benefit.** If Article XII is applicable to the Plan, the minimum benefit requirements of Section 12.4 may entitle certain Participants to minimum benefits under the Plan.
- (v) **Pre-1989 Accrued Benefit.** Each Participant who was an Eligible Employee on December 31, 1988, under the terms of the Plan as in effect on such date is entitled to a "1988 Minimum Benefit" which is payable in lieu of any other benefits under the Plan if such 1988 Minimum Benefit is larger than the benefits otherwise payable to or for the benefit of such Participant hereunder. Such 1988 Minimum Benefit will be an amount determined in accordance with the provisions of Section 2.1(a), except that such determination will not be subject to Section 2.11(e), and such determination will be made as of December 31, 1988, on the basis of the Participant's Average Monthly Compensation, Projected Benefit Service,

Primary Social Security Benefit and Actual Years of Service of such Participant as of such date as determined under the terms of the Plan as then in effect.

- (vi) **Pre-1990 Accrued Benefit.** Each Participant who was an Eligible Employee on December 31, 1989 under the terms of the Plan as in effect on such date is entitled to a "1989 Minimum Benefit," which is payable in lieu of any other benefits under the Plan if such 1989 Minimum Benefit is larger than the benefits otherwise payable to or for the benefit of such Participant hereunder. Such 1989 Minimum Benefit will be an amount determined as of December 31, 1989 in accordance with Section 2.1(a), except that for purposes of this subsection (b)(vi) the Average Monthly Compensation, the Projected Benefit Service, the Primary Social Security Benefit and the actual Years of Service of a Participant will be determined as of December 31, 1989 based upon the terms of the Plan as then in effect.
- (c) Nondiscrimination Limitations. The benefits accrued on behalf of any Participant who is a Highly Compensated Employee will be limited to no more than the maximum amount that may be so accrued on behalf of such Highly Compensated Employee under the Code Section 401(a)(4) nondiscrimination compliance testing method then in effect.

The benefit under any of Section 2.1(b)(i)-(vi) will be determined with reference to all of the provisions of the previous versions of this Plan referred to in such Section, including minimum benefit provisions similar to the other minimum benefit provisions of Section 2.1(b)(i)-(vi).

- **2.2.** "Actuarial Equivalent" means, for purposes of computing a Lump Sum or the payments under the forms of benefit described in Article VII and for purposes of computing any adjustments applicable under the Plan to reflect the commencement of any form of benefit payments on or after the Effective Date and at a time other than a Participant's Normal Retirement Date, the Pension Benefit having the same value as the Accrued Benefit based upon the applicable interest rates and mortality assumptions described below.
 - (a) The following interest rates and mortality assumptions will apply with respect to a Participant who incurs a Termination of Employment:
 - (i) Ten percent (10%) per annum compounded annually and applied as of the date of determination with respect to any computation of an optional form of benefit other than a Lump Sum;
 - (ii) For a Lump Sum, the Pension Benefit Guaranty Corporation ("PBGC") immediate or deferred annuity interest rate (whichever is applicable) in effect at the beginning of the Plan Year in which the Participant's benefit is paid hereunder, unless otherwise provided for in an applicable Supplement. Such PBGC rate will apply to Participants who incur a Termination of Employment prior to January 1, 1996. However, if the

Participant's Termination of Employment occurred during the 1995 Plan Year, such rate will be the lesser of the PBGC rate in effect on the first day of the 1995 Plan Year or January 1, 1996 if the distribution was not made prior to January 1, 1996. Effective for Terminations of Employment occurring after December 31, 1995, the applicable rate of interest based on the annual 30-year Treasury securities rate for October of the year prior to the Plan Year in which the Participant's benefit hereunder is paid (except as otherwise provided in the last paragraph of this Section 2.2(a)(ii)) will be used to calculate Lump Sums; provided, however, that for distributions made in 1996, the applicable interest rate will not be greater than the annual 30-year Treasury securities rate for November of 1995.

Beginning in February 2002, in light of the decision by the United States Department of Treasury to suspend issuance of 30-year bonds, and to therefore cease publication of the annual rate of interest on 30-year Treasury securities, the "annual rate of interest on 30-year Treasury securities" for a particular month will be based on such written guidance as may be issued from time to time by the Internal Revenue Service describing the appropriate 30-year Treasury interest rate to use for purposes of Code section 417(e).

For each Plan Year commencing with 2008, the interest rate shall be a blended rate, equal to the sum of the interest rate on 30-year Treasury securities, as determined in the preceding paragraphs, multiplied by the applicable percentage from the following table, plus the applicable segment rate, based upon the number of years until the Participant's Normal Retirement Date, as determined under Code section 417(e)(3) as amended by the Pension Protection Act of 2006 for October of the year preceding the commencement of the Plan Year, multiplied by the applicable percentage from the following table:

Plan Year	Percentage for 30-year Treasury security rate	Percentage for applicable segment rate
2008	80	20
2009	60	40
2010	40	60
2011	20	80
2012 and thereafter	0	100

Effective December 1, 2012, the applicable rate of interest shall be based on the rate in effect for August of the year prior to the Plan Year in which the Participant's benefit hereunder is paid; provided, however, that in the case of a Lump Sum paid prior to January 1, 2014, other than a Lump Sum paid pursuant to Supplement M, the applicable rate of interest shall be the

rate for either October or August of the year prior to the Plan Year in which the benefit is paid, whichever produces the larger Lump Sum.

- (b) The mortality assumption is determined in accordance with the 1971 Group Annuity Mortality Table, weighted 75 percent male and 25 percent female. Effective for Terminations of Employment occurring after December 31, 1995, the mortality assumption for purposes of the calculation of a Lump Sum payment will be the applicable mortality table prescribed by the Secretary of the Treasury pursuant to Code Section 417(e)(3)(A)(ii)(I). For purposes of the foregoing, the "applicable mortality table" for distributions with annuity start dates beginning on or after December 31, 2002, is the mortality table prescribed by the Secretary of the Treasury as set forth in Revenue Ruling 2001-62. The "applicable mortality table" for distributions with annuity start dates beginning during 2008 is the mortality table set forth in Revenue Ruling 2007-67, and for distributions with annuity start dates beginning in Plan Years after 2008 the applicable mortality table will be the mortality table specified for the Plan Year by the Secretary of the Treasury pursuant to Code section 417(e)(3) as amended by the Pension Protection Act of 2006, each which applicable mortality tables is incorporated herein by this reference.
- **2.3.** "Administrative Committee" means the committee which is responsible for administering the Plan in accordance with Article VIII.
- **2.4.** "American" means American Hospital Supply Corporation, the sponsor of the AHSC Plan prior to the Merger Date.
- **2.5.** "Average Monthly Compensation" means, effective for Participants whose Termination of Employment occurs after December 31, 1997, the amount determined under this Section. For Participants whose last Termination of Employment occurred prior to January 1, 1998, the rules in Supplement H will apply, along with the provisions of subsection (f) below. Prior to April 1, 2004, Average Monthly Compensation shall be determined in accordance with this Section 2.5 as in effect prior to this restatement (including all interpretations and administrative practices thereunder); provided that in no event shall any amendment to this Section 2.5 reduce any Participant's Accrued Benefit as of March 31, 2004).

Average Monthly Compensation is the Participant's Compensation paid while an Eligible Employee during the five consecutive Plan Years during the Participant's compensation history which produces the largest amount, divided by 60. A Participant's compensation history includes the ten consecutive Plan Years preceding his most recent Termination of Employment. However, if a Participant's Termination Date is on or after December 7 of a Plan Year, such Plan Year will be included in the Participant's ten-year compensation history. If a Participant has fewer than five consecutive Plan Years but at least one Plan Year in his Compensation history, the Compensation from those years will be divided by the number of months in those Plan Years.

A Participant's compensation history and Average Monthly Compensation will be determined without regard to the following "drop-out years" and the Compensation earned in such years:

- (i) the Participant's first partial Plan Year of employment as an Eligible Employee,
- (ii) any Plan Year in which the Participant was an Eligible Employee and was credited with less than 1,000 Hours of Service,
- (iii) any Plan Year in which the Participant performed no services for an Employer, and
- (iv) any Plan Year in which the Participant was employed at any time during such Plan Year by a Non-Participating Employer other than an Employer that has adopted the Baxter Healthcare Corporation of Puerto Rico Pension Plan as described in subsection (d), below, unless the Participant transferred from a Participating Employer to any such Non-Participating Employer on or after December 7 of the Plan Year, in which case the Plan Year will be included in determining Average Monthly Compensation, subject to clause (ii), above; provided, that this exclusion will only apply to Plan Years beginning with 2007 if the Non-Participating Employer to which the Participant is transferred is not a United States domestic Employer.

Except as provided in Section 2.11(a)(i)(A)(20), and subject to other provisions of the Plan, Compensation for any Plan Year will not include Compensation paid more than 30 days after the Participant's Termination of Employment.

Compensation will not be disregarded solely because it was used to calculate an Accrued Benefit that was previously paid in a lump sum.

The following special rules will apply in determining Average Monthly Compensation:

- (a) Participants with Less Than One Year of Compensation. If, after the drop-out years are disregarded, a Participant has less than one Plan Year of Compensation while an Eligible Employee, Average Monthly Compensation means the Participant's total Compensation as an Eligible Employee annualized based on the period over which such Compensation was earned and divided by twelve.
- (b) Disabled Participants. The Average Monthly Compensation of a Participant entitled to disability retirement benefits will be subject to the modification described in Section 4.4(a).
- (c) Imputed Compensation. Participants transferred to Allegiance Corporation on or about September 30, 1996 who consequently did not receive a full year of Compensation during the 1996 Plan Year will have their Average Monthly Compensation adjusted as set forth in Supplement D.
- (d) Transfers to/from Puerto Rico. If a Participant transfers from a Participating Employer to a Non-Participating Employer that has adopted the Baxter Healthcare Corporation of Puerto Rico Pension Plan, or transfers from such Non-

Participating Employer doing business in the Commonwealth of Puerto Rico to a Participating Employer, his Average Monthly Compensation will take into account both his Compensation and the Plan Years while employed by such Non-Participating Employer doing business in the Commonwealth of Puerto Rico and the Participating Employer as if he were at all times an Eligible Employee.

(e) Rehired Participants.

- (i) A Participant's Compensation in the year he is rehired will be disregarded for purposes of determining his Average Monthly Compensation, except as otherwise provided in clause (ii).
- (ii) If the Participant was rehired in the same year that he incurred a Termination of Employment, such Plan Year and the Compensation earned during such Plan Year may be included in determining Average Monthly Compensation, subject to the drop-out rules, above, but if such Plan Year is included the calculation of Average Monthly Compensation shall be adjusted to reflect the fraction of the Plan Year of rehire during which the Compensation was earned.
- (iii) A rehired Participant's Compensation earned as an Eligible Employee prior to his Termination of Employment shall be included (subject to the drop-out rules) regardless of whether he incurred a One-Year Break in Service prior to being rehired and regardless of whether he received a Lump Sum payment upon Termination of Employment. The Administrative Committee may establish procedures for estimating a Participant's Compensation during a prior period of employment if the actual amount cannot be ascertained.
- (iv) The provisions of this paragraph (e) will not apply to a Participant who is rehired after December 31, 2006.
- (f) Transition Rules. The rules in this Section will apply with respect to Participants whose Termination of Employment occurs after December 31, 1997. If a Participant commenced a new compensation history under the Plan prior to January 1, 1998 under the prior rules for determining Average Monthly Compensation because of non-continuous service (a "service break"), and the Compensation for the Plan Year following the Plan Year in which the service break ends was more than the Compensation for the Plan Year prior to the Plan Year in which the service break began, then the rules in this Section will apply only with respect to the compensation history following such service break. If the Compensation in the Plan Year following the Plan Year in which the service break ends is less than the Compensation in the Plan Year prior to the Plan Year in which the service break began, then the rules in this Section will apply to all Plan Years prior to such service break and to all Plan Years following such service break. However, the foregoing sentence will not apply if subsequent

- increases in Compensation result in a reduction in a Participant's Accrued Benefit in violation of Code Section 411(d)(6).
- (g) Rehires after Divestitures. In any case in which a Participant ceases to participate in the Plan as the result of a sale or other divestiture of the business that employed the Participant, and the Participant's Accrued Benefit and the assets attributable to such Accrued Benefit are transferred to another plan in connection with the transaction, if such Participant is subsequently rehired by an Employer the Compensation paid to such Employee prior to such transaction shall not be included in Average Monthly Compensation unless the Participant is rehired prior to January 1, 2007, and the Accrued Benefit and attributable assets are transferred back to the Plan.
- **2.6.** "Beneficiary" means the persons or trusts validly designated by a Participant or the Plan in accordance with Section 6.7 to receive any benefits payable on behalf of such Participant after his death.
- **2.7. "Benefit Reduction Factors"** means the factors utilized to adjust the amount of a Participant's benefits to reflect that such benefits are paid earlier than the Participant's Normal Retirement Date. The Benefit Reduction Factors utilized for purposes of the Plan will be determined as follows:
 - (a) Participants with 85 Points. Except as provided in (d) below, the benefits of a Participant will not be reduced to reflect early payment if the total number of the Participant's Points as of his Termination of Employment equals or exceeds 85 Points.
 - (b) Participants with At Least 65 but Fewer Than 85 Points and Disabled Participants. Except as provided in (d) below, the benefits of a Participant whose total number of Points as of his Termination of Employment equals 65 or more Points but less than 85 Points, or who qualifies for Disability Retirement under Section 4.4, will be calculated in accordance with Table 1 of Supplement A as of the date his payments commence.
 - (c) Reduction Factors Applicable to Participants with Fewer Than 65 Points. Except as provided in (d) below, a Participant who incurs a Termination of Employment prior to attaining 65 Points and elects to cause his Benefits to commence at or after attainment of 65 Points will be subject to the Benefit Reduction Factors described in Table 4 of Supplement A.
 - (d) Reduction Factors Applicable to Pre-1990 Accrued Benefits. In the case of a Participant whose Accrued Benefit is determined on the basis of his American Merger Benefit in accordance with Section 2.1(b)(i), the Benefit Reduction Factors are set forth in Table 3 of Supplement A. In the case of a Participant whose Accrued Benefit is determined on the basis of his 1988 Accrued Benefit in accordance with Section 2.1(b)(v) or on the basis of his 1989 Accrued Benefit in accordance with Section 2.1(b)(vi), the Benefit Reduction Factors are set forth in

Table 2 of Supplement A. However, if any Participant is age 55 upon Termination of Employment and his Accrued Benefit is based on the 1988 Accrued Benefit or the 1989 Accrued Benefit, the reduction factor will be .25% per month for each month that the Payment Date precedes the Participant's Normal Retirement Date. A Participant entitled to the American Merger Benefit is entitled to the .25% factor only if he attained age 55 as of November 25, 1985.

In no case will the benefit calculated above be less than the benefit to which a participant who did not incur a Termination of Employment prior to January 1, 1985 would be entitled on December 31, 1984 using the Actuarial Equivalent assumptions then in effect.

- **2.8. "Board of Directors"** means the Board of Directors of the Company.
- **2.9. "Code"** means the Internal Revenue Code of 1986, as amended.
- **2.10.** "Company" means Baxter International Inc.
- **2.11.** "Compensation" means the amount determined with respect to a Participant in accordance with the following alternative definitions:
 - (a) Compensation. Except as required by (b), (c) or (d) below, for each Eligible Employee, "Compensation" means the amounts paid by the Participating Employers during the Plan Year to such Eligible Employee for services as an Employee which is included in such Compensation under the rules set forth in Section 2.11(a)(i) below, other than such Compensation which is excluded under the rules set forth in Section 2.11(a)(ii) below. Compensation also includes amounts paid by a Non-Participating Employer that has adopted the Baxter Healthcare Corporation of Puerto Rico Pension Plan while an Eligible Employee is on the payroll of such Non-Participating Employer.
 - (i) Included Pay. For purposes of this subsection 2.11(a), an Eligible Employee's Compensation will include the items described in (A) and (B), below:
 - (A) The portion of such earnings of an Employee which are required to be reported as taxable income on Form W-2 (or which would be required to be so reported but for the fact that such Compensation is paid by an Employer in the Commonwealth of Puerto Rico), including:
 - bonuses paid pursuant to the Management Incentive Compensation Plan bonuses
 or any annual bonus plan adopted in replacement thereof; payments in lieu of
 salary increases; bonuses paid to sales representatives if included in the
 compensation plan; and other bonuses under bonus plans specifically designated
 by the Administrative

Committee as constituting Compensation hereunder, other than bonuses described in Section 2.11(a)(ii)(C)(7);

- 2. call in pay;
- 3. commission pay;
- 4. double time pay;
- 5. draws toward commissions;
- 6. funeral pay;
- 7. holiday pay;
- 8. jury duty pay;
- 9. lead pay;
- 10. mileage pay for long haul truckers;
- 11. military pay;
- 12. on-call (beeper) pay
- 13. overtime pay;
- 14. paid absences;
- 15. retroactive pay;
- 16. salary or other regular pay;
- 17. shift differentials;
- 18. sick pay or other short-term disability pay;
- 19. straight time pay;
- 20. severance pay as follows: (a) for an Eligible Employee whose Termination of Employment occurred prior to January 1, 1990, an amount equal to the actual amount of severance paid, not to exceed six months of severance pay; and (b) for Participants whose Termination of Employment occurred after December 31, 1989 and before May 1, 1996, six months of severance pay;
- 21. vacation pay.

- (B) the amount of any salary reduction or cash or deferred contributions made by such Eligible Employee under any plan maintained by the Participating Employers which satisfies the requirements of Code Section 125 (other than the amounts described in Sections 2.11(a)(ii) (C)(11) and (12) below), Code Section 401(k), or Code Section 132(f).
- (ii) **Excluded Pay.** For purposes of this Section 2.11(a), an Eligible Employee's Compensation will exclude:
 - (A) Amounts required to be reported on such form as imputed income arising from the Participating Employer's moving expense reimbursement policies, the Participating Employer's life insurance plans or the Participating Employer's other fringe benefit plans;
 - (B) Amounts paid to replace benefits not provided under any qualified plan due to the contribution or benefit limitations or non-discrimination restrictions; and
 - (C) The following amounts paid, accrued or imputed:
 - 1. attendance awards;
 - 2. automobile allowances;
 - 3. business expense reimbursements;
 - 4. cash prizes or awards;
 - 5. gifts;
 - 6. contest pay;
 - 7. deferred compensation, including deferred bonuses;
 - 8. discretionary awards;
 - 9. employee referral awards;
 - 10. executive perquisite allowances;
 - 11. flex credits;
 - 12. flex cash;
 - 13. hiring bonuses;
 - 14. income from sale of stock;

- 15. income from the exercise of stock options;
- 16. interest earnings on deferred compensation, including deferred bonuses;
- 17. invention fees and awards;
- 18. long term disability pay;
- 19. mortgage differential payments;
- 20. noncash prizes or awards;
- 21. pay for unused sick time;
- 22. performance shares;
- 23. promotional awards;
- 24. relocation expense reimbursements;
- 25. restricted stock rights;
- 26. retention bonuses;
- 27. severance pay, effective for Terminations of Employment occurring after April 30, 1996;
- 28. stock appreciation rights;
- 29. tax equalization payments to expatriates;
- 30. technical achievement awards;
- 31. travel allowances;
- 32. tuition reimbursements; and
- 33. workers' compensation benefits.
- (b) Compensation of Commissioned Sales Representatives. Except as provided in (c) and (d) below, the definition of Compensation set forth in Section 2.11(a) will apply with respect to an Eligible Employee who is a commissioned sales representative receiving Compensation without reimbursement for expenses under Pay Plan D, except that only eighty-five percent (85%) of the amounts included in Compensation will be recognized.
- (c) "Compensation" for Certain Purposes. For purposes of Section 13.12, and for purposes of determining whether an Employee is a Highly Compensated

Employee, "Compensation" means the compensation paid by an Employer during the Plan Year to an Employee for personal services rendered and which is reportable as taxable income on IRS Form W-2, plus the amount of any salary reduction or cash or deferred contributions made by such Eligible Employee under any plan maintained by the Participating Employers which satisfies the requirements of Code Section 125, Code Section 401(k), or Code Section 132(f). Anything else contained herein to the contrary notwithstanding, effective January 1, 2008, "Compensation" for such purposes shall not include any amount paid to an Employee after the Employee's termination of employment unless (i) such amount is paid by the later of the end of the year in which employment is terminated or two and one half months after the date of termination, (ii) such amount would otherwise have been included in the applicable definition of Compensation, and (iii) such amount constitutes either salary, wages (including overtime, shift differentials and similar amounts), commissions or bonuses that would have been paid prior to termination of employment if the Employee's employment had not terminated, payment for unused sick, vacation or other leave that the Employee would have been able to use if employment had continued, or payment of nonqualified deferred compensation that would have been paid at the same time had employment not terminated.

- (d) Maximum Amount of "Compensation." The annual Compensation for each Eligible Employee taken into account under the Plan in any Plan Year shall not exceed \$265,000, as adjusted for such Plan Year for cost-of-living as provided in Code Section 401(a)(17). If Compensation for any prior Plan Year is taken into account in determining an Employee's benefits accruing in the current Plan Year, the Compensation for that prior Plan Year will be subject to the compensation limit in effect for that prior Plan Year; provided that the compensation limit for Plan Years prior to 2003 shall be \$200,000.
- (e) Differential Military Pay. Any amount paid to a Participant while on active duty for a period of more than 30 days in the uniformed services of the United States, which represents all or a portion of the Compensation the Participant would have received if not on active duty, shall constitute Compensation for all purposes of the Plan.
- **2.12.** "Computation Period" means the Plan Year or, solely for purposes of determining Years of Eligibility Service under Section 2.50(c), either the twelve-month period commencing on the Employment Date or any Plan Year beginning with the first Plan Year commencing after the Employment Date.
 - **2.13.** "Death Benefit" means the benefit, if any, provided under Article VI of the Plan.
- **2.14.** "Disability" means a mental or physical condition which occurs prior to a Participant's Termination of Employment which entitles the Participant to disability benefits under the federal Social Security Act. To qualify as a Disability, the Participant must be

- 16 -

determined to be disabled by the Social Security Administration as of a date which falls on or before his Termination of Employment (determined without regard to Section 2.45(b))

- **2.15.** "Early Retirement Date" means the date of the Participant's Termination of Employment subsequent to the date he has accumulated 65 Points and five Years of Service as defined in Section 2.50(b) (or after he has attained age 55, in the case of a Participant who first became a Participant prior to January 1, 1990 and whose Benefit is determined on the basis of his Pre-1990 Accrued Benefit in accordance with Section 2.1(b)(i), (v) or (vi)).
 - **2.16.** "Effective Date" means January 1, 2016, except as expressly provided otherwise in the Plan.
- **2.17.** "Eligible Employee" means any Employee who was on the payroll of a Participating Employer on December 31, 2006, and whose Compensation constitutes wages from employment within the meaning of Sections 3121(a) and (b) of the Federal Insurance Contributions Act on and after the effective date of the adoption of the Plan by the Participating Employer. Eligible Employees will not include any of the following:
 - (a) Collective Bargaining Exclusion. An Employee who is a member of a unit of employees covered by a collective bargaining agreement if there is evidence that retirement benefits were the subject of good faith bargaining between representatives of such unit and an Employer;
 - (b) Excluded Classifications. (i) Employees employed in the Commonwealth of Puerto Rico or an Excluded Division listed on Supplement C; (ii) non-resident aliens with no U.S.-source income; (iii) non-resident aliens with U.S.-source income who have been assigned one of the benefit entitlement codes listed in Supplement E; (iv) any Employees (including U.S. citizens) accruing benefits under the Company's pension plan for international employees, (v) leased employees described in Code Section 414(n) or (vi) an independent contractor or self-employed individual. If an Employee is excluded from participation in the Plan as an independent contractor and is later reclassification to become a Participant, upon the completion of the eligibility requirements of Section 3.2, and will be credited with his Years of Service completed since the date as of which he became an employee for purposes of Section 2.50(b) but not for purposes of Section 2.50(a). If a Participant is reclassified for wage and hour purposes as an independent contractor or self-employed individual, any Accrued Benefit under the Plan attributable to service while the individual was improperly classified will be forfeited.
 - (c) Employees Hired after December 31, 2006. Anything else contained herein to the contrary notwithstanding, an Employee who was not employed by a Participating Employer on December 31, 2006, shall not be an Eligible Employee, and shall not become a Participant or accrue any benefit hereunder.

2.18. "Employee" means any person who is an employee of an Employer (as determined under the law of the relevant state or country) in active employment or on an approved leave of absence (including the period of Disability described in Section 4.4), including the period of time before which it became an Employer, but excluding the period of time after which it ceases to be an Employer. An individual who is considered a leased employee of an Employer under the provisions of Code Section 414(n)(2), as amended by the Small Business Job Protection Act of 1996, will be an Employee.

For purposes of the Plan, including Section 2.17(b)(v), "leased employee" means any person (other than a common-law employee of an Employer) who, under an agreement between an Employer and any other person (the "leasing organization"), has performed services for an Employer or for an Employer and related persons (determined in accordance with Code Section 414(n)(6)) on a substantially full-time basis for a period of at least one year, provided that the services are performed under the primary direction or control of an Employer. Contributions provided to a leased employee that are attributable to services performed for an Employer will be treated as provided by the Employer. The term "leased employee" will not include any person who would otherwise be a leased employee if (a) the person is covered by a money purchase pension plan providing (i) a nonintegrated employer contribution rate of at least 10% of compensation, as defined in Code Section 415(c)(3), but including amounts contributed in accordance with a salary reduction agreement that are excludable from the person's gross income under Code Section 125, 402(e)(3), 402(h), or 403(b), (ii) immediate participation, and (iii) full and immediate vesting; and (b) leased employees do not constitute more than 20% of the workforce of the Employer who are not Highly Compensated Employees.

2.19. "Employer" means:

- (a) Controlled Group. A Participating Employer and any corporation, trade or business, if it and the Participating Employer are members of a controlled group of corporations as defined in Code Section 414(b) or under common control as defined in Code Section 414(c); provided, however, that, solely for purposes of the provisions pertaining to maximum pensions set forth in Section 13.12, the standard of control under Code Sections 414(b) and 414(c) will be deemed to be "more than 50 percent" rather than "at least 80 percent";
- (b) Affiliated Service Group. A Participating Employer and an organization, if it and the Participating Employer are members of an affiliated service group as defined in Code Section 414(m); or
- (c) Other Related Organizations. A Participating Employer and any other organization described in applicable regulations issued under Code Section 414(o).
- **2.20. "Employment Date"** means the day a person is credited with his first Hour of Service, or in the case of an Employee who loses his prior Years of Eligibility Service, the first day on which the Employee is credited with an Hour of Service upon his rehire as an Employee.
 - **2.21.** "Entry Date" means January 1 and July 1 of each Plan Year, except as provided in Section 3.3(b).

- **2.22.** "ERISA" means the Employee Retirement Income Security Act of 1974, as amended.
- **2.23. "Excluded Division"** means a division or unit of a Participating Employer, the Employees of which are not Eligible Employees. Excluded Divisions are all divisions or units excluded from the Plan pursuant to Supplement C.
 - **2.24. "Highly Compensated Employee"** refers to an Employee if, for a Plan Year:
 - (a) The Employee was a five-percent owner (as defined in Code Section 416(i)) of the Employer at any time during that Plan Year or the preceding Plan Year; and
 - (b) for the preceding year, the Employee
 - (i) had Compensation in excess of \$120,000 (as adjusted pursuant to Code Section 414(q)(1)), and
 - (ii) was in the top 20% of Employees, ranked by Compensation paid during such Plan Year.

For purposes of paragraph (b)(ii) Compensation is not limited as provided in Code Section 401(a)(17). Employees who have not completed six months of service, who normally work less than 17½ hours per week, who normally work not more than six months during any Plan Year, who have not attained age 21 and union employees are excluded from the determination of the top 20% of Employees.

For purposes of this Section, non-resident aliens with no U.S.-source income are not considered Employees.

2.25. "Hour of Service" means:

- (a) Duty Hours. Each hour for which an Employee is directly or indirectly paid or entitled to payment by an Employer for the performance of duties.
- (b) Non-Duty Hours (Paid). Each hour for which an Employee is directly or indirectly paid or entitled to payment by an Employer for reasons (such as vacation, holidays, sickness, short-term disability, long-term disability, medical leave, family medical leave or jury duty) other than the performance of duties.
- (c) Non-Duty Hours (Unpaid). Each hour for which an Employee is not paid due to medical leave, family medical leave, approved leave of absence or on layoff. Up to a total of 501 Hours of Service will be credited under this subsection (c) to an Employee in a Computation Period on account of any single continuous period during which the Employee performs no duties; provided, however, that if such continuous period extends into the next Computation Period, up to 501 additional Hours of Service will be credited in such next Computation Period; and further provided that no Hours of Service will be credited under this subsection (c) for any period of time after the Employee's Termination of Employment.

- (d) Back-Pay Hours. Each hour for which no credit has been given under subsections (a), (b) or (c) above, but for which back pay, irrespective of mitigation of damages, has been either awarded or agreed to by an Employer.
- (e) Military Service Hours. To the extent not taken into account under another subsection of this Section, each hour of the normally scheduled work week during a period when the Employee is absent from employment with an Employer for voluntary or involuntary military service with the armed forces of the United States, provided that either such Employee is receiving Compensation from the Employer representing the differential between the Employee's normal Compensation and the amount paid to the Employee by the armed forces, or that the Employee returns to work within 90 days after his discharge date or within such longer period of time as may be prescribed by USERRA.
- (f) Worker's Compensation. No Hours of Service will be credited if payment is made solely to comply with applicable workers' compensation or disability insurance laws.
- (g) Disability. For purposes of Section 2.50, Hours of Service will only be counted during a period for which a Participant receives payments on account of a Disability to the extent provided in Section 4.4.
- (h) Intermittent Family Leave. An Employee will be credited with Hours of Service for each week in which he is on Intermittent Family Leave. Subsection (c) will not apply to such Employees. "Intermittent Family Leave" has the meaning given in the Employer's policies and procedures manual for an Employee who periodically needs time off for the treatment and care of himself or family members due to conditions which require ongoing medical treatment but which do not require the Employee to take an extended leave of absence to provide or obtain such care.
- (i) Maternity/Paternity Absences and FMLA Leave. An Employee who is absent from work due to a Maternity/Paternity Absence (as hereinafter defined) or due to an unpaid leave of absence for which credit is required pursuant to the Family Medical Leave Act of 1993, as amended, to be given for purposes of avoiding a break in service shall be treated as having completed certain Hours of Service for a limited period. The Employee will be treated as completing either (i) the number of Hours of Service that normally would have been credited but for the absence (i.e., 45 Hours of Service per week) or (ii) if the normal work hours are unknown, eight Hours of Service for each normal workday during the leave, to a maximum per Plan Year of 501 Hours of Service. The Hours of Service required to be credited under this subsection must be credited only to prevent a One-Year Break in Service in the Plan Year in which the absence begins for one of the permitted reasons or, if crediting in such year is not necessary to prevent a One-Year Break in Service in the Plan Year, in the following Plan Year. For this purposes, a "Maternity/Paternity Absence" means a paid or unpaid absence from employment (including an unapproved leave of absence) with an Employer by

reason of the pregnancy of the Employee; by reason of the birth of a child of the Employee; by reason of the placement of a child under age 18 in connection with the adoption of the child by the Employee (including a trial period prior to adoption); or for purposes of caring for a child immediately following birth or adoption. The Employee must prove to the satisfaction of the Administrative Committee that the absence meets the above requirements and must supply information concerning the length of the absence unless the Administrative Committee has access to relevant information without the Employee submitting it

The number of Hours of Service to be credited to Employees will be calculated based on 45 hours for each week for which the Employee would be entitled to at least one Hour of Service. In the case of a payment which is made or due on account of a period during which an Employee performs no duties and which results in the crediting of Hours of Service under subsection (b), (c) or (e) above, or in the case of an award or agreement for back pay made with respect to a period described in subsection (d) above, the number of Hours of Service to be credited will be in accordance with the provisions of the Rules and Regulations for Minimum Standards for Employee Pension Benefit Plans, U.S. Department of Labor, 29 C.F.R. Section 2530.200b-2(b) which are hereby incorporated by reference. Such rules and regulations will apply to subsection (c) above as if absences described in such Section were paid absences. Hours of Service will be credited to a Plan Year in accordance with the provisions of subsection (c) of the above-cited Department of Labor Regulations. Hours required to be credited for more than one reason under this Section which pertain to the same period of time will be credited only once.

For purposes of determining the Hours of Service for eligibility, Years of Service under Section 2.50(b) and Points under Section 2.40, an Employee employed by a Non-Participating Employer outside of the United States (i.e., not on a U.S. payroll) will be credited with 190 Hours of Service for each month during which he is employed in such capacity.

- **2.26.** "Investment Committee" means the committee which is responsible for directing the investment of the Trust Fund in accordance with Article VIII.
- **2.27.** "Investment Manager" means a fiduciary who meets the requirements of ERISA Section 3(38) and to whom the Investment Committee has delegated the responsibility for investment of a portion of the assets of the Trust Fund.
- **2.28.** "Joint and 50 Percent Survivor Annuity" means, except where specifically provided to the contrary in Sections 6.3(b) and 7.2(a), a reduced Pension Benefit payable monthly during the lifetime of the Participant with the provision that 50 percent of such monthly benefit will be payable to the Participant's surviving Spouse in monthly installments commencing on the first day of the month following the month in which the Participant dies, and continuing thereafter on the first day of the month during the remaining lifetime of the Spouse.
- **2.29. "Lump Sum"** means a lump sum payment in cash which is the Actuarial Equivalent of the Participant's Accrued Benefit.

- 21 -

- **2.30.** "Non-forfeitable" means, with respect to a Participant's Accrued Benefit, 100% of such Accrued Benefit, provided that the Participant:
 - (a) reached his Normal Retirement Date;
 - (b) completed five Years of Service if the Participant was employed by an Employer after December 31, 1988;
 - (c) completed ten Years of Service if the Participant was employed by an Employer prior to January 1, 1989, but not after December 31, 1988;
 - (d) is entitled to the American Merger Benefit described in Section 2.1(b)(i), (but such Non-forfeitable interest obtained by this subsection (d) will be limited to such American Merger Benefit); or
 - (e) is subject to accelerated vesting under Article V, but only with respect to the Accrued Benefit earned as of the date provided for in the resolution that grants full vesting of the Accrued Benefit.
 - **2.31.** "Normal Retirement Date" means the date on which a Participant attains age 65.
- **2.32.** "One-Year Break in Service" means a Computation Period (other than the first Computation Period used to determine Eligibility Years of Service) in which an Employee has fewer than 501 Hours of Service.
 - **2.33.** "Participant" means a person who is participating in the Plan pursuant to the provisions of Article III.
- **2.34.** "Participating Employer"; "Non-Participating Employer". A Participating Employer means the Company or any Employer on and after the effective date of its adoption of the Plan in accordance with Section 10.1. A Non-Participating Employer means any Employer which is not a Participating Employer. For all purposes of the Plan, an Employee who is employed by a Non-Participating Employer, but who is nevertheless eligible to participate in the Plan under a grandfather rule or similar action by the Administrative Committee, shall be treated as employed by a Participating Employer.
- **2.35.** "Participating Employer Contributions" means contributions by Participating Employers to the Trust under the terms of Section 9.8.
- **2.36. "Payment Date"** means the date as of which the payment of a retirement or disability benefit commences as set forth in Sections 4.1 through 4.5. If a Participant is rehired by an Employer prior to his Normal Retirement Date, any Payment Date determined under Sections 4.3, 4.4 or 4.5 will be disregarded.
- **2.37. "Pension Benefit"** means the benefit payable to a Participant under the provisions of Section 4.1, 4.2, 4.3, 4.4 or 4.5.

- **2.38.** "Plan" means the Baxter International Inc. and Subsidiaries Pension Plan (as amended), as set forth in this document and the attached Supplements.
 - **2.39.** "Plan Year" means the twelve consecutive month period beginning January 1 and ending December 31.
 - **2.40.** "Points" means, with respect to any Participant, a number which is equal to
 - (a) the sum of the number of full years of the Participant's attained age and the number of his Years of Service described in Section 2.50(a); plus
 - (b) the Participant's service with Non-Participating Employer(s) that would have constituted Years of Service under Section 2.50(a) as if such Non-Participating Employer(s) were Participating Employer(s) and as if the Participant's remuneration with such Employer(s) consisted of wages subject to the Federal Insurance Contributions Act while employed by such Employer; plus
 - (c) the Participant's service with an Employer after December 31, 2006, that would have constituted Years of Service if the Participant had not elected to cease accruing benefits as of December 31, 2006, pursuant to Section 3.1(c); minus
 - (d) any Years of Service under Section 2.50(a) attributable to prior Lump Sums of \$5,000 or less.

For purposes of Section 2.40(b), above, a Participant will receive credit for 190 Hours of Service for each calendar month in which he is employed by a Non-Participating Employer outside of the United States.

- **2.41.** "Primary Social Security Benefit" means the benefit amount specified in (a) below, subject to the provisions of (b), (c), (d) and (e) below:
 - (a) **General Rule**. The estimated monthly primary insurance amount that a Participant is or would be entitled to receive commencing at age 65, or at Termination of Employment, if later, under the Social Security Act, whether or not he applies for or actually receives such benefit. For purposes of the Plan, such estimated amount will be determined as of the last date that he receives Compensation ("Date of Determination") on the following basis.
 - (i) Social Security Act. The calculation is based on the provisions of the Social Security Act as in effect on the Date of Determination (regardless of any retroactive changes made by legislation enacted after said date). The factors for indexing wages, if any, and the table or formula that would be used to compute the Participant's benefit upon his attaining age 65, or his Date of Determination if later, will remain unchanged from those in effect on the Date of Determination.
 - (ii) Wages in Past Years. If submitted by the Participant as described in (c) below, the Participant's actual wages under Code Section 3121 for Past

Years; otherwise wages in Past Years are determined by projecting the Pivot Wage backwards from the Pivot Year by six percent (6%) per year..

- (iii) Wages in Future Years. Wages in all Future Years are assumed to equal the Pivot Wage;
- (iv) Definitions.
 - (A) "Pivot Year" is the calendar year preceding the calendar year of the Date of Determination.
 - (B) "Past Years" are those calendar years equal to or before the Pivot Year and equal to or after the later of the calendar year 1951 or the calendar year of the Participant's 22nd birthday.
 - (C) "Future Years" are those calendar years after the Pivot Year and before the later of the calendar year of the Date of Determination or the calendar year of the Participant's 65th birthday.
 - (D) "Annualized Compensation" for a calendar year is the actual Compensation received in the calendar year multiplied by a fraction, the numerator of which is the number of days in the calendar year and the denominator or which is the number of days in the calendar year included in the period or periods for which the Participant received Compensation.
 - (E) "Pivot Wage" is the Participant's Annualized Compensation for the Pivot Year if the Participant received any Compensation in the Pivot Year; otherwise it is the Annualized Compensation for the calendar year of the Date of Determination discounted by six percent (6%).
- (b) **Disability Exception**. The Primary Social Security Benefit for a Participant eligible to receive a disability retirement benefit under Section 4.4 will be calculated as described in subsection (a) above except that:
 - (i) The Date of Determination will be the earlier of the date he ceases accruing an Accrued Benefit under Section 4.4 or his Normal Retirement Date; and
 - (ii) The Participant's Pivot Year will be the calendar year next preceding the year in which his Disability arose for which a full year of such wages were paid.
- (c) **Proof of Actual Benefit Amounts**. The Participant may submit his actual wage history on such forms, in such manner and at such times as the Administrative Committee will require for the purpose of determining wages through the Participant's Termination of Employment. In no case will the submission of a

Participant's actual wage history cause him to receive a smaller benefit than he would have been entitled to using estimated earnings. A Participant will have up to six months to submit his actual wage history after being informed of the opportunity to do so. A Participant may not submit his actual wage history later than 30 days prior to his Payment Date. A Participant who has attained age 65 may submit an actual award letter from the Social Security Administration at least 30 days prior to his Payment Date and such award will be used to determine the offset under Section 2.1, if the offset is less than the amount that would have been offset using the assumptions set forth in this Section 2.41. The date payments commence under such award letter (or the date of such award letter if there is no commencement date) must be no earlier than twelve months prior to the Participant's Payment Date.

- (d) **Limitation on Offset**. Notwithstanding the provisions of subsections (a), (b), or (c) above, where it can be demonstrated that the dollar amount of a Participant's Primary Social Security Benefit determined above and offset by this Plan exceeds the amount permitted to be offset for the Plan to remain qualified under Code Section 401(a), based on the Participant's actual detailed earnings history under Code Section 3121, such offset will be accordingly reduced.
- (e) **Accrued Benefit Prior to January 1, 1988.** Notwithstanding any other provisions of this Section, the Accrued Benefit of a Participant who terminated employment prior to January 1, 1988 will be determined by the AHSC Plan or Travenol Plan as in effect immediately prior to such date.'
- **2.42. "Projected Benefit Service"** means a Participant's actual Years of Service plus the additional Years of Service he would be expected to earn if he continued in employment as an Eligible Employee who performs at least 1,000 hours of service each Plan Year until his Normal Retirement Date.
- **2.43.** "Single Life Annuity" means the form of payment of a Participant's Pension Benefit described in Section 7.2(b).
- 2.44. "Spouse" means the person who is married to the Participant at the relevant time. If a Participant was lawfully married under the laws of the state or other jurisdiction in which the marriage ceremony was performed (including a marriage to a person of the same gender), or if a Participant entered into a common law marriage that was valid in the state or other jurisdiction in which the Participant resided at the time, the person to whom the Participant was married shall continue to be considered the Participant's Spouse regardless of the laws of the state or other jurisdiction in which the Participant currently resides. The Administrative Committee is entitled to rely on a Participant's representation of his marital status. A Participant whose records indicate that he is married may establish that he was subsequently divorced or abandoned upon delivery of a court order evidencing the same to the Administrative Committee.
- **2.45.** "Termination of Employment" occurs when a person ceases to be an Employee. The Termination of Employment for an Employee incurring a separation from service prior to May 1, 1996 will occur as of the last day he is listed on an Employer's payroll.

- 25 -

The Termination of Employment of an Employee incurring a separation from service after April 30, 1996 will occur as of the date of such separation, regardless of whether any post-separation pay is owed to such Participant or included in such Participant's Compensation. The date of a Participant's Termination of Employment is subject to the provisions of Supplement D. The foregoing provisions also are subject to the following rules:

- (a) Transfers. A transfer of employment from a Participating Employer to any other Employer within the controlled group as defined in Section 2.19(a) will not constitute a Termination of Employment.
- (b) Disability. Effective for disabilities determined by the Social Security Administration to have occurred after September 30, 1996 a Participant's Termination of Employment, for purposes of determining whether an Employee is entitled to a Disability Pension Benefit under Section 4.4, occurs as of the end of the six-month period immediately following his separation from service with an Employer during which he is receiving disability pay. If no disability date is indicated by the Social Security Administration's award letter, then the disability date will be deemed to occur after September 30, 1996 as long as the Social Security award letter is dated after March 31, 1997. If the award letter is dated prior to April 1, 1997, the disability date will be deemed to occur prior to October 1, 1996 and the six-month period will be increased to twelve months. For purposes of determining his Accrued Benefit under Section 4.4(a), a Disabled Participant's Termination Date is his Normal Retirement Date, or if earlier, the date he elects to commence his Disability Retirement under Section 4.4(b).
- **2.46.** "Trust" means the legal entity resulting from the Trust Agreements entered into by or on behalf of the Participating Employers and the Trustee pursuant to which assets of the Plan are received, held, invested and distributed to or for the benefit of Participants, Spouses, and Beneficiaries.
- **2.47. "Trust Agreement"** means the agreements entered into by or on behalf of the Participating Employers and the Trustee establishing the Trust, as amended.
- **2.48.** "Trust Fund" means all assets held by the Trustee, Investment Managers and insurance institutions in accordance with the Trust Agreement and the Plan.
- **2.49.** "Trustee" means any individual(s) or corporation(s) designated in the Trust Agreement to execute the duties of the Trustee as set forth in the Trust Agreement.
- **2.50. "Year of Service"** means each Computation Period during which an Employee earns at least 1,000 Hours of Service.
 - (a) Benefit Accrual. For purposes of determining a Participant's Accrued Benefit and Projected Benefit Service, a Year of Service means a Year of Service (including periods of Disability to the extent provided in Section 4.4) earned while a Participant excluding:

- (i) Non-Participating Employer Service. All Years of Service due to employment with (a) an Employer prior to the date as of which it became a Participating Employer (except as otherwise agreed to by the Administrative Committee or its delegate in connection with such adoption); (b) a Non-Participating Employer; and (c) an Excluded Division of a Participating Employer.
- (ii) Service Prior to Participation. All Years of Service with a Participating Employer prior to becoming a Participant except that the following service prior to becoming a Participant will be recognized:
 - (A) Effective January 1, 1987, if an Eligible Employee becomes a Participant at any time during a Plan Year, his Hours of Service earned from January through the date participation commences in such Plan Year will count in determining whether a Year of Service is earned.
 - (B) Prior to January 1, 1986, the Plan was only applicable to Eligible Employees of American and its participating units. Effective as of such date, the Plan was extended to employees of the Company and its participating units. Accordingly, all Participants employed by the Company on or after January 1, 1986 will be credited with any Years of Service with respect to their employment with a Participating Employer prior to January 1, 1986, beginning with the January 1 or July 1 following the date such Participant attained age 18 and completed one Year of Service. Such Participants need not have been employed by the Company as of the date American was merged into the Company in order to receive such credit. Participants will receive credit for 190 Hours of Service for each month prior to January 1, 1986 during which the Eligible Employee was employed by a Participating Employer in lieu of the 45 Hours of Service equivalency set forth in Section 2.25.
 - (C) Effective for Participants who have not commenced their Pension Benefit as of the Effective Date but who became a Participant under the Plan's prior eligibility rules (age 30 and two Years of Service prior to January 1, 1976 and age 25 and one Year of Service between January 1, 1976 and January 1, 1985) such Participants are credited with Years of Service beginning on the January 1 or July 1 next following the date on which they attained the age and completed the service set forth in Section 3.1(b)(i) and (ii).
- (iii) Service Prior to Divestiture. In the case of a Participant who is rehired after ceasing to participate in the Plan as the result of a sale or other divestiture of the business that employed the Participant, in which the Participant's Accrued Benefit and the assets attributable to such Accrued

Benefit were transferred to another plan in connection with the transaction, all Years of Service earned prior to such transaction unless the Accrued Benefit and attributable assets are transferred back to the Plan, in which event the rules generally applicable to a rehired former Participant shall apply.

- (b) Vesting and Entitlement to Benefits. For purposes of determining (i) a Participant's Non-forfeitable interest in his Accrued Benefit, and (ii) entitlement to benefits under Articles IV and VI, a Year of Service includes a Year of Service as an Employee of an Employer. The following Years of Service will be recognized for purposes of this subsection (b):
 - (i) Years of Service earned with an entity prior to its becoming an Employer will be recognized only if the Employee was employed on the date the entity became an Employer. For purposes of this paragraph (i), an Employee will receive 190 Hours of Service per month from the most recent date of hire until the date of acquisition.
 - (ii) Prior to January 1, 1986, the Plan was only applicable to Eligible Employees of American. Effective as of such date, the Plan was extended to Eligible Employees of the Company. Accordingly, for all Participants whose Termination of Employment occurs after the Effective Date, such Participants will be credited with Years of Service with respect to their employment with an Employer prior to January 1, 1986. In determining such Years of Service, Participants will receive credit for 190 Hours of Service for each month prior to January 1, 1986 during which the Eligible Employee was employed by an Employer in lieu of the 45 Hours of Service equivalency set forth in Section 2.25.
- (c) Eligibility to Participate. For purposes of determining an Employee's eligibility to participate in the Plan a Year of Service (sometimes referred to as a Year of Eligibility Service) shall be calculated in the same manner as in subsection (b), except that the first Computation Period shall be the twelve month period commencing on the Employment Date rather than the Plan Year in which the Employment Date occurs.
- (d) Years of Service Disregarded. If an Employee does not have a Non-forfeitable interest in the Plan and incurs a One-Year Break in Service and thereafter returns to employment with an Employer on or after January 1, 1985, his prior Years of Service for vesting, eligibility and benefit service will be disregarded only if he has five consecutive One-Year Breaks in Service and if the number of One-Year Breaks in Service equals or exceeds his Years of Service for vesting purposes prior to the One-Year Break in Service.
- (e) Special Rule for Certain Rehired Participants. In the case of a Participant who is rehired by the Company after ceasing to participate in the Plan as the result of the Company's spin-off of the Caremark and Dade divisions that employed the

Participant, in which the Participant's Accrued Benefit and the assets attributable to such Accrued Benefit were transferred to another plan in connection with such transaction, all Years of Service earned prior to the spin-off shall be included for purposes of determining the Participant's total number of Points under Section 2.7; provided, however, that such Participant has a Termination of Employment on or after January 1, 2003.

ARTICLE III. Participation

- **3.1. Becoming a Participant**. An Eligible Employee may become a Participant in accordance with the following requirements:
 - (a) Continuation of Participation. Each Eligible Employee who was a Participant in the Plan immediately prior to the Effective Date will participate in the Plan on and after the Effective Date until his participation ceases in accordance with the Plan.
 - (b) Initial Participation. Each Eligible Employee who is not a Participant as described in (a) will become a Participant on the first Entry Date coincident with or next following the date he satisfies the following requirements:
 - (i) He has completed one Year of Service; and
 - (ii) He has attained age 21.
 - (c) Participation Freeze. Participation in the Plan is frozen effective December 31, 2006, and no Employees shall become Participants after such date, subject to the following:
 - (i) No Employee who was not employed by a Participating Employer on December 31, 2006, shall be eligible to Participate in the Plan.
 - (ii) An Eligible Employee who was employed by a Participating Employer on December 31, 2006, but who had not satisfied the requirements of paragraph (b) on such date, shall become a Participant on the first Entry Date after he satisfies such requirements, unless he elects not to become a Participant as provided in subparagraph (c)(iii) below.
 - (iii) Each Participant or Eligible Employee who either (A) was employed by a Participating Employer on December 31, 2006, and would not have been entitled to a Deferred Vested Benefit if he incurred a Termination of Employment on such date, or (B) is described in subparagraph (c)(ii) above, may irrevocably elect to either cease accruing benefits as of December 31, 2006, or not to become a Participant. The Accrued Benefit of a Participant who elects to cease accruing benefits shall thereafter be equal to his Accrued Benefit as of December 31, 2006, which shall not be adjusted for subsequent changes in his Average Monthly Compensation, Years of Service, Projected Benefit Service, or Primary Social Security Benefit, but such Participant shall continue to earn Years of Service for purposes of vesting, and Points, and his Accrued Benefit shall be payable upon retirement or other Termination of Employment as provided herein. An Eligible Employee who elects not to become a Participant shall thereafter be ineligible to become a Participant in the Plan. Such election shall be made in accordance with rules and procedures established by the

Administrative Committee, which rules and procedures shall specify the manner in which the election shall be made. Although it is anticipated that Participants and Eligible Employees who make such election will be eligible to receive additional contributions under the Baxter International Inc. and Subsidiaries Incentive Investment Plan ("IIP"), nothing contained herein shall be construed to limit the Company's authority to amend the IIP, or to permit a Participant to resume participation in this Plan should it do so. A Participant or Eligible Employee who fails to affirmatively make the election described above shall continue to accrue benefits, or shall become a Participant, as applicable, in accordance with the remaining terms of the Plan.

- 3.2. Ceasing to Be a Participant. Once an Eligible Employee has become a Participant in accordance with Section 3.1, he will remain a Participant in the Plan until the later of the date such Participant ceases to be an Eligible Employee (or elects to cease accruing benefits pursuant to Section 3.1(c)(iii)) or the date that all of the benefits to which the Participant is entitled under the Plan, if any, have been distributed for his benefit in accordance with the Plan.
- **3.3. Reemployment**. Reemployment following a Termination of Employment will affect the former Employee as follows:
 - (a) Eligible Employee and Prior Participant. An Eligible Employee who previously was a Participant prior to his Termination of Employment will automatically become a Participant on the date he again earns an Hour of Service as an Eligible Employee.
 - (b) Eligible Employee with One or More Years of Service. An Eligible Employee who previously satisfied the requirements of Section 3.1(b) but incurred a Termination of Employment before the next Entry Date, and who is rehired as an Eligible Employee with one Year of Eligibility Service, will automatically become a Participant on the later of (i) the Entry Date that would have applied without a Termination of Employment or (ii) the date he again earns an Hour of Service as an Eligible Employee.
 - (c) Eligible Employee without One or More Years of Service. An Eligible Employee who did not have a Year of Eligibility Service when he incurred a Termination of Employment must earn a Year of Eligibility Service based on the eligibility Computation Period described in Section 2.12.
 - (d) Re-Employment after December 31, 2006. Notwithstanding the foregoing, a Participant who incurs a Termination of Employment after December 31, 2006, and is subsequently re-employed, shall not be eligible to resume accruing benefits under the Plan, but such Participant's Accrued Benefit shall be fixed as of the date on which he incurred the Termination of Employment. An Eligible Employee who incurs a Termination of Employment after December 31, 2006, prior to

having become a Participant and who is subsequently re-employed shall not be an Eligible Employee and shall not become a Participant.

- **3.4. Change of Job Status.** An Employee who has satisfied the service requirement of Section 3.1(b)(i) but who is not a Participant because he is not an Eligible Employee will become a Participant on the later of (a) the Entry Date that would have applied had he been an Eligible Employee or (b) the date he becomes an Eligible Employee.
- 3.5. Transfers. A Participant who transfers to a Non-Participating Employer will continue to accrue Years of Service (as defined in Section 2.50(a)) after such transfer, unless the Non-Participating Employer is located outside of the United States, in which case he will not accrue any additional Years of Service after such transfer. If a Participant transferred to a Non-Participating Employer outside of the United States and is transferred back to a Participating Employer after December 31, 2006, he will not accrue any additional Years of Service, and his Accrued Benefit shall be fixed as of the date of the original transfer. For purposes of this Section 3.5, a Non-Participating Employer located in Puerto Rico shall be considered to be located outside of the United States.
- 3.6. International Employees. A Participant who becomes eligible to accrue benefits under the Company's pension plan for international employees will cease accruing any additional Years of Service (defined in Section 2.50(a)) effective as of the date he becomes a participant in such pension plan. If such a Participant ceases to accrue benefits under the pension plan for international employees after December 31, 2006, he will not accrue any additional Years of Service, and his Accrued Benefit shall be fixed as of the date he originally became eligible to accrue benefits under the pension plan for international employees.
- 3.7. Reemployment of Veterans. Notwithstanding any provision of this Plan to the contrary, benefit and service credit with respect to qualified military service will be provided in accordance with Code Section 414(u). If an Eligible Employee's or Participant's military service began prior to January 1, 2007, and Section 414(u) applies upon his re-employment after December 31, 2006, he shall resume or commence participation as required by Section 414(u), and if he would have been eligible to make the election described in Section 3.1(c)(iii) had he been employed on December 31, 2006, he shall be permitted to make the election upon his re-employment in accordance with rules and procedures established by the Administrative Committee. If he so elects, his Accrued Benefit shall thereafter be determined as of December 31, 2006, as if he had been employed through such date, in accordance with the requirements of Section 414(u). Effective January 1, 2009, the provisions of this Section shall also apply to a Participant who dies or becomes Disabled while performing qualified military service, as if such Participant had returned to employed on the day prior to the day on which he died or incurred such Disability.

ARTICLE IV.

Eligibility for and Amount of Pension Benefits

- **4.1. Normal Retirement**. A Participant who incurs a Termination of Employment in the month in which he attains his Normal Retirement Date is entitled to his Pension Benefit.
 - (a) Amount. The amount of the Participant's Pension Benefit will be his Accrued Benefit, or the Actuarial Equivalent of his Accrued Benefit if paid in one of the other forms of payment described in Article VII.
 - (b) Commencement. A Participant will begin receiving his Pension Benefit in the form provided for in Article VII commencing effective as of the Payment Date. The Payment Date is the first day of the second month next following his Normal Retirement Date. However, if a Participant notifies the Administrative Committee of his intent to retire at least 30 days prior to his Normal Retirement Date, the Payment Date is the first day of the month following his Normal Retirement Date. In order to receive a Pension Benefit, the Participant must provide the Administrative Committee with a complete and accurate application finalized at least ten days before the Payment Date or the Payment Date may be delayed to the first day of the following month.
- **4.2. Late Retirement.** A Participant who incurs a Termination of Employment during any month after the month in which his Normal Retirement Date occurs is entitled to a Pension Benefit.
 - (a) Amount. The amount of the Participant's Pension Benefit will be his Accrued Benefit, or the Actuarial Equivalent of his Accrued Benefit if paid in one of the other forms of payment described in Article VII.
 - (b) Commencement. A Participant will begin receiving his Pension Benefit in the form provided for in Article VII commencing effective as of the Payment Date. The Payment Date is the first day of the second month next following his Termination of Employment. However, if a Participant notifies the Administrative Committee of his intent to retire at least 30 days prior to his Termination of Employment, the Payment Date is the first day of the month following his Termination of Employment. In order to receive a Pension Benefit, the Participant must provide the Administrative Committee with a complete and accurate application finalized at least ten days before the Payment Date or the Payment Date may be delayed to the first day of the following month.
- **4.3. Early Retirement.** A Participant who incurs a Termination of Employment on or after his Early Retirement Date is entitled to a Pension Benefit.
 - (a) Amount. The amount of the Participant's Pension Benefit will be his Accrued Benefit determined as of his Early Retirement Date, or the Actuarial Equivalent of his Accrued Benefit, if paid in one of the other forms of payment described in Article VII. If the Participant elects to receive his Pension Benefit prior to his

- Normal Retirement Date, his Accrued Benefit will be reduced to the extent required by the Early Pension Benefit Reduction Factors applicable to such Participant as described in Section 2.7.
- (b) Commencement. A Participant will begin receiving his Pension Benefit in the form provided for in Article VII commencing effective as of the Payment Date. The Payment Date is the first day of the month following his Normal Retirement Date. The Participant may designate an earlier Payment Date which will be the first day of any month which is at least 30 days after the Participant delivers notice of such designation on such form, in such manner and at such times as the Administrative Committee will require. In order to receive a Pension Benefit, the Participant must provide the Administrative Committee with a complete and accurate application finalized at least ten days before the Payment Date or the Payment Date may be delayed to the first day of the following month.
- **4.4. Disability Retirement**. A Participant is entitled to a Disability Pension Benefit if he incurs a Termination of Employment described in Section 2.45(b)
 - (i) after he has been credited with five Years of Service under Section 2.50(b) and has been credited with 65 Points, or
 - (ii) after he has been credited with ten Years of Service under Section 2.50(b) if he became a Participant prior to January 1, 1990.
 - (b) Amount of Benefit, The amount of the Disability Pension Benefit will be the Participant's Accrued Benefit, or the Actuarial Equivalent of the Accrued Benefit if paid in one of the other forms of benefit described in Article VII. If the Participant elects to receive his Disability Pension Benefit prior to his Normal Retirement Date, his Accrued Benefit will be determined as if the Payment Date was the date of the Participant's Termination of Employment and will be reduced to the extent required by the Early Pension Benefit Reduction Factors applicable to such Participant as described in Section 2.7. The Accrued Benefit payable under this Section 4.4 will be determined by using the Participant's Years of Service earned at the time of his Disability, plus the Years of Service that he would have earned had he continued in employment as an Eligible Employee from the time of his Disability until his Normal Retirement Date, provided that such Participant is actually receiving disability payments from the Social Security Administration. The Accrued Benefit will be calculated using the Participant's Compensation in the Plan Year preceding the date as of which he is determined to be disabled by the Social Security Administration for the first year of Disability and for each year thereafter up to the earlier of the Participant's Termination of Employment or the first day of the month following his Normal Retirement Date in order to calculate the Participant's Average Monthly Compensation; provided, however, that for Participant's who are determined to be disabled on or after August 20, 2010, the Participant's Compensation for the Plan Year preceding the date as of which payment of disability benefits commences shall be used if it results in a higher Average Monthly Compensation.

- (c) Commencement of Benefits. Payment of the Accrued Benefit described in this Section 4.4 will commence effective on the Payment Date, in the form provided for under Article VII. The Payment Date is the first day of the month coincident with or next following the Participant's Normal Retirement Date. However, a Participant may elect an earlier Payment Date which may be the first day of any month following the Participant's accumulation of 65 Points if the Participant provides notice of such election to the Administrative Committee at least 30 days prior to such month. In order to receive a Pension Benefit, the Participant must provide the Administrative Committee with a complete and accurate application finalized at least ten days before the Payment Date or the Payment Date may be delayed to the first day of the following month.
 - A Participant eligible for a Disability Pension Benefit also may elect to receive a Lump Sum payment of his Accrued Benefit at any time when the Lump Sum value of such Accrued Benefit is \$5,000 or less, and, for purposes of determining such Lump Sum, the Participant will be deemed to have incurred a Termination of Employment on the Payment Date for such Lump Sum. Payment of any such Lump Sum will be subject to the provisions of Section 7.3.
- (d) Recovery. If a Participant who is earning an Accrued Benefit under this Section 4.4 recovers from his Disability prior to his Normal Retirement Date, such Participant will be deemed to have incurred a Termination of Employment as of the date of recovery, unless he is rehired as an Employee of an Employer at that time, in which case his prior Years of Service will be retained for all Plan purposes. If the Participant is rehired as an Eligible Employee, the Participant will continue accruing benefits under the provisions of the Plan applicable to all Eligible Employees. If such Participant is not reemployed by an Employer, he will become entitled to his Accrued Benefit under Section 4.3 or 4.5 depending upon his Points, Years of Service or age at the time of such Termination of Employment. The Participant's Accrued Benefit will be based on his Years of Service at the time of his Termination of Employment and his Average Monthly Compensation will be determined as set forth in 4.4(a). The Participant's Primary Social Security Benefit will be calculated based upon his Termination of Employment at the time of his recovery. For purposes of this subsection, a Participant will be deemed to be recovered from his Disability if he is no longer receiving disability payments from the Social Security Administration.
- **4.5. Deferred Vested Benefit**. A Participant who incurs a Termination of Employment for any reason other than Disability or death before his Early Retirement Date, but after completing at least five Years of Service as defined in Section 2.50(b), is entitled to a Pension Benefit.
 - (a) Amount. The amount of the Participant's Pension Benefit will be his Accrued Benefit determined at his Termination of Employment, or the Actuarial Equivalent of his Accrued Benefit, if paid in one of the other forms of payment described in Article VII. If the Participant elects to receive his Pension Benefit prior to his Normal Retirement Date, his Accrued Benefit will be reduced to the

extent required by the Early Pension Benefit Reduction Factors applicable to such Participant as described in Section 2.7.

(b) Commencement. Payment of the Accrued Benefit described in this Section 4.5 will commence effective on the Payment Date, in the form provided for under Article VII. The Payment Date is the first day of the month following his Normal Retirement Date. However, such Participant may designate an earlier Payment Date which may be the first day of any month following the date on which he attains an age which causes his Points to equal or exceed 65 if the Participant provides notice of such election to the Administrative Committee at least 30 days prior to such date. Such election will be made on such form, in such manner and at such times as the Administrative Committee will require. In order to receive a Pension Benefit, the Participant must provide the Administrative Committee with a complete and accurate application finalized at least ten days before the Payment Date or the Payment Date may be delayed to the first day of the following month.

A Participant entitled to a minimum Accrued Benefit under an applicable Supplement is fully vested in the portion of the Participant's Accrued Benefit, if any, which is described in such Supplement. Such Participant may request, after he attains age 55 but prior to his Normal Retirement Date, on such form, in such manner and at such times as the Administrative Committee will require, to begin receiving his Accrued Benefit commencing effective as of the first day of any month after he attains age 55. In such event, the amount of the benefit which would be paid beginning on his Normal Retirement Date will be reduced according to the Deferred Vested Benefit Reduction Factors described in Table 2 of Supplement A.

ARTICLE V.

Special Vesting Provisions for Divested Employees

- 5.1. Accelerated Vesting. In the event of a sale by the Company on or after the Effective Date of the stock or substantially all of the assets of a Participating Employer, or a separate business unit of a Participating Employer, so that the Participating Employer, or such business unit, ceases to be an Employer, the Administrative Committee, in its sole discretion, may determine that all or a portion of the affected Participants of said Participating Employer will be fully vested in their Accrued Benefit, determined on the date as of which the Participating Employer is no longer an Employer. If such employment ceases prior to the accumulation of 65 Points (or age 55 in the case of Participants who first become Participants prior to January 1, 1990), the Participant is entitled to the forms of payment and early commencement reduction factors applicable to a terminated Participant entitled to a deferred vested benefit under Section 4.5. If his employment ceases on or after accumulation of 65 Points (or age 55 in the case of Participants who first become Participants prior to January 1, 1990), the Participant is entitled to the forms of payment and early commencement reduction factors applicable to a terminated Participant entitled to an early retirement benefit under Section 4.3.
- **5.2. Re-employment.** If a Participant who fully vests under Section 5.1 is later rehired as an Eligible Employee, and that Participant received a distribution of his Accrued Benefit, his prior Years of Service for vesting purposes will be reinstated but no prior Years of Service for purposes of determining his Accrued Benefit will be reinstated. A repayment of the distribution is not required or permitted. The Pension Benefit to which the rehired Participant is entitled at his subsequent Termination of Employment will be paid under the terms and conditions applicable to all other Participants. If a Participant would be eligible for accelerated vesting under Section 5.1 if he was not rehired, and such Participant is rehired by the Employer before any benefit payments are made by the Plan, no such payment will be made and the Participant will not be entitled to the special vesting of Section 5.1. Such Participant will instead be subject to the regular vesting provisions as if no divestiture and Termination of Employment had ever occurred. The provisions of this Section 5.2 shall also apply to a Participant whose Accrued Benefit, and the assets attributable thereto, is transferred to another plan in connection with the divestiture as if such Participant had received a distribution of his Accrued Benefit, unless such Accrued Benefit is transferred back to the Plan upon rehire.

ARTICLE VI. Benefits after Death

- **6.1. Death after Payment Date**. The Death Benefit, if any, of a Participant who dies after his Payment Date under the Plan are those specified under the form in which his benefits were being paid at the time of his death.
- **6.2. Death before Payment Date.** Except as otherwise provided in this Article and in Supplement I, no benefits are payable on behalf of a Participant who dies before his Payment Date. Death Benefits that have been paid under a prior version of the Plan or which are currently being paid under such prior version will not be affected by the provisions of this Article.
- **6.3. Death Benefits Payable to Spouses.** A pre-retirement lifetime Death Benefit is payable under either (a) or (b) below to the surviving Spouse of a Participant who incurred a Termination of Employment after December 31, 1989 and who dies with a Non-forfeitable Accrued Benefit prior to the commencement of his Pension Benefits under the Plan. Section 6.3(a) and (b) of Supplement I applies in lieu of subsection 6.3(a) and (b) below to Participants who incurred a Termination of Employment prior to January 1, 1990 and have not be rehired as an Eligible Employee after December 31, 1989.
 - (a) Deferred Annuity. A deferred pre-retirement lifetime Death Benefit is payable to the surviving Spouse of a Participant who dies prior to attaining his 65th Point. The Death Benefit will commence effective as of the first day of the month following the month in which the Participant would have attained an age that, when combined with his Years of Service credited as Points would have credited him with his 65th Point. Such Death Benefit is based upon the assumption that the Participant had elected a Joint and 50 Percent Survivor Annuity form of payment and, if he had not previously incurred a Termination of Employment, that he had incurred a Termination of Employment immediately prior to his death. The amount of the monthly Death Benefit paid to such surviving Spouse is based on an Accrued Benefit determined under Section 2.1(a) as of the date such payment commences and is not actuarially reduced for commencement prior to age 65.
 - (b) Immediate Annuity. An immediate pre-retirement lifetime Death Benefit is payable to the surviving Spouse of a Participant who dies after attaining an age which, when combined with his Years of Service credited as Points as of the date of his death, equals or exceeds 65 Points. Such Benefit commences effective as of the first day of the month following the date of the Participant's death. The Death Benefit payable to the Participant's surviving Spouse is based upon the assumption that the Participant had elected a Joint and 100 Percent Survivor Annuity form of payment and, if he had not previously incurred a Termination of Employment, that he had incurred a Termination of Employment immediately prior to his death. The Death Benefit paid to the Participant's surviving Spouse is based on an Accrued Benefit determined under Section 2.1(a) as of the date of payment and is not actuarially reduced for early commencement prior to age 65.

- Compliance with Code Section 417. Death Benefits under this Section 6.3(a) and (b) (and not Supplement I) are based on the Accrued Benefit described in Section 2.1(a) only, even if the Participant has a frozen Accrued Benefit under Section 2.1(b) that would have paid a greater retirement benefit at age 65. In most cases, the Death Benefit, with the Points-based Early Retirement Date and full subsidy for early commencement will begin earlier and in greater amounts than partially subsidized or unsubsidized Death Benefits based on a frozen Accrued Benefit and an early retirement age of 55. However, for some married Participants with a frozen Accrued Benefit, the minimum qualified pre-retirement annuity required under Code Section 417 based on such Accrued Benefit may be greater in value, or may commence earlier than the Death Benefit in 6.3(a) or (b). In such a case, the surviving Spouse's benefit described in 6.3(a) or (b) will be paid at such earlier times and in such minimum amounts as are necessary to satisfy Code Section 417. For purposes of determining the minimum qualified pre-retirement survivor annuity based on a frozen Accrued Benefit, the Benefit Reduction Factors of Section 2.7(d) apply (with the .25% monthly reduction factor applicable if the Participant incurred a Termination of Employment after attaining age 55).
- 6.4. Survivor Benefits Payable to Non-Spouse Beneficiaries. A pre-retirement Death Benefit is payable to the Beneficiary of a Participant with a Non-forfeitable Accrued Benefit who incurred a Termination of Employment after December 31, 1989 and dies without a surviving Spouse or who dies with a surviving Spouse who has consented to the designation of a non-Spouse Beneficiary as provided in Section 6.7. Such Benefit will commence effective as of the first day of the month following the date of the Participant's death under (a) or (b) below. Section 6.4 of Supplement I applies in lieu of this Subsection 6.4 to Participants who incurred their last Termination of Employment prior to January 1, 1990.
 - (a) Calculation of Benefit. The pre-retirement Death Benefit payable under this Section 6.4 shall be the survivorship pension payable to his Beneficiary under a Ten-Year Certain and Life Annuity described in Section 7.2(c) (or an Actuarially Equivalent lump sum), calculated as if the Participant had
 - (i) incurred a Termination of Employment on his date of death if he had not already incurred a Termination of Employment prior to his death,
 - (ii) begun receiving the Annuity immediately if he had attained 65 Points at the time of his death, or, if he had not yet attained 65 Points, survived to the date he would have attained an age that, when combined with his Years of Service credited as Points, would have credited him with his 65th Point, and then begun receiving the annuity, in either case unreduced for early commencement, and
 - (iii) Died immediately after payment commenced.
 - (b) Form of Payment. The Beneficiary of a Participant who had not yet attained 65 Points at the time of his death shall receive a lump sum payment that is the

Actuarial Equivalent of the annuity described in Section 6.4(a), which shall be paid not later than the last day of the year that includes the fifth anniversary of the Participant's death. The Beneficiary of a Participant who had attained 65 Points at the time of his death shall also receive a lump sum payment, unless the Beneficiary elects to survivorship benefit payable under the Ten Year and Certain Life Annuity. Such election must be made, in accordance with procedures established by the Administrative Committee, in sufficient time to allow the first annuity payment to be paid not later than the last day of the year that includes the first anniversary of the Participant's death, and if no such election is made by such date the Beneficiary shall instead receive the lump sum payment described above.

- 6.5. Death Benefits for Disability Retirees. The surviving Spouse or Beneficiary of a Participant who incurred a Disability and who was accruing additional Benefit Service at his death under the provisions of Section 4.4 is entitled to a Death Benefit determined in accordance with Sections 6.3 or 6.4, as applicable. If the Participant incurred a Disability, accrued additional benefits under Section 4.4, recovered from his Disability, did not return to employment with the Employer, and died before his Payment Date, his surviving Spouse or Beneficiary is entitled to a Death Benefit determined in accordance with Sections 6.3 or 6.4, as applicable.
- **6.6. Pre-Retirement Death Benefits Payable in a Lump Sum**. If the present value of a Death Benefit payable in a form other than a Lump Sum is \$5,000 or less at the earliest date such Death Benefit may commence, such Death Benefit will be paid in a Lump Sum at such time, notwithstanding any provision of the Plan or election by the Participant or Beneficiary to the contrary. Any Lump Sum payable in accordance with this Section 6.5 to a Spouse is subject to the direct rollover provisions of Section 7.3.
- **6.7. Designation of Beneficiary.** Subject to Sections 7.4(d) and 13.4, a Participant's surviving Spouse is entitled to the Death Benefit, if any, which is payable with respect to the Participant. A Participant may designate a Beneficiary who is not the Participant's surviving Spouse, provided that such surviving Spouse consents to the designation. The designation of such Beneficiary can be made only after the notice required by Code Section 417(a)(3)(B) is provided to the Participant during the time periods specified in Code Section 417 and will be effective only if the consent of the surviving Spouse:
 - (i) is in writing;
 - (ii) acknowledges the effect of the Participant's designation of a Beneficiary other than the Spouse and the identity of such Beneficiary and
 - (iii) is witnessed by a notary public; provided, however, such consent will be deemed to have been granted where it is established to the satisfaction of the Administrative Committee that the consent of the Spouse cannot be obtained because (1) there is no surviving Spouse, (2) the surviving Spouse cannot be located, or (3) there exist such other circumstances as may be prescribed by regulations under ERISA or the Code. Consent of the surviving Spouse to a designated Beneficiary and any contingent

- 40 -

Beneficiary is irrevocable. Each change from one non-Spouse Beneficiary to another, or change of contingent Beneficiary, requires a new consent from the surviving Spouse.

If the Participant dies leaving no surviving Spouse and either (1) the Participant failed to file a valid beneficiary designation form, or (2) all persons designated on the beneficiary designation form have predeceased the Participant, the Participant's Death Benefit described in Section 6.4 will be paid in a lump sum to the Participant's estate.

- **6.8. Incapacitated Participants or Beneficiaries.** If a Participant or Beneficiary is incompetent or a minor, and a conservator, guardian, or other person legally charged with his care has been appointed, any benefits to which such Participant or Beneficiary is entitled is payable to such conservator, guardian, or other person legally charged with his care. The decision of the Administrative Committee in such matters is final, binding, and conclusive upon all affected or interested parties. Neither the Plan nor any representative of the Plan has any duty to see to the proper application of such payments.
- **6.9. Death During Military Service.** Notwithstanding any other provision of the Plan to the contrary, the death benefit payable to the surviving spouse or Beneficiary of a Participant who dies on or after January 1, 2007, while performing qualified military service as defined in Section 3.7, shall not be less than the death benefit that would have been payable had the Participant resumed employment immediately prior to his death and died while employed. Such Participant shall be credited with Hours of Service as if employed through the date of his death for purposes of vesting and entitlement to benefits pursuant to Section 2.50(b), but shall only be credited with any additional Accrued Benefit to the extent provided in Section 3.7.

ARTICLE VII.

Form and Payment of Pension Benefits and Death Benefits

- **7.1. Normal Form of Payment**. Pension Benefits under the Plan will commence effective as of the Payment Date and will be payable as follows:
 - (a) Joint and 50 Percent Survivor Annuity. A Participant who has a Spouse on his Payment Date will receive his Pension Benefit in the form of a Joint and 50 Percent Survivor Annuity (covering the Participant and his Spouse) unless the Participant elects not to receive such Joint and 50 Percent Survivor Annuity by electing in lieu thereof to receive payment under an available option described in Section 7.2.
 - (b) Life Annuity. A Participant who does not have a Spouse on his Payment Date will receive his Pension Benefit in the form of a Single Life Annuity unless he has properly elected to have his Pension Benefit paid in an optional form of payment under Section 7.2 in accordance with the procedures set forth in Section 7.4.
- **7.2. Optional Forms of Payment**. A Participant who is entitled to receive a Pension Benefit may elect to receive such Pension Benefit in one of the optional forms of payment described in this Section 7.2. All optional forms of payment will be the Actuarial Equivalent of the Participant's Pension Benefit. As of the Effective Date, the following optional forms of payment are available:
 - (a) Joint and 100 Percent, 75 Percent or 50 Percent Survivor Annuity. A reduced Pension Benefit payable monthly during the lifetime of the Participant with the provision that 100 percent, 75 percent or 50 percent (as elected by the Participant) of such monthly benefit will be payable to the Participant's Beneficiary in monthly installments commencing effective on the first day of the month following the month in which the Participant dies and continuing thereafter on the first day of each month during the remaining lifetime of such Beneficiary.
 - (b) Single Life Annuity. A Pension Benefit payable monthly during the lifetime of the Participant through the last monthly payment on or prior to such Participant's death.
 - (c) Ten-Year Certain and Life Annuity. A reduced Pension Benefit payable monthly during the lifetime of the Participant with the provision that if the Participant dies before receiving 120 payments, his Beneficiary will receive the same monthly payment as the Participant until a total of 120 payments have been made in the aggregate to the Participant and Beneficiary. If such Beneficiary should die prior to the aggregate payment of 120 payments, a lump sum which is Actuarially Equivalent to the remaining payments will be paid to the Beneficiary's estate.
 - (d) Special Lump Sum Option for Clintec Transferees. If the Plan received a transfer from the trustee of the Clintec Pension Plan on behalf of a Participant, such Participant may elect to receive a Lump Sum payment if the Lump Sum value of

his Accrued Benefit is not greater than \$10,000 at the time of the distribution from the Plan. The Lump Sum option under this subsection (d) will be subject to the provisions of Section 7.3(a) and (b) and Section 7.4(d) (if the Lump Sum value of the benefit is greater than \$5,000).

- 7.3. Lump Sum Cash-Out. Except as provided below, any Participant with a Non-forfeitable interest in the Plan having a Lump Sum value of \$5,000 or less on the Payment Date will receive a Lump Sum payment of the portion of his Accrued Benefit that is Non-forfeitable as soon as practicable. For purposes of this Section 7.3, if the Participant does not have a Non-forfeitable interest in his Accrued Benefit, such Participant will be deemed to have received a distribution of his entire Accrued Benefit of zero. Such Lump Sum payments will be subject to the following:
 - (a) Reemployment prior to Payment. If such a Participant is rehired by the Employer before any benefit payment is made by the Plan, no such payment will be made.
 - (b) Direct Rollovers of Lump Sums. Notwithstanding any provision of the Plan to the contrary that would otherwise limit a distributee's election under this Section, a distributee may elect, at the time and in the manner prescribed by the Administrative Committee, to have a Lump Sum paid directly to an eligible retirement plan specified by the distributee in a direct rollover. The following definitions will apply for purposes of this paragraph (b):
 - (i) Eligible Retirement Plan. An "Eligible Retirement Plan" is an individual retirement account described in Code Section 408(a), an individual retirement annuity described in Code Section 408(b), an annuity plan described in Code Section 403(a), or a qualified trust described in Code Section 401(a), that accepts the Distributee's eligible rollover distribution, an annuity contract described in Code Section 403(b), an eligible plan under Code Section 457(b) which is maintained by a state, political subdivision of a state, or any agency or instrumentality of a state or political subdivision of a state and which agrees to separately account for amounts transferred into such plan from the Plan and, effective January 1, 2008, a Roth IRA as defined in Code Section 408A. The definition of Eligible Retirement Plan will also apply in the case of a distribution to a surviving Spouse, or to a Spouse or former Spouse who is the alternate payee under a qualified domestic relation order, as defined in Code Section 414(p).
 - (ii) Distributee. A "Distributee" includes an Employee or former Employee. In addition, the Employee's or former Employee's surviving Spouse and the Employee's or former Employee's Spouse or former Spouse who is the alternate payee under a Qualified Domestic Relations Order are distributees with regard to the interest of the Spouse or former Spouse. Effective January 1, 2007, any Beneficiary who is a "designated beneficiary" for purposes of Code Section 401(a)(9) and the Regulations thereunder (or, to the extent provided in Regulations, that is a trust

- established for the benefit of one or more designated beneficiaries), shall also be a Distribute; provided that in the case of a Distributee who is not the Participant's Spouse or an alternate payee the term Eligible Retirement Plan shall mean only an individual retirement account or an individual retirement annuity that is treated as an inherited account or annuity under Code Section 408(c)(3)(B).
- (iii) Direct Rollover. A "direct rollover" is a payment by the Plan of an "Eligible Rollover Distribution" to the eligible retirement plan specified by a Distributee. An Eligible Rollover Distribution is the distribution of all or any portion of the present value of the Participant's Accrued Benefit, except that an Eligible Rollover Distribution does not include: (a) any distribution that is one of a series of substantially equal periodic payments (not less frequently than annually) made for the life (or life expectancy) of the Participant or the joint lives (or joint life expectancies) of the Participant and the Participant's designated beneficiary, or for a specified period of 10 years or more; (b) any distribution to the extent such distribution is required under Code Section 401(a)(9); or (c) any portion of any distribution that is not included in the recipient's gross income; provided that effective January 1, 2002, the portion that is not included in the recipient's gross income will be considered part of an eligible rollover distribution to the extent transferred to an eligible retirement plan that is described in paragraph (b)(i), or that is a defined contribution plan that agrees to separately account for the non-taxable portion of the distribution and the income attributable thereto.

In the event the amount of an Eligible Rollover Distribution payable to a Distributee who is an Employee or former Employee (but not to a surviving Spouse or alternate payee) prior to the Distributee's Normal Retirement Date exceeds \$1,000, if the Distributee does not elect to have such distribution paid directly to an Eligible Retirement Plan specified by the Distributee in a direct rollover or to receive the distribution directly, then the Plan will pay the distribution in a direct rollover to an Eligible Retirement Plan that is an individual retirement account described in Code Section 408(a) designated by the Administrative Committee.

- 7.4. Rules as to Election and Discontinuance of Optional Forms of Payment. A Participant's election of an optional form of payment specified in Section 7.2 will be subject to the following rules, to the extent appropriate:
 - (a) Written Elections. An election of an optional form of payment must be made on such forms, in such manner and at such times as the Administrative Committee will require.
 - (b) Election Period. A Participant may elect an optional form of payment any time during the 180-day period ending on his Payment Date. The Administrative

Committee will provide each Participant with a written explanation of the normal form of benefit under Section 7.1 Such explanation will advise such Participant:

- (A) of the circumstances in which the normal form of benefit will be provided unless the Participant has elected not to have benefits provided in that form;
- (B) of the availability of such election;
- (C) of the rights of the Spouse, if any; and
- (D) of the financial effect on a Participant's annuity of such election.

Such written notice will be provided at least 30, but not more than 180 days before the Payment Date. However, the written notice may be provided less than 30 days prior to the Payment Date (and may be provided after the Payment Date for Terminations of Employment occurring after December 31, 1996) provided that (i) the Participant is notified in writing of his right to a 30 day period to consider whether to waive the normal form of benefit and consent to a form of distribution other than the normal form of benefit; (ii) the Participant is permitted to revoke his election at any time prior to the Payment Date, or if later, the eighth day following the date he receives the written notice and (iii) no distribution is made to the Participant prior to the Payment Date, or if later, the eighth day following the date he receives the written notice. A Participant may elect to waive in writing (with the applicable spousal consent described in (d), below) his right to a full 30 days to consider whether to elect an optional form of benefit. If the Administrative Committee does not receive the Participant's election to waive the normal form of benefit by the first business day following the expiration of the 30 day period (or, if later, the Payment Date) then the Participant will be deemed to have consented to the normal form of benefit.

- (c) Revocation. Any election to receive an optional form of benefit pursuant to Section 7.2 may be revoked by a Participant by notifying the Administrative Committee in such manner as the Administrative Committee may require at any time during the 180-day period ending on the Payment Date. However, if the written explanation required in subsection (b) is provided less than 30 days before the Payment Date or after the Payment Date, a Participant may revoke his election any time up to the Payment Date, or if later, the eighth day following the date the written explanation is received by the Participant. Once revoked, an election for an optional form of benefit may again be made by submitting a new election to the Administrative Committee within the times specified in subsection (b).
- (d) Spousal Consent. Any election by a married Participant of an optional form of benefit described in Section 7.2 (other than Section 7.2(a) where the Spouse is the designated Beneficiary) must be consented to by the Participant's Spouse unless an optional form of payment is elected that provides a benefit to the Spouse that is equal to or greater than the Joint and 50 Percent Survivor Annuity. The consent

of a Participant's Spouse required under this subsection (d) will be effective only if such consent:

- (i) is in writing;
- (ii) acknowledges the effect of the Participant's designation of a Beneficiary other than the Spouse and the identity of such Beneficiary; and
- (iii) is witnessed by a notary public; provided, however, such consent will be deemed to have been granted where it is established to the satisfaction of the Administrative Committee that the consent of the Spouse cannot be obtained because (1) there is no Spouse, (2) the Spouse cannot be located, or (3) there exist such other circumstances as may be prescribed by regulations under ERISA or the Code. Consent of the Spouse to a designated Beneficiary and any contingent Beneficiary is irrevocable. Each change from one non-Spouse Beneficiary to another, or change of contingent Beneficiary, requires a new consent from the Spouse.
- (e) Death before Retirement. If a Participant who had elected an optional form of payment providing for a greater Death Benefit than would otherwise be paid under Article VI dies before his Payment Date, such optional form of payment will be in effect and any Death Benefit under Article VI will be canceled.
- (f) Death of Beneficiary before Payment Date. If the Beneficiary of a Participant who has elected an optional form of payment dies before the Participant's Payment Date, the optional form of payment will automatically be canceled and the Participant's Pension Benefit will be paid to him in the normal form unless a new election can be and is made by the Participant pursuant to the foregoing provisions of this Section.
- (g) Change of Form or Beneficiary after Option Effective. Except as otherwise provided below, a Participant may not change his optional form of payment or designate a new Beneficiary after his Payment Date (or, if later, after the last day on which he is permitted to revoke his election pursuant to subsection (c)), even if his Beneficiary dies or he is divorced from his Beneficiary. However, a Participant may designate a new Beneficiary to receive any unpaid portion of a Ten-Year Certain and Life Annuity option described in Section 7.2(c). If no Beneficiary is designated on the Participant's death, such unpaid portion will be payable to the Beneficiary specified under Section 6.7.
- (h) Special Rules for Retroactive Payment Dates. Effective January 1, 2004, if a Participant does not receive the written election described in subsection (a) until after the date that would otherwise be his Payment Date, he may elect (in accordance with procedures established by the Administrative Committee) to receive his benefit calculated as of the original Payment Date, provided that (A) the Participant receives a supplemental payment equal to the monthly annuity payments that would have been received had payment actually commenced on the

retroactive Payment Date, with an appropriate interest adjustment; (B) the person who is the Participant's Spouse on the date payment actually commences (if different from the person who is his Spouse on his Payment Date) consents to any optional form of benefit in accordance with subsection (d) (including an optional form described in Section 7.2(a) unless the survivorship payments to be received by the Spouse under such form are at least equal to the survivorship payments that would be received under one of the optional forms described in Section 7.2(a) with a Payment Date after the date the explanation was furnished; and (C) if the date on which payments commence is more than one year after the Payment Date, the benefit payments satisfy Section 13.12 based both upon the Payment Date and the date on which payments actually commence, in the latter case treating the supplemental payments (including the interest adjustment) as an additional benefit payment in the year of commencement.

- 7.5. Special Payment Limitations. Payment of benefits under the Plan to a Participant will commence not later than the sixtieth day after the end of the Plan Year in which the Participant attains his Normal Retirement Date or incurs a Termination of Employment, whichever occurs later. The following provisions will supersede any other provisions of this Plan:
 - (a) Maximum Payment Period. No optional form of payment will be permitted that causes the Pension Benefit to be paid over a period extending beyond the life or life expectancy of the Participant or the combined lives or life expectancies of the Participant and his Beneficiary as determined in accordance with applicable mortality tables contained in applicable federal regulations.
 - (b) Five-Year Limit. If the Participant dies without a Spouse and prior to commencement of benefits, no form of payment under the Plan will permit payments to continue beyond five years after the Participant's death except to the extent permitted by Code Section 401(a)(9).
 - (c) Age 70½ Benefit Payments. A Participant's Pension Benefit will commence as of the April 1 following the later of the Plan Year in which he attained age 70½ or the Plan Year in which he incurs a Termination of Employment; provided that if the Participant owned 5% or more of the outstanding voting stock of the Company or 5% or more of the value of all classes of outstanding stock at any time during the Plan Year in which he attained age 70½ his Pension Benefit will commence as of the following April 1 regardless of whether he is an Employee as of such date.

Notwithstanding anything in the Plan or this Section to the contrary, the provisions of the Plan will be interpreted, construed and administered in a manner that complies with, and benefits shall be paid in accordance with the requirements of Code Section 401(a)(9) and regulations thereunder, including the minimum distribution incidental benefit requirement of Code Section 401(a)(9)(G). Minimum distributions will be determined in accordance with Treasury Regulations Section 1.401(a)(9)-2 through Section 1.401(a)(9)-6, which are

incorporated herein by reference. The provisions of such final regulations will control over any provision of the Plan to the contrary.

- 7.6. Effect of Prior Lump Sums on Pension Benefits. Except as otherwise provided in an applicable Supplement, a Participant who is rehired after receiving his Pension Benefit in a Lump Sum will be entitled to a Pension Benefit upon Termination of Employment equal to his Accrued Benefit (taking into account all of his Years of Service), reduced by the portion of such Accrued Benefit represented by such Lump Sum. Such reduction will occur prior to any reductions for commencement of the Pension Benefit prior to the Participant's Normal Retirement Date and prior to the conversion of the Accrued Benefit into a form other than a Single Life Annuity. If the Lump Sum was not more than \$5,000, only Years of Service after rehire will be taken into account and no reduction will be made to the Accrued Benefit based on such Years of Service.
- 7.7. Effect of Participant Resuming Employment after Benefits Commence. If a Participant whose Pension Benefit has commenced is rehired by the Employer, the following rules will apply:
 - (a) Resumption of Employment prior to Age 65. If a Participant is rehired by Employer before his 65th birthday, his Pension Benefit payments will be discontinued and will not be paid or accrued during the period of such reemployment, his previous election of form of payment will be canceled, and he will have all Years of Service he had at the time of his Termination of Employment reinstated. Upon his subsequent Termination of Employment, his eligibility for a benefit and the amount of the benefit will be determined, calculated and paid as if he then first incurred a Termination of Employment based upon both reinstated Years of Service and any additional Years of Service credited, but such benefit will be actuarially reduced to recognize any Pension Benefit payments he received prior to his suspension. In no event will a Participant's Pension Benefit at his subsequent Termination of Employment be less than his benefit at his earlier Termination of Employment. However, if a Participant who is rehired as described above subsequently reaches his 65th birthday and is employed at a rate of fewer than ten hours per week, he is entitled to receive a Pension Benefit determined under Section 4.1. Such payments will continue every month thereafter until his rate of employment equals or exceeds ten hours per week, at which time his Pension Benefit payments will be suspended under the terms and conditions described below.
 - (b) Resumption of Employment after Age 65. If a participant is rehired by the Employer after his 65th birthday, at a rate of at least ten hours per week, his Pension Benefit payments will be discontinued and will not be paid and no payment obligation will accrue during the period of such reemployment. Such suspension of benefits will be done in accordance with Department of Labor Regulation §2530.203-3 and will include the notice described below. Such Participant will thereafter continue to accrue further benefits, and his previous election of form of payment will remain in effect and will determine what Death Benefit, if any, is payable under the Plan. Upon the Participant's subsequent

Termination of Employment, he will resume receiving payments in the same form as he elected at his earlier Termination of Employment, but such benefit amount will be increased to reflect the value of any additional Accrued Benefit. If a Participant who has his benefits suspended under this subsection (b) elected a Ten-Year Certain and Life Annuity, payments that are not made due to the suspension do not count against the 120-payment guarantee. If a Participant is rehired by the Employer after his 65th birthday and his rate of employment is fewer than ten hours per week, he will receive the same type and amount of his benefit payment which he was entitled to receive preceding his reemployment during such period of reemployment. Such payments will continue every month thereafter until his rate of employment equals or exceeds ten hours per week, at which time his Pension Benefit payments will be suspended as described above. If a Participant continues in employment with the Employer after his 65th birthday at a rate of at least ten hours per week, his Pension Benefit payments will not commence during the period of such employment. Such suspension of benefits will be done in accordance with Department of Labor Regulation §2530.203-3 and will include the notice described below. Such Participant will continue to accrue further benefits under the Plan. During such employment the Provisions of Article VI will remain in effect and will determine what Death Benefit is payable under the Plan. If a Participant continues in employment with the Employer after his 65th birthday and his rate of employment is fewer than ten hours per week, he will receive a Pension Benefit under Section 4.1 under the same terms and conditions as a Participant who incurred a Termination of Employment. Such payments will continue every month thereafter until his rate of employment equals or exceeds ten hours per week, at which time his Pension Benefit payments will be suspended as described above.

- (c) Notice of Benefit Suspension. If a Participant's benefits are to be suspended after age 65 due to either reemployment or continued employment, the Administrative Committee will notify the Participant by personal delivery or first class mail during the first calendar month in which the Plan withholds payments, that benefits are suspended. The notice will contain the following information:
 - (i) a general description of the reasons why payments are suspended;
 - (ii) a general description of Plan provisions relating to the suspension of benefits;
 - (iii) a copy of such Plan provisions;
 - (iv) a statement that applicable Department of Labor Regulations may be found in Section 2530.203-3 of the Code of Federal Regulations;
 - (v) a statement that a review of the suspension may be requested under the Plan's claims procedure; and

(vi) if the Plan requires a benefit resumption notice or verification by the Participant that his benefits should not be suspended, the procedure and forms for such purposes.

The Plan will adopt a procedure whereby a Participant may request a determination of whether specific contemplated employment after age 65 will result in the suspension of benefits.

7.8. Special One Time Election for Immediate Payment of Certain Deferred Vested Benefits.

- Each Eligible Participant (as hereinafter defined) shall be have the right to elect to begin receiving his Pension Benefit notwithstanding the fact that he has not attained his Early Retirement Age, and to receive his Pension Benefit in the form of a Lump Sum or Annuity, during an election period to be determined by the Plan Administrator, which election period shall be not fewer than thirty (30) days in length and shall end not later than December 31, 2012. Such election shall be offered only once, during the election period described above, shall not thereafter be available, and shall not be considered part of the Participants' Accrued Benefit after the end of the election period.
- (b) For purposes of this Section 7.8, the term "Eligible Participant" means any Participant who is identified by the Plan Administrator as having terminated employment on or prior to June 30, 2011, with a right to either an early retirement benefit or a deferred vested benefit pursuant to Section 4.3 or 4.5, who is not actively employed by the Company during the election period, whose benefit has not commenced (and who has not submitted an application for an early retirement pension) by the first day of the election period, and who (i) has not attained his Normal Retirement Age by the first day of the election period, (ii) is not a Spouse or Beneficiary of a deceased Participant, (iii) did not have a benefit under the BOC Group Cash Balance Retirement Plan, as described in Supplement J, regardless of whether such benefit has been cashed out, and (iv) did not terminate employment with a deferred vested benefit under the Immuno-U.S., Inc. Defined Benefit Plan, prior to the merger of such plan into the Plan effective December 31, 2003, as described in Supplement L. The list of Eligible Participants compiled by the Plan Administrator, who meet the above requirements and who are sent election materials, shall be final and binding, and no Participant who is not included on such list shall be an Eligible Participant even if he would otherwise be described in the preceding sentence unless otherwise determined by the Plan Administrator in its sole discretion, which shall be exercised on a uniform and non-discriminatory basis.
- (c) Each Eligible Participant who elects the immediate commencement of his Pension Benefit during the election period may also elect to waive the normal form of benefit and receive his Pension Benefit in the form of a Lump Sum equal to the present value of his Pension Benefit, payable not later than December 31, 2012. The Pension Benefit of an Eligible Participant who elects immediate commencement but who does not elect a Lump Sum, will be paid in the normal

form specified by Section 7.1 with a Payment Date of December 1, 2012, subject to the following:

- (i) If the Eligible Participant will have attained the age at which he is entitled to commence payment of his Pension Benefit without regard to this Section 7.8 on or before November 30, 2012, he may elect, in lieu of the normal form, to receive his Pension Benefit in any of the other optional forms provided by Section 7.2.
- (ii) If the Eligible Participant will not have attained the age at which he is entitled to commence payment of his Pension Benefit without regard to this Section 7.8 on or before November 30, 2012, but is married on his Payment Date, he may elect payment in the form of a 75% Joint and Survivor Annuity (but no other options).

For purposes of this Section 7.8, all Pension Benefits shall be calculated as otherwise provided in the Plan, except that (A) the Benefit Reduction Factors used to calculate the Pension Benefit of an Eligible Participant who does not elect a Lump Sum, but who has not yet attained the minimum age shown on the applicable Table in Supplement A, shall be calculated using the same actuarial assumptions used in calculating the applicable Table, and (B) the Lump Sum benefit shall be equal to the Actuarial Equivalent of the Participant's Normal Retirement Benefit paid at Normal Retirement Age, without any adjustment for the value of any subsidy that would be included if the Pension Benefit were paid in the form of an immediate annuity, regardless of whether the Participant would be eligible for an immediate annuity without regard to this Section 7.8.

- (d) If an Eligible Participant who is eligible for a Disability Pension pursuant to Section 4.4 elects immediate payment, his accrual of additional Years of Service during his period of Disability shall terminate on November 30, 2012, or such earlier date on which it is determined that he had recovered from the Disability. If an Eligible Participant notifies the Plan Administrator not later than the end of the election period that he has been determined to be disabled by the Social Security Administration, or has applied for such determination, then the Plan Administrator may pay such Eligible Participant's Pension Benefit and subsequently adjust the amount of the Pension Benefit based upon any additional Years of Service to which he may be entitled, but an Eligible Participant who elects a Lump Sum and does not so notify the Plan Administrator shall not be entitled to any additional benefit even if it is subsequently determined that he was disabled.
- (e) All elections pursuant to this Section 7.8 shall be made in accordance with the requirements of Section 7.4 (treating the Lump Sum option as if it were an optional form of benefit provided by Section 7.2), except that the election period described above shall be substituted for the 180 day period referred to in Section 7.4, and the Plan Administrator shall make such other changes to the procedures described in Section 7.4 as may be consistent with the intent of this Section 7.8

and applicable law. The Plan Administrator shall have the full authority and discretion to make all other determinations necessary or appropriate to implement the election provided for in this Section 7.8, in accordance with Article IX. Without limiting the generality of the foregoing, the Plan Administrator may permit Eligible Participants to elect to receive a Lump Sum payment, and make such payments by December 31, 2012, notwithstanding the fact that the Plan Administrator is still in the process of reviewing the Eligible Participant's service or compensation history or other factors relevant to the amount of his Pension Benefit, and make appropriate corrections to the amount so paid in years subsequent to 2012, may exclude groups of Participants other than those listed in Section 7.8(b) from the definition of Eligible Participant, on a nondiscriminatory basis, if it determines that the Plan's records are not sufficiently reliable to prepare election materials on a timely basis, or for similar reasons, and may determine the manner in which the election shall be made with respect to Eligible Participants whose Pension Benefit is or may be subject to a Qualified Domestic Relations Order."

ARTICLE VIII. Plan Committees

- 8.1. Membership of Administrative and Investment Committees. The Administrative Committee, consisting of at least three persons, will be appointed by the Compensation Committee of the Board of Directors. The Investment Committee, consisting of at least three persons, will be appointed by the Board of Directors or its delegate. The Secretary of the Company will certify to the Trustee from time to time the appointment to (and termination from) office of each member of the Administrative Committee and the Investment Committee and the persons, if any, who are selected as secretaries of the Administrative Committee and the Investment Committee by the members of such committees. The appointment of a member of either Committee and acceptance of such appointment by any person constitutes an agreement by and between the Company and such Committee member that the member, acting in concert with the other Committee members, will have and will exercise the powers and duties described herein, including, with respect to the Administrative Committee, the power and duty to interpret this Plan and determine the benefits to which Participants are entitled hereunder.
- **8.2.** Administrative Committee Powers and Duties. The Administrative Committee shall be the "plan administrator" for purposes of Code Section 414(g) and the "administrator" for purposes of Section 3(16)(A) of ERISA and will have such powers and duties necessary to discharge its duties hereunder, including, but not limited to, the following:
 - (a) Within its complete and unfettered discretion to construe and interpret the Plan and Trust Agreement provisions and to resolve all questions arising under the Plan including questions of Plan participation, eligibility for Benefits and the rights of Employees, Participants, Beneficiaries and other persons to benefits under the Plan and to determine the amount, manner and time of payment of any benefits hereunder;
 - (b) To prescribe procedures, rules and regulations to be followed by Employees, Participants, Beneficiaries and other persons or to be otherwise utilized in the efficient administration of the Plan consistent with the Trust;
 - (c) To make determinations as to the rights of Employees, Participants, Beneficiaries and other persons to benefits under the Plan and to afford any Participant or Beneficiary dissatisfied with such determination with rights pursuant to a claims procedure adopted by the Administrative Committee;
 - (d) To settle or compromise claims against the Plan by Participants, Beneficiaries and other persons;
 - (e) To enforce the Plan in accordance with the terms of the Plan and the Trust and to enforce its procedures, rules and regulations;
 - (f) To be responsible for the preparation and maintenance of records necessary to determine the rights and benefits of Employees, Participants and Beneficiaries or

- other persons under the Plan and the Trust and to request and receive from the Participating Employers such information necessary to prepare such records;
- (g) To prepare and distribute in such manner as it deems appropriate and to prepare and file with appropriate government agencies information, disclosures, descriptions and reporting documents regarding the Plan, and in the preparation and review of such reports the Administrative Committee is entitled to rely upon information supplied to it by the Employees, accountants, counsel, actuaries, the Investment Managers and any insurance institutions described in the Trust Agreement;
- (h) To appoint or employ individuals to assist in the administration of the Plan and other agents (corporate or individual) that the Administrative Committee deems advisable, including legal counsel and such clerical, medical, accounting, auditing, actuarial and other services as the Administrative Committee may require in carrying out the provisions of the Plan. However, no agent except an Investment Manager or fiduciary named in the Plan will be appointed or employed in a position that would require or permit him: (i) to exercise discretionary authority or control over the acquisition, disposition or management of Trust assets; (ii) to render investment advice for a fee; or (iii) to exercise discretionary authority or responsibility for Plan administration;
- (i) To cause to be prepared and to cause to be distributed, in such manner as the Administrative Committee determines to be appropriate, information explaining the Plan and Trust;
- (j) To furnish to the Participating Employers upon request such annual or other reports with respect to the administration of the Plan as are reasonable and appropriate;
- (k) To receive, review and keep on file (as it deems convenient or proper) reports of the financial condition, receipts and disbursements, and assets of the Trust;
- (l) To determine the method by which all notices and other documents required or permitted to be given to a Participant, Beneficiary or other person under the Plan or any applicable law shall be given, and the method by which Participants and Beneficiaries may exercise any elections permitted by the Plan, which methods may include the use of e-mail, interactive internet sites, voice response systems, and other electronic media to the extent permitted by applicable regulations;
- (m) To amend the Plan to the extent provided in Section 11.1(d); and
- (n) To discharge all other duties set forth in the Plan.

Except as otherwise specifically provided herein, the Administrative Committee has no power to add to, subtract from or modify any of the terms of the Plan, nor to change or add to any benefits provided by the Plan, nor to waive or fail to apply any requirements of eligibility for benefits under the Plan; provided, however, that nothing contained herein shall be construed to limit the

Administrative Committee's authority to interpret and construe the Plan. Procedures and policies adopted by the Administrative Committee may be inconsistent with any provision of the Plan that is ministerial or administrative in nature, and shall be deemed an amendment to the Plan to the extent of the inconsistency. The authority of the Administrative Committee may also be exercised in routine matters by the Corporate Vice President-Human Resources of the Company or persons acting under his authority, and any action taken by any such person within the apparent scope of his authority shall be presumed authorized and binding on all Participants, subject to the review of the Administrative Committee.

- **8.3. Investment Committee Powers and Duties**. The Investment Committee has such powers necessary to discharge its duties hereunder, including but not limited to the following:
 - (a) To establish and from time to time revise the investment policy of the Plan, to communicate and consult with the Company, the Administrative Committee and the Trustee and any Investment Manager or insurance institution regarding the investment policy applicable to the Plan as a whole or as to any individual investment fund;
 - (b) To supervise the performance by the Trustee and any Investment Manager or insurance institution regarding their responsibilities under the Plan and Trust. The Investment Committee will review and analyze performance information supplied by the Trustee and the Investment Managers or insurance institutions to the Investment Committee and/or any such performance information obtained independently by the Investment Committee and will report the results of such analysis to the Board of Directors or its delegate from time to time in such form and with such degree of frequency as the Investment Committee will determine proper. Such responsibilities of the Investment Committee with respect to supervision, review and analysis will be performed no less frequently than once each Plan Year and will ordinarily not be required more frequently than once each calendar quarter. The Trustee, Investment Managers and insurance institutions have been allocated the responsibility for day-to-day investment management of the Plan and Trust, and the responsibilities of the Investment Committee hereunder are not intended to relieve the Trustee, Investment Managers or insurance institutions of such ongoing investment management responsibilities;
 - (c) To instruct the Trustee, the Investment Managers and insurance institutions with respect to the proper application of contributions made under the Plan;
 - (d) To determine the proper allocation of investment responsibilities with respect to the assets of the Plan between the Trustee and any Investment Manager or insurance institution acting hereunder or under the terms of the Trust and to allocate fiduciary responsibilities among these parties;
 - (e) To the extent not provided to the contrary in the Trust Agreement, to appoint the Trustee and any Investment Managers or insurance institutions, to direct the establishment of any investment fund and to remove the Trustee and any

- Investment Managers or insurance institutions or appoint additional Trustees, Investment Managers or insurance institutions:
- (f) To review any accounts submitted by the Trustee and any Investment Managers or insurance institutions and to report to the Board of Directors or its delegate with respect to any such accounts;
- (g) Following the Administrative Committee's determination of the benefit rights of any Participant or Beneficiary, to collect information concerning such benefits and authorize and direct the Trustee with respect to the commencement, modification or cessation of such benefit payments;
- (h) To supervise the performance of fiduciary responsibilities by others, including the Trustee and any Investment Managers;
- (i) To appoint and utilize the services of administrative staff employees of the Company and the other Participating Employers for the performance of duties delegated to the Investment Committee hereunder and to rely upon information received from such staff employees; provided that in both cases the Investment Committee reasonably believes the performance of such services and the preparation of such information is within the competence of such staff employees;
- (j) To furnish to the Participating Employers, upon reasonable request, such annual or other reports as the Participating Employers deem necessary regarding the administration of the Plan; and
- (k) To employ reputable agents and to delegate to them any of the administrative powers or duties imposed upon the Investment Committee or the Participating Employers.
- **8.4. Conflicts of Interest**. No member of the Administrative Committee or the Investment Committee will participate in any action on matters involving solely such member's rights or benefits as a Participant under the Plan.
- **8.5. Compensation; Reimbursement**. No member of the Administrative Committee or the Investment Committee will receive compensation for his services, but the Participating Employers will reimburse him for any necessary expenses incurred in the discharge of his duties.
- **8.6. Standard of Care.** The Administrative Committee and the Investment Committee will perform their duties under this Plan in accordance with the terms of this document and the Trust Agreement solely in the interest of the Participants and for the exclusive purposes of providing retirement benefits to Participants and defraying the reasonable expenses of Plan administration and operation. The Administrative Committee and the Investment Committee will also perform their duties under this Plan with the care, skill, prudence and diligence under the circumstances then prevailing that a prudent man, acting in a like capacity and familiar with such matters, would use in the conduct of an enterprise of a like character and with like aims.

- **8.7. Action by Committees**. Action by each Plan Committee (i.e., the Administrative Committee and the Investment Committee) is subject to the following special rules:
 - (a) Each Committee may act by meeting or by document signed without meeting and documents may be signed through the use of a single document or concurrent documents;
 - (b) Each Committee will act by a majority, and such action will be as effective as if such action had been taken by all Committee members, provided that by majority action one or more Committee members or other persons may be authorized to act with respect to particular matters on behalf of all Committee members;
 - (c) Each Committee may, but is not required to, select a secretary, who may but need not be a Committee member, and the certificate of such secretary that the Committee has taken or authorized any action will be conclusive in favor of any person relying upon such certificate; and
 - (d) Each Committee may act through agents or other delegates and may retain legal counsel, auditors or other specialists to aid in the Committee's performance of its responsibilities.
- **8.8.** Resignation or Removal of Committee Member. Any person serving as an Administrative Committee member may resign from such Committee at any time by written notice to the Compensation Committee of the Board of Directors or may be removed by the Compensation Committee at any time by written notice to such member. Any person serving as an Investment Committee member may resign from such Committee at any time by written notice to the Board of Directors or its delegate or may be removed by the Board of Directors or its delegate at any time by written notice to such member. The Compensation Committee will fill any vacancy in the membership of the Administrative Committee as soon as practicable. The Board of Directors or its delegate will fill any vacancy in the membership of the Investment Committee as soon as practicable. Until any such vacancy is filled, the remaining members of the applicable Committee may exercise all of the powers, rights and duties conferred on such Committee.
- 8.9. Uniform Application of Rules by Administrative Committee. The Administrative Committee will apply all rules, regulations, procedures and decisions uniformly and consistently to all Employees and Participants similarly situated. Any ruling, regulation, procedure or decision of the Administrative Committee which is consistent with the provisions of the Plan or the Trust will be conclusive and binding upon all persons affected by it. There will be no appeal of any ruling by the Administrative Committee which is within its authority, except as provided in Section 8.10 below. When making a determination or a calculation, the Administrative Committee is entitled to rely on information supplied by the Participating Employer, Trustee, Investment Managers, insurance institutions, accountants and other professionals including legal counsel for the Company.
- **8.10. Claims Procedure**. Each person entitled to benefits under the Plan (the "Applicant") must submit a written claim for benefits to the Administrative Committee. Such

claim shall be filed not more than one year after the Applicant knows (or with the exercise of reasonable diligence would know) of the existence of a basis for a claim; provided that nothing herein shall be construed to permit the forfeiture of a Participant's benefit for failure to file a timely application for such benefit; and provided further that the Administrative Committee may waive or extend such requirement in its sole discretion. If a claim for benefits by the Applicant is denied, in whole or in part, the Administrative Committee will furnish the Applicant, within 90 days after receipt of such claim (or within 180 days after receipt if special circumstances require an extension of time), a written notice which specifies the reason for the denial, refers to the pertinent provisions of the Plan on which the denial is based, describes any additional material or information necessary for properly completing the claim and explains why such material or information is necessary, and explains the claim review procedures of this Section 8.10. Such notice will further describe that the Applicant has a right to bring a civil action under Section 502 of ERISA if his claim is denied after an appeal and review. Any Applicant whose claim is denied under the provisions described above, or who has not received from the Administrative Committee a response to his claim within the time periods specified in the provisions described above may request a review of the denied claim by written request to the Administrative Committee within 60 days after receiving notice of the denial. In connection with such request, the Applicant or his authorized representative may review pertinent documents and may submit issues and comments in writing. If such a request is made, the Administrative Committee will make a full and fair review of the denial of the claim and will make a decision not later than 60 days after receipt of the request, unless special circumstances (such as the need to hold a hearing) require an extension of time, in which case a decision will be made as soon as possible but not later than 120 days after receipt of the request for review, and written notice of the extension will be given to the Applicant before the commencement of the extension. The decision on review will be in writing and will include specific reasons for the decision and specific references to the pertinent provisions of the Plan on which the decision is based. Such notice will further describe that the Applicant has a right to bring a civil action under Section 502 of ERISA if his claim is denied after an appeal and review. No person entitled to benefits under the Plan will have any right to seek review of a denial of benefits, or to bring any action to enforce a claim for benefits, in any court prior to his filing a claim for benefits and exhausting all of his rights under this Section 8.10, or more than six months after receipt of the decision on review. Although not required to do so, an Applicant may choose to state the reason or reasons he believes he is entitled to benefits, and may choose to submit written evidence, during the initial claim process or review of claim denial process. However, failure to state any such reason or submit such evidence during the initial claim process or review of claim denial process, or by written notice to the Administrative Committee within 60 days of the date of the decision on the review of the claim denial, will permanently bar the Applicant, and his successors in interest, from raising such reason or submitting such evidence in any forum at any later date. An Applicant whose claim is denied initially or on review is entitled to receive, on request and free of charge, reasonable access to, and copies of, all documents, records, and other information relevant to such claim for benefits.

8.11. Correction of Errors. In the event that the Administrative Committee discovers that an error has been made in the calculation of a benefit, it will correct the error as soon as is administratively feasible. In the event of an underpayment, the Administrative Committee will either pay the amount of the underpayment in a single sum or will increase future monthly payments to the extent necessary to pay the underpayment within a reasonable period of time,

with interest at a reasonable rate not less than the interest rate specified in Section 2.2(a)(ii). If an overpayment was made, the Committee will reduce future monthly payments to the extent necessary to recover the overpayment within a reasonable period of time, using life expectancy as of the calculation date. If an overpayment was made in a Lump Sum, the Administrative Committee will seek reimbursement; provided, that if the Administrative Committee determines that the expense or burden of seeking recovery would be greater than the likely recovery, it may exercise discretion in determining not to pursue a recovery.

ARTICLE IX. Trust, the Trustee and Plan Financing

- **9.1. Trust Agreement**. The Trust Agreement provides for the administration of the Trust. The Trust Agreement, as amended, will continue in force and will be deemed to form a part of this Plan, and any and all rights or benefits which may accrue to any person under this Plan will be subject to all the terms and provisions of the Trust Agreement.
- **9.2. Selection of Trustee**. As provided in the Trust Agreement, the Investment Committee will have the power to remove the Trustee and to appoint a successor Trustee.
- **9.3. Trustee's Duties**. The powers, duties and responsibilities of the Trustee will be as stated in the Trust Agreement, and nothing contained in the Plan either expressly or by implication will be deemed to impose any additional powers, duties or responsibilities upon the Trustee. The foregoing notwithstanding, the Trustee will have no investment authority not specifically granted to the Trustee by the terms of the Trust and the Trustee will be bound by the instructions of the Investment Committee or Investment Manager as provided in Article VIII. All Participating Employer Contributions will be paid into the Trust and all benefits payable under the Plan will be paid from the Trust. No Participating Employer will have any rights or claims of any nature in or to the assets of the Trust Fund except the right to require the Trustee to hold, use, apply and pay such assets in its possession, in accordance with the directions of the Investment Committee for the exclusive benefit of the Participants, Spouses and Beneficiaries, except as otherwise provided in the Plan.
- **9.4. Trust Income**. The net income derived from the Trust will be accumulated and will from time to time be invested as a part of the Trust Fund.
- **9.5. Expenses**. Unless paid by the Participating Employers, all costs and expenses incurred in connection with the general administration of the Plan and Trust will be paid by the Trust.
- **9.6. Trust Entity**. The Trust will be a separate entity aside and apart from the Participating Employers or their assets. The Trust and the corpus and income thereof will in no event and in no manner whatsoever be subject to the rights or claims of any creditor of any Participating Employer.
- **9.7. Funding Policy**. The Investment Committee will establish and direct the implementation of a funding policy and method for the Plan which will be consistent with the objectives of the Plan and with the minimum funding standards established under Code Section 412.
- **9.8.** Participating Employer Contributions. Each Participating Employer will make contributions to the Trust to fund benefits of the Plan in such amounts and at such times as the Investment Committee, in accordance with the funding policy and method of the Plan, will from time to time direct; provided that any such contribution will be made not later than the due date for the Participating Employer's federal income tax return (including extensions) for the taxable year and Plan Year for which such contribution is made. All Participating Employer

Contributions are conditioned upon the qualification of the Plan under Code Section 401(a) and upon the deductibility of such contributions by the Participating Employer under Code Section 404. Participating Employer Contributions can be made in cash or in securities of the Company; provided that in no case will the Plan acquire or hold securities of the Company if the aggregate fair market value of the securities exceeds ten percent of the fair market value of the Plan assets or if the securities fail to qualify as "Qualifying Participating Employer Securities" under Section 407 of ERISA.

- **9.9. Forfeitures**. Forfeitures of benefits under the Plan arising for any reason will be applied to reduce the cost of the Plan and will not be used to increase the benefits under the Plan otherwise payable to or on behalf of Participants.
- **9.10. Exclusive Benefit of Participants**. Except to the extent provided below, all Participating Employer Contributions under the Plan will be paid to the Trustee and deposited in the Trust Fund and will be held, managed and distributed solely in the interest of the Participants, their Spouses and Beneficiaries for the exclusive purposes of providing benefits to such persons and paying all costs and expenses incurred in connection with the general administration of the Plan and Trust, to the extent such costs and expenses are not paid by the Participating Employers. Notwithstanding the foregoing, Participating Employer Contributions and the earnings thereon may be applied as follows:
 - (a) Non-Deductible Contribution Reversion. Participating Employer contributions are conditioned upon the deductibility of such contributions and if, and to the extent, deduction of a Participating Employer Contribution under Code Section 404 is disallowed, such Participating Employer Contributions, adjusted for any investment losses, will be returned to the Participating Employers within one year after the disallowance of the deduction, provided that this reversion provision will not be applicable to the extent that it is determined by the Administrative Committee that such reversion will adversely affect the qualified status of the Plan;
 - (b) Mistake of Fact Reversions. If, and to the extent, a Participating Employer Contribution is made through mistake of fact, such Participating Employer Contribution and any earnings on such contributions will be returned to the Participating Employers within one year of the payment of the contribution;
 - (c) Excess Assets Reversions. If any amounts arising out of the variation between expected actuarial requirements and actual requirements remain in the Trust Fund after termination of the Plan and if all liabilities of the Plan to persons entitled to benefits under the Plan have been satisfied in accordance with applicable law, such remaining amounts will be distributed to the Participating Employers in such amounts as the Investment Committee in its sole discretion will determine, provided such distribution is in accordance with applicable law.
 - (d) Exercise of Plan Sponsor or Settlor Authority. Notwithstanding the foregoing, the provisions of this Section 9.10 will not be applicable with respect to Plan design

decisions including decisions made pursuant to Section 1.3, Article VII or Article VIII, or with respect to exercises of any other Plan sponsor or settlor authority.

9.11. Benefits Payable Only from Trust Fund. All benefits provided by the Plan will be paid solely from the Trust Fund, and neither any Participating Employer nor any agent or representative of a Participating Employer will be liable in any manner for any such benefits.

ARTICLE X. Adoption and Withdrawal from Plan

- 10.1. Procedure for Adoption. Any Participating Employer may adopt the Plan for the benefit of its Employees as of a date specified. Any Employer that employs Participants shall be deemed to have adopted the Plan unless otherwise determined by the Administrative Committee. Notwithstanding any term or provision of the Plan to the contrary (including, but not limited to, terms and conditions concerning Years of Service, Compensation and amount of Pension Benefits), the terms and provisions as may be imposed with respect to such Employees in an applicable Supplement will govern. A partnership, limited liability company, or other noncorporate entity may be a Participating Employer, subject to such conditions as may be required by the Administrative Committee.
- **10.2. Procedure for Withdrawal**. Any Participating Employer may, with the consent of the Company, and subject to such conditions as may be imposed by the Company, terminate its adoption of the Plan.

ARTICLE XI. Amendment and Termination

11.1. Amendments.

- (a) Power to Amend. The Company will have the right at any time to amend in whole or in part any or all of the provisions of the Plan except as expressly set forth below:
 - (i) no amendment will increase the duties or liabilities of the Trustee without its written consent;
 - (ii) no amendment will have the effect of vesting in any Employer any interest in any funds, securities or other property subject to the terms of the Plan and Trust;
 - (iii) no amendment will authorize or permit at any time any part of the corpus or income of the Trust Fund to be used for or diverted to purposes other than for the purposes specified in the Plan, except as permitted by Section 11.14;
 - (iv) no amendment will reduce a Participant's Accrued Benefit, or the Non-forfeitable portion thereof, subject to Section 11.1(c);
 - (v) effective January 1, 2008, no amendment will shall increase the Plan's liabilities in violation of Section 13.13(b).

The Company's authority to amend the Plan has been delegated to the Administrative Committee to the extent provided in Section 11.1(d). The authority to amend the Plan in any respect (whether or not such amendment is within the authority delegated to the Administrative Committee) may also be exercised by the Board of Directors or any other person to whom the Board delegates such authority.

- (b) Effect of Amendment. If a person is not an Eligible Employee on or after the effective date of any amendment to the Plan, the amendment will be deemed as having no effect on the amount of such person's benefits unless the amendment specifically provides otherwise.
- (c) Restriction on Amendment. Except as otherwise permitted by Code Section 411 or regulations or other administrative guidance issued thereunder,
 - no amendment to the Plan shall reduce a Participant's Accrued Benefit as of the effective date of the amendment.
 - (ii) any amendment that affects the determination of the Non-forfeitable portion of a Participant's Accrued Benefit shall apply to a Participant who as of the date that is sixty days after the effective date of the amendment

has completed at least three Years of Vesting Service only if the effect of the amendment is to increase the Non-forfeitable portion of the Participant's Accrued Benefit (including the portion accrued after the effective date of the amendment). The preceding language concerning an amendment to the Plan's vesting schedule will also apply when a Plan with different requirements for non-forfeitability is merged into the Plan.

- (iii) the Plan will not be amended so as to eliminate or reduce an early retirement benefit or a retirement-type subsidy, or eliminate an optional form of payment, with respect to a Participant's Accrued Benefit as of the effective date of the Amendment; provided, however, the retirement-type subsidy need only be preserved for those Participants who before or after the date of the amendment (but before incurring a Termination of Employment) satisfy the eligibility requirements for the subsidy prior to the amendment. The foregoing limitations do not apply to benefit accrual occurring after the date of the amendment.
- (d) Authority of Administrative Committee. The Administrative Committee has been delegated the authority of the Company to adopt any amendments to the Plan as the Administrative Committee may determine to be necessary or appropriate, except that no amendment shall be made to any Plan without approval of the Board of Directors unless the Administrative Committee determines that such amendment will not significantly change the overall level of benefits provided by such Plan; significantly change the requirements for eligibility for participation in the Plan; or add any material new benefit that would significantly increase the cost of the Plan. In illustration but not limitation of the foregoing, the Administrative Committee is authorized to adopt any amendment to a Plan that it determines to be:
 - (i) an amendment that provides for the Plan to be adopted by any business entity acquired by the Company, including providing any special rules applicable to the employees of such business entity;
 - (ii) an amendment that the Administrative Committee determines to be of an administrative, ministerial or technical nature only;
 - (iii) an amendment that the Administrative Committee determines to be necessary or appropriate to carry out any amendment approved by, or other resolution adopted by, the Board;
 - (iv) an amendment that the Administrative Committee determines to be necessary or appropriate to comply with any applicable law, including without limitation any amendment required by the Internal Revenue Service as a condition to the issuance of a favorable determination letter with respect to the Plan, or necessary to conform the terms of the Plan to established administrative practices or procedures; or

(v) an amendment that the Administrative Committee determines to be necessary or appropriate to clarify or to resolve any inconsistency or ambiguity in the terms of the Plan.

The adoption by the Administrative Committee of any amendment to the Plan shall constitute conclusive evidence that the Administrative Committee has determined such amendment to be authorized under the terms of the foregoing resolution, which determination shall be conclusive and binding on all employees, participants, beneficiaries and other persons claiming any benefit under the Plan.

- 11.2. Termination. It is the expectation of the Company that it will continue the Plan and the payment of contributions hereunder indefinitely, but the continuation of the Plan and the payment of Participating Employer Contributions hereunder is not assumed as a contractual obligation of the Company or any other Participating Employer; and the right is reserved by the Company or any Participating Employer at any time to reduce, suspend or discontinue its contributions hereunder; provided, however, that the Participating Employer Contributions for any Plan Year accrued or determined prior to the end of such Plan Year will not after the end of such Plan Year be retroactively reduced, suspended or discontinued except as may be permitted by law. The Plan will terminate upon the occurrence of any of the following events:
 - (a) Business Form. Legal adjudication of the Company as bankrupt, a general assignment by the Company to or for the benefit of its creditors, or dissolution of the Company other than by form of or as a result of a reorganization where the business of the Company is continued; or
 - (b) Board of Directors Action. Termination of the Plan by the Board of Directors at any time when, in its judgment, business, financial or other good causes make such termination advisable, to become effective upon the execution and delivery by the Company to the Administrative Committee and Investment Committee and to the Trustee of a written resolution signed on its behalf by an officer of the Company and stating the fact of such termination and the date as of which it is to be effective.

Upon the termination of the Plan or partial termination of the Plan, the rights of affected persons to the benefits provided under the Plan which are not vested and Non-forfeitable as of the date of such termination or partial termination will be fully vested and Non-forfeitable to the extent then funded; provided, however, notwithstanding any other provision of the Plan, the rights of all persons entitled to vested and Non-forfeitable benefits under the Plan will be limited to the assets of the Trust Fund, and to the extent benefits are guaranteed by the PBGC, and no Participating Employer or Employer will have any obligation to make any contributions to pay any benefits under the Plan subsequent to a termination or partial termination of the Plan.

11.3. **Disposition of Fund on Termination**. Subject to the provisions of Section 9.10, upon an entire termination of the Plan the Trust Fund will be liquidated by making provisions (if not already so provided) for payments, after providing for the costs and expenses of the Plan and Trust Fund, to the extent the assets in the Trust Fund are sufficient therefor, in the order of precedence established under ERISA Section 4044; provided however, in the event the Plan terminates or there is a spin-off of part of the Plan (other than a de minimis spin-off permitted

under Treasury Regulations Section 1.414(1)-(n)(2)) within five years of the effective date of any de minimis merger of any defined benefit plan (the "Smaller Plan") into the Plan (the "Larger Plan") as permitted by Treasury Regulations Section 1.414(1)-1(h), there will be payable to Participants, on whose behalf assets and liabilities were transferred from the Smaller Plan to the Larger Plan, a special schedule of benefits (consisting of all the benefits that would be provided by the Smaller Plan on a termination basis just prior to the merger) in a priority category higher than the highest priority category payable under Section 4044 of ERISA. Plan assets will be allocated to that schedule in accordance with the allocation of assets to scheduled benefits in Treasury Regulations Section 1.414(1)-1(f)(3). The preceding provisions for payments will be made regardless of whether the Participant has incurred a Termination of Employment and regardless of the Participant's age.

11.4. **Disposition Medium**. The allocations, referred to in Section 11.3, may be implemented through the continuance of the Trust Fund (as a wasting trust), through a new Trust Fund, through the purchase of insurance company annuity contracts or by a combination of these methods. Notwithstanding the previous sentence, every Participant with an Accrued Benefit in excess of \$5,000 (or such higher amount permitted by applicable law) is entitled to receive his Accrued Benefit in the form of an insurance company annuity contract (with such contract including all optional forms of payment available under the Plan at plan termination, any retirement-type subsidy and qualified joint and survivor and pre-retirement survivor annuity features as are required by Code Section 411(d)(6) or Treasury regulations thereunder) unless such Participant elects a different form of payment. Any election by a married Participant of a form of payment other than an annuity contract with joint and survivor and pre-retirement survivor annuity features (with the Spouse designated as the survivor) must be consented to by the Participant and, to the extent applicable, the Spouse.

ARTICLE XII. Special Top-Heavy Rules

- **12.1. Application**. Notwithstanding any provisions of the Plan to the contrary, the provisions of this Article XII will apply and be effective for any Plan Year for which the Plan will be determined to be a "Top-Heavy Plan" as provided and defined herein.
 - **12.2. Special Terms.** For purposes of this Article XII, the following terms will have the following meanings:
 - (a) "Active Participant" means an Eligible Employee who is a Participant in the Plan at the time of reference.
 - (b) "Aggregate Benefit" means the sum of:
 - (i) the present value of the accrued benefit under each and all defined benefit plans in the Aggregation Group determined on each plan's individual Determination Date as if there were a termination of employment on the most recent date the plan is valued by an actuary for purposes of computing plan costs under Code Section 412 within the 12-month period ending on the Determination Date of each such plan, but with respect to the first plan year of any such plan determined by taking into account the estimated accrued benefit as of the Determination Date; provided, the actuarial assumptions to be applied for purposes of this paragraph (i) will be the same assumptions as those applied for purposes of determining the actuarial equivalents of optional benefits under the particular plan, except that the interest rate assumption will be five percent; nonproportional subsidies will be valued as required under Regulation 1.416-1;
 - (ii) the present value of the accrued benefit (i.e., account balances) under each and all defined contribution plans in the Aggregation Group, valued as of the valuation date coinciding with or immediately preceding the Determination Date of each such plan, including (A) contributions made after the valuation date but on or prior to the Determination Date, (B) with respect to the first plan year of any plan, any contribution made subsequent to the Determination Date but allocable as of any date in the first plan year, or (C) with respect to any defined contribution plan subject to Code Section 412, any contribution made after the Determination Date that is allocable as of a date on or prior to the Determination Date; and
 - (iii) the sum of each and all amounts distributed (other than a rollover or plan-to-plan transfer) from any Aggregation Group Plan, plus a rollover or plan-to-plan transfer initiated by the Employee and made to a plan which is not an Aggregation Group Plan within the Current Plan Year or within the preceding four plan years of any such plan, provided such amounts are not already included in the present value of the accrued benefits as of the

valuation date coincident wither immediately preceding the Determination Date. Effective January 1, 2002, the present value of the accrued benefits and amounts of account balances of an Employee as of the Determination Date will be increased by the distributions made with respect to the Employee under the Plan, and any plan aggregated with the Plan under Code Section 416(g)(2), during the one-year period ending on the determination date. The preceding sentence will also apply to distributions under a terminated plan which, had it not been terminated, would have been aggregated with the Plan under Section Code 416(g)(2)(A)(i). In the case of a distribution made for reason other than severance from employment, death, or disability, this provision will be applied by substituting "five-year period" for "one-year period". The accrued benefits of any individual who has not performed an hour of service for an Employer during the one-year period ending on the Determination Date will not be taken into account.

The Aggregate Benefit will not include the value of any rollover or plan-to-plan transfer, the contribution or transfer of which to an Aggregation Group Plan was initiated by a Participant, was from a plan which was not an Aggregation Group Plan and was made after December 31, 1983, nor will the Aggregate Benefit include the value of employee contributions which are deductible pursuant to Code Section 219.

- (c) "Aggregate Compensation" means one-twelfth of the average of an Active Participant's Compensation from a Participating Employer, as defined in Code Section 415, earned during the five-consecutive-Plan-Year period in which the Active Participant's compensation is the highest and during which the Plan is a Top-Heavy Plan, and during which the Active Participant earns a Year of Aggregate Service. If a person has less than five consecutive Plan Years of compensation, or if the Plan is not a Top-Heavy Plan during five consecutive Plan Years, "Aggregate Compensation" will mean one-twelfth of the average of the Active Participant's compensation earned during the Plan Years the Plan was a Top-Heavy Plan and during which the Active Participant earned a Year of Aggregate Service, divided by the number of such Plan Years. An Active Participant's Aggregate Compensation will exclude all compensation earned prior to January 1, 1984 or during any Plan Year the Plan is not a Top-Heavy Plan.
- (d) "Aggregation Group" means the Plan and one or more plans (including plans that terminated) which are described in Code Section 401(a), is an annuity contract described in Code Section 403(a) or is a simplified employee pension described in Code Section 408(k) maintained or adopted by the Employer in the Current Plan Year or one of the four preceding Plan Years which is either a "Required Aggregation Group" or a "Permissive Aggregation Group":
 - (i) **A "Required Aggregation Group"** means all Aggregation Group Plans in which either (A) a Key Employee participates or (B) which enable any

- Aggregation Group Plan in which a Key Employee participates to satisfy the requirements of Code Section 401(a)(4) or Code Section 410.
- (ii) A "Permissive Aggregation Group" means Aggregation Group Plans included in the Required Aggregation Group, plus one or more other Aggregation Group Plans, as designated by the Board of Directors of the Company in its sole discretion, which satisfy the requirements of Code Sections 401(a) (4) and 410 when considered with the other component plans of the Required Aggregation Group.
- (e) "Aggregation Group Plan" means the Plan and each other plan in the Aggregation Group.
- (f) "Current Plan Year" means (i) with respect to the Plan, the Plan Year in which the Determination Date occurs, and (ii) with respect to each other Aggregation Group Plan, the plan year of such other plan in which occurs the Determination Date of such other plan.
- (g) "Determination Date" means (i) with respect to the Plan and its Plan Year, the last day of the preceding Plan Year; or (ii) with respect to any other Aggregation Group Plan in any calendar year during which the Plan is not the only component plan of an Aggregation Group, the determination date of each plan in such Aggregation Group to occur during the calendar year as determined under the provisions of each such plan.
- (h) "Former Key Employee" means an Employee (including a terminated Employee) who is not a Key Employee in the Current Plan Year but who was a Key Employee at any time prior to the four preceding Plan Years.
- (i) "Key Employee" means an Employee or former terminated Employee (including any deceased Employee) who at any time during the Plan Year that includes the Determination Date is:
 - (i) An officer of the Employer whose compensation from a Participating Employer and an Employer during the Plan Year is greater than \$170,000 (as adjusted under Code Section 416(i)(1)); provided, however, that no more than the lesser of (A) 50 Employees, or (B) the greater of (1) three Employees or (2) ten percent (rounded to the next whole integer) of the greatest number of Employees during the Plan Year will be considered as officers for this purpose. Such officers considered will be those with the greatest annual compensation as an officer during the Plan Year;
 - (ii) A person who owns more than five percent of the value of outstanding stock of the Company or of any Employer or more than five percent of the total combined voting power of all stock of the Company or any Employer (considered separately) or 5%

(iii) A person who owns more than one percent of the value of the outstanding stock of a Participating Employer or of any Employer or more than one percent of the total combined voting power of all stock of a Participating Employer or of any Employer (considered separately) and whose total annual compensation from a Participating Employer and an Employer is in excess of \$150,000.

For the purposes of this Section, compensation will mean compensation as defined in Code Section 415(c)(3). The determination of who is a Key Employee will be made in accordance with Code Section 416(i)(1) and the applicable regulations and other guidance of general applicability issued thereunder. Any person who is a Key Employee under more than one of the three paragraphs of this Section 12.2(i) will have his Aggregate Benefit under the Aggregate Group Plans counted only once with respect to computing the Aggregate Benefit if Key Employees as of any Determination Date."

- (j) "Top-Heavy Plan" means the Plan with respect to any Plan Year if the Aggregate Benefit of all Key Employees or the Beneficiaries of Key Employees determined on the Determination Date is an amount in excess of 60 percent of the Aggregate Benefit of all persons who are Employees within the Current Plan Year (excluding Former Key Employees), plus the Aggregate Benefit of persons who have been Employees (but are not Former Key Employees) within the four preceding Plan Years, but who are not Employees in the Current Plan Year. With respect to any calendar year during which the Plan is not the only Aggregation Group Plan, the ratio determined under the preceding sentence will be computed based on the sum of the Aggregate Benefits of each Aggregation Group Plan totaled as of the last Determination Date of any Aggregation Group Plan to occur during the calendar year.
- (k) "Year of Aggregate Service" means a Plan Year in which an Active Participant earns at least 1,000 Hours of Service, provided that any Plan Year when the Plan is not a Top-Heavy Plan will be disregarded.
- 12.3. Vested Percentage. For any Plan Year that the Plan is a Top-Heavy Plan, the Non-forfeitable percentage of the Accrued benefit of any person who is an Employee for such Plan Year will be determined under the following table, where the first column is the Employee's Years of Service and the second column is the Employee's Non-forfeitable percentage in such Accrued benefit:

Years of Service	Vesting <u>Percentage</u>
Less than 2 years	0%
2 years but less than 3 years	20%
3 years but less than 4 years	40%
4 years but less than 5 years	60%
5 years or more	100%

For any Plan Year following a Plan Year in which the Plan was a Top-Heavy Plan but during which Plan Year the Plan is not a Top-Heavy Plan, the Employee's Non-forfeitable percentage in his accrued Benefit will be not less than the Non-forfeitable percentage as of the date the Plan was last a Top-Heavy Plan; provided that any Employee who has accrued not less than three Years of Service as of the first day of the Plan Year following the Plan Year the Plan was last determined to be a Top-Heavy Plan will have his Non-forfeitable percentage for future benefits as well as accrued benefits determined under this Section 12.3. For purposes of this Section 12.3, only Employees who perform an Hour of Service after the Plan becomes a Top-Heavy Plan are entitled to the special vesting provisions.

- 12.4. Minimum Benefit. Notwithstanding any provision of the Plan to the contrary, for any Plan Year that the Plan will be a Top-Heavy Plan, the Accrued Benefit under the Plan for an Active participant who has completed a Year of Aggregate Service during such Plan Year (regardless of whether he is still employed on the last day of the Plan Year) but who is neither a Key Employee nor a Former Key Employee will be not less than a monthly benefit payable to the Employee in the form of a Single Life Annuity commencing at his Normal Retirement Date or the attained age, if later, were the person to have a Termination of Employment on the Determination Date, considering his Aggregate Compensation and Years of Aggregate Service earned prior to the Determination Date in an amount equal to (a) reduced by (b):
 - (a) Is an amount equal to two percent of the Participant's Aggregate Compensation multiplied by the Participant's Years of Aggregate Service not to exceed ten such Years of Aggregate Service.
 - (b) Is the monthly benefit payable to the Participant in the form of a Single Life Annuity commencing at his Normal Retirement Date or the attained age, if later, were the person to have a Termination of Employment on the Determination Date, such monthly benefit being the Actuarial Equivalent of the accrued benefit of the Participant under any other Aggregation Group Plan, plus the additional amount, if any, of any distribution from the other Aggregation Group Plan which has not been recontributed to the other plan as of the Determination Date.

Any benefit paid under the Plan which satisfies the requirements of this Section 12.4 will be paid at such time, in such form, and subject to such conditions as benefits are otherwise payable under the Plan, except that any such benefit payable commencing after the person's Normal Retirement Date will be the Actuarial Equivalent of the person's benefit at his Normal Retirement Date.

For purposes of satisfying the minimum benefit requirements of Code Section 416(c)(1) and the Plan, in determining Years of Service, any Service with an Employer will be disregarded to the

extent that such Service occurs during a Plan Year when the Plan benefits (within the meaning of Code Section 410(b)) no Key Employee or former Key Employee.

- 12.5. Maximum Benefit Accrual. For any Plan Year that the Plan is a Top-Heavy Plan, the denominator of the "Defined Benefit Plan Fraction" and the denominator of the "Defined Contribution Plan Fraction" (as defined in Section 13.11) will be determined by substituting "1.0" for "1.25." The preceding sentence will not apply with respect to any Plan Year that the Plan is a Top-Heavy Plan if (a) Section 12.4 is applied by substituting "three percent" for "two percent" and (b) the Plan would not be a Top-Heavy Plan if "90 percent" were substituted for "60 percent" in Section 12.2(j). The first sentence of this Section 12.5 will not apply with respect to an Employee for any Plan Year during which he accrues no benefit under any plan of the Aggregation Group.
- **12.6. Termination of Top-Heavy Status**. If the Plan has been determined to be a Top-Heavy Plan for one or more Plan Years and thereafter ceases to be a Top-Heavy Plan, the provisions of this Article XII will cease to apply to the Plan effective as of the Determination Date on which the Plan is not a Top-Heavy Plan.

ARTICLE XIII. Miscellaneous Provisions

- 13.1. Company Merger. In the event any successor corporation to the Company by merger, consolidation, purchase or otherwise, will elect to adopt the Plan, such successor corporation will be substituted hereunder for the Company upon filing in writing with the Trustee its election to do so.
- 13.2. Plan Merger. The Plan will not merge or consolidate with, or transfer any assets or liabilities to, any other plan, unless each person entitled to benefits would receive a benefit immediately after the merger, consolidation or transfer (if the Plan were then terminated) which is equal to or greater than the benefit he would have been entitled to immediately before the merger, consolidation or transfer (if the Plan were then terminated).
- 13.3. Nonalienation of Benefits. Except as provided in Section 13.4, no benefit payable at any time under the Plan will be subject in any manner to alienation, sale, transfer, assignment, pledge, attachment, or other legal proceeding or processes, or encumbrance of any kind. Any attempt to alienate, sell, transfer, assign, pledge, attach or otherwise encumber any such benefits, whether currently or thereafter payable, will be void. No benefit, nor any fund which may be established for the payment of such benefits, will, in any manner, be liable for or subject to the debts or liabilities of any person entitled to such benefits. If any person will attempt to, or will, alienate, sell, transfer, assign, pledge or otherwise encumber his benefits under the Plan or, if by reason of his bankruptcy or other event happening at any time, such benefits would devolve upon any other person or would not be enjoyed by the person entitled thereto under the Plan, then the Administrative Committee, in its discretion, may terminate the interest in any such benefits of the person entitled thereto under the Plan and hold or apply them to or for the benefit of such person entitled thereto under the Plan in such manner as the Administrative Committee may deem proper.
- 13.4. Qualified Domestic Relations Orders. Notwithstanding the provisions of Article VI or Article VII, if a former Spouse, child or other dependent of a Participant (an "alternate payee" for purposes of this Section 13.4) is entitled to receive all or a portion of a Participant's Accrued Benefit pursuant to a Qualified Domestic Relations Order (as defined below), then the Participant's Accrued Benefit will be payable pursuant to such Qualified Domestic Relations Order and consistent with the Plan procedures identified below. The Participant's Accrued Benefit will be reduced to the extent necessary to reflect the time, manner and amount of such payments to such alternate payee(s) pursuant to a Qualified Domestic Relations Order. A "Qualified Domestic Relations Order" is a judgment, decree or order (including approval of a property settlement agreement) relating to the provision of child support, alimony payments or marital property rights to an alternate payee which is made pursuant to a state domestic relations law (including a community property law), which creates or recognizes the existence of an alternate payee's right to, or assigns to an alternate payee the right to, receive all or a portion of the benefits payable with respect to such Participant under the Plan, and which satisfies the other requirements of Code Section 414(p). The Administrative Committee will establish procedures to determine the qualified status of domestic relations orders and to administer distributions pursuant to Qualified Domestic Relations Orders.

- 13.5. No Employment Guarantee. Neither the establishment of the Plan nor any modification thereof, nor the creation of any fund or benefit, nor the payment of any benefits will be construed as giving to any Participant or any other person any legal or equitable right against the Participating Employers, the Administrative Committee, the Investment Committee, the Trustee or any Plan representative except as herein provided. Under no circumstances will the terms of employment with the Participating Employer of any Participant be modified or in any way affected hereby. The maintenance of this Plan will not constitute a contract of employment with the Participating Employer, nor will anything contained in the Plan be construed as such a contract. Participation in the Plan will not give any Participant a right to be retained as an Employee of the Participating Employer.
- **13.6. Termination of Employment**. When a person incurs a Termination of Employment, his right to benefits from the Plan will be determined only by the terms of the Plan.
- **13.7. Limitation on Vesting**. No person will have any vested right to benefits under the Plan until all of the applicable requirements for such benefits set forth in Article IV or V have been fulfilled, and then any such rights will be subject to the limitation of Section 11.2.
- **13.8. No Duplication of Benefits**. No benefits will be paid to any person under more than one provision of the Plan for the same period of time.
- **13.9. Source of Benefits**. All benefits payable under the Plan will be paid or provided for solely from the Trust, and the Participating Employers assume no liability or responsibility therefor.
- 13.10. Reduction for Overpayment. The Administrative Committee will, whenever it determines that a person has received benefit payments under the Plan in excess of the amount to which the person is entitled under the terms of the Plan, make reasonable attempts to collect such overpayment from the person. If the person to whom such overpayments were made does not, within a reasonable time, make the requested repayment to the Administrative Committee, and if the overpayment was due to an error by the Plan that the Participant would have no reasonable way of knowing was an error, the overpayment will be considered as an advance payment of benefits and the Administrative Committee will direct the Trustee to reduce future benefits until the overpayment has been recouped. Nothing contained herein shall be construed to limit the authority of the Administrative Committee to recover any overpayment by any method otherwise available at law or equity. Any payment made to a person in error shall be considered a separate fund held by such person in trust for the benefit of the Plan.
- 13.11. Limitations on Pension Benefits Payable to Highly Compensated Participants. In the event of Plan termination for any reason other than the failure to obtain Internal Revenue Service approval, the benefit of any highly compensated active or former Employee of the Participating Employer or any Related Participating Employer is limited to a benefit that is nondiscriminatory under Code Section 401(a)(4). In any other event, benefits distributed to any Participant who is one of the 25 most highly compensated active and former highly compensated Employees are restricted such that the annual payments are no greater than an amount equal to the payment that would be made on behalf of the Participant under a single

life annuity that is the Actuarial Equivalent of the sum of the Participant's Accrued Benefit and the Participant's other benefits under the Plan. The preceding sentence will not apply if:

- (a) after payment of the benefit to the Participant, the value of Plan assets equals or exceeds 110 percent of the value of current liabilities, as defined in Code Section 412(1)(7), or
- (b) the value of the benefits for the Participant is less than 1 percent of the value of current liabilities.

The limitations in this Section will automatically become inoperative and of no effect upon a ruling by the Internal Revenue Service that they are not required. If regulations are issued modifying the limitations described in this Section, the Plan will be amended in a timely fashion to incorporate such modified regulations; and, prior to such amendment, the Plan will be administered in accordance with the modified regulations. For purposes of this Section 13.11, the term "benefit" includes loans in excess of the amount set forth in Code Section 72(p)(2)(A), any periodic income, and any Death Benefits not provided for by insurance on the Participant's life.

13.12. Maximum Pensions. Notwithstanding any provisions of the Plan to the contrary, the Benefit to which a person is entitled at any time during any Plan Year will be subject to the provisions of Code Section 415, which are hereby incorporated by reference. If a Participant has an accrued benefit under a defined contribution plan or a defined benefit plan (other than this Plan) which is intended to meet the requirements of Code Section 401(a) and which is maintained by an Employer, benefits will be reduced under this Plan in order to satisfy the requirements of Code Sections 415(b)(1) or 415(e). For limitation years beginning on and after January 1, 2008, the limitations of Code Section 415 shall be applied in accordance with the final Treasury Regulations issued April 5, 2007, which are incorporated herein by this reference.

Benefit increases resulting from the increase in the limitations under Code Section 415(b) shall be provided to all current and former Participants with benefits limited by Section 415(b) who (a) have an Accrued Benefit under the Plan immediately prior to January 1, 2002, and (b) have not then commenced receiving benefit payments from the Plan. Notwithstanding any other Plan provisions to the contrary, for purposes of adjusting any benefit or limitation under Code Section 415(b)(2)(B), (C), or (D), the "applicable mortality table" for distributions with an annuity start date on or after December 31, 2002, is the mortality table prescribed by the Secretary of the Treasury as set forth in Revenue Ruling 2001-62.

13.13. Funding-Based Limitations on Benefits. Notwithstanding any other provision of the Plan to the contrary, effective January 1, 2008, the following limitations shall apply, as and to the extent required by the Pension Protection Act of 2006. This Section shall be interpreted and applied consistent with Code Section 436 and any guidance issued thereunder. For all purposes of this Section 13.13, the Plan's "funding target attainment percentage" and "adjusted funding target attainment percentage" shall have the meanings set forth in Code Section 436(j).

(a) Unpredictable Contingent Event Benefits

In accordance with Code Section 436(b), an "unpredictable contingent event benefit" (as defined in Code Section 436) to which a Participant would otherwise be entitled during any Plan Year shall not be provided to such Participant if the Plan's adjusted funding target attainment percentage for any Plan Year:

- (i) is less than 60%; or
- (ii) would be less than 60% taking into account the event for which such benefit is payable.

The preceding sentence shall cease to apply with respect to any Plan Year, effective as of the first day of such Plan Year, upon payment by the Company of a contribution (in addition to any minimum required contribution under Code Section 430) equal to: (A) in the case of subparagraph (i), the amount of the increase in the funding target of the Plan (under Code Section 430) for the Plan Year attributable to the event for which such benefit is payable; or (B) in the case of subparagraph (ii), the amount sufficient to result in a 'funding target attainment percentage' (as defined in Code Section 436) of 60%.

Notwithstanding the foregoing, any payments restricted under this paragraph shall not automatically resume upon the date on which such payments are no longer restricted, but shall resume only upon adoption of a Plan amendment that otherwise meets the requirements of this Section.

(b) Plan Amendments Increasing Liability for Benefits

In accordance with Code Section 436(c), an amendment to the Plan which would have the effect of increasing the Plan's liabilities by increasing benefits, establishing new benefits, changing the rate of benefit accrual or changing a vesting formula, may not take effect during any Plan Year in which the Plan's adjusted funding target attainment percentage:

- (i) is less than 80%; or
- (ii) would be less than 80% taking into account such amendment.

The preceding sentence shall cease to apply with respect to any Plan Year, effective as of the first day of such Plan Year (or if later, the effective date of the amendment), upon payment by the Company of a contribution (in addition to any minimum required contribution under Code Section 430)) equal to: (A) in the case of subparagraph (i), the amount of the increase in the funding target of the Plan (under Code Section 430) for the Plan Year attributable to the amendment; or (B) in the case of subparagraph (ii), the amount sufficient to result in a funding target attainment percentage of 80%. Notwithstanding the foregoing, this paragraph shall not apply to an amendment providing for an increase in benefits under a formula which is not based on a Participant's Compensation, but only if the rate

of such increase is not in excess of the contemporaneous rate of increase in average wages of all Participants subject to the amendment.

Notwithstanding the foregoing, any amendments restricted under this paragraph shall not automatically take effect upon the date on which such amendments are no longer restricted, but shall resume only upon the adoption of an additional Plan amendment that otherwise meets the requirements of this Section.

(c) Accelerated Benefit Distributions

- (i) In accordance with Code Section 436(d)(1), in any case in which the Plan's adjusted funding target attainment percentage for a Plan Year is less than 60%, the Plan may not pay any "prohibited payment" (as defined in subparagraph (iv) below) after the valuation date for the Plan Year.
- (ii) In accordance with Code Section 436(d)(2), during any period in which the Company is a debtor in a case under Title 11 of the United States Code, or similar Federal or State law, the Plan shall not pay any prohibited payment. The preceding sentence shall not apply on or after the date on which the enrolled actuary of the Plan certifies that the adjusted funding target attainment percentage of such Plan is not less than 100%.
- (iii) In accordance with Code Section 436(d)(3), in any case in which the Plan's adjusted funding target attainment percentage for a Plan Year is 60% or greater but less than 80%, the Plan shall not pay any prohibited payment after the valuation date for the Plan Year to the extent the amount of the payment exceeds the lesser of: (A) 50% of the amount of the payment which could be made without regard to the limits in Code Section 436(d)); or (B) the present value (determined under guidance prescribed by the Pension Benefit Guaranty Corporation, using the interest and mortality assumptions under Code Section 417(e)) of the maximum guarantee with respect to the Participant under ERISA Section 4022. In addition to the foregoing restriction as to payment amount, only one prohibited payment meeting the requirements of this paragraph may be made with respect to any Participant during any period of consecutive Plan Years to which the limitations under subparagraph (i) or (ii) of this paragraph apply. For purposes of the foregoing, a Participant and any Beneficiary on his behalf (including an alternate payee, as defined in Code Section 414(p)(8)) is treated as one Participant. As a result, if the Participant's Accrued Benefit is allocated to an alternate payee and one or more other persons, the amount under this subparagraph shall be allocated among such persons in the same manner as the Accrued Benefit is allocated, unless a QDRO provides otherwise.

A Participant or Beneficiary who elects an optional form of benefit that is not available as of the Participant's Annuity Starting Date due to the application of this subparagraph shall have the option either (A) to defer

payment to a later date (to the extent permitted under the Plan); or (B) to bifurcate the benefit into restricted and unrestricted portions. If the Participant elects to bifurcate payment of the benefit, with respect to the unrestricted portion, the Participant may elect any optional form of benefit then available under the Plan and, with respect to the restricted portion, the Participant may elect any optional form of benefit then available under the Plan that is not a prohibited payment. For purposes of this subparagraph, the "unrestricted portion" of the benefit is the lesser of (A) 50% of the benefit; or (B) the portion of the benefit that has a present value equal to the Pension Benefit Guaranty Corporation guarantee amount described in Treasury Regulation Section 1.436-(1)(d)(3)(iii)(C). Notwithstanding the foregoing, the form of a Participant's benefit that had commenced while the restriction under this paragraph applied shall not be adjusted to another form of payment on or after the date on which such payments are no longer restricted.

(iv) For purposes of this paragraph, a "prohibited payment" shall mean (A) any amount that the Participant elects to have paid in a lump sum pursuant to Section 7.2(d) or any Supplement (but not any amount paid as a lump sum pursuant to Section 7.3)), (B) any other payment in excess of the monthly amount paid under a life only annuity (plus any social security supplements paid in accordance with Code Section 411(a)(9)) to any Participant or Beneficiary whose annuity starting date (as defined in Code Section 417(f)(2)) occurs during a limitation period described in subparagraph (i) or (ii); (C) any payment for the purchase of an irrevocable commitment from an insurer to pay benefits; or (D) any other payment specified in the Treasury Regulations.

(d) Future Benefit Accruals

In accordance with Code Section 436(e), in any case in which the Plan's adjusted funding target attainment percentage for a Plan Year is less than 60%, benefit accruals under the Plan shall cease as of the valuation date for the Plan Year. The preceding sentence shall cease to apply with respect to any Plan Year, effective as of the first day of such Plan Year, upon payment by the Company of a contribution (in addition to any minimum required contributions under Code Section 430) equal to the amount sufficient to result in an adjusted funding target attainment percentage of 60% or more. Notwithstanding the foregoing, benefit accruals shall not automatically resume upon the date on which benefit accruals are no longer restricted, but shall resume only upon adoption of a Plan amendment that otherwise meets the requirements of this Section.

13.14. Indemnity. To the extent permitted by applicable law, and to the extent that they are not indemnified or saved harmless under any liability insurance contracts, any present or former Administrative Committee or Investment Committee members, officers, Employees or directors of the Participating Employers or their subsidiaries or affiliates, if any, and each of them will be indemnified and saved harmless by the Participating Employers from and against

any and all liabilities or allegations of liability to which they may be subjected by reason of any act done or omitted to be done in good faith in the administration of the Plan and Trust, including all expenses reasonably incurred in their defense in the event that the Participating Employers fail to provide such defense after having been requested to do so.

- **13.15. Gender and Number**. Words denoting the masculine gender will include the feminine and neuter genders and the singular will include the plural and the plural will include the singular wherever required by the context.
- 13.16. Severability. If any provision of the Plan is held illegal or invalid for any reason, such illegal or invalid provision will not affect the remaining provisions of the Plan, and the Plan will be construed and enforced as if such illegal or invalid provisions had never been contained in the Plan.
- **13.17. Headings**. The headings or articles are included solely for convenience of reference, and if there is any conflict between such headings and the text of the Plan, the text will control.
- **13.18. Uniform and Nondiscriminatory Treatment**. Except with respect to the Plan design decisions, including decisions made pursuant to Section 1.3, Article VII or Article VIII, or with respect to exercises of any other Plan sponsor or settlor authority, any discretion exercisable hereunder by the Company, a Participating Employer, or either Committee will be exercised in a uniform and nondiscriminatory manner.
- **13.19. Applicable Law**. The Plan and Trust will be construed in accordance with the provisions of ERISA and other applicable federal laws. To the extent not inconsistent with such laws, this Plan will be construed in accordance with the laws of Illinois.
- 13.20. Action by the Participating Employer. Action required or permitted to be taken by a Participating Employer may be taken by action of the board of directors (or the person or body exercising authority similar to that of a board of directors in the case of a noncorporate Participating Employer) of that Participating Employer or by a person or committee of persons authorized to act by said board. The Company's powers may be exercised by its Board of Directors or a person or committee of persons authorized to act by said Board or by a committee of said Board or by the Company's authorized officers or their delegates. The Company reserves the right to delegate a portion or all of its reserved authority as Plan sponsor or settlor of the Plan to any person, persons, committee or committees, including, but not limited to, its Compensation Committee, the Administrative Committee or the Investment Committee. Notwithstanding the other duties or responsibilities any such person or committee may have under this Plan or the Trust when acting in another capacity, any such delegated authority exercised by such person or committee will constitute the exercise of reserved Plan sponsor or settlor powers.
- **13.21. Participant Litigation.** In any action or proceeding regarding the Plan, Employees, Participants, Spouses or any other persons having or claiming to have an interest in this Plan will not be necessary parties and will not be entitled to any notice or process. Any final judgment which is not appealed or appealable and may be entered in any such action or proceeding will be binding and conclusive on the parties hereto and all persons having or

claiming to have any interest in this Plan. To the extent permitted by law, if a legal action is begun against the Company, any Participating Employer, the Administrative or Investment Committee or any member thereof, or any of their directors, officers, partners, members, managers, shareholders, employees, or agents, by or on behalf of any person and such action results adversely to such person or if a legal action arises because of conflicting claims to a Participant's or other person's benefits, the costs to such person of defending the action will be charged to the amounts, if any, which were involved in the action or were payable to the Participant or other person concerned. To the extent permitted by applicable law, acceptance of participation in this Plan will constitute a release of the Company, any Participating Employer, the Administrative and Investment Committees and all members thereof, or their respective directors, officers, partners, members, managers, shareholders, employees, or agents, from any and all liability and obligation not involving willful misconduct or gross neglect.

IN WITNESS WHEREOF, the Administrative Committee has caused this restatement of the Plan to be executed on the ____ day of January, 2016.

BAXTER INTERNATIONAL INC. ADMINISTRATIVE COMMITTEE

By: <u>/s/ Salvatore Dadouche</u>
Salvatore Dadouche
Administrative Committee Member

FIRST AMENDMENT TO THE BAXTER INTERNATIONAL INC. AND SUBSIDIARIES PENSION PLAN

(Amended and Restated as of January 1, 2016)

Pursuant to Section 11.1 of the Baxter International Inc. and Subsidiaries Pension Plan, as amended and restated effective as of January 1, 2016 (the "Plan"), the Plan is further amended as follows effective as of the date of adoption:

- 1. Effective as of January 1, 2016, a new Section 4.6 is added to Article IV of the Plan to read as follows:
- "4.6 **Deferral of Payment Date**. Notwithstanding the foregoing, a Participant who is entitled to a benefit under any of the foregoing provisions of this Article IV may elect to defer the Payment Date to not later than the latest date permitted by Section 7.5(c). A Participant who does not file a retirement application electing to retire on the Payment Date specified in the applicable provision of this Article IV will be deemed to have elected to defer his Payment Date until the date specified on an application that he subsequently files, and if he does not file a subsequent application, he will be deemed to have elected to defer his Payment Date until the latest date permitted by Section 7.5(c)."
- 2. Effective as of January 1, 2016, Section 7.5 is amended by revising the first sentence of to read as follows: "Unless a Participant otherwise elects (or is deemed to elect pursuant to Section 4.6), payment of benefits under the Plan to a Participant will commence not later than the sixtieth day after the end of the Plan Year in which the Participant attains his Normal Retirement Date or incurs a Termination of Employment, whichever occurs later." In addition, Section 7.5 is further amended by adding a new subsection (d) to the end of said Section to read as set forth below:
 - "(d) Inability to Locate Participants and Beneficiaries. In the event that the Administrative Committee is unable after a reasonable effort to locate a Participant or Beneficiary to whom a benefit is due by the date as of which payment is required to commence pursuant to the applicable provision of this Section 7.5, such benefit shall be forfeited as of the day immediately preceding the date on which payment is required to commence; provided that if such Participant or Beneficiary subsequently makes an application for such benefit, the forfeited benefit shall be restored, with appropriate adjustments. Nothing contained herein shall be construed to preclude the Administrative Committee from using any other method permitted by applicable law to satisfy the Plan's obligations to a missing Participant or Beneficiary, including use of the Pension Benefit Guaranty Corporation Missing Participant Program, if applicable."
- 3. Effective as of the date of adoption, Section 8.11 of the Plan is amended in its entirety to read as follows:
- ****8.11 Correction of Errors.** In the event that the Administrative Committee discovers that an error has been made in the calculation of a benefit, it will correct the error as soon as is administratively feasible. In the event of an underpayment, the Administrative Committee will

either pay the amount of the underpayment in a single sum or will increase future monthly payments to the extent necessary to pay the underpayment within a reasonable period of time, with interest at a reasonable rate not less than the interest rate specified in Section 2.2(a)(ii). If an overpayment was made, the Committee will reduce future monthly payments, either to the Participant or his Beneficiary or both, to the extent necessary to recover the overpayment within a reasonable period of time. If an overpayment cannot be recovered in a reasonable period of time through offsetting future benefits, including without limitation an overpayment made in a Lump Sum or a payment erroneously made after the death of a Participant or Beneficiary where no further benefits are payable, the Administrative Committee may exercise such other legal or equitable remedies as the Plan may have, including filing suit for reimbursement of the overpayment; provided, that if the Administrative Committee determines that the expense or burden of seeking recovery would be greater than the likely recovery, it may exercise discretion in determining not to pursue a recovery. Each Participant, Beneficiary or other person who receives an overpayment shall be deemed to hold such overpayment in trust for the Plan, and the Plan may recover such overpayment by an appropriate action in law or equity to the maximum extent permitted by law or equity."

- 4. Effective as of January 1, 2008 (except that the parenthetical phrase beginning "determined by not taking into account" in Section 131.13(c)(ii) shall be effective as of January 1, 2015), Section 13.13 of the Plan is amended in its entirety to read as follows:
- **"13.13 Funding-Based Limitations on Benefits**. Notwithstanding any other provision of the Plan to the contrary, effective January 1, 2008, the following limitations shall apply, as and to the extent required by the Pension Protection Act of 2006. This Section shall be interpreted and applied consistent with Code Section 436, Treas. Reg. §1.436-1 and any other guidance issued thereunder.

(a) Unpredictable Contingent Event Benefits

In accordance with Code Section 436(b), an "unpredictable contingent event benefit" (hereinafter defined) to which a Participant would otherwise be entitled during any Plan Year shall not be provided to such Participant if the Plan's AFTAP for any Plan Year:

- (i) is less than 60%; or
- (ii) would be less than 60% taking into account the event for which such benefit is payable.

The preceding sentence shall cease to apply with respect to any Plan Year, effective as of the first day of such Plan Year, upon payment by the Company of a contribution (in addition to any minimum required contribution under Code Section 430) equal to: (A) in the case of subparagraph (i), the amount of the increase in the funding target of the Plan (under Code Section 430) for the Plan Year attributable to the event for which such benefit is payable; or (B) in the case of subparagraph (ii), the amount sufficient to result in a 'AFTAP' (as defined in Code Section 436) of 60%.

Notwithstanding the foregoing, any payments restricted under this paragraph shall not automatically resume upon the date on which such payments are no longer restricted, but shall resume only upon adoption of a Plan amendment that otherwise meets the requirements of this Section.

For purposes of this Section 13.13, an unpredictable contingent event benefit means any benefit or increase in benefits to the extent the benefit or increase would not be payable but for the occurrence of an unpredictable contingent event. For this purpose, an unpredictable contingent event means a plant shutdown (whether full or partial) or similar event, or an event (including the absence of an event) other than the attainment of any age, performance of any service, receipt or derivation of any compensation, or the occurrence of death or disability, as more fully set forth in Treas. Reg. §1.436-1(j)(9).

(b) Plan Amendments Increasing Liability for Benefits

In accordance with Code Section 436(c), an amendment to the Plan which would have the effect of increasing the Plan's liabilities by increasing benefits, establishing new benefits, changing the rate of benefit accrual or changing a vesting formula, may not take effect during any Plan Year in which the Plan's AFTAP:

- (i) is less than 80%; or
- (ii) would be less than 80% taking into account such amendment.

The preceding sentence shall cease to apply with respect to any Plan Year, effective as of the first day of such Plan Year (or if later, the effective date of the amendment), upon payment by the Company of a contribution (in addition to any minimum required contribution under Code Section 430)) equal to: (A) in the case of subparagraph (i), the amount of the increase in the funding target of the Plan (under Code Section 430) for the Plan Year attributable to the amendment; or (B) in the case of subparagraph (ii), the amount sufficient to result in an AFTAP of 80%. Notwithstanding the foregoing, this paragraph shall not apply to an amendment providing for an increase in benefits under a formula which is not based on a Participant's Compensation, but only if the rate of such increase is not in excess of the contemporaneous rate of increase in average wages of all Participants subject to the amendment.

Notwithstanding the foregoing, any amendments restricted under this paragraph shall not automatically take effect upon the date on which such amendments are no longer restricted, but shall resume only upon the adoption of an additional Plan amendment that otherwise meets the requirements of this Section.

(c) Accelerated Benefit Distributions

(i) In accordance with Code Section 436(d)(1), in any case in which the Plan's AFTAP for a Plan Year is less than 60%, the Plan may not pay any

"prohibited payment" (as defined in subparagraph (iv) below) after the valuation date for the Plan Year.

- (ii) In accordance with Code Section 436(d)(2), during any period in which the Company is a debtor in a case under Title 11 of the United States Code, or similar Federal or State law, the Plan shall not pay any prohibited payment. The preceding sentence shall not apply on or after the date on which the enrolled actuary of the Plan certifies that the AFTAP of such Plan (determined by not taking into account any adjustment of segment rates pursuant to Code Section 430(h)(2)(C)(iv)) is not less than 100%.
- (iii) In accordance with Code Section 436(d)(3), in any case in which the Plan's AFTAP for a Plan Year is 60% or greater but less than 80%, the Plan shall not pay any prohibited payment after the valuation date for the Plan Year to the extent the amount of the payment exceeds the lesser of: (A) 50% of the amount of the payment which could be made without regard to the limits in Code Section 436(d)); or (B) the present value (determined under guidance prescribed by the Pension Benefit Guaranty Corporation, using the interest and mortality assumptions under Code Section 417(e)) of the maximum guarantee with respect to the Participant under ERISA Section 4022. In addition to the foregoing restriction as to payment amount, only one prohibited payment meeting the requirements of this paragraph may be made with respect to any Participant during any period of consecutive Plan Years to which the limitations under subparagraph (i) or (ii) of this paragraph apply. For purposes of the foregoing, a Participant and any Beneficiary on his behalf (including an alternate payee, as defined in Code Section 414(p)(8)) is treated as one Participant. As a result, if the Participant's Accrued Benefit is allocated to an alternate payee and one or more other persons, the amount under this subparagraph shall be allocated among such persons in the same manner as the Accrued Benefit is allocated, unless a QDRO provides otherwise.

A Participant or Beneficiary who elects an optional form of benefit that is not available as of the Participant's Annuity Starting Date (as defined for all purposes of this Section 13.13 pursuant to Treas. Reg. §1.436-1(j)(2)) due to the application of this subparagraph shall have the option either (A) to defer payment to a later date (to the extent permitted under the Plan); or (B) to bifurcate the benefit into restricted and unrestricted portions. If the Participant elects to bifurcate payment of the benefit, with respect to the unrestricted portion, the Participant may elect any optional form of benefit then available under the Plan and, with respect to the restricted portion, the Participant may elect any optional form of benefit then available under the Plan that is not a prohibited payment. For purposes of this subparagraph, the "unrestricted portion" of the benefit is the lesser of (A) 50% of the benefit; or (B) the portion of the benefit that has a present value equal to the Pension Benefit Guaranty Corporation guarantee amount described in Treasury Regulation Section 1.436-(1)(d)(3)(iii)(C). Notwithstanding the

foregoing, the form of a Participant's benefit that had commenced while the restriction under this paragraph applied shall not be adjusted to another form of payment on or after the date on which such payments are no longer restricted.

(iv) For purposes of this paragraph, a "prohibited payment" shall mean (A) any amount that the Participant elects to have paid in a lump sum pursuant to Section 7.2(d) or any Supplement (but not any amount paid as a lump sum pursuant to Section 7.3)), (B) any other payment in excess of the monthly amount paid under a life only annuity (plus any social security supplements paid in accordance with Code Section 411(a)(9)) to any Participant or Beneficiary whose annuity starting date (as defined in Code Section 417(f)(2)) occurs during a limitation period described in subparagraph (i) or (ii); (C) any payment for the purchase of an irrevocable commitment from an insurer to pay benefits; or (D) any other payment specified in the Treasury Regulations, all as set forth in Treas. Reg. §1.436-1(j)(6).

(d) Future Benefit Accruals

In accordance with Code Section 436(e), in any case in which the Plan's AFTAP for a Plan Year is less than 60%, benefit accruals under the Plan shall cease as of the valuation date for the Plan Year. The preceding sentence shall cease to apply with respect to any Plan Year, effective as of the first day of such Plan Year, upon payment by the Company of a contribution (in addition to any minimum required contributions under Code Section 430) equal to the amount sufficient to result in an AFTAP of 60% or more. Notwithstanding the foregoing, benefit accruals shall not automatically resume upon the date on which benefit accruals are no longer restricted, but shall resume only upon adoption of a Plan amendment that otherwise meets the requirements of this Section.

- (e) **Notice Requirement.** The Administrator shall provide written notice to all Participants and Beneficiaries in accordance with Section 101(j) of ERISA within 30 days after the Plan becomes subject to the limitations described in Section 13.13(a) or (c) or if the limitation described in Section 13.13(d) applies, within 30 days the date on which the AFTAP is determined (or presumed) to be less than 60%.
- (f) **Definition and Calculation of AFTAP.** For all purposes of this Section 13.13:
 - (i) "AFTAP" for a Plan Year shall mean the fraction (expressed as a percentage), the numerator of which is the adjusted plan assets for the Plan Year and the denominator of which is the adjusted funding target for the Plan Year, calculated in accordance with Treas. Reg. §1.436-1(j)(1).

- (ii) A "436 measurement date" shall mean the date used to determine the dates as of which certain of the restrictions set forth in this Section 13.13 begin or cease to apply, as determined under Treas. Reg. §1.436-1(j)(8).
- (iii) The Plan's AFTAP for each Plan Year shall be certified by the Plan's enrolled actuary in accordance with Treas. Reg. §1.436-1(h)(4). During certain periods, the Plan's presumed AFTAP shall be determined using the following presumptions, all as determined pursuant to Treas. Reg. §1.436-1(h):
 - (A) The Plan's AFTAP for one Plan Year shall be presumed to be its AFTAP for the following Plan Year until the AFTAP is certified for the following Plan Year, except as provided below.
 - (B) If the Plan's AFTAP has not been certified for a Plan Year by April 1 of such Plan Year, then its AFTAP shall be presumed to be its AFTAP for the preceding Plan Year reduced by 10 percentage points until its AFTAP for the current Plan Year is certified.
 - (C) If the Plan's AFTAP has not been certified by October 1 of a Plan Year, it shall be presumed to be less than 60%.
 - (D) During any period during which a presumed AFTAP is in effect, this Section 13.13 shall be applied as if the presumed AFTAP were the Plan's actual AFTAP, subject to the special rules set forth in Treas. Reg. §1.436-1(g).
- (iv) Notwithstanding the foregoing, pursuant to Section 203(a)(2) of the Preservation of Access to Care for Medicare Beneficiaries and Pension Relief Act of 2010 ("PRA 2010"), for purposes of (A) the limitation on benefit accruals under Section 13.13(d) and (B) the application of the limitations on prohibited payments under Section 13.13(c) to payments under a social security leveling option, the AFTAP for a Plan Year beginning on or after October 1, 2008, and before October 1, 2010, is the greater of the AFTAP for that Plan Year, determined without regard to section 203(a)(2) of PRA 2010, or the AFTAP for the Plan Year beginning after October 1, 2007, and before October 1, 2008.
- (g) Special Rules.
- (i) In the event of a termination of the Plan, any restrictions that were in effect immediately prior to the termination shall continue to apply, except that the limitations on prohibited payments pursuant to Section 13.13(c) shall not apply to prohibited payments that are made to carry out the termination of the Plan in accordance with applicable law.
- (ii) In any Plan Year in which one of the limitations set forth in this Section 13.13 would otherwise apply, the Company may avoid such limitations

either by electing to reduce the Plan's prefunding balance or funding standard carryover balance, or by making additional contributions, in accordance with and subject to Treas. Reg. §1.436-1(f)."

- 5. Effective January 1, 2017, Section I.e of Supplement J to the Plan is amended by striking the phrase "the lesser of" and substituting the phrase "the least of" in the introductory language, and adding a new subsection iii to the end of said Section to read as follows:
 - "iii. The monthly average of the third segment rate under Code Section 430(h) for the August preceding the applicable Plan Year."
- 6. Except as otherwise provided herein, the Plan shall remain in full force and effect.

[Signature on Following Page]

IN WITNESS WHEREOF, this First Amendment is adopted this 21st day of December, 2016.

BAXTER INTERNATIONAL INC. ADMINISTRATIVE COMMITTEE

By: <u>/s/ Salvatore Dadouche</u>
Salvatore Dadouche
Administrative Committee Member

[Signature Page for First Amendment to Baxter International Inc. and Subsidiaries Pension Plan]

BAXTER INTERNATIONAL INC. AND SUBSIDIARIES SUPPLEMENTAL PENSION PLAN

(Amended and Restated Effective January 1, 2015)

TABLE OF CONTENTS

ARTICLE I GENERAL1
1.1 Purpose and Effective Date 1
1.2 Plan Administration; Source of Benefit Payments 1
1.3 Limitation on Provisions 1
1.4 Inactive Participation 1
1.5 Plan Supplements 1
1.5 Baxalta Incorporated Spin-Off 2
· ·
ARTICLE II DEFINITIONS3
2.1 Accrued Benefit 3
2.2 Administrative Committee 3
2.3 Beneficiary 3
2.4 Benefit 3
2.5 Code 3
2.6 Controlled Group 3
2.7 Corporation 3
2.8 Deferred Compensation Plan 3
2.9 Effective Date 3
2.10 ERISA 3
2.11 Excess Benefit 3
2.12 Non-Participating Employer 3
2.13 Participant 3
2.14 Participating Employer 3
2.15 Pension Make-Whole Benefit 4
2.16 Pension Plan 4
2.17 Plan 4
2.17 Fian 4 2.18 Points 4
2.19 Qualified Benefit 4
2.20 Section 409A 4
2.21 Special Supplemental Benefit 4
2.22 Termination of Employment 4
ARTICLE III PARTICIPATION IN THE PLAN6
3.1 Eligibility 6
3.2 Restricted Participation 6
3.3 No Contract of Employment 6
3.4 Participation Freeze 6
ARTICLE IV AMOUNT AND PAYMENT OF PLAN BENEFITS8
4.1 Plan Benefits 8
4.2 Excess Benefit 8
4.3 Pension Make-Whole Benefit 8
4.4 Special Supplemental Benefits 8
4.5 Actuarial Equivalence 9
4.6 Time and Form of Payment 9
4.7 Death Benefits 12
= = =
4.8 Withholding Taxes 13
4.9 Compliance with Section 409A 13
4.10 Correction of Errors 13

TABLE OF CONTENTS

ARTICLE V ADMINISTRATION14	
5.1 Administrative Committee 14	
5.2 Administrative Committee Powers 14	
5.3 Effect of Administrative Committee Decisions	15
5.4 Claims Procedure 15	
5.5 Action by Administrative Committee 16	
5.6 Indemnity 16	
ARTICLE VI AMENDMENT AND TERMINATION17	
6.1 Amendment and Termination 17	
6.2 Successors and Assigns 17	
ARTICLE VII MISCELLANEOUS18	
7.1 Unfunded Plan 18	
7.2 Unsecured General Creditor 18	
7.3 Nonassignability 18	
7.4 Not a Contract of Employment 18	
7.5 Protective Provisions 18	
7.6 Governing Law 19	
7.7 Severability 19	
7.8 Notice 19	
7.9 Successors 19	
7.10 Action by Corporation 19	
7.11 Effect on Benefit Plans 19	

19

7.12 Participant Litigation

BAXTER INTERNATIONAL INC. AND SUBSIDIARIES SUPPLEMENTAL PENSION PLAN

(Amended and Restated Effective January 1, 2015)

ARTICLE I GENERAL

- 1.1 Purpose and Effective Date. Baxter International Inc. (the "Corporation") established the Baxter International Inc. and Subsidiaries Supplemental Pension Plan (the "Plan"), effective as of January 1, 1989, to assist in providing retirement and other benefits to certain employees of the Corporation and its affiliates which are in addition to those provided under the Baxter International Inc. and Subsidiaries Pension Plan (the "Pension Plan"). The Plan was previously amended effective January 1, 2005 and January 1, 2009. The following provisions constitute an amendment and restatement of the Plan effective as of January 1, 2015, the "Effective Date" of the Plan set forth herein. The Plan is intended to constitute an unfunded plan maintained primarily for the purpose of providing deferred compensation to a select group of management or highly compensated employees for purposes of ERISA.
- 1.2 <u>Plan Administration; Source of Benefit Payments</u>. The authority to control and manage the operation and administration of the Plan shall be vested in the Administrative Committee, as set forth in Article V. A Participating Employer's obligation under the Plan shall be reduced to the extent that any amounts due under the Plan are paid from one or more trusts, the assets of which are subject to the claims of general creditors of the Participating Employer or any affiliate thereof, provided, however, that nothing in the Plan shall require the Corporation or any Participating Employer to establish any trust to provide benefits under the Plan.
- 1.3 <u>Limitation on Provisions</u>. Any benefit payable under the Pension Plan shall be paid solely in accordance with the terms and conditions of the Pension Plan and nothing in the Plan shall operate or be construed in any way to modify, amend or affect the terms and provisions of the Pension Plan.
- 1.4 <u>Inactive Participation</u>. Except as otherwise specifically provided herein, the benefits, if any, payable to or on behalf of Participants who terminated employment with the Corporation and its affiliates prior to the Effective Date shall be determined in accordance with the terms of the Plan as in effect on such Termination of Employment; provided that any provision of the Plan that is required to be effective as of an earlier in order to comply with Section 409A of the Code shall be effective as of such date.
- 1.5 <u>Plan Supplements</u>. The provisions of the Plan as applied to any Participating Employer or Participant may be modified and/or supplemented from time to time by the adoption of one or more Supplements. In the event of any inconsistency between a Supplement and the Plan document, the terms of the Supplement shall govern; provided that no Supplement shall alter the provisions of Section 4.9 (except that a Supplement may provide that the portion of any Special Supplemental Benefit is not subject to the provisions of the Plan intended to comply with Section 409A to the extent such portion was accrued and vested on December 31, 2004, and that

the Special Supplemental Benefit is not materially modified after October 3, 2004) or otherwise cause the Plan to be administered in a manner that does not comply with Section 409A.

- Baxalta Incorporated Spin-Off. In connection with the Company's spin-off of its biopharmaceutical business, the Company underwent an internal reorganization and incorporated Baxalta Incorporated ("Baxalta") as a subsidiary of the Company. The Company anticipates that Baxalta shall be spun-off on or around July 1, 2015 (the "Spin-Off") pursuant to a Separation and Distribution Agreement (the "Agreement"). Effective as of the Spin-Off (the "Spin-Off Date"), Baxalta shall be an independent, publicly traded corporation which owns and operates the biopharmaceutical business previously owned and operated by the Company. As described in the Agreement and the Employee Matters Agreement, Baxalta shall establish the Baxalta Incorporated Supplemental Pension Plan (the "Baxalta Plan"), an unfunded nonqualified deferred compensation plan that generally mirrors the Plan, to provide for the benefits of the following Employees who were Participants immediately prior to the date such Employee transfers to employment with Baxalta from the Company:
 - (a) An Employee (including an Employee who is on an approved leave of absence from the Company) who transfers to employment with Baxalta, or one of its subsidiaries, from the Company on or before the Spin-Off Date;
 - (b) An Employee who transfers to employment with Baxalta, or one of its subsidiaries, from the Company in accordance with the Transition Services Agreement.

As described in the Agreement and the Employee Matters Agreement, Baxalta shall assume the obligation to pay all benefits accrued under the Plan for the benefit of each Employee who transfers to employment with Baxalta as described in this Section (a "Baxalta Participant"). As a result of the Spin-Off, a Baxalta Participant shall cease to have any benefit under the Plan following his transfer of employment to Baxalta, and shall instead participate under the Baxalta Plan. To the maximum extent permitted by Treas. Reg. §1.409A-1(h)(4), a Baxalta Participant shall not be considered to have incurred a Termination of Employment for purposes of the Plan or a separation from service as defined in Section 409A solely as a result of the transfer describe above.

ARTICLE II DEFINITIONS

- 2.1 <u>Accrued Benefit</u>. Accrued Benefit shall have the meaning ascribed to such term under the Pension Plan.
- 2.2 <u>Administrative Committee</u>. Administrative Committee shall have the meaning ascribed to such term under the Pension Plan.
- 2.3 <u>Beneficiary</u>. A Participant's Beneficiary shall be the Participant's beneficiary under the Pension Plan (or the person who would be the Participant's beneficiary under the Pension Plan if the Participant's Qualified Benefit were paid in the same form and at the same time as his Benefit hereunder).
- 2.4 <u>Benefit</u>. A Participant's Benefit means the sum of the Participant's Excess Benefit, Make-Whole Benefit, and Special Supplemental Benefit, if any, unless otherwise provided.
 - 2.5 Code. Code means the Internal Revenue Code of 1986, as amended.
- 2.6 <u>Controlled Group</u>. Controlled Group means the Corporation and all other business entities, whether or not incorporated, which, together with the Corporation, would be considered a single employer under section 414(b) or (c) of the Code.
 - 2.7 <u>Corporation</u>. Corporation has the meaning ascribed to such term in Section 1.1.
- 2.8 <u>Deferred Compensation Plan</u>. Deferred Compensation Plan means Baxter International Inc. and Subsidiaries Deferred Compensation Plan.
- 2.9 <u>Effective Date</u>. Effective Date means the effective date of this amendment and restatement of the Plan, which January 1, 2015. The original effective date of the Plan was January 1, 2002.
 - 2.10 ERISA. ERISA means the Employee Retirement Income Security Act of 1974, as amended.
 - 2.11 <u>Excess Benefit</u>. Excess Benefit means the benefit determined under Section 4.2.
- 2.12 <u>Non-Participating Employer</u>. A Non-Participating Employer means any Employer which is not a Participating Employer.
- 2.13 <u>Participant</u>. Participant means an employee of a Participating Employer who is eligible for an Excess Benefit, Pension Make-Whole Benefit or Special Supplemental Benefit, as set forth in Section 3.1.
- 2.14 <u>Participating Employer</u>. Participating Employer means the Corporation and any affiliate of the Corporation, which is a Participating Employer under the Pension Plan.

- 2.15 Pension Make-Whole Benefit. Pension Make-Whole Benefit means the benefit determined under Section 4.3.
- 2.16 <u>Pension Plan</u>. Pension Plan has the meaning ascribed to such term in Section 1.1.
- 2.17 <u>Plan.</u> Plan has the meaning ascribed to such term in Section 1.1.
- 2.18 <u>Points</u>. A Participant's Points shall be equal to the number of Points the Participant has accumulated as of any date under the terms of the Pension Plan as in effect on December 31, 2008, without regard to any amendments to the Pension Plan adopted after such date which directly or indirectly affect the Participant's Points.
- 2.19 <u>Qualified Benefit</u>. Qualified Benefit means the Participant's actual Accrued Benefit payable under the Pension Plan.
- 2.20 <u>Section 409A</u>. Section 409A means Section 409A of the Internal Revenue Code of 1986, as enacted by the American Jobs Creation Act of 2004 and as subsequently amended, and including all Treasury Regulations and other authoritative guidance issued pursuant thereto.
 - 2.21 <u>Special Supplemental Benefit</u>. Special Supplement Benefit means the benefit determined under Section 4.4.
- 2.22 <u>Termination of Employment.</u> A Termination of Employment occurs on the date on which a Participant incurs a separation from service as defined in Treasury Regulations issued pursuant to Section 409A. The following rules are intended to implement the requirements of Section 409A, and may be adjusted by the Administrative Committee as required to comply with guidance issued under Section 409A:
 - (a) The Participant shall not be considered to have separated from service so long as the Participant is on military leave, sick leave, or other bona fide leave of absence if the period of such leave does not exceed six months, or if longer, so long as the Participant retains a right to reemployment with a Participating Employer under an applicable statute or by contract.
 - (b) Regardless of whether his employment has been formally terminated, the Participant will be considered to have separated from service as of the date it is reasonably anticipated that no further services will be performed by the Participant for any Participating Employer, or that the level of bona fide services the Participant will perform after such date will permanently decrease to no more than 20 percent of the average level of bona fide services performed over the immediately preceding 36-month period (or the full period of employment if the Participant has been employed for less than 36 months). For purposes of the preceding test, during any paid leave of absence the Participant shall be considered to have been performing services at the level commensurate with the amount of compensation received, and unpaid leaves of absence shall be disregarded.

- (c) For purposes of determining whether the Participant has separated from service, all services provided for any Employer, or for any entity that is a member of the Controlled Group, shall be taken into account, whether provided as an employee or as a consultant or other independent contractor; provided that the Participant shall not be considered to have not separated from service solely by reason of service as a non-employee director of the Corporation or any other such entity. Solely for purposes of this Section 2.19, the term "Controlled Group" shall be modified by substituting "50 percent" for "80 percent" for all purposes of section 414(b) and (c) of the Code (and Section 1563 to the extent incorporated therein).
- A Participant who is employed by a Participating Employer, and continues to be employed by the Participating Employer following a stock sale, spin-off, or other transaction that causes the Participant's employer to cease to be a member of the Controlled Group, shall not be considered to have incurred a Termination of Employment as a result of such transaction. A Participant who ceases to be employed by the Corporation or any member of the Controlled Group as a result of a sale of substantially all of the assets constituting a division, facility, or separate line of business, shall be considered to have incurred a Termination of Employment unless the Corporation (or Participating Employer selling such assets) and the purchaser agree in writing, not later than the closing date of such transaction, that all Participants affected by such transaction shall not be considered to have incurred a Termination of Employment, and that the purchaser agrees to assume the obligation for payment of the Benefits of all such Participants in accordance with the Plan.

ARTICLE III PARTICIPATION IN THE PLAN

- 3.1 <u>Eligibility</u>. An employee of a Participating Employer shall become a Participant in the Plan on the first date such employee is eligible for an Excess Benefit, Pension Make-Whole Benefit or Special Supplemental Benefit, in accordance with the following:
 - (a) Each participant in the Pension Plan who has a fully vested interest in his or her Accrued Benefit under the Pension Plan and whose benefit under the Pension Plan is limited by reason of the application of Section 415 or Section 401(a)(17) of the Code shall be eligible for an Excess Benefit, determined in accordance with Section 4.2.
 - (b) Each participant in the Pension Plan who has a fully vested interest in his or her Accrued Benefit under the Pension Plan and who also is a participant in the Deferred Compensation Plan shall be eligible for a Pension Make-Whole Benefit, determined in accordance with Section 4.3.
 - (c) The Administrative Committee (or the person or persons delegated such authority by the Administrative Committee), in its sole discretion, shall designate the individuals, if any, who shall be eligible for Special Supplemental Benefits.
- 3.2 Restricted Participation. Notwithstanding any other provision of the Plan to the contrary, if the Administrative Committee determines that participation by one or more Participants shall cause the Plan as applied to any Participating Employer to be subject to Part 2, 3 or 4 of Subtitle B of Title I of ERISA, the entire interest of such Participants under the Plan shall be segregated from the Plan, and such Participants shall cease to have any interest under the Plan. In the event the Participant has died, the foregoing provisions of this Section 3.2 shall apply to the Participant's interest, if any, which is payable to the Participant's surviving spouse or other beneficiary.
- 3.3 No Contract of Employment. The Plan does not constitute a contract of employment, and participation in the Plan will not give any employee the right to be retained in the employ of the Corporation or any Participating Employer nor any right or claim to any benefit under the Plan, unless such right or claim has specifically accrued under the terms of the Plan.
- 3.4 <u>Participation Freeze</u>. Participation in the Plan is frozen effective December 31, 2006, and no Employees shall become Participants after such date, subject to the following:
 - (a) No Employee who is hired by a Participating Employer after December 31, 2006, or who was hired by a Non-Participating Employer prior to January 1, 2007, and transferred to a Participating Employer after December 31, 2006, shall be eligible to Participate in the Plan.
 - (b) An Eligible Employee who was employed by a Participating Employer on December 31, 2006, but who had not satisfied the requirements of Section 3.1(b)

- of the Pension Plan on such date, shall become a Participant on the first Entry Date after he satisfies such requirements, unless he elects not to become a Participant as provided in subparagraph (c) below.
- The election of a Participant or Eligible Employee (as defined in the Pension Plan) to either cease accruing benefits as of December 31, 2006, under the Pension Plan, or not to become a Participant in the Pension Plan after December 31, 2006, shall also apply under this Plan. The Plan Benefit of a Participant who elects under the Pension Plan to cease accruing benefits shall thereafter be equal to his Plan Benefit as of December 31, 2006, which shall not be adjusted for subsequent changes in his Average Monthly Compensation, Years of Service, Projected Benefit Service, or Primary Social Security Benefit, but such Participant shall continue to earn Years of Service for purposes of vesting, and Points, and his Benefit shall be payable as otherwise provided herein. An Eligible Employee who elects not to become a Participant in the Pension Plan shall thereafter be ineligible to become a Participant in this Plan.
- (d) Notwithstanding the foregoing, an Employee hired prior to December 31, 2006, and who did not elect to cease accruing benefits under the Pension Plan as of December 31, 2006, but who was not eligible for either an Excess Benefit or a Make-Whole Benefit prior to December 31, 2006, solely because his Qualified Benefit was not limited by the application of Section 415 or Section 401(a)(17) of the Code and he had not deferred any compensation under the Deferred Compensation Plan, shall be eligible to participate beginning with the first year in which he is eligible for either an Excess Benefit or a Make-Whole Benefit.
- (e) To the extent permitted by Section 4.4, an Employee not otherwise eligible to participate in the Plan may be eligible to receive a Special Supplemental Benefit.

ARTICLE IV AMOUNT AND PAYMENT OF PLAN BENEFITS

- 4.1 <u>Plan Benefits</u>. Eligible Participants under the Plan shall receive an Excess Benefit, Pension Make-Whole Benefit or Special Supplemental Benefit, in the amount and payable at the times set forth in the following provisions of this Article 4. Effective January 1, 2009, the amount of a Participant's Excess Benefit and Pension Make-Whole Benefit shall be calculated as if the Participant's Qualified Benefit had commenced as of the same date, and in the same form, as the Participant's Excess Benefit and Pension Make-Whole Benefit, regardless of when and in what form the Qualified Benefit is paid, and no adjustment shall be made to the Excess Benefit or Pension Make-Whole Benefit when the Qualified Benefit commences. To the extent a Supplemental Benefit is defined in whole or part by reference to the Qualified Benefit, the preceding sentence shall apply unless the terms of the Supplemental Benefit clearly provide for a different method of calculation.
- 4.2 Excess Benefit. As of any date, an eligible Participant's "Excess Benefit" under the Plan shall be an amount equal to the Qualified Benefit the Participant would be eligible for under the Pension Plan as of such date if such Qualified Benefit were determined without regard to limitations of Section 415 and Section 401(a)(17) of the Code, reduced by the Participant's Qualified Benefit as of such date. A Participant's Excess Benefit, if any, shall be paid at the time and in the form provided in Section 4.6.
- 4.3 <u>Pension Make-Whole Benefit</u>. As of any date, an eligible Participant's "Pension Make-Whole Benefit" under the Plan shall be an amount equal to:
 - the Qualified Benefit the Participant would be eligible for under the Pension Plan as of such date if such Qualified Benefit were determined (i) without exclusion of compensation deferred under the Deferred Compensation Plan, and (ii) without regard to the limitations of Code Sections 415 and 401(a)(17),

reduced by

(b) the sum of (i) the Participant's actual Qualified Benefit under the Pension Plan as of such date, and (ii) the amount of any Excess Benefit determined under Section 4.2 without regard to such deferred compensation.

A Participant's Pension Make-Whole Benefit, if any, shall be paid at the time and in the form provided in Section 4.6.

4.4 <u>Special Supplemental Benefits</u>. The amount, if any, of a Participant's "Special Supplemental Benefit" shall be determined by the Administrative Committee, shall be subject to such terms and conditions as the Administrative Committee may establish, and shall be payable at the times and in the form determined by the Administrative Committee. Effective as of January 1, 2005, the time and form of payment of any Special Supplemental Benefit shall be specified by the Administrative Committee at the time the Administrative Committee establishes the Participant's right to the Special Supplemental Benefit. In the event that any right to a Special Supplemental Benefit was not fully vested on December 31, 2004, and is not amended

not later than December 31, 2008 to specify the time and form of payment in a manner that satisfies the requirements of Section 409A, such Special Supplemental Benefit shall be paid in the form specified in Section 4.6. The Administrative Committee, in its sole discretion, may delegate its authority under this Section 4.4 to any person or persons in connection with the award of Special Supplemental Benefits to a particular Participant, a class of Participants, or all Participants. All rights to Special Supplemental Benefits shall be set forth in writing, which writing may include an employment contract or similar agreement, and a copy of all actions taken by the Administrative Committee or its delegate with respect to Special Supplemental Benefits under the Plan shall be sent to the Corporate Counsel in charge of the Company's employee benefit plans. Anything else contained herein to the contrary notwithstanding, no person shall have any right to a Special Supplemental Benefit in the absence of a written instrument setting forth the terms of such Special Supplemental Benefit.

- 4.5 <u>Actuarial Equivalence</u>. To the extent applicable, the benefits payable to any person under the Plan shall be determined by applying the appropriate interest rate and other actuarial assumptions set forth in the Pension Plan.
 - 4.6 <u>Time and Form of Payment.</u>
 - (a) The Benefit (excluding for all purposes of this Section 4.6 any Special Supplemental Benefit unless otherwise provided in Section 4.4) of the following Participants shall commence at the same time and be paid in the same manner as the Participant's Qualified Benefit: (i) Participants whose Qualified Benefit commences not later than December 31, 2008, and (ii) Participants whose entire Benefit was fully accrued and vested on December 31, 2004.
 - (b) The Benefit of a Participant who is not described in paragraph (a) shall become payable upon the later of the occurrence of the first day of the month following the Participant's Termination of Employment or, in the case of a Participant who was a Participant prior to December 31, 2008, a specified date, if any, elected by the Participant in accordance with paragraph (c) (in either case, the "Commencement Date"). Such Benefit shall be paid in the following form:
 - (i) If the actuarial present value of the Benefit as of the Commencement Date does not exceed \$50,000, the Benefit shall be paid in a lump sum equal to the actuarial present value, which payment shall be in full satisfaction of the Participant's right to the Benefit. Such payment shall be made not later than 90 days following the Commencement Date, subject to paragraph (d). For purposes of determining whether the present value of the Benefit exceeds \$50,000, any Special Supplemental Benefit shall be included if and only if the terms of the agreement creating the Special Supplemental Benefit provided for the Special Supplemental Benefit to be paid at the same time and in the same form as remainder of the Benefit not later than the later of the date the Participant first had a legally binding right to the Special Supplemental Benefit or December 31, 2008, and in such event the Special Supplemental Benefit shall be included notwithstanding any change in the terms of the Special Supplemental

Benefit after such date. If the preceding sentence does not apply, the Special Supplemental Benefit shall not be included in determining whether the present value exceeds \$50,000, and the provisions of this paragraph (b) shall be applied separately to the Special Supplemental Benefit.

- (ii) If the actuarial present value of the Benefit exceeds \$50,000 as of the Commencement Date, Benefit shall be paid in a monthly life annuity of the type set forth below. The first annuity payment shall be paid, subject to paragraph (d), on the first day of the month following the first month beginning with month that includes the Commencement Date in which the Participant's has accumulated at least 65 Points (the "Annuity Start Date.")
 - (A) If the Participant's Qualified Benefit commences as of or prior to the Annuity Start Date, his Benefit shall be paid in the same form of annuity as the Qualified Benefit.
 - (B) If the Participant's Qualified Benefit has not yet commenced by the Annuity Start Date, and the Participant is not married on the Annuity Start Date, his Benefit shall be paid in an annuity for the life of the Participant with no survivor benefits.
 - (C) If the Participant's Qualified Benefit has not yet commenced by the Annuity Start Date, and the Participant is married on Annuity Start Date, his Benefit shall be paid in an annuity that pays an actuarially reduced benefit to the Participant during the Participant's life, and pays 50% of such annuity to the Participant's spouse for the balance of the spouse's life if the spouse survives the Participant. No adjustment to such annuity shall be made if the Participant's spouse predeceases the Participant or the Participant and this spouse are divorced after the Annuity Start Date.
 - (D) The Administrative Committee may permit a Participant to elect a different form of annuity that is treated as a life annuity for purposes of Section 409A. Anything else contained herein to the contrary notwithstanding, all forms of life annuity shall be actuarially equivalent as defined in Section 409A, and any procedures adopted by the Administrative Committee to permit Participant's to elect different forms of annuity shall comply with the requirements of Section 409A.
- (c) Each person who was a Participant prior to January 1, 2009, and who is anticipated to have a Benefit accrued under this Plan as of December 31, 2008 (as determined by the Administrative Committee in its sole discretion) may elect a Commencement Date, which shall be either the first day of a specific month or the first day of the month following the date on which the Participant attains a

specified age. Such elections shall apply to the Participant's entire Benefit (including any Special Supplemental Benefit to be paid in the same form as the Benefit), and shall be made, in writing, in accordance with procedures specified by the Administrative Committee, not later than December 31, 2008, and shall not thereafter be revoked or changed; provided that no such election shall cause any amount to be paid in 2008 that would otherwise have been paid in a later year, or cause any amount that would otherwise have been paid in 2008 to be paid in a later year, and such elections shall otherwise comply with the requirements for transitional relief under IRS Notice 2007-86. An agreement (including a provision of an employment agreement) entered into between a Participant and the Corporation not later than December 31, 2008, that refers specifically to this Plan and specifies a time and/or form of payment of the Participant's Benefit shall constitute an election for purposes of this paragraph (c), and, in lieu of a specific date, may provide for the Participant's Commencement Date to occur upon the Participant's separation from service or the occurrence of any other event that satisfies the requirements of Section 409A.

- If a Participant's Commencement Date is the first day of the month following his Termination of Employment, and the Participant is a specified employee as hereinafter defined on the Commencement Date, payment of his Benefit shall be deferred until six months after his Termination of Employment, as described below. If payment is to be made in a lump sum (based on actuarial present value as of the Commencement Date), the lump sum shall be paid on the first day of the seventh month following the month that includes the Termination of Employment, and the amount shall be recalculated as of such date even if such recalculated amount exceeds \$50,000. If payment is to be made in an annuity, the first annuity payment shall be paid on the later of the first day of the seventh month following the month that includes the Termination of Employment or the Annuity Start Date, but if such date is later than the Annuity Start Date the annuity payments shall be calculated as of the Annuity Start Date, and the Participant shall receive a supplemental payment, with or following the first annuity payment, equal to the sum, without interest, of the annuity payments that would have been paid prior to such date but for this paragraph (d). For purposes of this paragraph (d), the term "specified employee" shall have the same meaning as in the Baxter International Inc. and Subsidiaries Deferred Compensation Plan.
- (e) Anything else contained herein to the contrary notwithstanding, the Administrative Committee at any time in its sole discretion may distribute to any Participant the entire actuarial present value of his Benefit (including any Special Supplemental Benefit) in a single lump sum in full satisfaction of his rights under the Plan, provided that the entire interest of the Participant in all other plans required to be aggregated with the Plan pursuant to Treas. Reg. §1.409A-1(c)(2) is also distributed and that the total amount distributed does not exceed the limit in effect under Section 402(g) of the Code at the time of distribution.

- 4.7 Death Benefits.
- (a) If a Participant whose Benefit is payable in an annuity dies after the Annuity Start Date, the only death benefit payable shall be the survivorship benefit, if any, payable under the applicable form of annuity.
- (b) If a Participant whose Benefit is payable in a lump sum dies after his Commencement Date but before actual payment of his Benefit (including but not limited to a Participant whose benefit is deferred pursuant to Section 4.6(d)), the lump sum payment shall be made to his Beneficiary as soon as practical, but not more than 90 days after the date of his death.
- (c) If a Participant either dies prior to his Commencement Date, or after his Commencement Date but prior to his Annuity Start Date if his Benefit is payable as an annuity, and if his Beneficiary is entitled to a pre-retirement survivor annuity under the Pension Plan (or would be entitled to a preretirement survivorship benefit but for the fact that payment of his Qualified Benefit had commenced at the time of his death), his Beneficiary shall be entitled to a preretirement survivor benefit (the "Survivor Benefit") under the terms of this paragraph (c). The Survivor Benefit shall be paid on the first day of the first month following the month that includes the Participant's death in which the Participant either had completed 65 Points, or would have completed 65 Points had he not died. The Survivor Benefit shall be paid in a single lump sum equal to the actuarial present value of the excess of (i) the amount of the preretirement survivor annuity that would be paid to the Beneficiary under the Pension Plan if the Participant's Benefit were calculated with the adjustments described in Sections 4.2 and 4.3 (and included the Special Supplemental Benefit, if applicable), over (ii) the amount of pretirement survivor annuity actually payable under the Pension Plan, in both cases calculated as if payment of the preretirement survivor annuity under the Pension Plan commenced on the date of payment of the Survivor Benefit.
- (d) Notwithstanding the foregoing, if a Participant whose benefit is paid in the form of an annuity and whose Benefit is deferred pursuant to Section 4.6(d) dies after his Annuity Start Date but before the date to which payment of his benefit is deferred, his Beneficiary shall not receive a Survivor Benefit under paragraph (c), but shall instead receive whatever survivorship benefits are provided by the Participant's form of annuity, determined as if payment had commenced on the Annuity Start Date, and in addition shall receive a payment equal to the annuity payments that would have been paid prior to the Participant's death but for the requirement of Section 4.6(d).
- (e) Except as otherwise provided in this Section 4.7, no person shall receive any form of death or survivorship benefits following the death of a Participant, whether before or after his Commencement Date.

- 4.8 <u>Withholding Taxes</u>. Benefits and payments under the Plan are subject to the withholding of all applicable taxes. Notwithstanding any provision of the Plan to the contrary, a Participant's initial benefit payment under the Plan shall be in an amount sufficient pay any remaining employment tax required to be withheld with respect to Plan benefits. To the extent such amount is in excess of the first distribution that would otherwise have been made based on the form of benefit elected by the Participant, subsequent payments will not begin until the aggregated payments that would have been made under the form of benefit elected by the Participant exceed the amount of such initial distribution.
- Compliance with Section 409A. Anything else in this Plan to the contrary notwithstanding, effective January 1, 2005, the Plan is intended to comply in all regards with Section 409A and shall be so construed and administered. Without limiting the generality of the preceding sentence, (i) in no event shall any benefit under the Plan be paid at any time other than under the terms of the Plan as in effect on the date on which the Participant first acquires a legal right to such benefit (whether or not vested), whether by amendment of the Plan, exercise of the Administrative Committee's discretion, or otherwise, except as permitted by Section 409A, and (ii) in the event that the Administrative Committee, in its sole discretion, determines that any time or form of payment provided for in the Plan, or the existence of a right to elect a time or form of distribution (including without limitation the payment of benefits in the same form elected by a Participant under the Pension Plan), would cause the Plan to fail to meet the requirements of Section 409A, or otherwise cause Participants to be subject to any adverse federal income tax consequences, such provision shall to the maximum extent permitted by law be deemed amended to the extent required to comply with Section 409A, or the Plan shall be construed as if such provision were not included therein. The restrictions of Section 409A shall apply to the entire benefit of a Participant if any portion of the Participant's benefit was accrued or vested on or after January 1, 2005, but shall not apply to a Participant whose entire benefit was accrued and vested prior to such date.
- 4.10 Correction of Errors. The Administrative Committee shall have the authority to correct any error in the calculation of Benefits, regardless of the reason for the error and regardless of whether payment of the Benefit has commenced. By his participation in the Plan and acceptance of Benefits hereunder, each Participant agrees that he will promptly repay to the Plan any Benefit or other payment that exceeds the amount to which he was entitled under the Plan (an "excess payment"), and will hold any excess payment, and any proceeds of any excess payment, or property acquired with any excess payment, in trust for the benefit of the Plan, which trust shall remain in effect, and shall continue to apply to any excess payment, proceeds or other property even if transferred to a third party, until the total amount of the excess payment has been repaid to the Plan. The Administrative Committee may, on behalf of the Plan, commence an action to enforce such trust, or take any other available action in law or equity, including setting off any other amount owed to the Participant, to recover such excess payment.

ARTICLE V ADMINISTRATION

- 5.1 <u>Administrative Committee</u>. The Plan is administered by the Administrative Committee, which is the "administrator" for purposes of Section 3(16)(A) of the Employee Retirement Income Security Act of 1974, as amended ("ERISA"). Baxter has appointed the members of the Administrative Committee to administer the Plan. Members of the Administrative Committee may be Participants in the Plan.
- 5.2 <u>Administrative Committee Powers</u>. The Administrative Committee has such powers as may be necessary to discharge its duties hereunder, including, but not by way of limitation, the following powers, rights and duties:
 - (a) <u>Interpretation of Plan</u>. The Administrative Committee has the power, right and duty to construe, interpret and enforce the Plan provisions and to determine all questions arising under the Plan including, but not by way of limitation, questions of Plan participation, eligibility for Plan benefits and the rights of employees, Participants, Beneficiaries and other persons to benefits under the Plan and to determine the amount, manner and time of payment of any benefits hereunder;
 - (b) Plan Procedures. The Administrative Committee has the power, right and duty to adopt procedures, rules, regulations and forms to be followed by employees, Participants, Beneficiaries and other persons or to be otherwise utilized in the efficient administration of the Plan which may alter any procedural provision of the Plan without the necessity of an amendment, and which procedures may provide for any election or consent to be made, or any other action to be taken (including without limitation filing claims and requesting review of denied claims), by electronic mail, internet website, telephone or voice response system or other electronic method to the extent permitted by applicable law;
 - (c) <u>Benefit Determinations</u>. The Administrative Committee has the power, right and duty to make determinations as to the rights of employees, Participants, Beneficiaries and other persons to benefits under the Plan and to afford any Participant or beneficiary dissatisfied with such determination with rights pursuant to a claims procedure adopted by the Committee; and
 - (d) <u>Allocation of Duties</u>. The Administrative Committee is empowered to employ agents (who may also be employees of Baxter) and to delegate to them any of the administrative duties imposed upon the Administrative Committee or Baxter.
 - (e) <u>Plan Amendments</u>. The Administrative Committee has the power and right, at any time, to amend or supplement the Plan. Notwithstanding the foregoing provisions of this Section 5.2(e), no amendment of the Plan shall reduce the benefit to which a Participant would be entitled if he had terminated employment immediately prior to the adoption of the resolution amending the Plan; provided, however, the Administrative Committee or Corporation, as applicable, may

amend the Plan at any time to take effect retroactively or otherwise, as deemed necessary or advisable for purposes of conforming the Plan to any present or future law, regulations or rulings relating to plans of this or a similar nature.

- 5.3 <u>Effect of Administrative Committee Decisions.</u> Any ruling, regulation, procedure or decision of the Administrative Committee will be conclusive and binding upon all persons affected by it. There will be no appeal from any ruling by the Administrative Committee which is within its authority, except as provided in Section 5.4 below. When making a determination or a calculation, the Administrative Committee will be entitled to rely on information supplied by any Employer, accountants and other professionals including, but not by way of limitation, legal counsel for Baxter or any Employer.
- 5.4 <u>Claims Procedure</u>. Each person entitled to benefits under the Plan (the "Applicant") must submit a written claim for benefits to the Administrative Committee. Such claim shall be filed not more than one year after the Applicant knows, or with the exercise of reasonable diligence would know, if the basis for the claim. A formal claim shall not be required for the distribution of a Participant's Accounts in the ordinary course of business, but in any case a claim that relates to a dispute over the amount of a distribution shall be filed not more than one year after payment of the distribution commences. The Administrative Committee may, in its sole discretion accept a claim that is filed late if it determines that special circumstances warrant acceptance of the claim.

If a claim for benefits by the Applicant is denied, in whole or in part, the Administrative Committee, or its delegate, shall furnish the Applicant within 90 days after receipt of such claim, a written notice which specifies the reason for the denial, refers to the pertinent provisions of the Plan on which the denial is based, describes any additional material or information necessary for properly completing the claim and explains why such material or information is necessary, and explains the claim review procedures of this Section 5.4. Such notice will further describe that the Applicant has a right to bring a civil action under Section 502 of ERISA if his claim is denied after an appeal and review. The 90 day period may be extended by up to an additional 90 days if special circumstances required, in which event the Applicant shall be notified in writing by the end of the initial 90 day period of the reason for the extension and an estimate of when the claim will be processed.

Any Applicant whose claim is denied under the provisions described above, or who has not received from the Administrative Committee a response to his claim within the time periods specified in the provisions described above may request a review of the denied claim by written request to the Administrative Committee within 60 days after receiving notice of the denial. If such a request is made, the Administrative Committee shall make a full and fair review of the denial of the claim and shall make a decision not later than 60 days after receipt of the request, unless special circumstances (such as the need to hold a hearing) require an extension of time, in which case a decision shall be made as soon as possible but not later than 120 days after receipt of the request for review, and written notice of the reason for the extension and an estimate of when the review will be complete shall be given to the Applicant before the commencement of the extension. The decision on review shall be in writing and shall include specific reasons for the decision and specific references to the pertinent provisions of the Plan on which the decision

is based. Such notice will further describe that the Applicant has a right to bring a civil action under Section 502 of ERISA.

No person entitled to benefits under the Plan shall have any right to seek review of a denial of benefits, or to bring any action to enforce a claim for benefits, in any court or administrative agency prior to his filing a claim for benefits and exhausting all of his rights under this Section 5.4, or more than 180 days after he receives the Administrative Committee's decision on review of the denial of his claim. Although not required to do so, an Applicant, or his representative, may choose to state the reason or reasons he believes he is entitled to benefits, and may choose to submit written evidence, during the initial claim process or review of claim denial process. However, failure to state any such reason or submit such evidence during the initial claim process or review of claim denial process, shall permanently bar the Applicant, and his successors in interest, from raising such reason or submitting such evidence in any forum at any later date. An Applicant whose claim is denied initially or on review is entitled to receive, on request and free of charge, reasonable access to, and copies of, all documents, records, and other information relevant to such claim for benefits.

- 5.5 <u>Action by Administrative Committee</u>. Action by the Administrative Committee will be subject to the following special rules:
 - (a) <u>Meetings and Documents</u>. The Administrative Committee may act by meeting or by document signed without meeting and documents may be signed through the use of a single document or concurrent documents.
 - (b) <u>Action by Majority</u>. The Administrative Committee will act by a majority decision which action will be as effective as if such action had been taken by all Administrative Committee members, provided that by majority action one or more Administrative Committee members or other persons may be authorized to act with respect to particular matters on behalf of all Administrative Committee members.
 - (c) <u>Resolving Deadlocks</u>. If there is an equal division among the Administrative Committee members with respect to any question a disinterested party may be selected by a majority vote to decide the matter. Any decision by such disinterested party will be binding.
- 5.6 Indemnity. To the extent permitted by applicable law and to the extent that they are not indemnified or saved harmless under any liability insurance contracts, any present or former Administrative Committee members, officers, or directors of Baxter, the Employers or their subsidiaries or affiliates, if any, will be indemnified and saved harmless by the Employers from and against any and all liabilities or allegations of liability to which they may be subjected by reason of any act done or omitted to be done in good faith in the administration of the Plan, including all expenses reasonably incurred in their defense in the event that Baxter fails to provide such defense after having been requested in writing to do so.

ARTICLE VI AMENDMENT AND TERMINATION

- Amendment and Termination. As indicated in Section 5.2 above, the Administrative Committee may, at any time, amend or supplement the Plan. The Board of Directors of the Corporation may, at any time, terminate the Plan. Notwithstanding the foregoing provisions of Sections 5.2 or 6.1, neither an amendment or termination of the Plan shall reduce the benefit to which a Participant would be entitled if he had terminated employment immediately prior to the adoption of the resolution amending or terminating the Plan; provided, however, the Administrative Committee or Corporation, as applicable, may amend or terminate the Plan at any time to take effect retroactively or otherwise, as deemed necessary or advisable for purposes of conforming the Plan to any present or future law, regulations or rulings relating to plans of this or a similar nature. Upon termination of the Plan, all benefits accrued through the date of termination shall be paid as provided herein; provided that the Administrative Committee may, to the extent permitted under Section 409A, provide for the payment of actuarially equivalent lump sums in full satisfaction of some or all of the accrued benefits.
- 6.2 <u>Successors and Assigns</u>. The obligations of the Corporation and the Participating Employers under the Plan shall be binding upon any assignee or successor in interest thereto.

ARTICLE VII MISCELLANEOUS

- 7.1 <u>Unfunded Plan</u>. This Plan is intended to be an unfunded retirement plan maintained primarily to provide retirement benefits for a select group of management or highly compensated employees. All credited amounts are unfunded, general obligations of the appropriate Participating Employer. This Plan is not intended to create an investment contract, but to provide retirement benefits to eligible employees who participate in the Plan. Eligible employees are members of a select group of management or are highly compensated employees, who, by virtue of their position with a Participating Employer, are uniquely informed as to such Participating Employer's operations and have the ability to affect materially Participating Employer's profitability and operations.
- 7.2 <u>Unsecured General Creditor</u>. In the event of a Participating Employer's insolvency, Participants and their Beneficiaries, heirs, successors and assigns will have no legal or equitable rights, interest or claims in any property or assets of such Participating Employer, nor will they be Beneficiaries of, or have any rights, claims or interests in any life insurance policies, annuity contracts or the proceeds therefrom owned or which may be acquired by such Participating Employer (the "Policies") greater than those of any other unsecured general creditors. In that event, any and all of the Participating Employer's assets and Policies will be, and remain, the general, unpledged, unrestricted assets of Participating Employer. Participating Employer's obligation under the Plan will be merely that of an unfunded and unsecured promise of Participating Employer to pay money in the future.
- 7.3 Nonassignability. Neither a Participant nor any other person will have any right to commute, sell, assign, transfer, pledge, anticipate, mortgage or otherwise encumber, transfer, hypothecate or convey in advance of actual receipt the amounts, if any, payable hereunder, or any part thereof, which are, and all rights to which are, expressly declared to be nonassignable and nontransferable. No part of the amounts payable will, prior to actual payment, be subject to seizure or sequestration for the payment of any debts, judgments, alimony or separate maintenance owed by a Participant or any other person, nor be transferable by operation of law in the event of a Participant's or any other person's bankruptcy or insolvency.
- 7.4 Not a Contract of Employment. The terms and conditions of this Plan will not be deemed to constitute a contract of employment between a Participant and such Participant's Participating Employer, and neither the Participant nor the Participant's beneficiary will have any rights against such Participant's Participating Employer except as may otherwise be specifically provided herein. Moreover, nothing in this Plan is deemed to give a Participant the right to be retained in the service of his or her Participating Employer or to interfere with the right of such Participating Employer to discipline or discharge him or her at any time.
- 7.5 <u>Protective Provisions</u>. A Participant will cooperate with the Corporation by furnishing any and all information requested by the Corporation, in order to facilitate the payment of benefits hereunder.

- 7.6 Governing Law. The provisions of this Plan will be construed and interpreted according to the laws of the State of Illinois, to the extent not preempted by ERISA.
- 7.7 <u>Severability</u>. In the event any provision of the Plan is held invalid or illegal for any reason, any illegality or invalidity will not affect the remaining parts of the Plan, but the Plan will be construed and enforced as if the illegal or invalid provision had never been inserted, and the Corporation will have the privilege and opportunity to correct and remedy such questions of illegality or invalidity by amendment as provided in the Plan, including, but not by way of limitation, the opportunity to construe and enforce the Plan as if such illegal and invalid provision had never been inserted herein.
- 7.8 Notice. Any notice or filing required or permitted to be given to the Corporation or the Administrative Committee under the Plan will be sufficient if in writing and hand delivered, or sent by registered or certified mail to any member of the Administrative Committee, or to the Corporation's Chief Financial Officer and, if mailed, will be addressed to the principal executive offices of the Corporation. Notice to a Participant or beneficiary may be hand delivered or mailed to the Participant or beneficiary at his or her most recent address as listed in the employment records of the Corporation. Notices will be deemed given as of the date of delivery or mailing or, if delivery is made by certified or registered mail, as of the date shown on the receipt for registration or certification. Any person entitled to notice hereunder may waive such notice.
- 7.9 <u>Successors</u>. The obligations of the Corporation and the Participating Employers under the Plan shall be binding upon any assignee or successor in interest thereto. The provisions of this Plan will bind and inure to the benefit of the Corporation and the Participating Employers, the Participants and Beneficiaries, and their respective successors, heirs and assigns. The term successors as used herein will include any corporate or other business entity, which, whether by merger, consolidation, purchase or otherwise acquires all or substantially all of the business and assets of the Corporation, and successors of any such corporation or other business entity.
- 7.10 Action by Corporation. Except as otherwise provided herein, any action required of or permitted by the Corporation under the Plan will be by resolution of the Compensation Committee or any person or persons authorized by resolution of the Compensation Committee.
- 7.11 <u>Effect on Benefit Plans</u>. Amounts paid under this Plan, will not by operation of this Plan be considered to be compensation for the purposes of any benefit plan maintained by any Participating Employer. The treatment of such amounts under other employee benefit plans will be determined pursuant to the provisions of such plans.
- 7.12 <u>Participant Litigation</u>. In any action or proceeding regarding the Plan, employees or former employees of the Corporation or a Participating Employer, Participants, Beneficiaries or any other persons having or claiming to have an interest in this Plan will not be necessary parties and will not be entitled to any notice or process. Any final judgment which is not appealed or appealable and may be entered in any such action or proceeding will be binding and conclusive on the parties hereto and all persons having or claiming to have any interest in this Plan. To the extent permitted by law, if a legal action is begun against the Corporation, a

Participating Employer, the Administrative Committee, or any member of the Administrative Committee by or on behalf of any person and such action results adversely to such person or if a legal action arises because of conflicting claims to a Participant's or other person's benefits, the costs to such person of defending the action will be charged to the amounts, if any, which were involved in the action or were payable to the Participant or other person concerned. To the extent permitted by applicable law, acceptance of participation in this Plan will constitute a release of the Corporation, each Participating Employer, the Administrative Committee and each member thereof, and their respective agents from any and all liability and obligation not involving willful misconduct or gross neglect.

* **

IN WITNESS WHEREOF, the undersigned has caused this Amended and Restated Plan to be executed this 18th day of December, 2014.

BAXTER INTERNATIONAL INC. ADMINISTRATIVE COMMITTEE

By:/s/ Salvatore Dadouche

Salvatore Dadouche

Administrative Committee Member

- 20 -

BAXTER INTERNATIONAL INC. AND SUBSIDIARIES

DEFERRED COMPENSATION PLAN

(Amended and restated effective January 1, 2015)

TABLE OF CONTENTS

ARTICLE I	— PURPOSE, EFFECTIVE DATE, EMPLOYER1
1.1	l Purpose1
1.2	Effective Date1
1.3	3 Employer1
1.4	Spin-Off of Baxalta Incorporated1
ARTICLE II	I — DEFINITIONS3
2.1	1 Accounts3
2.2	2 Administrative Committee3
2.3	Beneficiary3
2.4	4 Bonus3
2.5	5 Bonus Deferral3
2.6	6 Code3
2.7	7 Compensation3
2.8	8 Compensation Committee3
2.9	9 Deferral Election Form4
2.1	10 Distribution Election Form4
2.1	11 Eligible Employee4
2.1	12 Employer4
2.1	13 Employer Non-Matching Contribution5
2.1	14 Excess Matching Contribution5
2.1	15 Matching Contribution5
2.1	16 Participant5
2.1	Pay Deferral Contribution5
2.1	18 Plan Year5
2.1	19 Section 409A5
2.2	Termination of Employment5
2.2	21 Unforeseeable Emergency6
2.2	Vesting7
ARTICLE II	II — ELIGIBILITY FOR CONTRIBUTIONS AND DEFERRALS
3.1	1 Excess Matching Contributions8
3.2	2 Bonus Deferral Elections8
3.3	Pay Deferral Elections9
3.4	Somatogen Acquisition Deferral Election9
3.5	5 Discretionary Contributions10
3.6	Employer Non-Matching Contribution10
3.7	7 Contributions Following Military Service10
3.8	8 Mid-Year Deferral Elections10
3.9	Gambro Plans10
ARTICLE Γ	V — CREDITING OF ACCOUNTS12
4.1	1 Crediting of Accounts12
4.2	
4.3	

4.4	Vesting13
ARTICLE V —	DISTRIBUTION OF BENEFITS14
5.1	Distribution of Benefits14
5.2	Distribution14
5.3	Effect of Payment17
5.4	Taxation of Plan Benefits17
5.5	Withholding and Payroll Taxes17
5.6	Distribution Due to Unforeseeable Emergency17
5.7	Distribution Due to Inclusion in Taxable Income 18
5.8	Distribution of De Minimis Amounts18
5.9	Correction of Errors18
ARTICLE VI —	- BENEFICIARY DESIGNATION19
6.1	Beneficiary Designation19
6.2	Amendments to Beneficiary Designation19
6.3	No Beneficiary Designation19
6.4	Form of Payment to Beneficiary19
ARTICLE VII –	– ADMINISTRATION20
7.1	Administrative Committee 20
7.2	Administrative Committee Powers20
7.3	Effect of Administrative Committee Decisions21
7.4	Claims Procedure21
7.5	Action by Administrative Committee22
7.6	Indemnity22
ARTICLE VIII -	— AMENDMENT AND TERMINATION OF PLAN23
8.1	Amendment23
8.2	Right to Terminate24
8.3	Payment at Termination24
ARTICLE IX —	- MISCELLANEOUS25
9.1	Unfunded Plan25
9.2	Unsecured General Creditor25
9.3	Nonassignability25
9.4	Not a Contract of Employment26
9.5	Protective Provisions26
9.6	Governing Law26
9.7	Severability26
9.8	Notice26
9.9	Successors26
9.10	Action by Baxter27
9.11	Effect on Benefit Plans27
9.12	Participant Litigation 27

APPENDIX A — PARTICIPATING EMPLOYERS28

APPENDIX B — SUPPLEMENTAL PAY DEFERRALS UNDER SECTION 3.329

APPENDIX C — DISCRETIONARY EMPLOYER CONTRIBUTIONS UNDER SECTION 3.530

 $APPENDIX\ D\ -SPECIAL\ DISTRIBUTION\ PROVISIONS\ APPLICABLE\ TO\ AMOUNTS\ TRANSFERRED\ FROM\ GAMBRO\ PLANS31$

BAXTER INTERNATIONAL INC. AND SUBSIDIARIES DEFERRED COMPENSATION PLAN

(Amended and Restated Effective January 1, 2015)

ARTICLE I — PURPOSE, EFFECTIVE DATE, EMPLOYER

1.1 Purpose.

The Baxter International Inc. and Subsidiaries Deferred Compensation Plan (the "Plan") has been adopted by Baxter International Inc. ("Baxter"). The Plan is intended to be an unfunded arrangement to provide deferred compensation for the benefit of a select group of management and highly compensated employees. The Plan is designed to enable eligible participants to defer compensation and receive matching contributions under the provisions of the Baxter International Inc. and Subsidiaries Incentive Investment Plan ("IIP"), a tax-qualified defined contribution plan, in excess of the limitations imposed by the Internal Revenue Code ("Code"). Baxter amended and restated the Plan effective January 1, 1998, in part to combine the Plan and the Baxter International Inc. and Subsidiaries Incentive Investment Excess Plan, and amended and restated the Plan again effective January 1, 2002, January 1, 2005, January 1, 2007, and January 1, 2009. The Plan is hereby further amended and restated effective January 1, 2015. Capitalized terms not defined in this Plan are deemed to have the meaning given them in the IIP.

1.2 Effective Date.

The effective date of this restatement is January 1, 2015, except as otherwise provided herein; provided that any provision of the Plan that is required to be effective as of an earlier in order to comply with Section 409A of the Code shall be effective as of such date.

1.3 Employer.

The Plan is adopted for the benefit of a select group of management or highly compensated employees of Baxter or of any subsidiaries or affiliates of Baxter, as set forth below. The Plan may be adopted by any subsidiaries or affiliates of Baxter with the consent of the Administrative Committee. Participating Employers are listed on Appendix A as attached and updated from time to time.

1.4 Baxalta Incorporated Spin-Off.

In connection with the Company's spin-off of its biopharmaceutical business, the Company underwent an internal reorganization and incorporated Baxalta Incorporated ("Baxalta") as a subsidiary of the Company. The Company anticipates that Baxalta shall be spun-off on or around July 1, 2015 (the "Spin-Off") pursuant to a Separation and Distribution Agreement (the "Agreement"). Effective as of the Spin-Off (the "Spin-Off Date"), Baxalta shall be an independent, publicly traded corporation which owns and operates the biopharmaceutical business previously owned and operated by the Company. As described in the Agreement and the Employee Matters Agreement, Baxalta shall establish the Baxalta Incorporated Deferred Compensation Plan (the "Baxalta Plan"), an unfunded nonqualified deferred compensation plan that generally mirrors the Plan, to provide for the benefits of the following Employees who were

Participants immediately prior to the date such Employee transfers to employment with Baxalta from the Company:

- (a) An Employee (including an Employee who is on an approved leave of absence from the Company) who transfers to employment with Baxalta, or one of its subsidiaries, from the Company on or before the Spin-Off Date;
- (b) An Employee who transfers to employment with Baxalta, or one of its subsidiaries, from the Company in accordance with the Transition Services Agreement.

As described in the Agreement and the Employee Matters Agreement, Baxalta shall assume the obligation to pay all benefits accrued under the Plan for the benefit of each Employee who transfers to employment with Baxalta as described in this Section (a "Baxalta Participant"). As a result of the Spin-Off, a Baxalta Participant shall cease to have any benefit under the Plan following his transfer of employment to Baxalta, and shall instead participate under the Baxalta Plan. To the maximum extent permitted by Treas. Reg. §1.409A-1(h)(4), a Baxalta Participant shall not be considered to have incurred a Termination of Employment for purposes of the Plan or a separation from service as defined in Section 409A solely as a result of the transfer describe above.

ARTICLE II — DEFINITIONS

2.1 Accounts.

Accounts mean the sum of the Participant's Excess Matching Contribution Account balance, Bonus Deferral Account balance, Pay Deferral Account balance, and Deferred Compensation Account balance.

2.2 Administrative Committee.

For purposes of the Plan, Administrative Committee has the same meaning as the Administrative Committee in the IIP.

2.3 Beneficiary.

A Participant's Beneficiary, as defined in Article VI, is the Beneficiary designated to receive the Participant's Accounts, if any, from the Plan, upon the death of the Participant.

2.4 Bonus.

The term Bonus means those bonuses that are included in the definition of Compensation in the IIP and also includes any other bonus which is approved by the Administrative Committee and listed on Attachment A to this Plan. Attachment A may be updated from time to time to accurately reflect the approved bonuses for purpose of this definition.

2.5 Bonus Deferral.

The Bonus Deferral is the amount of the Participant's Bonus which the Participant elected to defer and contribute to the Plan which, but for such election, would have otherwise been paid to him/her.

2.6 *Code.*

The Code shall mean the Internal Revenue Code of 1986, as amended.

2.7 Compensation.

For purposes of the Plan, Compensation has the same meaning as Compensation in the IIP without regard to Section 401(a)(17) of the Code, except that the Bonuses deferred under the Plan are included in Compensation in the Plan Year in which such amounts would be paid if they were not deferred and not in the Plan Year in which such amounts are actually paid. In no event shall Compensation include any amount payable after a Participant has terminated employment.

2.8 Compensation Committee.

The Compensation Committee of the Board of Directors of Baxter.

2.9 Deferral Election Form.

The form which a Participant must complete and return to the Administrative Committee or its designee, in accordance with the rules and procedures as may be established by the Administrative Committee, in order to elect to defer a portion of his or her Bonus into the Plan and to designate his or her Pay Deferral Election.

2.10 Distribution Election Form.

The form which a Participant must complete and return to the Administrative Committee or its designee, in accordance with the rules and procedures as may be established by the Administrative Committee. This form is to be used by both (a) Participants who are not eligible to defer a portion of their Bonus or make a Pay

Deferral Contribution to the Plan; and (b) Participants who are electing distributions with respect to a Deferred Compensation Account.

2.11 Eligible Employee.

For any Plan Year, an Eligible Employee is anyone who:

- (a) is a Corporate Officer of Baxter, a member of Baxter's Global Leadership Team and/or is participant in the Baxter International Inc. Long Term Incentive Plan for the Plan Year to which deferrals relate;
- (b) is a former participant in the Baxter International Inc. Long Term Incentive Plan;
- (c) for Plan Years prior to 2005, was a participant in the IIP whose Matching Contributions to the IIP for the Plan Year were limited because of the application of the Code, provided he or she met the eligibility rules under Section 3.1 as in effect for such Plan Year;
- (d) solely for purposes of Section 3.5, is designated by the Administrative Committee to be a Participant in the Plan and eligible to receive discretionary benefits under Section 3.5 of the Plan for the Plan Year, subject to the terms and conditions imposed by the Administrative Committee in accordance with Section 3.5; or
- (e) for Plan Years subsequent to 2006, and solely for purposes of Section 3.6, is eligible to receive an Employer Non-Matching Contribution into the IIP for the Plan Year and has Compensation for the Plan Year in excess of the limitations of Section 401(a)(17) of the Code. An Employee who has never previously been an Eligible Employee shall be treated as becoming an Eligible Employee on the last day of the first Plan Year in which he meets the requirements of this paragraph (d).

2.12 Employer.

The term Employer means Baxter and any entity that is a member of a controlled group or affiliated service group that includes Baxter, or is otherwise required to be considered as a single employer with Baxter under Section 414 of the Code. A "Participating Employer" is an

Employer that has adopted the Plan for the benefit of its Eligible Employees as provided in Section 1.3, and a Non-Participating Employer is an Employer that is not a Participating Employer.

2.13 Employer Non-Matching Contribution.

The term Employer Non-Matching Contribution has the same meaning in the Plan as it does in the IIP.

2.14 Excess Matching Contribution.

The Excess Matching Contribution is the difference between the Matching Contributions allocated to a Participant's IIP Account during the Plan Year and the amount that would have been allocated if the limitations of Sections 415, 401(k), 402(g), 401(m) or 401(a)(17) of the Code, were disregarded.

2.15 Matching Contribution.

The term Matching Contribution has the same meaning in the Plan as it does in the IIP.

2.16 Participant.

A Participant is any Eligible Employee who has an Account balance in the Plan.

2.17 Pay Deferral Contribution.

The term Pay Deferral Contribution has the same meaning as Pay Deferral Contribution in the IIP. The Pay Deferral Contribution is the amount of the Participant's Compensation, which the Participant elected to defer into the Plan which, but for such election, would have otherwise been paid to him/her.

2.18 Plan Year.

The Plan Year is the calendar year.

2.19 Section 409A.

Section 409A means Section 409A of the Code, as enacted by the American Jobs Creation Act of 2004 and as interpreted by Treasury Regulations or other authority issued thereunder.

2.20 Termination of Employment.

For purposes of the Plan, Termination of Employment has the same meaning as Termination of Employment in the IIP; provided that for purposes of determining when a Participant's benefit becomes payable, Termination of Employment shall not be considered to have occurred until the Participant incurs a separation from service as defined in Treasury Regulations issued pursuant to Section 409A. The following rules are intended to implement the requirements of Section 409A, and may be adjusted by the Administrative Committee as required to comply with any guidance issued under Section 409A:

- (a) The Participant shall not be considered to have separated from service so long as the Participant is on military leave, sick leave, or other bona fide leave of absence if the period of such leave does not exceed six months, or if longer, so long as the Participant retains a right to reemployment with a Participating Employer under an applicable statute or by contract.
- (b) Regardless of whether his employment has been formally terminated, the Participant will be considered to have separated from service as of the date it is reasonably anticipated that no further services will be performed by the Participant for any Participating Employer, or that the level of bona fide services the Participant will perform after such date will permanently decrease to no more than 20 percent of the average level of bona fide services performed over the immediately preceding 36-month period (or the full period of employment if the Participant has been employed for less than 36 months). For purposes of the preceding test, during any paid leave of absence the Participant shall be considered to have been performing services at the level commensurate with the amount of compensation received, and unpaid leaves of absence shall be disregarded.
- (c) For purposes of determining whether the Participant has separated from service, all services provided for any Employer, or for any entity that is a member of the Controlled Group, shall be taken into account, whether provided as an employee or as a consultant or other independent contractor; provided that the Participant shall not be considered to have not separated from service solely by reason of service as a non-employee director of the Corporation or any other such entity. Solely for purposes of this Section 2.19, the term "Controlled Group" shall be modified by substituting "50 percent" for "80 percent" for all purposes of section 414(b) and (c) of the Code (and Section 1563 to the extent incorporated therein).
- (d) A Participant who is employed by a Participating Employer, and continues to be employed by the Participating Employer following a stock sale, spin-off, or other transaction that causes the Participant's employer to cease to be a member of the Controlled Group, shall not be considered to have incurred a Termination of Employment as a result of such transaction. A Participant who ceases to be employed by the Corporation or any member of the Controlled Group as a result of a sale of substantially all of the assets constituting a division, facility, or separate line of business, shall be considered to have incurred a Termination of Employment unless the Corporation (or Participating Employer selling such assets) and the purchaser agree in writing, not later than the closing date of such transaction, that all Participants affected by such transaction shall not be considered to have incurred a Termination of Employment, and that the purchaser agrees to assume the obligation for payment of the Benefits of all such Participants in accordance with the Plan.

2.21 Unforeseeable Emergency.

A severe financial hardship resulting from a sudden or unexpected illness or accident of the Participant or one of his or her dependents, loss of the Participant's property due to casualty or similar extraordinary and unforeseeable circumstances arising as a result of one or more recent events beyond the control of the Participant, as determined by the Administrative Committee.

2.22 Vesting.

For purposes of the Plan, Vesting has the same meaning as Vesting in the IIP.

ARTICLE III — ELIGIBILITY FOR CONTRIBUTIONS AND DEFERRALS

3.1 Excess Matching Contributions.

The Excess Matching Contributions Account of an Eligible Employee who makes either a Bonus Deferral or Pay Deferral Election for a Plan Year shall be credited with Excess Matching Contributions equal to the lesser of the total amount deferred pursuant to Sections 3.2 and 3.3 (not including any Supplemental Pay Deferral) for the Plan Year or three and one half percent (3.5%) of the excess of the Participant's total Compensation for the Plan Year over the portion of the Participant's Compensation taken into account under the IIP for the Plan Year. Prior to January 1, 2005, an Employee was an Eligible Employee if the Employee's Matching Contributions under the IIP were less than 3% of the Employee's Compensation for the Plan Year, regardless of whether the Employee were eligible to make Bonus Deferral and/or Pay Deferral Elections. A Participant who was an Eligible Employee solely by reason of this Section 3.1 for one or more Plan Years prior to 2005 shall continue to be a Participant with respect to his Excess Matching Contribution Account until it is distributed.

3.2 Bonus Deferral Elections.

An Eligible Employee for a Plan Year may elect to defer all or a portion of his or her Bonus for the Plan Year through the Plan until his or her Termination of Employment, or such other time as specified on his or her Deferral Election Form, by completing a Deferral Election Form in accordance with applicable rules and procedures established by the Administrative Committee. A Participant may elect to defer up to 100% of his or her Bonus, in whole percentages. Beginning January 1 of the year to which the Deferral Election Form applies, the Deferral Election Form is irrevocable, except as provided in Section 5.6. The Deferral Election Form must be filed in accordance with the rules established by the Administrative Committee, at the time set forth below:

- (a) Deferral Election Forms must be filed prior before January 1 of the Plan Year in which the Bonus is earned, except as hereinafter provided.
- (b) The Administrative Committee may permit an employee who becomes an Eligible Employee for the first time during a Plan Year to make an election to defer his or her Bonus for such Plan Year not more than 30 days after becoming an Eligible Employee, which Bonus Deferral Election shall apply only to the portion of the Bonus earned after the election is made. An Eligible Employee shall not be eligible to make the election within the first 30 days after becoming an Eligible Employee if the employee has been a participant (other than through accrual of earnings on amounts previously deferred) in any account balance deferred compensation arrangement sponsored by any Employer during the 24 month period prior to the date he or she becomes an Eligible Employee, unless the employee received a distribution of his or her entire balance in such plan during such 24 month period, and immediately prior to such distribution was not eligible to continue to participate in such plan.
- (c) The Administrative Committee may also permit Eligible Employees to make an election to defer their Bonuses not later than six months prior to the end of the Bonus

determination period, provided that the Administrative Committee determines that the Bonus satisfies the requirements for performance based compensation under Section 409A of the Code.

3.3 Pay Deferral Elections.

An Eligible Employee may make a Pay Deferral Election under the Plan if he or she elects to defer a portion of his or her Compensation under the IIP for a Plan Year, and the amount of Compensation that he or she has elected to defer exceeds the amount that is permitted to be deferred under the IIP by reason of the annual contribution limit under Section 415 or 402(g) of the Code (as increased, if applicable, by the limit on catch-up contributions pursuant to Section 414(v) of the Code), or the fact that the Eligible Employee's Compensation exceeds the annual limit under Section 401(a)(17) of the Code, A Pay Deferral Election shall be made by the last day of the Plan Year preceding the Plan Year to which it relates, in accordance with applicable rules and procedures established by the Administrative Committee, and shall thereafter be irrevocable (except as provided in Section 5.6), except that the Administrative Committee may permit an employee who first becomes an Eligible Employee during a Plan Year, and who meets the requirements of Section 3.2(b), to make an Pay Deferral Election not more than 30 days after becoming an Eligible Employee, which Pay Deferral Election shall apply prospectively only. If an Eligible Employee makes a Pay Deferral Election for a Plan Year, then all amounts that the Eligible Employee elected to defer under the IIP (based upon the Eligible Employee's IIP deferral election at the beginning of the Plan Year, which cannot be changed during the Plan Year) that exceed one or more of the limits described above shall instead be credited to his or her Account in this Plan, commencing with the first payment of Compensation that would cause the amount deferred to exceed such limits. Notwithstanding the foregoing provisions of this Section 3.3, the Administrative Committee, in its sole discretion, may permit a Participant to defer a percentage of his or her Compensation to the Plan for any Plan Year that exceeds the percentage that the Participant elects to defer under the IIP for such Plan Year (a "Supplemental Pay Deferral"), provided that the Supplemental Pay Deferral election is made when the Participant is otherwise eligible to make a Pay Deferral Election as described above and is thereafter irrevocable (except as provided in Section 5.6). To the extent that the Administrative Committee exercises its discretionary authority under the prior sentence, such exercise shall be reflected in Appendix B to the Plan which shall identify each Participant designated as eligible to make Supplemental Pay Deferrals, specify the Plan Year(s) for which Supplemental Pay Deferrals may be made, and reflect any other conditions and limitations applicable with respect to such Supplemental Pay Deferrals. In no event shall Supplemental Pay Deferrals be eligible for Excess Matching Contributions.

3.4 Somatogen Acquisition Deferral Election.

Any former employee of Somatogen, Inc. who became an employee of Baxter International Inc. as of the closing date of the merger agreement between Baxter and Somatogen and who completed a Special Deferral Enrollment Form shall have such form recognized as a valid election under the Plan. Deferrals authorized under this section shall be treated as deferrals authorized under Section 3.2 for purposes of accounting and distribution.

3.5 Discretionary Contributions.

The Administrative Committee may, in its sole discretion, specify such additional amounts in the form of employer contributions to be credited to the Account of a Participant or another employee who is a member of a select group of management and highly compensated employees, subject to such terms and conditions as the Administrative Committee may establish. To the extent that the Administrative Committee exercises its discretionary authority under this Section 3.5, such exercise shall be reflected in Appendix C to the Plan, which shall identify each Participant credited with such discretionary employer contributions, specify the Plan Year(s) for which contributions relate, and reflect any other limitations applicable with respect to such discretionary contributions, including any applicable Vesting requirements. Discretionary employer contributions authorized under this section shall be treated as deferrals authorized under Section 3.2 for purposes of accounting and distribution.

3.6 Employer Non-Matching Contribution.

For any Plan Year after 2006, an Eligible Employee who (i) is eligible to receive an Employer Non-Matching Contribution into the IIP for the Plan Year and (ii) has Compensation for the Plan Year in excess of the limitations of Section 401(a)(17) of the Code, shall receive a contribution equal to 3% of the Eligible Employee's Compensation in excess of the limitations of Section 401(a)(17) of the Code.

3.7 Contributions Following Military Service.

A Participant who incurs a Termination of Employment, or a leave of absence, in order to serve in the armed forces of the United States, who is entitled to re-employment rights under the Uniformed Services Employment and Reemployment Rights Act ("USERRA"), and who is re-employed during the period in which such re-employment rights are protected, shall be entitled to increase the percentage of his or her Compensation subject to a Pay Deferral Election in order to make up the Pay Deferral Contributions missed during the period of military service, in accordance with rules established by the Administrative Committee in accordance with USERRA and Section 409A. Such a Participant shall also be entitled to receive the same amount of Excess Matching Contributions he or she would have received had the additional Pay Deferral Contributions been made during the period of military service. A Participant who is otherwise eligible for Employer Non-Matching Contributions shall be entitled to receive the Employer Non-Matching Contributions he or she would have received had he or she been employed at the same rate of Compensation during the period of military service, which shall be credited to the Deferred Compensation Account not later than 90 days after re-employment.

3.8 Mid-Year Deferral Elections.

If a person becomes an Eligible Employee as defined in Section 2.11(a) during a plan, and such person has not been eligible to elect to defer compensation under this Plan or any other nonqualified deferred compensation maintained by any Employer for the period of twenty-four months ending on the date he becomes an Eligible Employee (or the period beginning on the date any balance in this Plan or any such other plan was distributed to him and ending on the date he becomes an Eligible Employee), such Eligible Employee may be eligible to make either a Pay

Deferral Election or Bonus Deferral Election in accordance with this Section. After the Compensation Committee has identified the persons who have become Eligible Employees as defined in Section 2.11(a), the Administrative Committee shall establish a period during which such elections may be made, which election period shall be no more than thirty (30) days in length and end on or prior to the first day of a payroll period specified by the Administrative Committee. During such election period, each such Eligible Employee may make a Pay Deferral Election that will apply to all Compensation earned during such Plan Year commencing with such specified payroll period. During such period, each such Eligible Employee may also make a Bonus Deferral Election for his/her Bonus for such Plan Year, provided that, unless such election is also permissible pursuant to Section 3.5(c), the portion of his/her Bonus deferred shall not exceed a fraction, the numerator of which is the number of days in the Plan Year commencing with the first day of the specified payroll period and the denominator of which is the total number of days in the Plan Year. The Administrative Committee may also permit other deferral elections to be made in the Plan Year during which a person first becomes an Eligible Employee in accordance with Treas. Reg. §1.409A-2(a)(7).

3.9 Gambro Plans.

Effective as of January 1, 2015, the Plan shall assume the liability to pay all compensation deferred as of such date pursuant to the Gambro Renal Products, Inc., Executive Retirement Plan and the Gambro Renal Products, Inc. Voluntary Deferral Plan (collectively the "Gambro Plans"). A separate Gambro Plan Account shall be established for each Employee who had an account in either of the Gambro Plans on December 31, 2014, and each such Employee shall be considered a Participant with respect to such Gambro Plan Account, regardless of whether he is otherwise an Eligible Employee, but shall be eligible to be credited with further deferrals and allocations of Employer contributions only if he otherwise satisfies the requirements of the Plan. Anything else contained in the Plan to the contrary notwithstanding, all Gambro Plan Account balances shall be distributed only at the time and in the manner provided in the applicable Gambro Plan, as summarized in Appendix D, and no change to the time or form of distribution shall be made by reason of the assumption by the Plan of the balances under the Gambro Plans, except as otherwise determined by the Administrative Committee and to the extent permitted by Section 409A. Commencing with 2015, the earnings on the Gambro Plan Accounts shall be determined under Section 4.2, subject to such transitional procedures as the Administrative Committee. A Participant's Beneficiary with respect to his Gambro Plan Account shall be the person designed as beneficiary pursuant to the applicable Gambro Plan until changed by the Participant in accordance with the Plan.

ARTICLE IV — CREDITING OF ACCOUNTS

4.1 Crediting of Accounts.

- A. *Excess Matching Contribution Account.* An account equal to the Excess Matching Contributions, if any, of each Participant made for Plan Years prior to 2002, as adjusted for investment return under Section 4.2 and distributions under Article V.
- B. *Bonus Deferral Account.* An account equal to the Bonus Deferrals, if any, of each Participant made for Plan Years prior to 2002, as adjusted for investment return under Section 4.2 and distributions under Article V.
- C. Pay Deferral Account. An account equal to the Pay Deferral Contributions and Supplemental Pay Deferrals, if any, of each Participant made for Plan Years prior to 2002, as adjusted for investment return under Section 4.2 and distributions under Article V.
- D. *Deferred Compensation Account.* An account equal to the Excess Matching Contributions, Pay Deferral Contributions, Bonus Deferrals, Supplemental Pay Deferrals and Employer Non-Matching Contributions made for the 2002 Plan Year and thereafter, as adjusted for investment return under Section 4.2 and distributions under Article V.

Notwithstanding the foregoing provisions of this Section 4.1, if elected by the Participant in accordance with rules established by the Administrative Committee, the Participant may elect to have his or her Excess Matching Contributions, Pay Deferral Contributions, Bonus Deferrals and Supplemental Pay Deferrals made for the 2001 Plan Year, if any, credited to his or her Deferred Compensation Account under paragraph D, instead of to the Excess Matching Contribution Account, Bonus Deferral Account and Pay Deferral Account described in paragraphs A, B and C.

Further, effective January 1, 2002, notwithstanding the forgoing provisions of this Section 4.1, if elected by the Participant in accordance with rules established by the Administrative Committee, the Participant may make a one-time election to have amounts credited to his or her Excess Matching Contribution Account, Bonus Deferral Account and Pay Deferral Account (including Supplemental Pay Deferrals) credited to his or her Deferred Compensation Account under paragraph D, provided however, that such election is made prior to 2002 and such amounts are not scheduled to be distributed in 2001.

4.2 Earnings.

Each Participant's Accounts will be adjusted for investment return, on a daily basis, in accordance with the following provisions of this Section 4.2:

A. Amounts in a Participant's Excess Matching Account, Bonus Deferral Account and Pay Deferral Account will be credited with earnings at a rate determined by the Administrative Committee from time to time. Until the Administrative Committee determines otherwise, such earnings will be credited at the same rate as the Stable Income Fund in the IIP.

B. Amounts in a Participant's Deferred Compensation Account shall be adjusted upward or downward to reflect the investment return that would have been realized had such amounts been invested in one or more investments selected by the Participant from among the assumed investment alternatives designated by the Administrative Committee for use under the Plan. Prior to the first day of each month, or at such other times as the Administrative Committee may permit, Participants may change the assumed investment alternatives in which their Deferred Compensation Account will be deemed invested for such Plan Year. Participant elections of assumed investment alternatives shall be made at the time and in the form determined by the Administrative Committee, and shall be subject to such other restrictions and limitations as the Administrative Committee shall determine. In the event that a Participant fails to make an investment election, his or her Deferred Compensation Account shall be credited with earnings in the same manner as provided in paragraph A above.

4.3 Account Statements.

Account Statements will be generated effective at such intervals as the Administrative Committee may determine and transmitted to each Participant as soon as administratively feasible. Account Statements will reflect all Account activity during the reporting period, including Account contributions, distributions and earnings credits.

4.4 Vesting.

Subject to Sections 9.1 and 9.2, and any Vesting requirements specified by the Administrative Committee with respect to Discretionary Contributions, a Participant is always 100% Vested in his or her Accounts in the Plan at all times; provided, however, that if a Participant who incurs a Termination of Employment is not 100% Vested in his or her Employer Non-Matching Contribution Account in the IIP, the portion of his or her Deferred Compensation Account attributable to Employer Non-Matching Contributions and the earnings thereon shall be forfeited, and no Participating Employer shall have any obligation to the Participant with respect to such portion.

ARTICLE V — DISTRIBUTION OF BENEFITS

5.1 Distribution of Benefits.

Subject to Section 5.2, distribution of a Participant's Accounts, if any, will commence in accordance with the Participant's Distribution Election Form or Deferral Election Form as soon as administratively feasible after the Participant's Termination of Employment. Any spousal consent requirements under the IIP will not apply to distributions under the Plan.

Anything else in this Plan to the contrary notwithstanding, effective October 22, 2004, (i) in no event shall the distribution of any Account be accelerated to a time earlier than which it would otherwise have been paid, whether by amendment of the Plan, exercise of the Administrative Committee's discretion, or otherwise, except as permitted by Treasury Regulations issued pursuant to Section 409A, and (ii) in the event that the Administrative Committee, in its sole discretion, determines that any time or form of distribution provided for in the Plan, or the existence of a right to elect a different time or form of distribution, would cause the Plan to fail to meet the requirements of Section 409A, or otherwise cause Participants to be subject to any adverse federal income tax consequences, the Administrative Committee shall amend the Plan to modify or remove the form of distribution or election right. The distribution restrictions under Section 409A shall apply to Participant's entire account balances under the Plan, whether deferred before or after January 1, 2005. Notwithstanding the foregoing, the Administrative Committee may give Participants a one-time opportunity to change the time and/or form of payment of their Accounts by a written irrevocable election made not later than December 31, 2008, subject to such terms and conditions as the Administrative Committee may require; provided that no such election shall cause any amount to be paid in 2008 that would otherwise have been paid in a later year, or cause any amount that would otherwise have been paid in 2008 to be paid in a later year, and such elections shall otherwise comply with the requirements for transitional relief under IRS Notice 2007-86.

5.2 Distribution.

- A. Deferral Election Form. A Participant's Excess Matching Contribution Account, Bonus Deferral Account and Pay Deferral Account will be paid in accordance with the form of payment designated in the Participant's Deferral Election Form. The Deferral Election Form shall not be used to elect forms of distribution with respect to deferrals for Plan Years after 2001 (or 2000, with respect to a Participant electing to have his or her deferrals credited to the Deferred Compensation Account for Plan Year 2000 under Section 4.1).
- B. Distribution Election Form Termination of Employment. A Participant's Deferred Compensation Account and, if the Participant is not eligible for Pay Deferrals or Bonus Deferrals, his or her Excess Matching Contribution Account, will be paid after the Participant's Termination of Employment, in accordance with the form of payment designated in such Participant's Distribution Election Form. Distribution Election Forms shall be filed in accordance with rules established by the Administrative Committee, subject to the following:
 - (a) Prior to January 1, 2007, only one Distribution Election Form could be submitted with respect to distribution of a Participant's Deferred Compensation Account

- following Termination of Employment. Any such Distribution Election Form filed prior to January 1, 2007, shall remain in effect shall apply to the Participant's entire Deferred Compensation Account (and Excess Matching Contribution Account if applicable) balance at his or her Termination of Employment.
- (b) Effective January 1, 2007, a Participant who has not previously been described in paragraph (c) may submit a Distribution Election Form at the time he or she first makes a Bonus Deferral or Pay Deferral Election pursuant to Section 3.2 or 3.3. Except as otherwise provided in subparagraph (c) below, only one Distribution Election Form shall be filed, which shall apply to the Participant's entire Deferred Compensation Account balance at his or her Termination of Employment. A Distribution Election Form must be filed by the end of the period for making the Participant's first Bonus Deferral or Pay Deferral Election, and if the Participant fails to file a Distribution Election Form at such time his or her entire Deferred Compensation Account balance shall be distributed in a lump sum at his or her Termination of Employment, or in accordance with a Distribution Election Form previously filed pursuant to subparagraph (c) if applicable.
- An Employee who first becomes an Eligible Employee pursuant to Section 3.1, 3.5, or 3.6 on or after January 1, 2007, and who meets the requirements of Section 3.2(b), may file a Distribution Election Form not later than 30 days after his or her first day of eligibility. Except as provided in the following sentence, only one Distribution Election Form shall be filed, which shall apply to the Participant's entire Deferred Compensation Account balance at his or her Termination of Employment, and if the Participant fails to file a Distribution Election Form at such time his or her entire Deferred Compensation Account balance shall be distributed in a lump sum at his or her Termination of Employment. Notwithstanding the foregoing, if such a Participant subsequently becomes eligible to make a Bonus Deferral or Pay Deferral Election, he or she may file a new Distribution Election Form pursuant to subparagraph (b) above. In such event, the portion of the Participant's Deferred Compensation Account that represents amounts credited to the Deferred Compensation Account under all provisions of Article III beginning with the first Plan Year to which the Bonus Deferral or Pay Deferral Election applies (and all earnings thereon) shall be distributed in accordance with such Distribution Election Form, and the remaining portion of the Deferred Compensation Account shall continue to be governed by this subparagraph (c).
- C. Forms of Distribution. The forms of distribution are:
- (a) a lump sum payment, or
- (b) annual installments of at least 2 years, but not to exceed 15 years.

If annual installments are elected, the amount of each installment will be equal to the remaining balance in the Participant's Account prior to payment of the installment, divided by the remaining number of installments to be paid (including the installment being calculated).

Except as provided below, effective January 1, 2007, lump sum payments will be paid, and annual installments will commence, in the first quarter of the Plan Year as specified in the Participant's Deferral Election Form or Distribution Election Form (or, if the Distribution Election Form provides for payments following a Termination of Employment, in the first quarter of the Plan Year following the Plan Year in which the Termination of Employment occurs). Subsequent installments will be paid annually in the first quarter of subsequent Plan Years. In the case of installment payments which commenced prior to January 1, 2007, the installment that would otherwise have been paid in the third quarter of 2007 shall be paid in the first quarter, and all installments shall thereafter be paid in the first quarter of subsequent years.

If a Participant does not elect a form of distribution by the time the Deferral Election Form or the Distribution Election Form is required to be completed, the Participant's election will default to a lump sum payment in the first quarter of the Plan Year following the Plan Year in which the Participant incurs a Termination of Employment.

- D. Special Rules. Notwithstanding the foregoing:
- (a) A Participant whose Accounts under the Plan total less than \$50,000 as of the last day of the Plan Year in which he or she incurs a Termination of Employment will receive lump sum payment of his or her Accounts in the first quarter of the Plan Year following the Plan Year in which the Participant incurs a Termination of Employment.
- (b) If a Participant who has made a Bonus Deferral election for a Plan Year incurs a Termination of Employment during the Plan Year, but is still eligible for a Bonus for the Plan Year, the deferred portion of his or her Bonus shall be distributed during March of the subsequent year, regardless of the form of distribution otherwise elected, and shall not be taken into account in determining whether the Participant's Account Balance is less than \$50,000. Such amount shall not be credited with any earnings unless paragraph (c) applies.
- (c) Anything else contained herein to the contrary notwithstanding, in no event shall any payment of a benefit made in connection with the Termination of Employment of a "specified employee", as hereinafter defined, be made until at least six months following such Termination of Employment, and any amounts that would otherwise have been paid during such six month period shall be accumulated and paid in a lump sum, without interest, on the first business day following the expiration of such period. For purposes of this Plan, the term "specified employee shall have the meaning set forth in Treas. Reg. §1.409A-1(i), using the safe harbor definition of compensation contained in Treas. Reg. §1.415(c)-2(d)(4) (compensation required to be reported on Form W-2 plus elective deferrals) and excluding compensation paid to a nonresident alien that is not effectively connected with the conduct of a trade or business within the

United States shall be excluded. The status of Participants as specified employees shall be determined as of December 31 of each year, and if a Participant is determined to be a specified employee on any December 31, the restriction of clause (ii) shall apply if and only if he incurs a termination of employment at any time during the twelve month period commencing on the following February 1.

5.3 Effect of Payment.

Payment to the person or trust reasonably and in good faith determined by the Administrative Committee to be the Participant's Beneficiary will completely discharge any obligations Baxter or any other Employer may have under the Plan. If a Plan benefit is payable to a minor or a person declared to be incompetent or to a person the Administrative Committee in good faith believes to be incompetent or incapable of handling the disposition of property, the Administrative Committee may direct payment of such Plan benefit to the guardian, legal representative or person having the care and custody of such minor and such decision by the Administrative Committee is binding on all parties. The Administrative Committee may initiate whatever action it deems appropriate to ensure that benefits are properly paid to an appropriate guardian.

The Administrative Committee may require proof of incompetence, minority, incapacity or guardianship as it may deem appropriate prior to distribution of the Plan benefit. Such distribution will completely discharge the Administrative Committee and the Employer from all liability with respect to such benefit.

5.4 Taxation of Plan Benefits.

It is intended that each Participant will be taxed on amounts credited to him or her under the Plan at the time such amounts are received, and the provisions of the Plan will be interpreted consistent with that intention.

5.5 Withholding and Payroll Taxes.

Baxter will withhold from payments made hereunder any taxes required to be withheld for the payment of taxes to the Federal, or any state or local government.

5.6 Distribution Due to Unforeseeable Emergency.

Upon written request of a Participant and the showing of Unforeseeable Emergency, the Administrative Committee may authorize distribution of all or a portion of the Participant's Accounts, and or the acceleration of any installment payments being made from the Plan, but only to the extent reasonably necessary to relieve the Unforeseeable Emergency, taking into account the tax imposed on such distribution. In any event, payment may not be made to the extent such Unforeseeable Emergency is or may be satisfied through reimbursement by insurance or otherwise, including, but not limited to, liquidation of the Participant's assets, to the extent that such liquidation would not in and of itself cause severe financial hardship. If a Participant has an Unforeseeable Emergency, the Participant's Pay Deferral Election and Bonus Deferral Election, if any, shall be revoked for the Plan Year (and no subsequent Pay Deferral or Bonus Deferral may be made for the same Plan Year), and the additional income resulting from

such revocation shall be taken into account in determining the amount of distribution reasonably necessary to relieve the Unforeseeable Emergency. A Participant shall not be required to take any hardship withdrawal or loan to which he is entitled under the IIP or any other tax qualified retirement plan as a condition of receiving a distribution pursuant to this Section 5.6, but if a Participant receives a hardship withdrawal from the IIP or any other tax-qualified §401(k) plan maintained by an Employer and the terms of such plan require a suspension of the Participant's deferrals for six months following the date of the distribution, then the Participant's Deferral Elections shall be permanently revoked with respect to any compensation paid or payable to the Participant during such six month period.

5.7 Distribution Due to Inclusion in Taxable Income.

In the event that any portion of a Participant's Account is included in his or her taxable income prior to distribution pursuant to Section 409A, the amount so included shall be distributed to the Participant as soon as administratively possible.

5.8 Distribution of De Minimis Amounts.

The Administrative Committee may at any time direct that the entire balance of a Participant's Account be distributed to the Participant in full liquidation of his or her benefit under the Plan; provide that the Participant's entire account balance in all other separate account nonqualified deferred compensation plans maintained by any Employer is also distributed at the same time, and that the total amount so distributed (including all such other plans) does not exceed the limit in effect under Section 402(g) of the Code at the time of the distribution.

5.9 Correction of Errors.

The Administrative Committee shall have the authority to correct any error in the calculation of a Participant's Account or the amount distributed to a Participant, regardless of the reason for the error and regardless of whether distribution of the Account has commenced. By his participation in the Plan and acceptance of benefits hereunder, each Participant agrees that he will promptly repay to the Plan any payment that exceeds the amount to which he was entitled under the Plan (an "excess payment"), and will hold any excess payment, and any proceeds of any excess payment, or property acquired with any excess payment, in trust for the benefit of the Plan, which trust shall remain in effect, and shall continue to apply to any excess payment, proceeds or other property even if transferred to a third party, until the total amount of the excess payment has been repaid to the Plan. The Administrative Committee may, on behalf of the Plan, commence an action to enforce such trust, or take any other available action in law or equity, including setting off any other amount owed to the Participant, to recover such excess payment.

ARTICLE VI — BENEFICIARY DESIGNATION

6.1 Beneficiary Designation.

Each Participant has the right to designate one or more persons or trusts as the Participant's Beneficiary, primary as well as secondary, to whom benefits under this Plan will be paid in the event of the Participant's death prior to complete distribution to the Participant of the benefits due under the Plan. Each Beneficiary designation will be in a written form prescribed by the Administrative Committee and will be effective only when filed with the Administrative Committee during the Participant's lifetime.

6.2 Amendments to Beneficiary Designation.

Any Beneficiary designation may be changed by a Participant without the consent of any Beneficiary by the filing of a new Beneficiary designation with the Administrative Committee. Filing a Beneficiary designation as to any benefits available under the Plan revokes all prior Beneficiary designations effective as of the date such Beneficiary designation is received by the Administrative Committee. If a Participant's Accounts are community property, any Beneficiary designation will be valid or effective only as permitted under applicable law.

6.3 No Beneficiary Designation.

In the absence of an effective Beneficiary designation, or if all Beneficiaries predecease the Participant, the Participant's estate will be the Beneficiary. If a Beneficiary dies after the Participant and before payment of benefits under this Plan has been completed, and no secondary Beneficiary has been designated to receive such Beneficiary's share, the remaining benefits will be payable to the Beneficiary's estate.

6.4 Form of Payment to Beneficiary.

The Account of a Participant who dies prior to Termination of Employment shall be paid to his or her Beneficiary in a single lump sum as soon as administratively feasible following the date of death, regardless of the form of payment elected by the Participant The Account of a Participant who dies after Termination of Employment, but before his or her Account has been fully distributed, shall be distributed in the same manner and at the same time as it would have been distributed to the Participant, except that the six month delay in distributions to a specified employee pursuant to the last paragraph of Section 5.2 shall not apply to the Beneficiary of a specified employee who dies during the six month period following his or her Termination of Employment.

ARTICLE VII — ADMINISTRATION

7.1 Administrative Committee.

The Plan is administrator for purposes of Section 3(16)(A) of the Employee Retirement Income Security Act of 1974, as amended ("ERISA"). Baxter has appointed the members of the Administrative Committee to administer the Plan. Members of the Administrative Committee may be Participants in the Plan.

7.2 Administrative Committee Powers.

The Administrative Committee has such powers as may be necessary to discharge its duties hereunder, including, but not by way of limitation, the following powers, rights and duties:

- (a) <u>Interpretation of Plan</u>. The Administrative Committee has the power, right and duty to construe, interpret and enforce the Plan provisions and to determine all questions arising under the Plan including, but not by way of limitation, questions of Plan participation, eligibility for Plan benefits and the rights of employees, Participants, Beneficiaries and other persons to benefits under the Plan and to determine the amount, manner and time of payment of any benefits hereunder;
- (b) Plan Procedures. The Administrative Committee has the power, right and duty to adopt procedures, rules, regulations and forms to be followed by employees, Participants, Beneficiaries and other persons or to be otherwise utilized in the efficient administration of the Plan which may alter any procedural provision of the Plan without the necessity of an amendment, and which procedures may provide for any election or consent to be made (including without limitation the filing of a Deferral Election Form or Distribution Election Form), or any other action to be taken (including without limitation filing claims and requesting review of denied claims), by electronic mail, internet website, telephone or voice response system or other electronic method to the extent permitted by applicable law;
- (c) <u>Benefit Determinations</u>. The Administrative Committee has the power, right and duty to make determinations as to the rights of employees, Participants, Beneficiaries and other persons to benefits under the Plan and to afford any Participant or Beneficiary dissatisfied with such determination with rights pursuant to a claims procedure adopted by the Committee; and
- (d) <u>Allocation of Duties</u>. The Administrative Committee is empowered to employ agents (who may also be employees of Baxter) and to delegate to them any of the administrative duties imposed upon the Administrative Committee or Baxter.
- (e) <u>Plan Amendments</u>. The Administrative Committee is empowered to amend the Plan as provided in Section 8.1(b).

7.3 Effect of Administrative Committee Decisions.

Any ruling, regulation, procedure or decision of the Administrative Committee will be conclusive and binding upon all persons affected by it. There will be no appeal from any ruling by the Administrative Committee which is within its authority, except as provided in Section 7.4 below. When making a determination or a calculation, the Administrative Committee will be entitled to rely on information supplied by any Employer, accountants and other professionals including, but not by way of limitation, legal counsel for Baxter or any Employer.

7.4 Claims Procedure.

Each person entitled to benefits under the Plan (the "Applicant") must submit a written claim for benefits to the Administrative Committee. Such claim shall be filed not more than one year after the Applicant knows, or with the exercise of reasonable diligence would know, if the basis for the claim. A formal claim shall not be required for the distribution of a Participant's Accounts in the ordinary course of business, but in any case a claim that relates to a dispute over the amount of a distribution shall be filed not more than one year after the distribution is paid. The Administrative Committee may, in its sole discretion (and notwithstanding the first sentence of Section 7.3) accept a claim that is filed late if it determines that special circumstances warrant acceptance of the claim.

If a claim for benefits by the Applicant is denied, in whole or in part, the Administrative Committee, or its delegate, shall furnish the Applicant within 90 days after receipt of such claim, a written notice which specifies the reason for the denial, refers to the pertinent provisions of the Plan on which the denial is based, describes any additional material or information necessary for properly completing the claim and explains why such material or information is necessary, and explains the claim review procedures of this Section 7.4. Such notice will further describe that the Applicant has a right to bring a civil action under Section 502 of ERISA if his or her claim is denied after an appeal and review. The 90 day period may be extended by up to an additional 90 days if special circumstances required, in which event the Applicant shall be notified in writing by the end of the initial 90 day period of the reason for the extension and an estimate of when the claim will be processed.

Any Applicant whose claim is denied under the provisions described above, or who has not received from the Administrative Committee a response to his or her claim within the time periods specified in the provisions described above may request a review of the denied claim by written request to the Administrative Committee within 60 days after receiving notice of the denial. If such a request is made, the Administrative Committee shall make a full and fair review of the denial of the claim and shall make a decision not later than 60 days after receipt of the request, unless special circumstances (such as the need to hold a hearing) require an extension of time, in which case a decision shall be made as soon as possible but not later than 120 days after receipt of the request for review, and written notice of the reason for the extension and an estimate of when the review will be complete shall be given to the Applicant before the commencement of the extension. The decision on review shall be in writing and shall include specific reasons for the decision and specific references to the pertinent provisions of the Plan on which the decision is based. Such notice will further describe that the Applicant has a right to bring a civil action under Section 502 of ERISA.

No person entitled to benefits under the Plan shall have any right to seek review of a denial of benefits, or to bring any action to enforce a claim for benefits, in any court or administrative agency prior to his or her filing a claim for benefits and exhausting all of his or her rights under this Section 7.4, or more than 180 days after he receives the Administrative Committee's decision on review of the denial of his or her claim. Although not required to do so, an Applicant, or his or her representative, may choose to state the reason or reasons he believes he is entitled to benefits, and may choose to submit written evidence, during the initial claim process or review of claim denial process. However, failure to state any such reason or submit such evidence during the initial claim process or review of claim denial process, shall permanently bar the Applicant, and his or her successors in interest, from raising such reason or submitting such evidence in any forum at any later date. An Applicant whose claim is denied initially or on review is entitled to receive, on request and free of charge, reasonable access to, and copies of, all documents, records, and other information relevant to such claim for benefits.

7.5 Action by Administrative Committee.

Action by the Administrative Committee will be subject to the following special rules:

- (a) <u>Meetings and Documents</u>. The Administrative Committee may act by meeting or by document signed without meeting and documents may be signed through the use of a single document or concurrent documents.
- (b) Action by Majority. The Administrative Committee will act by a majority decision which action will be as effective as if such action had been taken by all Administrative Committee members, provided that by majority action one or more Administrative Committee members or other persons may be authorized to act with respect to particular matters on behalf of all Administrative Committee members.
- (c) <u>Resolving Deadlocks</u>. If there is an equal division among the Administrative Committee members with respect to any question a disinterested party may be selected by a majority vote to decide the matter. Any decision by such disinterested party will be binding.

7.6 Indemnity.

To the extent permitted by applicable law and to the extent that they are not indemnified or saved harmless under any liability insurance contracts, any present or former Administrative Committee members, officers, or directors of Baxter, the Employers or their subsidiaries or affiliates, if any, will be indemnified and saved harmless by the Employers from and against any and all liabilities or allegations of liability to which they may be subjected by reason of any act done or omitted to be done in good faith in the administration of the Plan, including all expenses reasonably incurred in their defense in the event that Baxter fails to provide such defense after having been requested in writing to do so.

ARTICLE VIII — AMENDMENT AND TERMINATION OF PLAN

8.1 Amendment.

- (a) The Compensation Committee may amend the Plan at any time, except that no amendment will decrease or restrict the Accounts of Participants and Beneficiaries at the time of the amendment. The Company's authority to amend the Plan has been delegated to the Administrative Committee to the extent provided in Section 8.1(b). The authority to amend the Plan in any respect (whether or not such amendment is within the authority delegated to the Administrative Committee) may also be exercised by the Board of Directors, the Compensation Committee or any other person to whom the Board or Compensation Committee delegates such authority.
- (b) The Administrative Committee has been delegated the authority to adopt any amendments to the Plan as the Administrative Committee may determine to be necessary or appropriate, except that no amendment shall be made to any Plan without approval of the Compensation Committee unless the Administrative Committee determines that such amendment will not significantly change the overall level of benefits provided by such Plan; significantly change the requirements for eligibility for participation in the Plan; or add any material new benefit that would significantly increase the cost of the Plan. In illustration but not limitation of the foregoing, the Administrative Committee is authorized to adopt any amendment to a Plan that it determines to be:
 - (i) an amendment that provides for the Plan to be adopted by any business entity acquired by the Company, including providing any special rules applicable to the employees of such business entity;
 - (ii) an amendment that the Administrative Committee determines to be of an administrative, ministerial or technical nature only;
 - (iii) an amendment that the Administrative Committee determines to be necessary or appropriate to carry out any amendment approved by, or other resolution adopted by, the Board;
 - (iv) an amendment that the Administrative Committee determines to be necessary or appropriate to comply with any applicable law, or necessary to conform the terms of the Plan to established administrative practices or procedures; or
 - (v) an amendment that the Administrative Committee determines to be necessary or appropriate to clarify or to resolve any inconsistency or ambiguity in the terms of the Plan.

The adoption by the Administrative Committee of any amendment to the Plan shall constitute conclusive evidence that the Administrative Committee has determined such amendment to be authorized under the terms of the foregoing resolution, which determination shall be conclusive and binding on all employees, participants, beneficiaries and other persons claiming any benefit under the Plan.

8.2 Right to Terminate.

The Compensation Committee may at any time terminate the Plan. Any Employer may terminate its participation in the Plan by notice to Baxter. The Plan may also be terminated with respect to a group of Eligible Employees only (including, effective January 1, 2005, Participants who are Eligible Employees solely by reason of Section 3.1), and the provisions of Section 8.3 shall apply to such group of Eligible Employees only.

8.3 Payment at Termination.

If the Plan is terminated all Accounts shall continue to be held and distributed in accordance with the terms of the Plan; provided that the Administrative Committee may, to the extent permitted under Section 409A, provide for the immediate distribution of Accounts.

ARTICLE IX — MISCELLANEOUS

9.1 Unfunded Plan.

This Plan is intended to be an unfunded retirement plan maintained primarily to provide retirement benefits for a select group of management or highly compensated employees. All credited amounts are unfunded, general obligations of the appropriate Employer. This Plan is not intended to create an investment contract, but to provide retirement benefits to eligible employees who participate in the Plan. Eligible employees are members of a select group of management or are highly compensated employees, who, by virtue of their position with an Employer, are uniquely informed as to such Employer's operations and have the ability to affect materially Employer's profitability and operations. The Administrative Committee, but shall not be obligated to, establish one or more trusts of the type commonly referred to as "rabbi trusts" and cause funds representing all or a portion of the amounts deferred under the Plan to be deposited in such trusts, provided that the terms of such trusts provide that, upon the insolvency of the an Employer, the funds held in such trusts are subject to the claims of the Employer's creditors, in accordance with applicable Internal Revenue Service guidance. The Administrative Committee may cause Baxter to assume the obligations of the employer under any such trust previously established with respect to either or both of the Gambro Plans, and such trust shall be treated as a trust established pursuant to this Section 9.1 in accordance with its terms. The Administrative Committee may, on behalf of Baxter, enter into agreements establishing any such trusts, amend or terminate such trusts, and remove or replace the trustees of such trusts, including any trusts established under the Gambro Plans, in accordance with their terms.

9.2 Unsecured General Creditor.

In the event of an Employer's insolvency, Participants and their Beneficiaries, heirs, successors and assigns will have no legal or equitable rights, interest or claims in any property or assets of such Employer, nor will they be Beneficiaries of, or have any rights, claims or interests in any life insurance policies, annuity contracts or the proceeds therefrom owned or which may be acquired by such Employer (the "Policies"), or any trust established pursuant to Section 9.1, greater than those of any other unsecured general creditors. In that event, any and all of the Employer's assets and Policies will be, and remain, the general, unpledged, unrestricted assets of Employer. Employer's obligation under the Plan will be merely that of an unfunded and unsecured promise of Employer to pay money in the future.

9.3 Nonassignability.

Neither a Participant nor any other person will have any right to commute, sell, assign, transfer, pledge, anticipate, mortgage or otherwise encumber, transfer, hypothecate or convey in advance of actual receipt the amounts, if any, payable hereunder, or any part thereof, which are, and all rights to which are, expressly declared to be nonassignable and nontransferable. No part of the amounts payable will, prior to actual payment, be subject to seizure or sequestration for the payment of any debts, judgments, alimony or separate maintenance owed by a Participant or any other person, nor be transferable by operation of law in the event of a Participant's or any other person's bankruptcy or insolvency. Nothing contained herein will preclude an Employer from

offsetting any amount owed to it by a Participant against payments to such Participant or his or her Beneficiary.

9.4 Not a Contract of Employment.

The terms and conditions of this Plan will not be deemed to constitute a contract of employment between a Participant and such Participant's Employer, and neither the Participant nor the Participant's Beneficiary will have any rights against such Participant's Employer except as may otherwise be specifically provided herein. Moreover, nothing in this Plan is deemed to give a Participant the right to be retained in the service of his or her Employer or to interfere with the right of such Employer to discipline or discharge him or her at any time.

9.5 Protective Provisions.

A Participant will cooperate with Baxter by furnishing any and all information requested by Baxter, in order to facilitate the payment of benefits hereunder.

9.6 Governing Law.

The provisions of this Plan will be construed and interpreted according to the laws of the State of Illinois, to the extent not preempted by ERISA.

9.7 Severability.

In the event any provision of the Plan is held invalid or illegal for any reason, any illegality or invalidity will not affect the remaining parts of the Plan, but the Plan will be construed and enforced as if the illegal or invalid provision had never been inserted, and Baxter will have the privilege and opportunity to correct and remedy such questions of illegality or invalidity by amendment as provided in the Plan, including, but not by way of limitation, the opportunity to construe and enforce the Plan as if such illegal and invalid provision had never been inserted herein.

9.8 Notice.

Any notice or filing required or permitted to be given to Baxter or the Administrative Committee under the Plan will be sufficient if in writing and hand delivered, or sent by registered or certified mail to any member of the Administrative Committee, or to Baxter's Chief Financial Officer and, if mailed, will be addressed to the principal executive offices of Baxter. Notice to a Participant or Beneficiary may be hand delivered or mailed to the Participant or Beneficiary at his or her most recent address as listed in the employment records of Baxter. Notices will be deemed given as of the date of delivery or mailing or, if delivery is made by certified or registered mail, as of the date shown on the receipt for registration or certification. Any person entitled to notice hereunder may waive such notice.

9.9 Successors.

The provisions of this Plan will bind and inure to the benefit of Baxter, each Employer, the Participants and Beneficiaries, and their respective successors, heirs and assigns. The term

successors as used herein will include any corporate or other business entity, which, whether by merger, consolidation, purchase or otherwise acquires all or substantially all of the business and assets of Baxter, and successors of any such corporation or other business entity.

9.10 Action by Baxter.

Except as otherwise provided herein, any action required of or permitted by Baxter under the Plan will be by resolution of the Compensation Committee or any person or persons authorized by resolution of the Compensation Committee.

9.11 Effect on Benefit Plans.

Amounts paid under this Plan, will not by operation of this Plan be considered to be compensation for the purposes of any benefit plan maintained by any Employer. The treatment of such amounts under other employee benefit plans will be determined pursuant to the provisions of such plans.

9.12 Participant Litigation.

In any action or proceeding regarding the Plan, employees or former employees of Baxter or an Employer, Participants, Beneficiaries or any other persons having or claiming to have an interest in this Plan will not be necessary parties and will not be entitled to any notice or process. Any final judgment which is not appealed or appealable and may be entered in any such action or proceeding will be binding and conclusive on the parties hereto and all persons having or claiming to have any interest in this Plan. To the extent permitted by law, if a legal action is begun against Baxter, an Employer, the Administrative Committee, or any member of the Administrative Committee by or on behalf of any person and such action results adversely to such person or if a legal action arises because of conflicting claims to a Participant's or other person's benefits, the costs to such person of defending the action will be charged to the amounts, if any, which were involved in the action or were payable to the Participant or other person concerned. To the extent permitted by applicable law, acceptance of participation in this Plan will constitute a release of Baxter, each Employer, the Administrative Committee and each member thereof, and their respective agents from any and all liability and obligation not involving willful misconduct or gross neglect.

* * *

IN WITNESS WHEREOF, the undersigned has caused this Amended and Restated Plan to be executed this 18th day of December, 2014.

BAXTER INTERNATIONAL INC. ADMINISTRATIVE COMMITTEE

By: /s/ Salvatore Dadouche

Salvatore Dadouche
Administrative Committee Member

28

APPENDIX A

PARTICIPATING EMPLOYERS

Participating Employers in the Plan include all participating Employers in the Baxter International Inc. and Subsidiaries Incentive Investment Plan.

APPENDIX B

SUPPLEMENTAL PAY DEFERRALS UNDER SECTION 3.3

None.

APPENDIX C

DISCRETIONARY EMPLOYER CONTRIBUTIONS UNDER SECTION 3.5

Contribution on Behalf of Dagmar Bjorkeson

- 1. Not more than thirty (30) days after the date on which Dagmar Bjorkeson ("Bjorkeson") commences employment with Baxter, Baxter shall credit a discretionary contribution in the amount of one million dollars (\$1,000,000.00) to a special employer contribution account (the "Account").
- 2. In the event Bjorkeson voluntarily resigns from employment with Baxter prior to the first anniversary of Bjorkeson's first day of employment, the entire balance in the Account, including earnings, shall be forfeited.
- 3. In the event Bjorkeson voluntarily resigns from employment with Baxter on or after the first anniversary of Bjorkeson's first day of employment, but prior to the second anniversary of Bjorkeson's first day of employment, fifty percent (50%) of the balance in the Account on the date of resignation, including earnings, shall be forfeited.
- 4. Bjorkeson may elect the manner of distribution of the Account by completing a Deferral Election Form in accordance with Section 5.2 of the Plan not later than Bjorkeson's first day of employment, which Deferral Election Form shall subsequently apply to all other amounts deferred by Bjorkeson in the manner provided by Section 5.2.B(b) of the Plan.
- 5. Upon the anticipated spin-off by Baxter of Baxalta, it is anticipated that Bjorkeson will be employed by Baxalta. The transfer of Bjorkeson's employment from Baxter to Baxalta shall not be considered a termination of employment, and the Account shall remain subject to forfeiture as provided above if Bjorkeson's employment by Baxalta is terminated by resignation prior to the first anniversary or second anniversary, as applicable, of the first day of Bjorkeson's employment by Baxter.
- 6. Except as otherwise provided above, the Account shall be treated in the same manner as any other bonus deferral under the Plan, and the establishment of the Account shall not otherwise affect Bjorkeson's eligibility to participate in the Plan.

APPENDIX D

SPECIAL DISTRIBUTION PROVISIONS APPLICABLE TO AMOUNTS TRANSFERRED FROM GAMBRO PLANS

Forms of Distribution under the Voluntary Deferral Plan

- <u>D.1</u> The following terms shall govern the manner in which accounts transferred from the Gambro Renal Products, Inc., Voluntary Deferral Plan ("VDP Accounts") are distributed. In the event of any conflict between the provisions of this Section D.1 and the terms of the Gambro Plan, the terms of the Gambro Plan shall govern and control to the extent required to avoid a violation of Code Section 409A.
- (a) <u>General.</u> VDP Accounts shall be paid out in cash. At the time of the deferral election with respect to a Plan Year, the Participant shall elect a payout alternative for that Plan Year's Deferral (adjusted for hypothetical investment returns).
- (b) <u>Deferral Elections.</u> The Participant shall make an election to receive or commence receipt of the payout:
 - (1) <u>Year One After Retirement:</u> in the year following the Participant's Retirement (as defined in Section 2(t)),
 - (2) <u>Year Two After Retirement:</u> in the second year following the Participant's Retirement (as defined below), or
 - (3) Year Certain: for an Initial Deferral Election made by a Participant, in a year certain at least three (3) years after the date the Initial Deferral Election is executed for all Compensation other than a Participant's Annual Incentive and at least four (4) years after the Initial Deferral Election is executed for a Participant's Annual Incentive (to the extent that Participant is or was permitted to make an Initial Deferral Election with respect to Participant's Annual Incentive) and for all other deferral elections, in a year certain at least three (3) years beyond the year for which the deferral election is effective for all Compensation other than a Participant's Annual Incentive and at least four (4) years beyond the year for which the deferral election is effective for a Participant's Annual Incentive (to the extent that Participant is or was permitted to make a deferral election with respect to Participant's Annual Incentive). The election timing provisions of this Section 6.3(b)(3) described above apply for elections made in 2009 and subsequent Plan Years for deferrals of compensation earned in 2010 and subsequent Plan Years; provided, however, that deferral elections will not be permitted with respect to 2012 (or any subsequent Plan Year) Annual Incentives.

The Participant shall make a deferral election to receive distributions in the form of one (1) lump-sum cash payment or five (5), ten (10), fifteen (15), or twenty (20) annual installments. Any lump-sum cash payment shall be paid in the January of the year specified by the deferral election. If applicable, the first installment payment shall be payable in the January of the year specified by the deferral election and each subsequent

installment payment shall be paid each subsequent January until the amount subject to the installment election is paid out in full. Effective January 1, 2009, each installment shall be equal to the remaining balance(s) on January 19 or the first business day following January 19, if January 19 is not a business day, of the portion(s) of the VDP Account (less applicable surrender charges) subject to the deferral election(s), divided by the total number of installments yet to be made (i.e., the denominator in the year of the first of ten installments shall be ten (10), in the year of the second installment shall be nine (9), etc.) with the last installment payment equal to the remaining value. Each payment described in this paragraph is intended to be a separate payment for purposes of Code Section 409A and Treas. Reg. Section 1.409A-2(b)(2).

- (c) <u>Time and Manner of Payment.</u> The time and manner of payment shall be based on the earliest of the following to occur, as follows:
 - (1) Retirement (as defined below):
 - (i) For amounts deferred in 2010 (excluding Deferral of the Annual Incentive paid in 2010 for the 2009 Plan Year) and subsequent Plan Years, a Participant shall commence receipt of the payout as elected by the Participant in accordance with the payment method(s) designated by the deferral election(s) previously made by the Participant (Year One After Retirement, Year Two After Retirement and/or in a Year Certain).
 - (ii) For amounts deferred after 2004 and prior to 2010 (including Deferral of the Annual Incentive paid in 2010 for the 2009 Plan Year), payment shall be made in accordance with Section D.1(c)(5)
 - (2) <u>Becoming Disabled:</u> For amounts deferred in 2005 and subsequent Plan Years, if a Participant becomes Disabled, payment shall be made in accordance with Section 6.3(c)(5). For amounts deferred prior to 2005, if a Participant becomes Disabled, a lump-sum cash payment of the VDP Account shall be made, or installment payments shall commence and be paid over the period(s) designated by the Participant's deferral election(s) (if the Participant has attained age fifty-nine and one-half (59-1/2) and so elected), within sixty (60) days following the Participant becoming Disabled.
 - (3) <u>Death:</u> In the event of a Participant's death, a lump-sum cash payment of the VDP Account shall be made, within sixty (60) days following such death.
 - (4) <u>Year Certain:</u> In the event a Participant has elected a year certain payout, such Participant shall commence receipt of the payout as elected by the Participant on the deferral election(s) previously made by the Participant in accordance with Section 6.3(b)(3).
 - (5) Other Separation from Service:
 - (i) For amounts deferred in 2010 (excluding Deferral of the Annual Incentive paid in 2010 for the 2009 Plan Year) and subsequent Plan Years, if a

Participant has a Separation from Service prior to an event described in Sections 6.3(c)(1), (2), (3) or (4) above, he or she shall receive a lump-sum cash payment of the VDP Account in January of the year immediately following such Separation from Service; *provided, however*, that any amount deferred in 2010 or a subsequent Plan Year that is subject to a Year Certain election described in Section 6.3(b)(3) (whether or not payment of such amount has commenced) shall continue to be subject to such Year Certain election and shall be paid out in accordance with such Year Certain election.

- (ii) For amounts deferred after 2004 and prior to 2010 (including Deferral of the Annual Incentive paid in 2010 for the 2009 Plan Year), if a Participant has a Separation from Service for any reason other than death, he or she shall receive a lump-sum cash payment of the VDP Account in January of the year immediately following such Separation from Service. Furthermore, if a Participant has commenced installment payments under a Year Certain payout, the remaining installment payments shall be made in the January of the year immediately following such Separation from Service.
- Failure to Make Deferral Election. In the event that a Participant has a Separation from Service and such Participant has not made the deferral election described in Section D.1(b) with respect to all or a portion of his or her VDP Account, such Participant shall receive a lump-sum cash payment of the VDP Account (to the extent of the portion of such VDP Account for which no deferral election has been made) in January of the year immediately following such Separation from Service. A Participant shall be deemed not to have made a deferral election pursuant to Section 6.3(b) if payment in accordance with an election or purported election made by the Participant would subject a distribution under this Plan to additional tax or interest under Section 409A of the Code.
- (d) <u>Death.</u> Notwithstanding Section D.1(b) through (c) above, in the event of a Participant's death prior to the commencement of distributions or during the installment period, only a lump-sum cash payment of the remaining balance of the entire VDP Account shall be made, within sixty (60) days after such death.
- (e) <u>Certain 2005 Through 2009 Deferrals.</u> Effective January 1, 2009, notwithstanding D.1(c)(1) and (4), any Deferral Subaccount with respect to Deferrals made for the Plan Years 2005 through 2009 (including the Deferral of the Annual Incentive paid in 2010 for the 2009 Plan Year) shall, in the event that a Participant would have received a distribution of such Deferral Sub-account due to Retirement, be paid in a lump-sum cash payment in January of the year immediately following Participant's Separation from Service (other than a Separation from Service due to death of the Participant). Furthermore, if a Participant has commenced installment payments under a Year Certain payout, the remaining installment payments shall be made in the January of the year immediately following such Separation from Service. This Section 6.3(g) does not apply to distributions due to a Participant's death.

- Amounts Deferred Prior to 2005. For amounts deferred prior to 2005, distributions shall be made in accordance with the Gambro Voluntary Deferral Plan as it existed as of October 3, 2004 which required that a Participant's deferral accounts shall be paid out in cash and in accordance with a Participant's deferral election(s) except as provided in the next sentence. If a Participant's Termination of Service (as defined in the Gambro Voluntary Deferral Plan as it existed as of October 3, 2004) is for any reason other than Retirement (as defined in the Gambro Voluntary Deferral Plan as it existed as of October 3, 2004), he or she shall automatically receive a lump-sum payment in the January of the year immediately following the Participant's Termination of Service (as defined in the Gambro Voluntary Deferral Plan as it existed as of October 3, 2004). Notwithstanding the foregoing, for amounts deferred prior to 2005, if a Participant becomes Disabled, a lump-sum cash payment shall be made, or installment payments shall commence and be paid over the periods designated by the Participant's deferral election(s) (if the Participant has attained age fifty-nine and one-half (59-1/2) and so elected), within sixty (60) days of the Participant becoming Disabled.
- (g) For purposes of this Section D.1, "Retirement" means :
 - (1) effective January 1, 2010, for amounts deferred in 2010 (excluding Deferral of the Annual Incentive paid in 2010 for the 2009 Plan Year) and subsequent Plan Years, a Separation from Service on or after a Participant's attainment of age fifty-nine and one-half (59 1/2);
 - (2) for amounts deferred in 2009 (including Deferral of the Annual Incentive paid in 2010 for the 2009 Plan Year), that a Participant has voluntarily had a Separation from Service on or after his attainment of age fifty-nine and one-half (59 1/2);
 - (3) for amounts deferred after 2004 but prior to 2009, that a Participant has voluntarily terminated employment with the Employer on or after his or her attainment of age fifty-nine and one-half (59 ½), and does not continue to provide services to the Employer (defined terms in this Section 2(t)(3) have the meanings as defined in the Gambro Voluntary Deferral Plan (As Amended and Restated January 1, 2005), as subsequently amended), and,
 - (4) for amounts deferred prior to 2005, the Participant has reached the age of fifty-nine and one-half (59-1/2) and has voluntarily terminated employment with the Employer (defined terms in this Section 2(t)(4) have the meanings as defined in the Gambro Voluntary Deferral Plan as it existed as of October 3, 2004).

Forms of Distribution under the Executive Retirement Plan

<u>D.2</u> The following terms shall govern the manner in which accounts transferred from the Gambro Renal Products, Inc., Executive Retirement Plan ("GERP Accounts") are distributed. In the event of any conflict between the provisions of this Section D.2 and the terms of the Gambro Plan, the terms of the Gambro Plan shall govern and control to the extent required to avoid a violation of Code Section409A.

- (a) GERP Accounts shall be paid out in cash. Upon a Separation from Service which includes a Participant becoming Disabled but excludes a Separation from Service due to the death of the Participant, the Participant shall be paid in a lump-sum equal to the balance of his or her GERP Account as of the date of distribution, but only to the extent such GERP Account is vested. Such payment shall be made on or after January 1 and on or before March 15 of the year following the Participant's Separation from Service (or if the Participant has attained an age equal to or greater than sixty (60) upon his or her Separation from Service, such payment shall be made within sixty (60) days after such Separation from Service), with the exact payment date being at the sole discretion of the Employer.
- (b) Notwithstanding the foregoing, if a Participant dies, Participant's GERP Account shall be distributed in a lump-sum payment equal to the balance of his or her GERP Account as of the date of distribution, but only to the extent such GERP Account is vested, within sixty (60) days following the date Participant died, with the exact payment date being at the sole discretion of the Employer. Notwithstanding the foregoing, if a Participant becomes Disabled, the balance of the Participant's pre-2005 sub-accounts shall be distributed in a lump-sum payment equal to the aggregate balances of such subaccounts as of the date of distribution within sixty (60) days following the date the Participant becomes Disabled.

FIRST AMENDMENT TO THE BAXTER INTERNATIONAL INC. AND SUBSIDIARIES DEFERRED COMPENSATION PLAN

(Amended and Restated as of January 1, 2015)

Pursuant to Section 8.1 of the Baxter International Inc. and Subsidiaries Deferred Compensation Plan, as amended and restated effective as of January 1, 2015 (the "Plan"), the Plan is further amended as follows, effective May 11, 2016:

- 1. Section 4.2 is amended by the addition of a new paragraph C at the end thereof to read as follows:
 - "C. Participants whose Deferred Compensation Account is deemed invested in the Company Stock Fund shall not be permitted to elect to have their account balance determined as if their shares had been exchanged for shares of Baxalta Incorporated pursuant to the exchange offer set forth in the prospectus, a preliminary copy of which was filed by the Company with the SEC on March 21, 2016, and the balances of Deferred Compensation Accounts deemed to be invested in the Company Stock Fund shall be determined without regard to such exchange offer."
- 3. Section 4.2 is further amended by the addition of a new paragraph D at the end thereof to read as follows:
 - "D. On January 11, 2016, Baxalta Incorporated ("Baxalta") announced that it had entered into an Agreement and Plan of Merger with Shire plc ("Shire"), pursuant to which Shire will acquire Baxalta (the "Merger"), subject to the satisfaction or waiver of certain conditions, and the common stock of Baxalta will be converted into a combination of American Depositary Shares with respect to the common stock of Shire (the "Shire ADSs") at the rate of .1482 Shire ADSs and \$18 in cash (the "Cash Consideration") for each share of common stock of Baxalta. Each Participant whose Deferred Compensation Account is deemed to be invested in the Baxalta common stock fund at the closing of the Merger will be deemed to have received a number of Shire ADSs and Cash Consideration based upon the number of shares of Baxalta common stock in which the Account is deemed invested, and thereafter the Baxalta common stock fund will be converted into a fund consisting of Shire ADSs, and each such Participant's deemed share of the Cash Consideration will be transferred into the Stable Income Fund. The fund consisting of Shire ADSs will be closed to new investments, but each such Participant may transfer his or her share of the Cash Consideration from the Stable Income Fund into any other open fund in accordance with Paragraph B."
- 3. This amendment shall be immediately effective upon its approval by the Administrative Committee, subject to the following. In the event the exchange offer described in the amendment made by Section 1 above is not consummated due to the termination of the exchange offer, in accordance with the terms described in the prospectus, the amendment made by Section 1 shall be null and void. In the event the Merger described in the

amendment made by Section 2 above is not consummated due to the termination of the Merger agreement between Baxalta and Shire, the amendment made by Section 2 shall be null and void. In the event that neither the exchange offer nor the Merger is consummated, as referenced above, this entire amendment shall be null and void. Except as otherwise provided herein, the Plan shall remain in full force and effect.

BAXTER INTERNATIONAL INC. ADMINISTRATIVE COMMITTEE

By: <u>/s/ Salvatore Dadouche</u>
Salvatore Dadouche
Administrative Committee Member

SECOND AMENDMENT TO THE BAXTER INTERNATIONAL INC. AND SUBSIDIARIES DEFERRED COMPENSATION PLAN

(Amended and Restated as of January 1, 2015)

Pursuant to Section 8.1 of the Baxter International Inc. and Subsidiaries Deferred Compensation Plan, as amended and restated effective as of January 1, 2015 (the "Plan"), the Plan, as previously amended by the First Amendment thereto, is further amended as follows,:

- 1. Section 2.11(a) is amended to read as follows, effective as of the date on which this Amendment is executed:
 - "(a) Is a Corporate Officer of Baxter, a participant in the Baxter International Inc. Long Term Incentive Plan (the "LTIP") or an individual designated to participate in the LTIP as of the next annual appointment cycle for the Plan Year for which deferrals relate, provided however that in the event an individual is permitted by the Administrator to make a deferral election for a Plan Year based upon his status as a designated participant in the LTIP for a Plan Year but is not appointed to the LTIP during the annual appointment cycle that occurs during such Plan Year, then such individual's deferral election shall remain in effect for such Plan Year, but such individual shall not be considered an Eligible Employee for any subsequent Plan Year unless he independently satisfies the requirements of this Section 2.11(a) for such Plan Year;"
- 5. Except as otherwise provided above, the Plan shall remain in full force and effect.

[Signature on Following Page]

IN WITNESS WHEREOF, this Second Amendment is adopted this 21st day of December, 2016.

BAXTER INTERNATIONAL INC. ADMINISTRATIVE COMMITTEE

By: <u>/s/ Salvatore Dadouche</u>
Salvatore Dadouche
Administrative Committee Member

[Signature Page for Second Amendment to Baxter International Inc. and Subsidiaries Deferred Compensation Plan]

FIRST AMENDMENT TO BAXTER INTERNATIONAL INC. EMPLOYEE STOCK PURCHASE PLAN (As Amended and Restated Effective July 1, 2011)

WHEREAS, Baxter International Inc. ("Baxter") maintains the Baxter International Inc. Employee Stock Purchase Plan (the "Plan");

WHEREAS, Section 3.03(b) of the Plan imposes a limit on the maximum number of shares that can be purchased by an Eligible Employee in a U.S. Offering, as defined in the Plan ("Limit");

WHEREAS, the administrative practice has been to impose the same Limit on International Offerings, as defined in the Plan;

WHEREAS, the Administrative Committee has consistently exercised its authority over administrative matters of the Plan, including necessary amendments, pursuant to the delegation of such authority by the Compensation Committee of the Board of Directors of Baxter on November 9, 2015;

WHEREAS, it is appropriate for the terms of the Plan to reflect the administrative practice; and

NOW, THEREFORE, the Plan is hereby amended as follows, effective as of July 1, 2011:

1. Section 3.03 of the Plan is amended by the addition of the following paragraph immediately following Section 3.03(c):

"Except as otherwise determined by the Committee, the limitation set forth in Section 3.03(b) shall also apply to International Offerings, and the Committee shall have the authority to impose other limitations on the number of shares that may be purchased pursuant to any U.S. or International Offering as it determines to be necessary to comply with any applicable law or otherwise appropriate."

IN WITNESS WHEREOF, the undersigned has executed this First Amendment on behalf of Baxter International Inc. this 15th day of July, 2016.

BAXTER INTERNATIONAL INC.

By: /s/ Jeanne K. Mason

Jeanne K. Mason Corporate Vice President, Human Resources

BAXTER INTERNATIONAL INC.

Non-Employee Director Compensation Plan

(As amended and restated effective January 1, 2017)

Terms and Conditions

1. Purpose

This Non-Employee Director Compensation Plan (the "<u>Plan</u>") is adopted by the Board of Directors (the "<u>Board</u>") of Baxter International Inc. ("<u>Baxter</u>"). This Plan is adopted pursuant to the Baxter International Inc. 2015 Incentive Plan (the "<u>2015 Incentive Plan</u>"), for the purposes stated in the 2015 Incentive Plan. Capitalized terms defined in the 2015 Incentive Plan that are used without being defined in the Plan will have the same meaning as in the 2015 Incentive Plan.

2. Participants

Each member of the Board who is not an employee of Baxter or any of its subsidiaries shall participate in the Plan (a "<u>Participant</u>"). A Participant shall acknowledge his or her agreement to the terms and conditions of the Plan by countersigning the Plan.

3. <u>Unrestricted Shares of Stock</u>

- 3.1 Subject to Section 3.4, on the date of Baxter's annual meeting of stockholders in each year beginning with the Annual Meeting held on May 2, 2017 (the "Annual Meeting"), and subject to availability of Shares under the 2015 Incentive Plan and except as set forth in Section 4, each Participant upon completion of the Annual Meeting shall, automatically and without necessity of any action by the Board or any committee thereof, receive the number of Full Value Awards in the form of Shares of Baxter Common Stock, par value \$0.01 per Share, ("Unrestricted Shares") equal to the quotient of (A) \$180,000 divided by (B) the Fair Market Value of a Share on the date of grant rounded up or down to the nearest whole number (the "Annual Unrestricted Share Grant Amount").
- 3.2 Each Participant elected or appointed on a date other than the date of an Annual Meeting shall, on the date of such election or appointment and automatically and without necessity of any action by the Board or any committee thereof, receive the number of Unrestricted Shares equal to the product of (A) the Annual Unrestricted Share Grant Amount (as defined in Section 3.1, subject to adjustment in accordance with the 2015 Incentive Plan) for the Unrestricted Shares awarded on the date of the immediately preceding Annual Meeting, *multiplied by* (B) the quotient of (i) the number of full calendar months before the next Annual Meeting divided by (ii) 12, rounded up or down to the nearest whole number. The number of Unrestricted Shares granted under this Section 3.2 shall not exceed the number available under the 2015 Incentive Plan on the date of grant.
- 3.3 All Unrestricted Shares shall be granted with no vesting restrictions.

3.4 If a Participant ceases service as a member of the Board for any reason, other than death or disability and except in connection with any qualifying retirement (as set forth in Section 4.9), prior to the date that is six months after the grant date of Unrestricted Shares, Participant hereby agrees (i) to return any Unrestricted Shares which the Participant has not otherwise sold or transferred prior to the date of Participant's departure from the Board (the "Termination Date") to Baxter and (ii) with respect to any Unrestricted Shares that the Participant has sold or otherwise transferred prior to the Termination Date, to make a cash payment to Baxter equal to the amount of the net proceeds received from the sale, disposition or transfer of the Unrestricted Shares, less any taxes withheld at time of grant within ten Business Days of the Termination Date. Additionally, Participant hereby agrees to promptly notify Baxter of any sale, transfer or other disposition of any Unrestricted Shares that occurs prior to six months after to the applicable grant date.

4. Options

- 4.1 Prior to the date of the Annual Meeting and subject to availability of Shares under the 2015 Incentive Plan, each Participant shall be permitted to elect to receive 50% of the Unrestricted Share value such Participant would otherwise receive in accordance with Section 3 (or \$90,000) in the form of Stock Options ("Stock Options") at such Annual Meeting; provided, that Participant must comply with any related election procedures established by Baxter (including any applicable election deadline provided by Baxter). The number of Stock Options to be granted to a Participant under this Section 4.1 will be determined based on a Black-Scholes or other option valuation model in the discretion of the Board or the Compensation Committee of the Board (the "Committee"). If a Participant duly elects to receive Stock Options at the applicable Annual Meeting in accordance with the terms and conditions established by Baxter and this Section 4.1, such Participant shall receive the remaining 50% of such Unrestricted Share value (or \$90,000) in the form of Unrestricted Shares in accordance with the terms of Section 3.
- 4.2 The purchase price for each Share subject to a Stock Option shall be the Fair Market Value of a Share on the date of grant. The terms of each Stock Option will be as set forth in this Plan and the 2015 Incentive Plan. To the extent that any provision of the Plan is inconsistent with the 2015 Incentive Plan, the 2015 Incentive Plan shall control. The Stock Options are not intended to qualify as Incentive Stock Options within the meaning of Section 422 of the United States Internal Revenue Code.
- **4.3** Stock Options shall be immediately exercisable on the date of grant.
- **4.4** After a Stock Option becomes exercisable and until it expires, it may be exercised in whole or in part, in the manner specified by Baxter. Under no circumstances may a Stock Option be exercised after it has expired. Shares may be used to pay the purchase price for Shares to be acquired upon exercise of a Stock Option or fulfill any tax withholding obligation, subject to any requirements or restrictions specified by Baxter.

- 4.5 Except as provided in Sections 4.7, 4.8, and 4.9, if a Participant ceases service as a member of the Board for any reason, other than death or disability and except in connection with any qualifying retirement (as set forth in Section 4.9), prior to the date that is six months after the grant date of Stock Options, Participant hereby acknowledges and agrees that (i) any unexercised Stock Options shall be automatically cancelled without any further action of the Participant as of the applicable Termination Date and (ii) with respect to any Stock Options that the Participant has exercised on or prior to the Termination Date, Baxter shall be entitled to a cash repayment equal to the amount of net gain recognized from the exercise of the Stock Options, less any taxes withheld at time of exercise and Participant shall make any such payment within ten Business Days of the Termination Date.
- 4.6 If a participant ceases service as a member of the Board more than six months after the applicable grant date of the Stock Options (a "Qualifying Departure"), the Stock Options will not expire but will remain exercisable. Subject to Sections 4.7, 4.8, 4.9 and 4.10, the Stock Options previously granted under the Plan to any Participant who has made a Qualifying Departure will expire three months after the applicable Termination Date, unless the Participant dies or becomes disabled during such three month period in which case the Stock Option will expire on the first anniversary of the date the Participant ceased serving as a member of the Board.
- **4.7** If a Participant dies while serving as a member of the Board, his or her Stock Options will not expire and will remain fully exercisable. Subject to Sections 4.9 and 4.10, the Stock Option will expire on the fifth anniversary of the Participant's death.
- 4.8 If a Participant becomes disabled and unable to continue service as a member of the Board, his or her Stock Options will not expire and will remain fully exercisable. Subject to Sections 4.9 and 4.10, the Stock Option will expire on the fifth anniversary of the date the Participant ceases service as a member of the Board.
- 4.9 If a Participant who has served as a member of the Board for a continuous period of at least ten years or who is at least 72 years of age ceases to serve as a member of the Board (including without limitation by reason of death or disability), his or her Stock Options will not expire and will remain fully exercisable. Subject to Section 4.10, the Stock Options will expire on the fifth anniversary of the date the Participant ceases service as a member of the Board.
- **4.10** Notwithstanding any other provision in these terms and conditions to the contrary, Stock Options that have not previously expired or been cancelled in accordance with Section 4.5 will expire at the close of business on the tenth anniversary of the date of grant. If a Stock Option would expire on a date that is not a Business Day, it will expire at the close of business on the last Business Day preceding that date. A "Business Day" is any day on which the Shares are traded on the New York Stock Exchange.

- 4.11 An exercisable Stock Option may only be exercised by the Participant, his or her legal representative, or a person to whom the Participant's rights in the Stock Option are transferred by will or the laws of descent and distribution or in accordance with rules and procedures established by the Committee.
- 4.12 The Board or the Committee may, in its sole discretion and without receiving permission from any Participant, substitute SARs for any or all outstanding Stock Options granted under this Plan. Upon the grant of substitute SARs, the related Stock Options replaced by the substitute SARs shall be cancelled. The grant price of the substitute SAR shall be equal to the Exercise Price of the related Stock Option, the term of the substitute SAR shall not exceed the term of the related Stock Option, and the terms and conditions applicable to the substitute SAR shall otherwise be substantially the same as those applicable to the related Stock Option replaced by the substitute SAR.
- **4.13** To the extent that a Stock Option has not been exercised on the date the Stock Option would otherwise expire pursuant to Section 4.5, 4.6, 4.7, 4.8, 4.9 or 4.10, and the Fair Market Value of the Common Stock on such date exceeds the exercise price, Baxter may (but shall not be obligated to), on behalf of the Participant, direct that the Stock Option be exercised and the Shares of Common Stock sold, with the proceeds used to pay the exercise price and any applicable tax withholding, and the remaining proceeds credited to the Participant's account, or take such other action as the Committee may determine; provided that in no event shall Baxter, any member of the Committee, or any person acting on their behalf have any liability to a Participant for failing to take any such action.

5. <u>Cash Compensation</u>

- 5.1 Except as provided in the following sentence, Baxter shall pay each Participant a meeting fee of \$2,000 for each meeting of any committee of the Board attended. Except as provided in the following section, participants acting as the chairperson of any committee of the Board shall receive an annual cash retainer of \$15,000 for each committee chaired by him or her. A participant acting as the chairperson of the Audit Committee shall receive an annual cash retainer of \$20,000. Amounts payable within this Section 5.1 shall be paid quarterly in arrears and are payable if the Participant attends in person, by conference telephone, or by any other means permitted by the Delaware General Corporation Law and Baxter's Bylaws, as amended and restated. For the purposes of determining the amount of such quarterly payment(s), a Participant must be a chairperson of a committee of the Board on or prior to the 15th day of a month in order to be entitled to receive such payment(s) with respect to that month.
- 5.2 Baxter shall pay each Participant a total annual cash retainer of \$85,000 per calendar year ("<u>Annual Cash Retainer</u>"). Baxter shall pay an additional annual cash retainer of \$50,000 per calendar year to the Lead Director ("<u>Lead Director Retainer</u>"). Both the Annual Cash Retainer and Lead Director Retainer shall be paid quarterly in arrears. For purposes of determining the amount of such quarterly payment(s), a

Participant and/or the Lead Director must be a member of the Board on or prior to the 15th day of a month in order to be entitled to receive such payment(s) with respect to that month.

5.3 Participants shall be eligible to defer payment of cash compensation otherwise payable under this Section 5 pursuant to the terms and conditions of the Baxter Non-Employee Director Deferred Compensation Plan.

6. Availability of Shares

If on any grant date, the number of Shares which would otherwise be granted in the form of Unrestricted Shares or subject to Stock Options (as duly elected in accordance with Section 4.1) granted under the Plan shall exceed the number of Shares then remaining available under the 2015 Incentive Plan, the available Shares shall be allocated among the Participants in proportion to the number of Shares subject to Stock Options (as duly elected in accordance with Section 4.1) and Unrestricted Shares that Participants would otherwise be entitled to receive.

7. Change in Control

Notwithstanding any other provision of the 2015 Incentive Plan or this Plan, if a Change in Control occurs then all Awards previously unvested shall become immediately vested and exercisable. For purposes of the Plan, a "Change in Control" means the first to occur of any of the following: (i) any Person is or becomes the beneficial owner (as defined in Rule 13d-3 under the Exchange Act), directly or indirectly, of securities of Baxter (not including in the securities beneficially owned by such Person any securities acquired directly from Baxter or its Affiliates) representing 30% or more of the combined voting power of Baxter's then outstanding securities, excluding any Person who becomes such a beneficial owner in connection with a merger or consolidation of Baxter or any direct or indirect subsidiary of Baxter with any other corporation immediately following which the individuals who comprise the Board immediately prior thereto constitute at least a majority of the board of directors of (A) any parent of Baxter or the entity surviving such merger or consolidation or (B) if there is no such parent, of Baxter or such surviving entity; (ii) the following individuals cease for any reason to constitute a majority of the number of directors then serving: individuals who, on the date of grant, constitute the Board and any new director (other than a director whose initial assumption of office is in connection with an actual or threatened election contest, including but not limited to a consent solicitation, relating to the election of directors of Baxter) whose appointment or election by the Board or nomination for election by Baxter's shareholders was approved or recommended by a vote of at least two-thirds (2/3) of the directors then still in office who either were directors on the date of grant or whose appointment, election or nomination for election was previously so approved or recommended; (iii) there is consummated a merger or consolidation of Baxter or any direct or indirect subsidiary of Baxter with any other corporation or other entity, other than a merger or consolidation immediately following which the individuals who comprise the Board immediately prior thereto constitute at least a majority of the board of directors of (A) any parent of Baxter or the entity surviving such

merger or consolidation or (B) if there is no such parent, of Baxter or such surviving entity; or (iv) the shareholders of Baxter approve a plan of complete liquidation or dissolution of Baxter or there is consummated an agreement for the sale or disposition by Baxter of all or substantially all of Baxter's assets, other than a sale or disposition by Baxter of all or substantially all of Baxter's assets immediately following which the individuals who comprise the Board immediately prior thereto constitute at least a majority of the board of directors of (A) any parent of Baxter or of the entity to which such assets are sold or disposed or (B) if there is no such parent, of Baxter or such entity.

8. General Provisions

- 8.1 Subject to the limitations contained in Section 9 of the 2015 Incentive Plan, the Board or the Committee may, at any time and in any manner, amend, suspend, or terminate the Plan or any Award outstanding under the Plan.
- **8.2** Participation in the Plan does not give any Participant any right to continue as a member of the Board for any period of time or any right or claim to any benefit unless such right or claim has specifically accrued hereunder.

Acknow	ledged and Agreed to:	
PARTI	CIPANT	
Ву:		
Name:		

Baxter International Inc. and Subsidiaries

Computation of Ratio of Earnings to Fixed Charges (unaudited — in millions, except ratios)

years ended December 31		2016	2015	2014	2013	2012
Income from continuing operations before income taxes		4,954	\$ 428	\$ 490	\$ 375	\$ 703
Fixed charges						
Interest costs(1)		107	197	237	225	165
Estimated interest in rentals(2)		58	70	83	71	67
Fixed charges as defined		165	267	320	296	232
Adjustments to income from continuing operations						
Interest costs capitalized		(18)	(51)	(70)	(70)	(52)
Net (gains) losses of less than majority-owned affiliates, net of dividends		2	(1)	(19)	(1)	1
Income from continuing operations as adjusted	\$	5,103	\$ 643	\$ 721	\$ 600	\$ 884
Ratio of earnings from continuing operations to fixed charges		30.93	2.41	2.25	2.03	3.81

⁽¹⁾ Excludes interest on uncertain tax positions.

⁽²⁾ Represents the estimated interest portion of rents.

BAXTER INTERNATIONAL INC.

The following is a list of subsidiaries of Baxter International Inc. as of December 31, 2016, omitting some subsidiaries which, when considered in the aggregate, would not constitute a significant subsidiary. Where ownership of a subsidiary is less than 100% by Baxter International Inc. or a Baxter International Inc. subsidiary, such has been noted by designating the percentage of ownership.

Domestic Subsidiary	Incorporation	
Baxter Colorado Holding Inc.	Colorado	
Baxter Corporation Englewood	Colorado	
Baxter Export Corporation	Nevada	
Baxter Global Holdings II Inc.	Delaware	
Baxter Healthcare Corporation	Delaware	
Baxter Holding Services Company	Delaware	
Baxter Holdings LLC	Delaware	
Baxter Pharmaceutical Solutions LLC	Delaware	
Baxter Sales and Distribution LLC	Delaware	
Baxter World Trade Corporation	Delaware	
Gambro Renal Products, Inc.	Colorado	
GRP US Holding Corp.	Colorado	
Laboratorios Baxter S.A.	Delaware	
Synovis Life Technologies, Inc.	Minnesota	
•		
Foreign Subsidiary	Incorporation	
Baxter Healthcare Pty Ltd	Australia	
Baxter S.A.	Belgium	
Baxter World Trade SPRL	Belgium	
Baxter Hospitalar Ltda.	Brazil	
Baxter Corporation	Canada	
Baxter (China) Investment Co., Ltd	China	
Baxter Healthcare (Guangzhou) Company Ltd	China	87.5%
Baxter Healthcare (Suzhou) Company Ltd	China	
Baxter Deutschland GmbH	Germany	
Baxter Deutschland Holding GmbH	Germany	
Baxter Oncology GmbH	Germany	
Gambro Dialysatoren GmbH	Germany	
Baxter (India) Private Limited	India	
Baxter S.p.A.	Italy	
Baxter World Trade Italy S.R.L.	Italy	
Bieffe Medital S.p.A.	Italy	
Baxter Holdings Limited	Japan	
Baxter Limited	Japan	
Baxter Holding Mexico, S. de R.L. de C.V.	Mexico	
Baxter S.A. de C.V.	Mexico	
Baxter Healthcare Asia PTE Ltd	Singapore	
Baxter Pacific Investments Pte Ltd	Singapore	
Baxter Incorporated	South Korea	
Baxter Holding AB	Sweden	
Gambro AB	Sweden	
Gambro Lundia AB	Sweden	
Indap Holding AB	Sweden	
Baxter AG	Switzerland	
Baxter Healthcare Holding GmbH	Switzerland	
Baxter Healthcare SA	Switzerland	
Baxter Holding B.V.	The Netherlands	
Baxter Netherlands Holding B.V.	The Netherlands	
ApaTech Limited	United Kingdom	
Baxter Healthcare Limited	United Kingdom	

CONSENT OF INDEPENDENT REGISTERED PUBLIC ACCOUNTING FIRM

We hereby consent to the incorporation by reference in the Registration Statements on Form S-8 (Nos. 33-28428, 33-54069, 333-10520, 333-43563, 333-47019, 333-71553, 333-80403, 333-88257, 333-48906, 333-62820, 333-102140, 333-104420, 333-104021, 333-105032, 333-143063, 333-174401, 333-206700 and 333-206701), and on Form S-3 (No. 333-207810) of Baxter International Inc. of our report dated February 23, 2017 relating to the financial statements, the financial statement schedule and the effectiveness of internal control over financial reporting, which appears in this Form 10-K.

/s/ PricewaterhouseCoopers LLP Chicago, Illinois February 23, 2017

Certification of Chief Executive Officer Pursuant to Rules 13a-14(a) and 15d-14(a) of the Securities Exchange Act of 1934, as Amended

I, José E. Almeida, certify that:

- 1. I have reviewed this Annual Report on Form 10-K of Baxter International Inc.;
- 2. Based on my knowledge, this report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this report;
- 3. Based on my knowledge, the financial statements, and other financial information included in this report, fairly present in all material respects the financial condition, results of operations and cash flows of the registrant as of, and for, the periods presented in this report;
- 4. The registrant's other certifying officer and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15(e)) and internal control over financial reporting (as defined in Exchange Act Rules 13a-15(f) and 15d-15(f)) for the registrant and have:
 - (a) Designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this report is being prepared;
 - (b) Designed such internal control over financial reporting, or caused such internal control over financial reporting to be designed under our supervision, to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles;
 - (c) Evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in this report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this report based on such evaluation; and
 - (d) Disclosed in this report any change in the registrant's internal control over financial reporting that occurred during the registrant's most recent fiscal quarter (the registrant's fourth fiscal quarter in the case of an annual report) that has materially affected, or is reasonably likely to materially affect, the registrant's internal control over financial reporting; and
- 5. The registrant's other certifying officer and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrant's auditors and the audit committee of the registrant's board of directors (or persons performing the equivalent functions):
 - (a) All significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrant's ability to record, process, summarize and report financial information; and
 - (b) Any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal control over financial reporting.

/s/ José E. Almeida

José E. Almeida Chairman of the Board and Chief Executive Officer

Date: February 23, 2017

Certification of Chief Financial Officer Pursuant to Rules 13a-14(a) and 15d-14(a) of the Securities Exchange Act of 1934, as Amended

I, James K. Saccaro, certify that:

- 1. I have reviewed this Annual Report on Form 10-K of Baxter International Inc.;
- 2. Based on my knowledge, this report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this report;
- 3. Based on my knowledge, the financial statements, and other financial information included in this report, fairly present in all material respects the financial condition, results of operations and cash flows of the registrant as of, and for, the periods presented in this report;
- 4. The registrant's other certifying officer and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15(e)) and internal control over financial reporting (as defined in Exchange Act Rules 13a-15(f) and 15d-15(f)) for the registrant and have:
 - (a) Designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this report is being prepared;
 - (b) Designed such internal control over financial reporting, or caused such internal control over financial reporting to be designed under our supervision, to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles;
 - (c) Evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in this report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this report based on such evaluation; and
 - (d) Disclosed in this report any change in the registrant's internal control over financial reporting that occurred during the registrant's most recent fiscal quarter (the registrant's fourth fiscal quarter in the case of an annual report) that has materially affected, or is reasonably likely to materially affect, the registrant's internal control over financial reporting; and
- 5. The registrant's other certifying officer and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrant's auditors and the audit committee of the registrant's board of directors (or persons performing the equivalent functions):
 - (a) All significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrant's ability to record, process, summarize and report financial information; and
 - (b) Any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal control over financial reporting.

/s/ James K. Saccaro

James K. Saccaro Corporate Vice President and Chief Financial Officer

Date: February 23, 2017

Certification of Chief Executive Officer Pursuant to 18 U.S.C. Section 1350, as Adopted Pursuant to Section 906 of the Sarbanes-Oxley Act of 2002

José E. Almeida, as Chairman of the Board and Chief Executive Officer of Baxter International Inc. (the "Company"), certifies, pursuant to 18 U.S.C. Section 1350, as adopted pursuant to Section 906 of the Sarbanes-Oxley Act of 2002, that:

- (1) the Company's Annual Report on Form 10-K for the year ended December 31, 2016 as filed with the Securities and Exchange Commission on the date hereof (the "Report") fully complies with the requirements of Section 13(a) or 15(d) of the Securities Exchange Act of 1934; and
- (2) the information contained in the Report fairly presents, in all material respects, the financial condition and results of operations of the Company.

/s/ José E. Almeida

José E. Almeida Chairman of the Board and Chief Executive Officer

February 23, 2017

Certification of Chief Financial Officer Pursuant to 18 U.S.C. Section 1350, as Adopted Pursuant to Section 906 of the Sarbanes-Oxley Act of 2002

James K. Saccaro, as Corporate Vice President and Chief Financial Officer of Baxter International Inc. (the "Company"), certifies, pursuant to 18 U.S.C. Section 1350, as adopted pursuant to Section 906 of the Sarbanes-Oxley Act of 2002, that:

- (1) the Company's Annual Report on Form 10-K for the year ended December 31, 2016 as filed with the Securities and Exchange Commission on the date hereof (the "Report") fully complies with the requirements of Section 13(a) or 15(d) of the Securities Exchange Act of 1934; and
- (2) the information contained in the Report fairly presents, in all material respects, the financial condition and results of operations of the Company.

/s/ James K. Saccaro

James K. Saccaro Corporate Vice President and Chief Financial Officer

February 23, 2017